

Since January 2020 Elsevier has created a COVID-19 resource centre with free information in English and Mandarin on the novel coronavirus COVID-19. The COVID-19 resource centre is hosted on Elsevier Connect, the company's public news and information website.

Elsevier hereby grants permission to make all its COVID-19-related research that is available on the COVID-19 resource centre - including this research content - immediately available in PubMed Central and other publicly funded repositories, such as the WHO COVID database with rights for unrestricted research re-use and analyses in any form or by any means with acknowledgement of the original source. These permissions are granted for free by Elsevier for as long as the COVID-19 resource centre remains active.

Chemistry, biochemistry and biology of sialic acids

Roland Schauer¹ and Johannis P. Kamerling²

¹Biochemisches Institut, Christian-Albrechts-Universität zu Kiel, Germany, ²Bijvoet Center, Department of Bio-Organic Chemistry, Utrecht University, The Netherlands

List of Abbreviations

Abbreviation	s used for sialic acids (Sia) are included	GlcA	D-glucuronic acid
in Tables 1 a	ind 13.	GlcNAc	N-acetyl-D-glucosamine
Alt	altrose	GlcN	D-glucosamine
Ara	L-arabinose	GD3	disialoganglioside
Asn	L-asparagine	GM1	monosialoganglioside
Asp	L-aspartate	Gul	gulose
CDP	cytidine diphosphate	Нер	heptose
Cer	ceramide	HPLC	high-performance liquid
CI	chemical ionization		chromatography
CMP	cytidine monophosphate	HPTLC	high-performance thin-layer
CoA	coenzyme A		chromatography
CTP	cytidine triphosphate	IgG	immunoglobulin G
C7-Neu5Ac	Neu5Ac missing the C8-C9 part	IgM	immunoglobulin M
C8-Neu5Ac	Neu5Ac missing the C9 part	Leu	L-leucine
Da	Dalton	Man	D-mannose
DFP	diisopropylfluorophosphate	ManNAc	N-acetyl-D-mannosamine
DIG	digoxigenin	MS	mass spectrometry
DMB	1,2-diamino-4,5-	MU	4-methylumbelliferyl
	methylenedioxybenzene	NMK	nucleoside monophosphate kinase
DNA	desoxyribonucleic acid	NMR	nuclear magnetic resonance
EI	electron impact	NOESY	nuclear Overhauser enhancement
EPR	electron paramagnetic resonance		spectroscopy
FAB	fast atom bombardment	P, Pi	phosphate
Fuc	L-fucose	PAD	pulsed amperometric detection
FucN	L-fucosamine	PEP	phosphoenolpyruvate
Gal	D-galactose	PK	pyruvate kinase
GalNAc-ol	N-acetylgalactosaminitol	PPase	inorganic phosphatase
GalANGro	N-galacturonyl-2-aminoglycerol	PPi	pyrophosphate
GalNAc	N-acetyl-D-galactosamine	PYR	pyruvate
GDP	guanosine diphosphate	ROESY	rotating-frame nuclear Overhauser
Glc	D-glucose		enhancement spectroscopy
GLC	gas-liquid chromatography	QuiNAc	N-acetyl-D-quinovose

S	sulfate	Thr	L-threonine
SDS/PAGE	sodium dodecyl sulfate/polyacrylamide	TOCSY	total correlation spectroscopy
	gel electrophoresis	UDP	uridine diphosphate
Ser	L-serine	UV	ultraviolet
TADH	Thermoanaerobium brockii alcohol	Xyl	D-xylose
	dehydrogenase	2D	two-dimensional
TFA	trifluoroacetic acid		
TLC	thin-layer chromatography		

1. Introduction

Since the discovery of N-acetylneuraminic acid, the most universal sialic acid, at the end of the 1930s [1,2] as well as the structural and stereochemical elucidation of its free and bound forms at the end of the 1960s (reviewed in ref. [3]), there has been a continual increase in the number of sialic acid types (1994: more than 40) recognized to occur in a variety of living organisms. It is now generally accepted that naturally occurring sialic acids are monosaccharides which influence many important biological and pathological phenomena.

In previous articles [3–5], the former literature has been extensively reviewed with respect to a number of chemical, biological, metabolic, functional as well as historical aspects. Since 1982 [4,5], the proliferation of the literature on the chemistry, biochemistry and (molecular) biology of sialic acids has accelerated dramatically, and several short reviews dealing with specific aspects of sialic acids have appeared [6–12].

The aim of this chapter is to collate a mixture of, what is in our opinion, relevant information published before 1982 and new data appeared since then. Because of the profusion of biochemical and biological data published in the last few years, it was necessary to select those reports which, we believe, reflect potential trends in the future development of sialobiology?

2. General characteristics of sialic acid

In Table 1 a survey of the 43 naturally occurring members of the sialic acid family [13– 51], together with their abbreviations and typical biological sources, is presented. The general name "sialic acid" is derived from the Greek "sialos", meaning saliva. Taking into account the Rules for Carbohydrate Nomenclature, as recommended by the IUPAC-IUBMB Joint Commission on Biochemical Nomenclature (1995), the mother-molecule neuraminic acid, which does not occur in free form in nature, due to its immediate cyclization to form an internal Schiff base, is systematically named 5-amino-3,5-dideoxy-D-glycero-Dgalacto-non-2-ulosonic acid (Fig. 1), and abbreviated as Neu, whereby the D-notation is implied in the trivial name. Chemically, this nine-carbon-containing monosaccharide is a 2-keto-carboxylic acid, a deoxysugar, and an aminosugar. The amino group is generally *N*-acetylated (5-acetamido-3,5-dideoxy-D-glycero-D-galacto-non-2-ulopyranosonic acid;

Name	Abbreviation	Typical biological sources	Reference(s)
Neuraminic acid	Neu	does not occur in free form; only found in gangliosides	[13,14]
N-Acetylneuraminic acid	Neu5Ac	oligosaccharides, polysaccharides and glycoconjugates from man, higher animals, and microorganisms; free form in body fluids	[13,15–18]
5-N-Acetyl-4-O-acetyl-neuraminic acid	Neu4,5Ac ₂	ungulate, monotreme and guinea pig oligosaccharides and glycoconjugates	[15,16,18–21]
5-N-Acetyl-7-O-acetyl-neuraminic acid	Neu5,7Ac ₂	animal glycoconjugates; bacterial polysaccharides	[15,16,18,22-24]
5-N-Acetyl-8-O-acetyl-neuraminic acid	Neu5,8Ac ₂	bovine glycoproteins; bacterial polysaccharides	[15,22–24]
5-N-Acetyl-9-O-acetyl-neuraminic acid	Neu5,9Ac ₂	human and animal glycoconjugates; bacterial (lipo)polysaccharides	[15,16,18,22-26]
5-N-Acetyl-4,9-di-O-acetyl-neuraminic acid	Neu4,5,9Ac ₃	equine glycoproteins	[15]
5-N-Acetyl-7,9-di-O-acetyl-neuraminic acid	Neu5,7,9Ac ₃	bovine glycoproteins and bacterial lipopolysaccharides	[15,16,18,22,23, 26,27]
5-N-Acetyl-8,9-di-O-acetyl-neuraminic acid	Neu5,8,9Ac ₃	bovine glycoproteins	[15,16,18,22,23]
5-N-Acetyl-7,8,9-tri-O-acetyl-neuraminic acid	Neu5,7,8,9Ac4	bovine glycoproteins	[15,22]
5-N-Acetyl-9-O-L-lactyl-neuraminic acid	Neu5Ac9Lt	human and animal glycoproteins; free form in body fluids	[15,28,29]
5-N-Acetyl-4-O-acetyl-9-O-lactyl-neuraminic acid	Neu4,5Ac ₂ 9Lt	equine glycoproteins	[15,30]

 Table 1

 Survey of established structures of naturally occurring members of the sialic acid family^a

continued on next page 5

Table 1, continued			
Name	Abbreviation	Typical biological sources	Reference(s)
5-N-Acetyl-8-O-methyl-neuraminic acid	Neu5Ac8Me	starfish glycoconjugates	[15,31]
5-N-Acetyl-9-O-acetyl-8-O-methyl-neuraminic acid	Neu5,9Ac ₂ 8Me	starfish glycoconjugates	[31]
5-N-Acetyl-8-O-sulpho-neuraminic acid	Neu5Ac8S	sea urchin glycolipids	[32]
5-N-Acetyl-9-O-phosphoro-neuraminic acid	Neu5Ac9P	biosynthetic intermediate to Neu5Ac	[33]
5-N-Acetyl-2-deoxy-2,3-didehydro-neuraminic acid	Neu2en5Ac	body fluids and tissues	[15-17,34]
5-N-Acetyl-9-O-acetyl-2-deoxy-2,3-didehydro-neuraminic acid	Neu2en5,9Ac ₂	urine and tissues	[35]
5-N-Acetyl-2-deoxy-2,3-didehydro-9-O-lactyl-neuraminic acid	Neu2en5Ac9Lt	urine and tissues	[35]
5-N-Acetyl-2,7-anhydro-neuraminic acid	Neu2,7an5Ac	urine, wet cerumen, leech	[36–39]
N-Glycolylneuraminic acid	Neu5Gc	glycoconjugates from most higher animals	[13,15,16,18,25, 40]
4-O-Acetyl-5-N-glycolyl-neuraminic acid	Neu4Ac5Gc	ungulate glycoconjugates	[15,16,18]
7-O-Acetyl-5-N-glycolyl-neuraminic acid	Neu7Ac5Gc	glycoconjugates from most higher animals	[15,22]
8-O-Acetyl-5-N-glycolyl-neuraminic acid	Neu8Ac5Gc	bovine glycoproteins	[36]
9-O-Acetyl-5-N-glycolyl-neuraminic acid	Neu9Ac5Gc	glycoconjugates from most higher animals	[15,16,18,22]
7,9-Di-O-acetyl-5-N-glycolyl-neuraminic acid	Neu7,9Ac ₂ 5Gc	bovine glycoproteins	[15,22]
8,9-Di-O-acetyl-5-N-glycolyl-neuraminic acid	Neu8,9Ac ₂ 5Gc	bovine glycoproteins	[15,22]
7,8,9-Tri-O-acetyl-5-N-glycolyl-neuraminic acid	Neu7,8,9Ac ₃ 5Gc	bovine glycoproteins	[15,22]
5-N-glycolyl-9-O-lactyl-neuraminic acid	Neu5Gc9Lt	porcine glycoproteins	[36]
5-N-glycolyl-8-O-methyl-neuraminic acid	Neu5Gc8Me	starfish glycoconjugates	[15,31]
9-O-Acetyl-5-N-glycolyl-8-O-methyl-neuraminic acid	Neu9Ac5Gc8Me	starfish glycoconjugates	[31]
7,9-Di-O-acetyl-5-N-glycolyl-8-O-methyl-neuraminic acid	Neu7,9Ac ₂ 5Gc8Me	starfish glycoconjugates	[41]

continued on next page

Table 1, continued			
Name	Abbreviation	Typical biological sources	Reference(s)
5-N-Glycolyl-8-O-sulpho-neuraminic acid	Neu5Gc8S	sea urchin glycolipids	[42]
N-(O-Acetyl)glycolylneuraminic acid	Neu5GcAc	rat thrombocytes	[43]
N-(O-Methyl)glycolylneuraminic acid	Neu5GcMe	starfish glycolipids	[44]
2-Deoxy-2,3-didehydro-5-N-glycolyl-neuraminic acid	Neu2en5Gc	body fluids and tissues	[45]
9-O-Acetyl-2-deoxy-2,3-didehydro-5-N-glycolyl-neuraminic acid	Neu2en9Ac5Gc	urine	[35]
2-Deoxy-2,3-didehydro-5-N-glycolyl-9-O-lactyl-neuraminic acid	Neu2en5Gc9Lt	urine and tissues	[35]
2-Deoxy-2,3-didehydro-5-N-glycolyl-8-O-methyl-neuraminic acid	Neu2en5Gc8Me	starfish	[35]
2,7-Anhydro-5-N-glycolyl-neuraminic acid	Neu2,7an5Gc	rat urine	[36,38]
2,7-Anhydro-5-N-glycolyl-8-O-methyl-neuraminic acid	Neu2,7an5Gc8Me	starfish	[36]
2-Keto-3-deoxynononic acid	Kdn	fish, amphibian and mammalian glycoconjugates; bacterial polysaccharides	[46–50]
9-O-Acetyl-2-keto-3-deoxynononic acid	Kdn9Ac	fish glycoconjugates	[51]

^a If possible, typical biological sources have been indicated. In the case of a sialic acid structure being proven by mass spectrometry and/or NMR spectroscopy, reference numbers refer in general to such studies or to review articles including this sialic acid.



Fig. 1. Chemical structures for simple sialic acids in different views. (a) 5-amino-3,5-dideoxy-D-glycero-D-galacto-non-2-ulosonic acid (Neu, open chain, Fischer projection formula); (b) 5-acetamido-3,5-dideoxy-D-glycero- α -D-galacto-non-2-ulopyranosonic acid (α -Neu5Ac, Fischer projection formula, note that C7 is the anomeric reference atom); (c) α -Neu5Ac (Haworth projection formula); (d) α -Neu5Ac (${}^{2}C_{5}$ chair conformation); (e) 3-deoxy-D-glycero- β -D-galacto-non-2-ulopyranosonic acid (β -Kdn, ${}^{2}C_{5}$ chair conformation). Note that the D-notation is part of the trivial name.



Fig. 2. Chemical structure of cytidine 5'-(5-acetamido-3,5-dideoxy-D-glycero-β-D-galacto-non-2-ulopyranosylonate monophosphate (CMP-β-Neu5Ac).

N-acetylneuraminic acid; Neu5Ac) (Fig. 1) or N-glycolylated (5-hydroxyacetamido-3,5-dideoxy-D-glycero-D-galacto-non-2-ulopyranosonic acid; N-glycolylneuraminic acid; Neu5Gc). As is evident from Table 1, the hydroxyl groups may be free, esterified (acetylated, lactylated, sulfated, phosphorylated) or etherified (methylated). In the case where a sialic acid bears a hydroxy group instead of an amino group at C5, so-called deaminated neuraminic acid, it is systematically named 3-deoxy-D-glycero-D-galacto-non-2-ulopyranosonic acid (Fig. 1), and is abbreviated as Kdn (2-keto-3-deoxynononic acid; also in this case the D-notation is part of the trivial name). Sialic acids are relatively strong acids, e.g. Neu5Ac has a pK_a value found in the range of 2.2–3.0 in various studies with an average of 2.6. This strong acidity is responsible for processes such as autohydrolysis of sialic-acid-containing carbohydrate chains. Conformationally, sialic acids adopt the ${}^{2}C_{5}$ chair conformation, having the glycerol side chain in an equatorial orientation [52]. Due to hydrogen bonding of HO7 and HO8, leading to a trans-orientation of these groups, the glycerol side chain is not as flexible as one might expect. This has consequences for the course of the mild periodate oxidative degradation of this moiety in the case of substitution at C9 [53]. Free sialic acids have mainly the β -anomeric ring structure (>93%), whereas glycoconjugate-bound sialic acids occur specifically in the α -anomeric form [54]. In nucleotide-bound sialic acids, i.e. CMP-Neu5Ac (Fig. 2), a β-configuration for the glycosidic bond is present [55]. Crystalline Neu5Ac occurs specifically in the β-anomeric form [56].

With respect to *O*-acetylation patterns in sialic acids, it is worthwhile to mention that spontaneous migration of *O*-acetyl groups can occur between C7, C8 and C9. At pH values at which no significant de-*O*-acetylation is observed (e.g. physiological pH values), Neu5,7Ac₂ can readily transform into Neu5,9Ac₂, whereas Neu5,7,9Ac₃ yields an equilibrium of Neu5,7,9Ac₃ and Neu5,8,9Ac₃ in a molar ratio of approximately 1:1; Neu4,5Ac₂ does not give rise to *O*-acetyl migrations [23]. Also starting from α -Neu5,8,9Ac₃ 4-aminophenylthio-glycoside a 1:1 equilibrium between the 8,9- and 7,9-di-*O*-acetylated derivatives is established [57]. 9-*O*-Acetylated *N*-acylneuraminic acids have been found in both glycoconjugates and oligo/polysaccharides of different



(a)



(b)

Fig. 3. Chemical structures of (a) 5-N-acetyl-2-deoxy-2,3-didehydro-neuraminic acid (Neu2en5Ac); (b) 5-N-acetyl-2,7-anhydro-neuraminic acid (Neu2,7an5Ac); and (c) the tautomers of 5-N-acetyl-4,8-anhydro-neuraminic acid (Neu4,8an5Ac).

biological origin (Table 1). The same is true for 7-O-acetylated N-acylneuraminic acids. It should be emphasized that Neu5,9Ac₂ and other side-chain-O-acylated sialic acids occur in man [58] (see also sections 8.4.2 and 8.4.3), e.g. in human colon. Ungulates form the major source for 4-O-acetylated N-acylneuraminic acids [13], whereas minor sources are monotremes [19], guinea pigs [21] and humans [59]. In an evaluation of the naturally occurring O-acylation patterns, it is evident that O-acyl groups are most frequently found at C9 of both Neu5Ac and Neu5Gc. O-Methylated and O-sulfated N-acylneuraminic acids have only been found in lower animals, e.g. in echinoderms [13,31,42].

5-N-Acyl-2-deoxy-2,3-didehydro-neuraminic acids like Neu2en5Ac (5-acetamido-2,6anhydro-3,5-dideoxy-D-glycero-D-galacto-non-2-enonic acid, Fig. 3) occur in free form in nature. Moreover, they can be generated from corresponding CMP-N-acylneuraminic acids in a non-enzymatic elimination reaction, occurring under physiological and, much faster, under alkaline conditions [34,35]. Small amounts of Neu2en5Ac are formed by a water elimination side reaction from Neu5Ac during influenza-B-virus-sialidase-catalyzed desialylations of sialoglycoconjugates [60].

5-N-Acetyl-2,7-anhydro-neuraminic acid (Neu2,7an5Ac, Fig. 3), which occurs in free form in nature [36,37], can be generated from Neu5Ac(α 2-3)Gal(β 1- containing glycoconjugates using a sialidase isolated from the leech *Macrobdella decora* [38,61].

It is clear that such a sialidase has a highly unusual specificity (see sections 5.1, 8.4.6 and 9.2). The methyl ester of Neu2,7an5Ac is formed as a very minor by-product during methanolysis [62].

5-N-Acetyl-4,8-anhydro-neuraminic acid (Neu4,8an5Ac, Fig. 3), having CH₂-CO-COOH as a side chain of a pyranose ring with the ⁷C₄ conformation, does not occur as such in nature. Initially, it has been isolated from an acid hydrolysate of edible bird's nest substance [63]. Furthermore, Neu4,8an5Ac was detected in a sialic acid mixture obtained after acid hydrolysis of equine serum, whereas incubation of Neu4,5Ac₂ under alkaline conditions showed, in addition to de-*O*-acetylation, a partial conversion into Neu4,8an5Ac [41]. Heating of solid sodium Neu5Ac (3 h, 140°C) or a solution of Neu5Ac at pH > 11 or at pH 2.0 (30 min, 80°C) also yielded Neu4,8an5Ac (sodium salt) as a major degradation product, probably in two tautomeric forms [64].

Hyperexcretion of free Neu5Ac in urine, defined as sialuria, has been observed in several mentally retarded patients [65–68] (see section 10.5).

3. Occurrence of sialic acids in biomolecules

Members of the sialic acid family occur mainly in bound form in higher animals, from the echinoderms onwards in evolution, but also in some viruses, various bacteria, protozoa, and pathogenic fungi [8,13,69]. Generally, they are constituents of glycoproteins [20,21,25,40,46–48,51,70–107] (Table 2), glycolipids [13,31,44,108–124] (Table 3), and oligosaccharides [16,19,25,76,125–155] (Table 4), usually occurring as terminal monosaccharide units, and of homo- and heteropolysaccharides [24,26,27,49,156–185] (Table 5). However, glycosylphosphatidylinositol membrane anchors have also been shown to contain sialic acid [186]. Neu5Ac- and Neu5Gc-containing glycans frequently occur together, whereby ratio differences reflect species specificity, tissue specificity or physiological fluid specificity. Also Kdn has been found together with Neu5Ac/5Gc in specific glycoconjugates (Table 2). It should be noted that the presence of Neu5Gc in normal human tissue and soluble glycoproteins has not been established conclusively [13,187,188].

As is evident from Table 2, Neu5Ac/5Gc- and Kdn-containing elements in Nand O-glycoprotein glycans do occur in many different microenvironments, however, with a restricted glycosidic linkage specificity. In general, N-acylneuraminic acids are α -2,3- or α -2,6-linked with D-galactose (Gal), α -2,3- or α -2,6-linked with N-acetyl-Dgalactosamine (GalNAc), α -2,6-linked with N-acetyl-D-glucosamine (GlcNAc), or α -2,8linked with other N-acylneuraminic acids. These types of glycosidic linkages have been firmly established by different analytical methods, including NMR spectroscopy [76,77]. The sialic-acid-containing N-linked carbohydrate chains form part of the complex (Nacetyllactosamine and N,N'-diacetyllactosediamine subtypes; mono-, di-, tri-, tri'-, and tetraantennae) or the hybrid type of structures [189]. For the mucin-type O-glycans, the N-acylneuraminic-acid-containing sequences are extensions of most of the wellestablished core types [190]. In addition, terminal Neu5Ac(α 2–4)Gal [75], terminal Neu5Ac(α 2–4)GlcNAc [96], and internal Fuc(1–4)Neu5Gc [97] sequences have been reported, and only one example of a terminal Neu5Ac(α 2–9)Neu5Ac(α 2–0 dimer has

Table 2	
Sialic-acid-containing elements in N- and	O-glycoproteins ^a

Partial structure	N	0	Ref(s).
Neu5Ac(α 2-3)Gal(β 1-3)GlcNAc(β 1-	+		[70,71]
$Neu5Ac(\alpha 2-3)Gal(\beta 1-3)[Neu5Ac(\alpha 2-6)]GlcNAc(\beta 1-2)[Neu5Ac(\alpha 2-6)]GlcNAc(\alpha 2-6)]GlcNAc(\alpha 2-6)[Neu5Ac(\alpha 2-6)]GlcNAc(\alpha 2-6)]GlcNAc(\alpha 2-6)[Neu5Ac(\alpha 2-6)]GlcNAC(\alpha 2-6)]GlcNAC(\alpha 2-6)[Neu5Ac(\alpha 2-6)]GlcN$	+		[70,72,73]
Neu5Ac(α 2–3)Gal(β 1–3)[Fuc(α 1–4)]GicNAc(β 1–		+	[74]
Neu5Ac(α 2–4)Gal(β 1–3)[Neu5Ac(α 2–6)]GlcNAc(β 1-	+		[75]
Gal(β1–3)[Neu5Ac(α2–6)]GlcNAc(β1-	+		[70,73]
Neu5Ac(α2–3)Gal(β1–4)GlcNAc(β1-	+	+	[76,77]
Neu5Ac(α 2-3)Gal(β 1-4)[Fuc(α 1-3)]GlcNAc(β 1-	+	+	[77,78]
Neu5Ac(α 2-3)[Gal(β 1-4)]Gal(β 1-4)GlcNAc(β 1-	+		[79,80]
$Neu5Ac(\alpha 2-3)[GalNAc(\beta 1-4)]Gal(\beta 1-4)GlcNAc(\beta 1-4)GlcN$	+	+	[77,81]
Neu5Ac(α 2-3)[6S]Gal(β 1-4)GlcNAc(β 1-	+		[82]
Neu4,5Ac ₂ (α 2-3)Gal(β 1-4)GlcNAc(β 1-	+		[83]
Neu4,5Ac ₂ (α 2-3)[Gal(β 1-4)]Gal(β 1-4)GlcNAc(β 1-	+		[79]
Neu5,9Ac ₂ (α 2-3)Gal(β 1-4)GlcNAc(β 1-	+		[25,84]
Neu5Gc(α2–3)Gal(β1–4)GlcNAc(β1-	+		[25,84]
Neu5Ac(α 2–4)Gal(β 1–4)GlcNAc(β 1-	+		[75]
Neu5Ac(α2–6)Gal(β1–4)GlcNAc(β1-	+	+	[76,77]
Neu5Ac(α 2–6)Gal(β 1–4)[6S]GlcNAc(β 1-	+		[85]
$Neu5Ac(\alpha 2-6)[GalNAc(\alpha 1-3)]Gal(\beta 1-4)GlcNAc(\beta 1-6)[GalNAc(\alpha 1-3)]Gal(\beta 1-6)[GalNAc(\alpha 1-6)][GalNAc(\alpha 1-6)][Ga$		+	[86]
Neu4,5Ac ₂ (α 2–6)Gal(β 1–4)GlcNAc(β 1-	+		[20,21,87,88]
Neu5Gc(α2–6)Gal(β1–4)GlcNAc(β1-	+		[40]
$Neu5Ac(\alpha 2-3){Gal(\beta 1-4)}_{1-2}Gal(\beta 1-4)GlcNAc(\beta 1-4)G$	+		[80,89]
$\pm Neu5Ac(\alpha 2-3)Gal(\beta 1-4)[\pm Neu5Ac(\alpha 2-3)]Gal(\beta 1-4)[\pm Fuc(\alpha 1-3)]GlcNAc(\beta 1-3)]GlcN$	+		[90]
Neu5Ac($\alpha 2$ -3)Gal($\beta 1$ -3)GalNAc($\alpha 1$ -		+	[77]
Neu5Ac(α 2–3)Gal(β 1–3)GalNAc(α 1–O)Ser/Thr		+	[77]
$Neu5Ac(\alpha 2-8)Neu5Ac(\alpha 2-3)Gal(\beta 1-3)GalNAc(\alpha 1-0)Ser/Thr$		+	[91]
$Neu5Ac(\alpha 2-3)[GalNAc(\beta 1-4)]Gal(\beta 1-3)GalNAc(\alpha 1-0)Ser/Thr$		+	[77]
Neu5Gc(α 2–3)Gal(β 1–3)GalNAc(α 1–O)Ser/Thr		+	[77]
Neu5Ac(α2–3)GalNAc(β1–4)GlcNAc(β1-	+		[92]
Neu5Ac($\alpha 2$ -6)GalNAc($\beta 1$ -4)GlcNAc($\beta 1$ -	+		[93,94]
$Neu5Ac(\alpha 2-3)[GalNAc(\beta 1-4)]GalNAc(\beta 1-3)Gal(\beta 1-4)Gal(\beta 1-4)Gag$		+	[95]
$Neu5Gc(\alpha 2-3)[GalNAc(\beta 1-4)]GalNAc(\beta 1-3)Gal(\beta 1-4)Gal(\beta 1-4)Gag$		+	[77]
Neu5Ac(α 2–6)GalNAc(α 1–O)Ser/Thr		+	[77]
Neu5Gc(α 2–6)GalNAc(α 1–O)Ser/Thr		+	[77]
Neu5Ac(α 2–4)GlcNAc(β 1–3)[Neu5Ac(α 2–6)]Gal(β 1–		+	[96]
-Fuc(1-4)Neu5Gc(2-		+	[97]
Neu5Ac(α2–9)Neu5Ac(α2–3/6)Gal(β1–4)GlcNAc(β1-	+		[98]
${\text{Neu5Ac}(\alpha 2-8)}_n \text{Neu5Ac}(\alpha 2-3) \text{Gal}(\beta 1-4) \text{GlcNAc}(\beta 1-4)$	+		[99]
${\text{Neu5Ac}(\alpha 2-8)}_n \text{Neu5Ac}(\alpha 2-6) \text{GalNAc}(\alpha 1-O) \text{Ser}/\text{Thr}$		+	[77]

continued on next page

Partial structure	N	0	Ref(s).
$\{\operatorname{Neu5Gc}(\alpha 2-8)\}_n \operatorname{Neu5Gc}(\alpha 2-6) \operatorname{GalNAc}(\alpha 1-0) \operatorname{Ser/Thr}$		+	[77]
$\{\text{Neu5Ac}(\alpha 2-8)\}_n$ - with 4Ac, 7Ac, 9Ac or 9Lt		+	[100]
${\text{Neu5Gc}(\alpha 2-8)}_n$ - with 4Ac, 7Ac or 9Ac		+	[51,100]
$\{\text{Neu5Ac/5Gc}(\alpha 2-8)\}_n$ - with 4Ac, 7Ac or 9Ac		+	[100]
${\operatorname{Neu5Gc}(\alpha 2-O5)}_n$ -		+	[101]
$Kdn(\alpha 2-3)Gal(\beta 1-4)GlcNAc(\beta 1-$	+		[102]
$Kdn(\alpha 2-3)Gal(\beta 1-3)GalNAc(\alpha 1-3)GalNAc(\alpha 1-0)Ser/Thr$		+	[47]
$Kdn(\alpha 2-3)[GalNAc(\beta 1-4)]GalNAc(\beta 1-3)Gal(\beta 1-4)Gal(\beta $		+	[77]
$Kdn(\alpha 2-6)GalNAc(\alpha 1-O)Ser/Thr$		+	[48,77,103]
$\{Kdn(\alpha 2-8)\}_n Kdn(\alpha 2-6) GalNAc(\alpha 1-O) Ser/Thr$		+	[47]
$Fuc(\alpha 1-4)[Fuc(\alpha 1-5)]Kdn(\alpha 2-6)GalNAc(\alpha 1-O)Ser/Thr$		+	[104]
-Fuc(a1-4)Kdn(a2-6)GalNAc(a1-O)Ser/Thr		+	[105]
$Kdn(\alpha 2-6)[Kdn(\alpha 2-3)]GalNAc(\alpha 1-O)Thr$		+	[106]
$Kdn(\alpha 2-3)GalNAc(\alpha 1-O)Thr$		+	[106a]
$Fuc(\alpha 1-4)[{Fuc(\alpha 1-5)}_{0-1}]Kdn(\alpha 2-3)Gal(\beta 1-$		+	[104]
$\{Kdn(\alpha 2-8)\}_n$ - with 4Ac, 7Ac or 9Ac		+	[107]
$Kdn(\alpha 2-8){Neu5Ac(\alpha 2-8)}_n$ - with $\pm Ac$		+	[107]
$Kdn(\alpha 2-8){Neu5Gc(\alpha 2-8)}_n$ - with $\pm Ac$		+	[46]
$Kdn(\alpha 2-8)\{Neu5Ac/5Gc(\alpha 2-8)\}_n$ - with $\pm Ac$		÷	[107]
$Kdn9Ac(\alpha 2-8)\{Neu5Gc(\alpha 2-8)\}_{n}$		+	[51]

Table 2, continued

^a In the case of a specific fragment being established by ¹H NMR spectroscopy, the reference refers to such a study or to a review that includes the fragment. S means sulfate.

been described [98]. In a number of structures not only sialic acid is responsible for the acidic character of the carbohydrate chain, but also sulfate. Here, an unusual example is the Neu5Ac(α 2-3)[6S]Gal(β 1-4)GlcNAc element [82]. The list of sialic-acid-containing sequences, in which *N*-acylneuraminic acid is replaced by Kdn is continuously growing (Table 2). Interestingly, also structural elements occur which have not been found so far for *N*-acylneuraminic acids, e.g. Kdn(α 2-6)[Kdn(α 2-3)]GalNAc(α 1-O)Thr [106].

General structural information with respect to glycoprotein glycans is presented in volume 29a of the New Comprehensive Biochemistry series, and detailed information with respect to poly-*N*-acylneuraminyl-containing glycoproteins and Kdn-containing glycoproteins in the present volume 29b. Several of the sialic-acid-containing sequences present in N- and O-glycoprotein glycans, also occur in glycolipids, milk and urinary oligosaccharides, and in (lipo)polysaccharides of different biological origin. An impression of this overlap in structures can be obtained from an inspection of the structural data in Tables 3–5, summarizing sialic-acid-containing elements of glycolipids, structures of milk and urinary sialo-oligosaccharides, and structures or elements of sialic-acid-containing (lipo)polysaccharides, respectively. It is interesting to note that

Table 3 Compilation of sialic-acid-containing fragments in glycolipids^a

Structure	References
Neu5Ac(α 2-3)Gal(β 1-1')Cer	[108]
${\text{Neu5Ac}(\alpha 2-8)}_{0-1}$ Neu5Ac $(\alpha 2-6)$ Glc $(\beta 1-1')$ Cer	[13]
Neu5Gc/5Gc8S(α 2–6)Glc(β 1–1')Cer	[13,109]
Neu5Ac(α 2–6)Glc(β 1–8)Neu5Ac(α 2–6)Glc(β 1–1')Cer	[13]
$Neu5Ac(\alpha 2-8)Neu5Ac(\alpha 2-6)Glc(\beta 1-6)Glc(\beta 1-1')Cer$	[13]
$Neu5Ac/5Gc/4Ac5Gc/5Gc8Me/5GcMe(\alpha 2-3)Gal(\beta 1-4)Glc(\beta 1-1')Cer$	[44,108]
$Neu5Ac/5,7Ac_2/5,9Ac_2/5,7,9Ac_3/5Ac8S/5Gc(\alpha 2-8) Neu5Ac(\alpha 2-3)Gal(\beta 1-4)Glc(\beta 1-1')Cer$	[13,108,110]
Neu5Ac/5Gc($\alpha 2$ -8)Neu5Gc($\alpha 2$ -3)Gal($\beta 1$ -4)Glc($\beta 1$ -1')Cer	[108]
$Neu5Ac(\alpha 2-8)Neu5Ac(\alpha 2-8)Neu5Ac(\alpha 2-3)Gal(\beta 1-4)Glc(\beta 1-1')Cer$	[108]
$Neu5,9Ac_2(\alpha 2-8)Neu5Ac(\alpha 2-8)Neu5Ac(\alpha 2-3)Gal(\beta 1-4)Glc(\beta 1-1')Cer$	[111]
$GalNAc(\beta 1-4)[Neu5Ac/5Ac8S/5Gc/5Gc8S(\alpha 2-3)]Gal(\beta 1-4)Glc(\beta 1-1')Cer$	[108]
$Gal(\beta 1-3)GalNAc(\beta 1-4)[Neu(\alpha 2-3)]Gal(\beta 1-4)Glc(\beta 1-1')Cer$	[112]
$[R-3]GalNAc(\beta 1-4)[Neu5Ac/5Gc(\alpha 2-3)]Gal(\beta 1-4)Glc(\beta 1-1')Cer^{b}$	[108]
$[R-3]GalNAc(\beta 1-4)[\{Neu5Ac(\alpha 2-8)\}_{1-2}Neu5Ac(\alpha 2-3)]Gal(\beta 1-4)Glc(\beta 1-1')Cer^{c}$	[108]
$[R-3]GalNAc(\beta 1-4)[Neu5,9Ac_2(\alpha 2-8)\{Neu5Ac(\alpha 2-8)\}_{0-1}Neu5Ac(\alpha 2-3)]Gal(\beta 1-4)Glc(\beta 1-1')Cer^{c}$	[108,113]
$[R-3]GalNAc(\beta 1-4)[\{Neu5Gc(\alpha 2-8)\}_{1-2}Neu5Gc(\alpha 2-3)]Gal(\beta 1-4)Glc(\beta 1-1')Cer^{b}$	[108]
$[R-3]GalNAc(\beta 1-4)[Neu5Ac(\alpha 2-8)Neu5Gc(\alpha 2-3)]Gal(\beta 1-4)Glc(\beta 1-1')Cer^{b}$	[114]
$Ara(1-6)Gal(\beta 1-4)[\{Gal(\beta 1-8)\}_{0-1}]Neu5Gc/5Gc8Me(\alpha 2-3)Gal(\beta 1-4)Glc(\beta 1-1')Cer$	[13]
Neu5Gc8Me(2-3)GalNAc(1-3)Gal(1-4)Glc(1-1')Cer	[115]
$Neu5Gc8Me(\alpha 2-3)[Neu5Gc8Me(\alpha 2-6)]GalNAc(\beta 1-3)Gal(\beta 1-4)Glc(\beta 1-1')Cer$	[44,115]
Neu5Ac/5,9Ac ₂ /5Gc(α 2–3)Gal(β 1–3)GalNAc(β 1-	[108]
${Neu5Ac(\alpha 2-8)}_{1-2}Neu5Ac(\alpha 2-3)Gal(\beta 1-3)GalNAc(\beta 1-$	[108,116]
Neu5Gc(a2-8)Neu5Gc(a2-3)Gal(b1-3)GalNAc(b1-	[117]
$Neu5Ac(\alpha 2-3)[Neu5Ac(\alpha 2-6)]Gal(\beta 1-3)GalNAc(\beta 1-6)]Gal(\beta 1-6)]G$	[108]

continued on next page

Structure	References
Neu5Ac(α2–3)[[R-3]GalNAc(β1–4)]Gal(β1–3)GalNAc(β1- ^c	[108]
Neu5Gc(α2–3)[GalNAc(β1–4)]Gal(β1–3)GalNAc(β1-	[108]
$\{\text{Neu5Ac}(\alpha 2-8)\}_{0-1}\text{Neu5Ac}(\alpha 2-3)\text{Gal}(\beta 1-3)[\{\text{Neu5Ac}(\alpha 2-8)\}_{0-1}\text{Neu5Ac}(\alpha 2-6)]\text{GalNAc}(\beta 1-3)]$	[108]
$Neu5Ac(\alpha 2-3)Gal(\beta 1-3)[Neu5Ac(\alpha 2-6)]GalNAc(\beta 1-4)[\{Neu5Ac(\alpha 2-8)\}_{0-1}Neu5Ac(\alpha 2-3)]Gal(\beta 1-4)Glc(\beta 1-1')Cer(\beta 1-1)]Gal(\beta 1-4)Glc(\beta 1-1)]Gal(\beta 1-4)Glc(\beta 1-1)]Gal(\beta 1-4)Glc(\beta 1-1)]Gal(\beta 1-4)[\{Neu5Ac(\alpha 2-8)\}_{0-1}Neu5Ac(\alpha 2-3)]Gal(\beta 1-4)Glc(\beta 1-1)]Gal(\beta 1-4)Glc(\beta 1-4)[\{Neu5Ac(\alpha 2-8)\}_{0-1}Neu5Ac(\alpha 2-3)]Gal(\beta 1-4)Glc(\beta 1-1)]Gal(\beta 1-4)[\{Neu5Ac(\alpha 2-8)\}_{0-1}Neu5Ac(\alpha 2-3)]Gal(\beta 1-4)Glc(\beta 1-1)]Gal(\beta 1-4)Glc(\beta 1-4)[\{Neu5Ac(\alpha 2-8)\}_{0-1}Neu5Ac(\alpha 2-3)]Gal(\beta 1-4)Glc(\beta 1-1)]Gal(\beta 1-4)Glc(\beta 1-1)]Gal(\beta 1-4)Glc(\beta 1-4)[Ac(\alpha 2-8)]Gal(\beta 1-4)Glc(\beta 1-4)]Gal(\beta 1-4)Glc(\beta 1-4)[Ac(\alpha 2-8)]Gal(\beta 1-4)Glc(\beta 1-4)]Gal(\beta 1-4)Glc(\beta 1-4)[Ac(\alpha 2-8)]Gal(\beta 1-4)Glc(\beta 1-4)]Gal(\beta 1-4)[Ac(\alpha 2-8)]Gal(\beta 1-4)Glc(\beta 1-4)[Ac(\alpha 2-8)]Gal(\beta 1-4)Glc(\beta 1-4)]Gal(\beta 1-4)[Ac(\alpha 2-8)]Gal(\beta 1-4)Glc(\beta 1-4)[Ac(\alpha 2-8)]Gal(\beta 1-4)Glc(\beta 1-4)]Gal(\beta 1-4)[Ac(\alpha 2-8)]Gal(\beta 1-4)[Ac(\alpha 2-8)]Gal(\alpha 2-8)]Gal(\alpha 2-8)[Ac(\alpha 2-8)]Gal(\alpha 2-8)$	[118,119]
Neu5Ac(α 2–6)Gal(β 1–4)GlcNAc(β 1-	[108]
Neu5Ac/5Gc($\alpha 2$ -3)Gal($\beta 1$ -4)GlcNAc($\beta 1$ -	[108]
${\rm Neu5Ac}(\alpha 2-8){\rm I}_{1-2}{\rm Neu5Ac}(\alpha 2-3){\rm Gal}(\beta 1-4){\rm GlcNAc}(\beta $	[108]
$Neu5Ac/5Gc(\alpha 2-8)Neu5Gc(\alpha 2-3)Gal(\beta 1-4)GlcNAc(\beta 1-4)GlcAc(\beta 1-4$	[120]
$Neu5Ac(\alpha 2-3)[GalNAc(\beta 1-4)]Gal(\beta 1-4)GlcNAc(\beta 1-4)GlcN$	[108]
Neu5Ac($\alpha 2$ -3)Gal($\beta 1$ -4)[Fuc($\alpha 1$ -3)]GlcNAc($\beta 1$ -	[108]
Neu5Ac(α 2-3)Gal(β 1-3)GlcNAc(β 1-	[108]
Neu5Ac(α 2–3)[GalNAc(β 1–4)]Gal(β 1–3)GlcNAc(β 1–	[121]
$Neu5Ac(\alpha 2-3)Gal(\beta 1-3)[Fuc(\alpha 1-4)]GlcNAc(\beta 1-4)]$	[108]
$Neu5Ac(\alpha 2-3)Gal(\beta 1-3)[Neu5Ac(\alpha 2-6)][R-4]GlcNAc(\beta 1-c)$	[108]
Neu5Ac(α 2-3)GalNAc(β 1-3)Gal(α 1-	[108]
Neu5Ac(α 2-3)GalNAc(β 1-3)Gal(β 1-	[108]
$\{\operatorname{Neu5Gc8Me}(\alpha 2-\operatorname{O5})\}_n$ -	[31]
$Kdn(\alpha 2-3)Gal(\beta 1-4)Glc(\beta 1-1')Cer$	[122]
$[R-3]GalNAc(\beta 1-4)[Kdn(\alpha 2-3)]Gal(\beta 1-4)Glc(\beta 1-1')Cer^{b}$	[123,124]
$Kdn/Kdn9Ac(\alpha 2-3)Gal(\beta 1-3)GalNAc(\beta 1-$	[123,124]
$Kdn(\alpha 2-3)Gal(\beta 1-3)[Neu5Ac/Kdn(\alpha 2-6)]GalNAc(\beta 1-$	[124]

Table 3, continued

^a S, sulfate; Cer, ceramide. ^b R, saccharide.

^c R, H or saccharide.

Table 4 Sialic-acid-containing human milk oligosaccharides

Structures	References
Neu5Ac(α 2–3)Gal(β 1–4)Glc ^{a–e}	[16,125–127]
Neu5Ac(α 2–6)Gal(β 1–4)Glc ^{a,b}	[16,125-127]
Neu5Ac(α 2–3)Gal(β 1–3)GlcNAc(β 1–3)Gal(β 1–4)Glc	[125,126,128,129]
$Neu5Ac(\alpha 2-3)Gal(\beta 1-3)[Neu5Ac(\alpha 2-6)]GlcNAc(\beta 1-3)Gal(\beta 1-4)Glc^{a,f}$	[125,128-130]
Neu5Ac(α2-3)Gal(β1-3)[Fuc(α1-4)]GlcNAc(β1-3)Gal(β1-4)Glc	[128,131,132]
$Neu5Ac(\alpha 2-3)Gal(\beta 1-3)[Fuc(\alpha 1-4)][Neu5Ac(\alpha 2-6)]GlcNAc(\beta 1-3)Gal(\beta 1-4)Glc$	[129,130]
Neu5Ac(α 2–3)[Fuc(α 1–2)]Gal(β 1–3)GlcNAc(β 1–3)Gal(β 1–4)Glc	[133]
Gal(β 1-3)[Neu5Ac(α 2-6)]GlcNAc(β 1-3)Gal(β 1-4)Glc ^a	[125,126,128]
$Gal(\beta 1-3)[Neu5Ac(\alpha 2-4)]GlcNAc(\beta 1-3)Gal(\beta 1-4)Glc$	[134]
Fuc(α1-2)Gal(β1-3)[Neu5Ac(α2-6)]GlcNAc(β1-3)Gal(β1-4)Glc	[128,131,135]
Neu5Ac(α 2–6)Gal(β 1–4)GlcNAc(β 1–3)Gal(β 1–4)Glc ^a	[125,126,128]
$Neu5Ac(\alpha 2-3)Gal(\beta 1-3)GlcNAc(\beta 1-3){Gal(\beta 1-4)[Fuc(\alpha 1-3)]GlcNAc(\beta 1-6)}Gal(\beta 1-4)Glc$	[136]
Neu5Ac(α2-3)Gal(β1-3)GlcNAc(β1-3)[Neu5Ac(α2-6)Gal(β1-4)GlcNAc(β1-6)]Gal(β1-4)Glc	[129]
Neu5Ac(α2-3)Gal(β1-3)[Neu5Ac(α2-6)]GlcNAc(β1-3)[Gal(β1-4)GlcNAc(β1-6)]Gal(β1-4)Glc	[129]
Neu5Ac(α2–3)Gal(β1–3)[Neu5Ac(α2–6)]GlcNAc(β1–3)[Fuc(α1–3)Gal(β1–4)GlcNAc(β1–6)]Gal(β1–4)Glc	[137]
Neu5Ac(α2–3)Gal(β1–3)[Neu5Ac(α2–6)]GlcNAc(β1–3){Gal(β1–4)[Fuc(α1–3)]GlcNAc(β1–6)}Gal(β1–4)Glc	[129,130,137]
Neu5Ac(α2–3)Gal(β1–3)[Neu5Ac(α2–6)]GlcNAc(β1–3)[Neu5Ac(α2–6)Gal(β1–4)GlcNAc(β1–6)]Gal(β1–4)Glc	[138]
Neu5Ac(α2–3)Gal(β1–3)[Fuc(α1–4)]GlcNAc(β1–3)[Gal(β1–4)GlcNAc(β1–6)]Gal(β1–4)Glc	[139]
$Neu5Ac(\alpha 2-3)Gal(\beta 1-3)[Fuc(\alpha 1-4)]GlcNAc(\beta 1-3){Gal(\beta 1-4)[Fuc(\alpha 1-3)]GlcNAc(\beta 1-6)}Gal(\beta 1-4)GlcAc(\beta 1-6)]Gal(\beta 1-4)GlcAc(\beta 1-6)]Gal(\beta 1-6)$	[139]
Neu5Ac(α2-3)Gal(β1-3)[Fuc(α1-4)]GlcNAc(β1-3)[Neu5Ac(α2-6)Gal(β1-4)GlcNAc(β1-6)]Gal(β1-4)Glc	[139]
$Neu5Ac(\alpha 2-3)Gal(\beta 1-3)[Fuc(\alpha 1-4)]GlcNAc(\beta 1-3)[Gal(\beta 1-4)GlcNAc(\beta 1-3)Gal(\beta 1-4)GlcNAc(\beta 1-6)]Gal(\beta 1-4)GlcNAc(\beta 1-6)]Gal(\beta 1-4)GlcNAc(\beta 1-6)]Gal(\beta 1-4)GlcNAc(\beta 1-6)]Gal(\beta $	[140]
$Neu5Ac(\alpha 2-3)Gal(\beta 1-3)[Fuc(\alpha 1-4)]GlcNAc(\beta 1-3)\{Gal(\beta 1-4)[Fuc(\alpha 1-3)]GlcNAc(\beta 1-3)Gal(\beta 1-4)GlcNAc(\beta 1-6)\}Gal(\beta 1-4)GlcNAc(\beta 1-6)\}Gal(\beta 1-4)GlcNAc(\beta 1-6)\}Gal(\beta 1-4)GlcNAc(\beta 1-6)\}Gal(\beta 1-4)GlcNAc(\beta 1-6)\}Gal(\beta 1-6)Gal(\beta 1-6)$	[141]
$Neu5Ac(\alpha 2-3)Gal(\beta 1-3)[Fuc(\alpha 1-4)]GlcNAc(\beta 1-3)[Gal(\beta 1-3)GlcNAc(\beta 1-3)Gal(\beta 1-4)GlcNAc(\beta 1-6)]Gal(\beta 1-4)GlcNAc(\beta 1-6)]Gal(\beta 1-4)GlcNAc(\beta 1-6)]Gal(\beta 1-4)GlcNAc(\beta 1-6)]Gal(\beta $	[141]
$Neu5Ac(\alpha 2-3)Gal(\beta 1-3)[Fuc(\alpha 1-4)]GlcNAc(\beta 1-3){Gal(\beta 1-3)[Fuc(\alpha 1-4)]GlcNAc(\beta 1-3)Gal(\beta 1-4)GlcNAc(\beta 1-6)}Gal(\beta 1-4)GlcNAc(\beta 1-6)]Gal(\beta 1-4)GlcNAc(\beta 1-6)]Gal(\beta 1-4)GlcNAc(\beta 1-6)]Gal(\beta 1-4)GlcNAc(\beta 1-6)]Gal(\beta 1-6)]Gal$	[141]

continued on next page

Table 4,	continued
----------	-----------

Structures	References
Neu5Ac(α2-3)Gal(β1-3)[Fuc(α1-4)]GlcNAc(β1-3){Gal(β1-3)GlcNAc(β1-3)Gal(β1-4)[Fuc(α1-3)]GlcNAc(β1-6)}Gal(β1-4)Glc	[141]
$Neu5Ac(\alpha 2-3)Gal(\beta 1-3)[Fuc(\alpha 1-4)]GlcNAc(\beta 1-3){Fuc(\alpha 1-2)Gal(\beta 1-3)GlcNAc(\beta 1-3)Gal(\beta 1-4)[Fuc(\alpha 1-3)]GlcNAc(\beta 1-6)}Gal(\beta 1-4)GlcNAc(\beta 1-6)]Gal(\beta 1-6$	[140]
Gal(β1-3)GlcNAc(β1-3)[Neu5Ac(α2-6)Gal(β1-4)GlcNAc(β1-6)]Gal(β1-4)Glc	[125,126]
$Gal(\beta 1-3)[Neu5Ac(\alpha 2-6)]GlcNAc(\beta 1-3){Gal(\beta 1-4)[Fuc(\alpha 1-3)]GlcNAc(\beta 1-6)}Gal(\beta 1-4)GlcAc(\beta 1-6)]Gal(\beta 1-4)GlcAc(\beta 1-6)]Gal(\beta 1-6$	[136]
Gal(β1-3)[Neu5Ac(α2-4)]GlcNAc(β1-3)[Gal(β1-3)GlcNAc(β1-6)]Gal(β1-4)Glc	[134]
Gal(β1-3)[Neu5Ac(α2-4)]GlcNAc(β1-3)[Fuc(α1-2)Gal(β1-3)GlcNAc(β1-6)]Gal(β1-4)Glc	[134]
$Gal(\beta 1-3)[Fuc(\alpha 1-4)]GlcNAc(\beta 1-3)[Neu5Ac(\alpha 2-6)Gal(\beta 1-4)GlcNAc(\beta 1-6)]Gal(\beta 1-4)GlcNAc(\beta 1-6)]Gal(\beta 1-4)GlcNAc(\beta 1-6)]Gal(\beta 1-6)[Gal(\beta 1-6)]Gal(\beta 1-6)[Gal(\beta 1-6)]Gal(\beta 1-6)[Gal(\beta 1-6)]Gal(\beta 1-6)[Gal(\beta 1-6)]Gal(\beta 1-6)]Gal(\beta 1-6)[Gal(\beta 1-6)]Gal(\beta 1-6$	[136]
$Gal(\beta1-3)[Fuc(\alpha1-4)]GlcNAc(\beta1-3)\{Neu5Ac(\alpha2-3)Gal(\beta1-4)[Fuc(\alpha1-3)]GlcNAc(\beta1-6)\}Gal(\beta1-4)GlcAc(\beta1-6)\}Gal(\beta1-4)GlcAc(\beta1-6)]Gal(\beta1-6)]Gal(\beta1-6)[Fuc(\alpha1-6)]Gal(\beta1-6)]Gal(\beta1-6)]Gal(\beta1-6)[Fuc(\alpha1-6)]Gal(\beta1-6)]Gal(\beta1-6)]Gal(\beta1-6)[Fuc(\alpha1-6)]Gal(\beta1-6)]Gal(\beta1-6)]Gal(\beta1-6)[Fuc(\alpha1-6)]Gal(\beta1-6)]Gal(\beta1-6)]Gal(\beta1-6)[Fuc(\alpha1-6)]Gal(\beta1-6)]Gal(\beta1-6)]Gal(\beta1-6)[Fuc(\alpha1-6)]Gal(\beta1-6)]Gal(\beta1-6)]Gal(\beta1-6)[Fuc(\alpha1-6)]Gal(\beta1-6)]Gal(\beta1-6)[Fuc(\alpha1-6)]Gal(\beta1-6)]Gal(\beta1-6)[Fuc(\alpha1-6)]Gal(\beta1-6)]Gal(\beta1-6)]Gal(\beta1-6)[Fuc(\alpha1-6)]Gal(\beta1-6)]Gal(\beta1-6)[Fuc(\alpha1-6)]Gal(\beta1-6)]Gal(\beta1-6)[Fuc(\alpha1-6)]Gal(\beta1-6)]Gal(\beta1-6)[Fuc(\alpha1-6)]Gal(\beta1-6)]Gal(\beta1-6)[Fuc(\alpha1-6)]Gal(\beta1-6)]Gal(\beta1-6)[Fuc(\alpha1-6)]Gal(\beta1-6)]Gal(\beta1-6)[Fuc(\alpha1-6)]Gal(\beta1-6)[Fuc(\alpha1-6)]Gal(\beta1-6)]Gal(\beta1-6)[Fuc(\alpha1-6)]Gal($	[142]
$Fuc(\alpha 1-2)Gal(\beta 1-3)GlcNAc(\beta 1-3)[Neu5Ac(\alpha 2-6)Gal(\beta 1-4)GlcNAc(\beta 1-6)]Gal(\beta 1-4)Glc$	[126,143]
$Fuc(\alpha 1-2)Gal(\beta 1-3)GlcNAc(\beta 1-3)\{Neu5Ac(\alpha 2-3)Gal(\beta 1-4)[Fuc(\alpha 1-3)]GlcNAc(\beta 1-6)\}Gal(\beta 1-4)GlcAc(\beta 1-6)\}Gal(\beta 1-4)GlcAc(\beta 1-6)\}Gal(\beta 1-6)Gal(\beta 1-6)\}Gal(\beta 1-6)Gal(\beta 1-6)Gal($	[142]
$Fuc(\alpha 1-2)Gal(\beta 1-3)[Fuc(\alpha 1-4)]GlcNAc(\beta 1-3)[Neu5Ac(\alpha 2-6)Gal(\beta 1-4)GlcNAc(\beta 1-6)]Gal(\beta 1-4)GlcNAc(\beta 1-6)]Gal(\beta 1-4)GlcNAc(\beta 1-6)]Gal(\beta 1-4)GlcNAc(\beta 1-6)]Gal(\beta 1-4)GlcNAc(\beta 1-6)]Gal(\beta $	[136]
Gal(β1-4)GlcNAc(β1-3)[Neu5Ac(α2-6)Gal(β1-4)GlcNAc(β1-6)]Gal(β1-4)Glc	[125]
Neu5Ac(α2–6)Gal(β1–4)GlcNAc(β1–3)[Gal(β1–4)GlcNAc(β1–6)]Gal(β1–4)Glc	[126,144]
$Neu5Ac(\alpha 2-6)Gal(\beta 1-4)GlcNAc(\beta 1-3){Gal(\beta 1-4)[Fuc(\alpha 1-3)]GlcNAc(\beta 1-6)}Gal(\beta 1-4)Glc$	[126]
$Neu5Ac(\alpha 2-6)Gal(\beta 1-4)GlcNAc(\beta 1-3){Fuc(\alpha 1-2)Gal(\beta 1-4)[Fuc(\alpha 1-3)]GlcNAc(\beta 1-6)}Gal(\beta 1-4)Glc$	[126]
Neu5Ac(α2-6)Gal(β1-4)GlcNAc(β1-3)[Neu5Ac(α2-6)Gal(β1-4)GlcNAc(β1-6)]Gal(β1-4)Glc	[130]
Neu5Ac(α2-6/3)Gal(β1-4)[Fuc(α1-3)]GlcNAc(β1-3)[Neu5Ac(α2-3/6)Gal(β1-4)GlcNAc(β1-6)]Gal(β1-4)Glc	[137]
Neu5Ac(α 2-3)Gal(β 1-4)[Fuc(α 1-3)]Glc	[126]
Neu5Ac(α2-6)Gal(β1-4)GlcNAc(β1-3)Gal(β1-4)[Fuc(α1-3)]Glc	[126,145]
Neu5Ac(α2-3)Gal(β1-3)[Neu5Ac(α2-6)]GlcNAc(β1-3)Gal(β1-4)[Fuc(α1-3)]Glc	[130]
$Neu5Ac(\alpha 2-3)Gal(\beta 1-3){Gal(\beta 1-4)[Fuc(\alpha 1-3)]GlcNAc(\beta 1-6)}Gal(\beta 1-4)Glc$	[136]
Neu5Ac(α 2-3)Gal(β 1-3)[Fuc(α 1-4)]GlcNAc	[146]
Neu5Ac(α 2–3)Gal(β 1–3)[Fuc(α 1–4)]GlcNAc(β 1–3)Gal	[146]

Table	4,	notes
-------	----	-------

^a Also occurring in feces of preterm infants fed on breast milk [147].

^b Also occurring in bovine colostrum and human (pregnancy) urine [148].

Other sialyloligosaccharides in bovine colostrum are: Neu5Ac($\alpha 2-3$)Gal, Neu5Gc($\alpha 2-3$)Gal($\beta 1-4$)Glc, Neu5Ac($\alpha 2-3$)Gal($\beta 1-4$)Glc, Neu5Ac($\alpha 2-3$)Gal($\beta 1-4$)Glc, Neu5Ac($\alpha 2-6$)Gal($\beta 1-4$)GlcNAc, Neu5Ac($\alpha 2-6$)Gal($\beta 1-4$)GlcNAc($\alpha 1-P$, Neu5Ac($\alpha 2-6$)Gal($\beta 1-4$)GlcNAc, Neu5Ac($\alpha 2-$

Other sialyloligosaccharides in human (pregnancy) urine are: Neu5Ac(α 2-3)Gal(β 1-4)GlcNAc, Neu5Ac(α 2-6)Gal(β 1-4)GlcNAc, Neu5Ac(α 2-3)Gal(β 1-4)GlcNAc(α 1-P, Neu5Ac(α 2-6)Gal(β 1-4)GlcNAc(α 1-P, Neu5Ac(α 2-6)Gal(β 1-4)GlcNAc(α 1-P, Neu5Ac(α 2-6)Gal(β 1-3)[Neu5Ac(α 2-6)]GalNAc(α 1-P, Neu5Ac(α 2-3)Gal(β 1-3)[Neu5Ac(α 2-6)]GalNAc(α 1-P, Neu5Ac(α 2-3)Gal(β 1-3)[Neu5Ac(α 2-6)]GalNAc(α 1-P, Neu5Ac(α 2-3)Gal(β 1-3)[Neu5Ac(α 2-3)Gal(β 1-3)[Neu5Ac(α 2-6)]GalNAc(α 1-P, Neu5Ac(α 2-3)Gal(β 1-3)[Neu5Ac(α 2-3)Gal(β 1-3)[Neu5Ac(α 2-6)]GalNAc(α 1-P, Neu5Ac(α 2-3)Gal(β 1-3)[Neu5Ac(α 2-3)Gal(β 1-3)[Neu5Ac(α 2-6)]GalNAc(α 1-P, Neu5Ac(α 2-3)Gal(β 1-3)[Neu5Ac(α 2-3)Gal(β 1-3)[Neu5Ac(α 2-6)]GalNAc(α 1-P, Neu5Ac(α 2-3)Gal(β 1-3)[Neu5Ac(α 2-3)Gal(β 1-3)[Neu5Ac(α 2-6)]GalNAc(α 1-P, Neu5Ac(α 2-3)Gal(β 1-3)[Neu5Ac(α 2-3)[Neu5Ac(α 2-3)Gal(β 1-3)[Neu5Ac(α 2-3)[Neu5Ac(α 2-3)[Neu

The urinary carbohydrates, except the inositol derivatives and the tetrasaccharide, do also occur in hemofiltrates of patients with end-stage renal disease; in addition Neu5Ac($\alpha 2$ -8)Neu5Ac, Neu5Ac($\alpha 2$ -3)Gal($\beta 1$ -3)GalNAc($\alpha 1$ -O)Ser-Leu and Neu5Ac($\alpha 2$ -3)Gal($\beta 1$ -4)Xyl($\beta 1$ -O)Ser have been detected [149,150]. For NMR data of Neu5Ac($\alpha 2$ -3)Gal($\beta 1$ -4)Glc, Neu5Ac($\alpha 2$ -3)Gal($\beta 1$ -4)GlcNAc, and Neu5Ac($\alpha 2$ -6)Gal($\beta 1$ -4)GlcNAc, see also [151].

 $\label{eq:constraint} Urine \ of \ patients \ with \ aspartylgly cosaminuria \ contains \ Neu5Ac(\alpha 2-3)Gal(\beta 1-4)GlcNAc(\beta 1-N)Asn, \ Neu5Ac(\alpha 2-6)Gal(\beta 1-4)GlcNAc(\beta 1-N)Asn, \ and \ Neu5Ac(\alpha 2-3)Gal(\beta 1-4)GlcNAc(\beta 1-N)Asn, \ and \$

Urine of patients with β -mannosidosis contains Neu5Ac(α 2-6)Man(β 1-4)GlcNAc [153].

^c Neu5Ac(α2-3)[6S]Gal(β1-4)Glc occurs in rat milk and mammary gland [154].

^d Neu5,9Ac₂(α 2–3)Gal(β 1–4)Glc occurs in rat urine [25].

^e Neu4,5Ac₂(α2-3)Gal(β1-4)Glc occurs in monotreme milk [19].

^f Occurs also in human pregnancy urine [155].

for both glycoprotein and glycolipid glycans a sequence has been found comprising the oligomerization of Neu5Gc residues through their anomeric centers and N-glycolyl moieties, Neu5Gc($\alpha 2$ -O5)Neu5Gc($\alpha 2$ -O5)Neu5Gc($\alpha 2$ -[31,101]; in the case of the glycolipid material the Neu5Gc residues are also 8-O-methylated [31]. In microbial polysaccharides, besides internal 8- and 9-substituted Neu5Ac residues, also internal 4- and 7-substituted Neu5Ac units have been frequently found (Table 5). It should be noted that some of the sialic-acid-containing glycan fragments and polysialic acids are specifically found in lipopolysaccharides and capsular polysaccharides of pathogenic bacteria, leading to severe problems in the development of suitable vaccines. A typical source for sialo-oligosaccharides generated from N- and O-linked glycans is the urine of sialidosis and I-cell disease patients [191–199], though this is not discussed in detail here (see section 10.5, and volume 30 of the New Comprehensive Biochemistry series). Patients with other inborn errors of metabolism, like aspartylglycosaminuria [76,152] or β -mannosidosis [153] excrete small amounts of structurally unusual sialooligosaccharides, of which the formulae have been included in the footnotes of Table 4. Finally, sequence information of the already mentioned glycosylphosphatidylinositol anchor is available, showing that the glycan core consists of Mana-Mana-Man-[Neu5Ac-Gal-GalNAc)Man-GlcN-inositol [186].

The phenomenon of intramolecular lactone formation, often reported for polysialic acid [200-202] and for gangliosides [203-206], has not been detected so far in glycoprotein sialoglycans. In the case of a Neu5Ac($\alpha 2$ -8)Neu5Ac sequence, lactonization affords a Neu5Ac(α 2–8,1–9)Neu5Ac element (Fig. 4), whereby the COOH group of one residue reacts with HO9 of an adjacent residue, to give a six-membered ring. Similarly Neu5Ac(α 2–9)Neu5Ac can be converted into Neu5Ac(α 2–9,1–8)Neu5Ac, and Neu5Ac($\alpha 2$ -3)Gal into Neu5Ac($\alpha 2$ -3,1-2)Gal or Neu5Ac($\alpha 2$ -3,1-4)Gal. In a polysialic acid chain of α -2,8-linked Neu5Ac residues the α -2,8/1,9-lactonization can be effected under relatively mild conditions, like mild acid treatment [200], yielding a water-insoluble polymer. The Neu5Ac(α 2-9,1-8)Neu5Ac formation in a polysialic acid chain can only be realized by carbodiimide treatment [201], illustrating a more difficult condensation with the secondary HO8 group. The difference in reactivity between the primary and the secondary OH group has been nicely illustrated for the alternating α -2,8/ α -2,9polysialic acid of E. coli K92 (Table 5) [202]. For gangliosides, it has been stated that ganglioside lactones occur also as such in nature [206], and that in this way the negative charge of a ganglioside under physiological conditions can be modulated. Lactonization makes the oligosaccharide chain also more rigid, which may have important biological implications. Treatment with carbodiimide or glacial acetic acid can even convert a Neu5Ac($\alpha 2$ -8)Neu5Ac($\alpha 2$ -3)Gal($\beta 1$ - into a Neu5Ac($\alpha 2$ -8,1-9)Neu5Ac($\alpha 2$ -3,1-2)Gal(β 1- sequence [204,205].

Over the years Neu5Ac has been prepared by a variety of methods. Several biological sources have been explored to isolate this sialic acid in high amounts. Among them are edible birds nest substance [207], urine of sialuria patients [65], colominic acid {Neu5Ac($\alpha 2$ -8)}_n produced by *E. coli* strains [208], and hen's egg chalaza, egg-yolk membranes and delipidated egg yolk [209,210]. The large scale organic synthesis of Neu5Ac is still complicated (see section 6.1). However, recent biotechnological routes, using sialate-pyruvate lyase (aldolase; see section 9.5), readily yield large amounts of

Structure/element	Species	Reference(s)	
Capsular polysacharides (CPS)		··· ···	
$\{-8\}$ Neu5Ac(α 2-8)Neu5Ac(α 2- $\}_n$	Neisseria meningitidis B	[24]	
	Escherichia coli K235	[156]	
	Pasteurella haemolytica A-2	[157]	
	Moraxella nonliquefaciens	[158]	
$\{-8\}$ Neu5Ac($\alpha 2$ -8)Neu5Ac($\alpha 2$ - $\}_n$ (with 7Ac or 9Ac)	Escherichia coli K1	[24,159]	
$\{-9\}$ Neu5Ac($\alpha 2$ -9)Neu5Ac($\alpha 2$ - $\}_n$ (with 7Ac or 8Ac)	Neisseria meningitidis C	[24]	
$\{-8\}$ Neu5Ac(α 2-9)Neu5Ac(α 2- $\}_n$	Escherichia coli K92	[160]	
$\{-4\}$ Neu5Ac(α 2-6)Gal(α 1- $\}_n$	Neisseria meningitidis W135	[24]	
$\{-4\}$ Neu5Ac(α 2-6)Glc(α 1- $\}_n$ (with Ac)	Neisseria meningitidis Y	[24]	
$\{-3\}GalNAc(\beta 1-4)Gal(\alpha 1-4)Neu5,9Ac_2(\alpha 2-3)Gal(\beta 1-\}_n$	Escherichia coli K9	[26,161]	
$\{-4\}$ Glc(β 1-4)[Neu5Ac(α 2-3)Gal(β 1-4)GlcNAc(β 1-3)]Gal(β 1- $\}_n$	Group B Streptococcus Ia	[24]	
$\{-4\}Glc(\beta 1-4)[Neu5Ac(\alpha 2-3)Gal(\beta 1-3)GlcNAc(\beta 1-3)]Gal(\beta 1-\}_n$	Group B Streptococcus Ib	[24]	
$ \{-4)GlcNAc(\beta 1-3)[Gal(\beta 1-6)]Gal(\beta 1-4)Glc(\beta 1-3)Glc(\beta 1-2)[Neu5Ac(\alpha 2-3)]Gal(\beta 1-\}_n $	Group B Streptococcus II	[24]	
$\{-4\}Glc(\beta1-6)[Neu5Ac(\alpha2-3)Gal(\beta1-4)]GlcNAc(\beta1-3)Gal(\beta1-\}_n$	Group B Streptococcus III	[162]	
$\{-4\}Glc(\alpha 1-4)[Neu5Ac(\alpha 2-3)Gal(\beta 1-4)GlcNAc(\beta 1-6)]Gal(\beta 1-4)Glc(\beta 1-\}_{n}$	Group B Streptococcus IV	[163]	
$\{-4\}Glc(\beta 1-4)[Neu5Ac(\alpha 2-3)Gal(\beta 1-4)GlcNAc(\beta 1-6)]Glc(\alpha 1-4)[Glc(\beta 1-3)]Gal(\beta 1-\}_n$	Group B Streptococcus V	[164]	
$ \{-4)Glc(\beta 1-6)[Neu5Ac(\alpha 2-3)Gal(\beta 1-3)]Glc(\beta 1-3)Gal(\beta 1-3)]Glc(\beta 1-3)[Glc(\beta 1-3)Gal(\beta 1-3)]Glc(\beta 1-3)]Glc(\beta 1-3)[Glc(\beta 1-3)Gal(\beta 1-3)]Glc(\beta 1-3)[Glc(\beta 1-3)[Glc(\beta 1-3)Gal(\beta 1-3)]Glc(\beta 1-3)[Glc(\beta 1-3)[Glc(\beta 1-3)Gal(\beta 1-3)]Glc(\beta 1-3)[Glc(\beta 1$	Group B Streptococcus VI	[165]	
$\{-4\}Glc(\beta 1-4)[Neu5Ac(\alpha 2-3)Gal(\beta 1-4)GlcNAc(\beta 1-6)]Glc(\alpha 1-4)Gal(\beta 1-\}_n$	Group B Streptococcus VII	[165a]	
$\{-3)Glc(\beta 1-2)[Kdn(\alpha 2-4)][3Ac]GlcA(\alpha 1-3)Man(\alpha 1-3)Glc(\alpha 1-3)_n$	Klebsiella ozaenae K4 (2211)	[49]	
Lipopolysaccharides (LPS) ^a			
$\{-3\}GalNAc(\beta 1-7)Neu5Ac(\alpha 2-3)[Glc(\alpha 1-2)]Glc(\beta 1-\}_n$	Escherichia coli O24	[166,167]	
$\{-3\}$ GlcNAc(β 1-7)Neu5Ac(α 2-3)[Gal(α 1-2)]Glc(β 1- $\}_n$	Escherichia coli O56	[166,167]	
$\{-3\}$ GalNAc(β 1-4)Gal(α 1-4)Neu5,7/8,9Ac ₃ (α 2-3)Gal(β 1- $\}_n$ (also with 9Ac)	Escherichia coli O104	[26,27]	

 Table 5

 Survey of structures or elements of sialic-acid-containing microbial polysaccharides

continued on next page

Structure/element	Species	Reference(s)	
$\{-3)[\pm 6Ac]GlcNAc(\beta 1-7)Neu5Ac(\alpha 2-3)[CH_3(NH)C-2]FucN(\alpha 1-\}_n$	Salmonella arizonae O21	[168]	
$\{-4GalANGro(\alpha 1-4)Neu5Ac(\alpha 2-3)GalANGro(\beta 1-3)QuiNAc(\beta 1-)\}^{b}$	Vibrio cholerae H11 (non-O1)	[169]	
$ \{-3\}[Glc(\alpha 1-6)]GalNAc(\beta 1-4)Neu5Ac(\alpha 2-6)Glc(\alpha 1-6)[Glc(\alpha 1-4)Gal(\beta 1-6)Glc(\beta 1-3)]Gal(\beta 1-3)]Gal(\beta 1-3)Gal(\beta 1-3)]Gal(\beta 1-3)Gal(\beta 1-3)Gal(\beta 1-3)Gal(\beta 1-3)]Gal(\beta 1-3)Gal(\beta $	Hafnia alvei O2	[170]	
Core regions of lipopolysaccharides (LOS) ^a			
$Neu5Ac(\alpha 2-3)Gal(\beta 1-4)GlcNAc(\beta 1-3)Gal(\beta 1-4)Glc(\beta 1-4)Hep-$	Neisseria meningitidis	[171–173]	
$Neu5Ac(\alpha 2-3/6)Gal(\beta 1-4)GlcNAc(\beta 1-3)Gal(\beta 1-4)Hep-$	Haemophilus ducreyi	[174]	
$Neu5Ac(\alpha 2-3)[GalNAc(\beta 1-4)]Gal(\beta 1-3)[Gal(\alpha 1-2)]Gal(\beta 1-3)Hep-$	Campylobacter jejuni Ol	[175]	
Neu5Ac(α 2-3)Gal(β 1-3)[Gal(α 1-2)]Gal(β 1-3)Hep-	Campylobacter jejuni O2	[176]	
$Neu5Ac(\alpha 2-3)Gal(\beta 1-3)GalNAc(\beta 1-4)[Neu5Ac(\alpha 2-3)]Gal(\beta 1-3)Hep-$	Campylobacter jejuni O4	[175]	
$Neu5Ac(\alpha 2-3)[GalNAc(\beta 1-4)]Gal(\beta 1-4)Glc(\beta 1-2)Hep-$	Campylobacter jejuni O23 and O36	[175]	
Gal(α1–6)Glc(β1–7)Neu5Ac	Rhodobacter capsulatus 37b4	[177]	

Table 5, continued

^a Some other species for which it has been demonstrated that their LPS or LOS contains sialic acid are: *Rhizobium meliloti* M11S (LPS, Neu5Ac/Neu5,9Ac₂) [178]; *Salmonella toucra* (LPS, terminal and 4-linked Neu5Ac) [179]; *Salmonella djakarta* (LPS, terminal Neu5Ac) [180]; *Salmonella isaszeg* (LPS, terminal and 4-linked Neu5Ac) [180]; *Citrobacter freundii* O37 (LPS, terminal and 7-linked Neu5Ac) [179]; *Pasteurella haemolytica* 1 and 5 (LPS, Neu5Ac) [181]; *Haemophilus influenzae* (LOS, terminal Neu5Ac) [182,183]; *Neisseria gonorrhoeae* (LOS, terminal Neu5Ac) [184,185]. Note that for *E. coli* O24 and O56 and *H. alvei* O2 also terminal Neu5Ac has been found [179].

^b GalANGro, *N*-galacturonyl-2-aminoglycerol.



Fig. 4. Lactonization of a Neu5Ac(α 2–8)Neu5Ac(α 2–sequence, affording a Neu5Ac(α 2–8,1–9)Neu5Ac(α 2–fragment.

this sialic acid (see section 6.1). Efficient procedures to prepare Neu5Gc from colominic acid via de-*N*-acetylation, *N*-acryloylation and reductive ozonolysis, followed by acid or enzymatic hydrolysis of the formed {Neu5Gc($\alpha 2$ -8)}_n has appeared in refs. [211,212]. Also porcine submandibular gland is a good source for the large scale preparation of Neu5Gc, whereas Neu5,9Ac₂ can be prepared from bovine submandibular gland [213].

Besides the occurrence of sialic acids [(O-acetylated) Neu5Ac and Kdn] in microbial polysaccharides (Table 5), some lipopolysaccharides have shown to contain sialic-acidlike monosaccharides. They differ from sialic acids in the presence of an additional amino function at C7, a deoxy function at C9, and in the configuration at the chiral centers: 5,7-diamino-3,5,7,9-tetradeoxy-L-glycero-L-manno-non-2-ulosonic acid and 5,7-diamino-3,5,7,9-tetradeoxy-L-glycero-D-galacto-non-2-ulosonic acid [214,214a]. So far, the two amino groups have been found to be substituted in different combinations, yielding acetamido, formamido, (R)-3-hydroxybutyramido, 4-hydroxybutyramido, or acetamidino functions. Also O-acetylation can occur, whereas the (R)-3-hydroxybutyryl group at C7 can be used to link monosaccharides in a polysaccharide chain. Typical species are Pseudomonas aeruginosa strains [214], Shigella boydii type 7 [214], Vibrio salmonicida [214a], Vibrio cholerae O2 [215], Vibrio alginolyticus strain 945-80 [216], Salmonella arizonae O61 [217], Yersinia ruckeri O1 [218], Legionella pneumophila strain 1 [219,220], and Pseudomonas fluorescens ATCC 49271 [220a]. For a review, see ref. [220], but it should be noted that the absolute configuration of 5,7-diamino-3,5,7,9tetradeoxy-L-glycero-D-galacto-non-2-ulosonic acid was earlier assigned as D-glycero-Lgalacto-[214,214a,216-219,220a].

4. Screening of biological materials for the presence of sialic acid

For the staining of sialic acids in tissues and cells, a great variety of techniques is available. Classical histochemistry of sialic-acid-containing glycoconjugates makes use of either binding of cationized dyes (e.g. Alcian blue, cationized ferritin, ruthenium red) or selective periodate oxidation (derivatization of generated aldehyde groups of sialic acids with e.g. *p*-dimethylaminobenzaldehyde, dansylhydrazine, rhodamine, biotin/ferritin-conjugated avidin). The second approach is strongly dependent on side-chain modifications [53]. Comprehensive reviews on this subject have appeared [6,221].

Nowadays, specific lectins are frequently used to detect the presence of glycosidically bound sialic acids in complex carbohydrates. However, the specificity of lectins is generally broad, and positive information has always to be checked in control experiments using e.g. sialidases in the presence and absence of inhibitors. For the histochemical analysis, lectins may be conjugated e.g. with gold particles, peroxidase, rhodamine or fluorescein isothiocyanate [6]. A large series of lectins which recognize sialic acid have been demonstrated to occur in nature, and most of their biological sources have been summarized in refs. [6,11,222]; see also refs. [223–227]. For updated reviews focusing on lectins, see the present volume 29b of the New Comprehensive Biochemistry series. Generally, lectins have been isolated from lower animals such as prawns, snails, crabs, spiders, scorpions, lobsters, slugs, oysters, but also from plants, rat brain and B cells. While some of these lectins bind to both Neu5Ac and Neu5Gc, others are specific for Neu5Ac. *O*-Acetylation may also influence the binding strength of the lectin, both in a positive and in a negative manner (see section 10.3).

For introductory glycoprotein analysis, in answering questions like "what monosaccharides are in the glycoprotein glycans?", two sialic-acid-specific plant lectins, having also a glycosidic linkage specificity, have been included in commercially available kits. These lectins are the agglutinins from *Maackia amurensis* and *Sambucus nigra*, being diagnostic for Neu5Ac(α 2–3)Gal and Neu5Ac(α 2–6)Gal/GalNAc elements, respectively. For screening purposes, the lectins are labelled with digoxigenin-succinyl- ε -amidocaproic acid hydrazide (DIG, a spacer-linked steroid hapten digoxigenin). After SDS/PAGE of the (glyco)protein mixture and Western blotting, sialoglycoprotein bands with α -2,3and/or α -2,6-linked Neu5Ac can be labelled by one or both of these DIG-labelled lectins, whereby the detection is carried out in an enzyme immuno-assay using a digoxigeninspecific antibody conjugated to alkaline phosphatase, followed by color development with 5-bromo-4-chloro-3-indolyl-phosphate/4-nitroblue tetrazolium chloride [228,229].

Viruses and antibodies can also be used for the detection of sialic acid in complex carbohydrate systems. The sialic-acid-binding properties of a number of viruses have been established, and it has been shown that influenza C virus, bovine corona virus and encephalomyelitis virus recognize 9-O-acetylated sialic acids (see also sections 8.4.2, 9.1 and 10.3). Microtiter-plate and nitrocellulose-membrane assays have been developed that use the hemagglutinin (receptor) and the receptor destroying activities (sialate 9-Oacetylesterase) of the influenza C virus to specifically detect bound 9-O-acetylated sialic acids in glycoproteins [230,231]. Although the recognition site for Neu5,9Ac₂ and the esterase activity are located on the same viral glycoprotein, these activities can be separated using different temperatures for the binding (4°C) and the enzyme reaction (4-methylumbelliferyl acetate or α -naphtyl acetate as substrates; 20–37°C). Other approaches are based on binding of the virus to immobilized ligands, and detection of the virus with monoclonal antibodies, whereby the esterase was selectively inactivated by the use of diisopropylfluorophosphate (DFP) [232]. Virus particles have also been labelled with radioactive isotopes [232] or biotin [233]. The application of the assay with virus particles for the staining of O-acetylated sialic acids in tissue sections will be described in section 8.4.2. Recently, a new technique using a soluble chimera of the hemagglutininesterase portion of the hemagglutinin-esterase-fusion-glycoprotein from influenza C virus and the Fc portion of human IgG, has been reported [234,235]. Such a chimera retains 264

both its recognition and enzymatic functions, and also has the binding properties of the Fc portion of IgG. The probing can be carried out on blots and TLC plates taking into account precautions for the recognition and esterase activities (this has to be inhibited by DFP in the test) as discussed above.

An interesting electrochemical method for the determination of bound sialic acid has been developed, making use of a potentiometric four-channel thick-film sensor [236]. The sialidase sensor consists of a bilayer of a membrane containing *Clostridium perfringens* sialidase immobilized in a poly(vinyl acetate)–polyethylene copolymer, which is placed on top of an H⁺-selective poly(vinyl chloride)–poly(vinyl acetate) indicator membrane. The enzyme-induced release of bound sialic acid leads to a concomitant decrease in pK_a of the carboxyl function of sialic acid. This decrease affords a local pH change inside the sialidase-containing sensor membrane, which is monitored by the H⁺-selective indicator membrane. The pH optimum of the sialidase sensor was pH 4 for sialyllactose, mucin and colominic acid.

Finally, TLC analysis may also be of great help in the screening of biological materials for the presence of sialic acids. This item will be discussed in section 5.3.1.

5. Isolation and analysis of sialic acids

5.1. Liberation

The characterization of the type of sialic acid in sialoglycoconjugates is frequently carried out after release and (partial) isolation. The cleavage of sialic acid from sialic-acid-containing material is mainly performed by two methods, namely, acid hydrolysis and enzymatic hydrolysis. Both approaches have advantages and disadvantages. In the case of acid hydrolysis, problems arise with respect to de-*O*-esterification, which complicates quantitative analysis procedures. With respect to the enzymatic hydrolysis with sialidases, linkage specificity as well as a reduced or complete lack of susceptibility have to be taken into account. Moreover, in most cases much lower amounts of sialic acids are released by sialidases than by acid hydrolysis, which may be due to the different accessibility of the sialic acids in the biomolecules to be analyzed. Additionally, in work-up procedures and analyses, pH values below 4 and over 6 should be avoided to prevent migration of the *O*-acetyl group at C7/C8 and hydrolysis of *O*-acetyl groups as much as possible [23,57].

Several approaches have been reported for the effective acid hydrolysis of the labile α -2,3-, α -2,6- and α -2,8-linkages. All these procedures suffer from being not optimal in giving the real spectrum of sialic acids originally present in the sialoglycoconjugate under study, especially in the case of a mixture of (*O*-acylated) *N*-acylneuraminic acids. Terminal sialic acids are released in high yield (and low destruction) using a two-step acid hydrolysis procedure comprising treatment with formic acid (pH 2, 1 h, 70°C), followed by HCl (pH 1, 1 h, 80°C). After each step, the liberated sialic acids must be recovered by centrifugation, ultrafiltration or dialysis [11,12,213,237]. It should be noted that in the case of a spectrum of (*O*-acylated) *N*-acylneuraminic acids, the supernatant, ultrafiltrate or diffusate of the formic acid hydrolysis contains the majority of the *O*-acylated *N*-acylneuraminic acid, whereas that of the HCl hydrolysis contains

Source of sialidase	pH optimum	α-2,3 ^b	α-2,6°	α-2,8 ^d	α-2,3 °	Reference
Arthrobacter ureafaciens	4.4–5.5 ^f	100	167	53	12	[5]
Vibrio cholerae	5.0-6.5 ^g	100	53	31	25	[5]
Clostridium perfringens ^h	5.57.2 ⁱ	100	44	44	20	[5]
Salmonella typhimurium	5.07.0 ^j	100	0.4	n.d. ^{k,I}	n.d.	[247]
Newcastle disease virus	5.0-6.0	100	0.2	78	11	[5]

Table 6 Substrate specificity of commercially available sialidases^a

activity).

^b Neu5Ac(α 2–3)lactose.

^c Neu5Ac(α 2–6)lactose.

^d Neu5Ac(α 2-8)Neu5Ac(α 2-3)lactose.

^e Neu5Gc($\alpha 2$ -3)lactose.

^f Sialyllactose, pH 5.0-5.5, and colominic acid,

pH 4.3-4.5.

g Sialyllactose, pH 5.0-5.5.

personal communication) (see also section 9.2). Sialyllactose, pH 5.0-5.1 (acetate buffer) and pH 5.8-6.0 (phosphate buffer). ^j Depending on the buffer system used.

k n.d., not determined.

¹ {-8)Neu5Ac($\alpha 2$ -8)Neu5Ac($\alpha 2$ -}, 0.1% and

 $\{-9\}$ Neu5Ac($\alpha 2-9$)Neu5Ac($\alpha 2-\}_n$, 0.08%.

mostly Neu5Ac and Neu5Gc. In the case of low-molecular-mass substances, isolations can be carried out by gel-permeation chromatography. Although these conditions do not lead to significant de-N-acylation, de-O-acylation has been shown to occur to an extent of about 30-50%. One has to consider that milder acidic conditions result in incomplete release of sialic acids. Acid hydrolysis with acetic acid (2 M, 3 h, 80°C) as suggested in ref. [238] did not improve the yield of O-acylated N-acylneuraminic acids [11]. When focusing on sialic acid analysis, the use of H_2SO_4 (0.05 M) is not recommended because of work-up problems. In connection with the HPLC analysis of Neu5Ac and Neu5Gc, microwave hydrolysis in 2 M acetic acid has shown to be an interesting alternative [239].

In the methanolysis procedure, as used for the standard quantitative monosaccharide GLC analysis of glycoconjugates [240,241], 1 M methanolic HCl (24 h, 85°C) is applied. However, under these conditions released sialic acids are completely de-N,O-acylated, which makes this approach unsuitable for the characterization of different types of sialic acid. It is, however, a reliable approach for the determination of the total amount of a mixture of (O-acetylated) N-acylneuraminic acids. When using a milder methanolysis procedure (0.05 M methanolic HCl, 1 h, 80°C) the de-N-acylation but not the de-O-acylation is strongly reduced [242].

In the quantitative determination of N-acylneuraminic acids in poly-N-acylneuraminicacid-containing glycoproteins, the release of free sialic acid was shown to be optimal using a combined mild acid hydrolysis (0.1 M TFA, 3 h, 80°C)/subsequent mild methanolysis (0.05 M methanolic HCl, 1 h, 80°C) [243]. This method is also advised for the analysis of the Kdn content in Kdn-containing glycoproteins. In contrast, optimal release of Kdn from poly-Kdn-containing glycoproteins is obtained by mild methanolysis with a longer incubation time [243]. In the latter case the standard conditions of methanolysis also give good results.

The enzymatic release of N-acylneuraminic acids can be carried out under such mild conditions (low temperature, pH 5-6), that destruction, migration, and de-O-acylation is kept at a minimum. In Table 6 the substrate specificity of commercially available sialidases from Arthrobacter ureafaciens, Vibrio cholerae, Clostridium perfringens. Salmonella typhimurium, and Newcastle disease virus, using simple N-acylneuraminyllactose substrates is compared (for comprehensive reviews on sialidases, see refs. [5,33,244-246]). It is evident that the sialidases show different ratios of cleavage rates for the α -2,3-, α -2,6- and α -2,8-linkages. The finding of a strong preference of the Newcastle disease virus sialidase for α -2,3-linkages holds also for other viral sialidases, such as those from fowl plague virus and influenza A_2 virus. The latter enzyme also has a low specificity for α -2,8-linkages [5]. Among the bacterial sialidases, A. ureafaciens sialidase has a certain preference for α -2,6linkages. The S. typhimurium sialidase is the only bacterial sialidase with a viral sialidase-like pronounced preference for α -2,3-linkages [247]. Recently, two sialidases from *Bacteroides fragilis* having a higher preference for the cleavage of α -2,8linkages, when compared with α -2,3- and α -2,6-linkages, were isolated [248]. The sialidase, recently isolated from *Macrobdella* leech, cleaves only Neu5Ac(α 2–3)Gal linkages [61]. In general, Neu5Ac residues are released faster than Neu5Gc residues. In a study using different N- and O-glycoproteins with α -2,3- and/or α -2,6-linked N-acylneuraminic acids as substrates [antifreeze glycoprotein, ovine submandibular gland glycoprotein, α_1 -acid glycoprotein; Neu5Ac(α_2 -6)GalNAc(α_1 -O)Thr/Ser, Neu5Ac/5Gc(α_2 -6)[Gal(β 1-3)]GalNAc(α 1-O)Thr, Neu5Ac/5Gc(α 2-3)Gal(β 1-3)GalNAc(α 1-O)Thr, Neu- $5Ac(\alpha 2-3)Gal(\beta 1-4)GlcNAc(\beta 1-O)R;$ Neu $5Ac/5Gc(\alpha 2-6)Gal(\beta 1-4)GlcNAc(\beta 1-O)R]$ and sialidases from A. ureafaciens, V. cholerae, C. perfringens, Newcastle disease virus, fowl plague virus, and influenza A2 virus, roughly similar patterns of substrate specificity as for sialyllactoses were found. However, it was demonstrated that the core oligosaccharide and/or the protein structure may also influence the rate of release for different glycosidic linkages [249]. In the case of S. typhimurium sialidase, also N-glycoprotein α -2,3-sialoglycans were susceptible to efficient cleavage, but not mucin O-glycoprotein sialoglycans [247]. The most recently discovered sialic acid, Neu5GcAc, could not be released with V. cholerae sialidase [43]. A comparison of the different commercially available sialidases shows that the C. perfringens sialidase iso-enzyme with a molecular mass of about 63 kDa has the broadest specificity. It should be noted that C. perfringens in fact produces two sialidases, the larger of which (63 kDa) is commercially available (P. Roggentin, personal communication) (see also section 9.2). As described for the acid hydrolysis procedure, the work-up of enzymatically released sialic acids can be achieved employing various methods, depending on the starting sialoglycoconjugate material.

4-O-Acetylated neuraminic acids in any glycoconjugate are resistant to most sialidases tested so far; only viral sialidases show a low but significant activity. With the exception of *Streptococcus sanguis* sialidase [250], the presence of O-acetyl substituents at C7–C9 leads to a reduced rate of cleavage by all sialidases, so that prolonged incubations are necessary for an efficient release [251]. More information has been collected in a detailed study with bacterial and viral sialidases and 4-methylumbelliferyl (MU) α -glycosides of 4-, 7-, and 9-O-acetylated N-acetylneuraminic acids as substrates [252]. In contrast to

the other sialidases tested, the fowl plague virus sialidase catalyzes a slow release of Neu4,5Ac₂ from α -Neu4,5Ac₂-MU.

The recent finding of {Neu5Gc($\alpha 2$ -O5)}_n elements in a glycoprotein [101] initiated a kinetic study of the enzymatic and non-enzymatic hydrolysis of Neu5Gc($\alpha 2$ -O5)Neu5Gc and Neu5Gc($\alpha 2$ -8)Neu5Gc [253]. It turned out that at pH < 3.8 the rate of acid hydrolysis of the unusual α -2,O5-linkage was greater than that of the normal α -2,8-linkage. However, at pH > 3.8 reverse results were obtained; Neu5Gc($\alpha 2$ -8)Neu5Gc released a small but detectable amount of Neu5Gc even at pH 6. The α -2,O5-linkage was only partially cleaved by *C. perfringens* and *V. cholerae* sialidases, and was essentially resistant to *A. ureafaciens* sialidase.

The detection of sialidases that can release Kdn is so far highly limited. The liver of the loach *Misgurnus fossilis* was found to contain a sialidase capable of releasing both Neu5Ac and Kdn from sialoglycoconjugates [39]. While the sialidases investigated so far in detail require an NH-acyl group at C5 for full activity, the loach enzyme can handle both NH-acyl and OH functions at C5. The rainbow trout also turned out to be a useful source for the isolation of a sialidase, active in releasing both Neu5Ac and Kdn [254]. Recently, a sialidase was isolated from *Sphingobacterium multivorum* that specifically released α -2,3-, α -2,6- and α -2,8-linked Kdn; Neu5Ac and Neu5Gc were not liberated [255].

Before fractionation, pools of free sialic acids can be freed from contaminants by several methods, including ion-exchange chromatography, cellulose chromatography, reversed-phase chromatography, preparative TLC [6,11,12]. Generally, one of the purification procedures for the pool of sialic acids comprises Dowex ion-exchange chromatography at low temperature. After passage through a cation-exchange resin (Dowex 50W-X8, H⁺-form), the pool of sialic acids is adsorbed to an anion-exchange resin (Dowex 2-X8 or 1-X8, HCOO⁻-form). Elution from the anion-exchange resin is generally carried out with 0-2 M formic acid. The ion-exchange chromatography should be carried out rapidly, as prolonged contact of acylneuraminic acids with the resin or the solvent systems used may result in degradation, *O*-acetyl migration and/or de-*O*-acetylation. After rotary evaporation or lyophilization, the sialic acid pools are stored at -20° C.

In general, mixtures of sialo-oligosaccharides from {Neu5Ac($\alpha 2-8$)}_n, {Neu5Ac($\alpha 2-9$)}_n, {-8)Neu5Ac($\alpha 2-9$)Neu5Ac($\alpha 2-3$)_n, {Neu5Gc($\alpha 2-8$)}_n, and {Kdn($\alpha 2-8$)}_n, are generated by limited acid hydrolysis [211,256,257]. Depending on the polysialic acid, different conditions have been applied. Also attention has been paid to the intramolecular self-cleavage of polysialic acids such as {Neu5Ac($\alpha 2-8$)}_n [258]. Adjacent COOH groups with a high pK_a (3.9–5.5) act as proton donors for general acid catalysis. The lability is seen under mild acidic conditions, that can be encountered in various physiological situations. {Neu5Ac($\alpha 2-8$)}_{>3} is substantially more unstable than {Neu5Ac($\alpha 2-8$)}₂₋₃.

A highly useful enzyme for the depolymerization of polysialyl carbohydrate chains, yielding oligosialyl compounds, is endo-sialidase (endo-N) produced by infection of *E. coli* K1 with a lytic bacteriophage [9,259,260]. The enzyme is specific in cleaving α -2,8-linkages, and requires at least five Neu5Ac or Neu5Gc residues for activity. A limited digestion of {Neu5Ac(α 2-8)}_n-R affords mainly {Neu5Ac(α 2-8)}₄, with some {Neu5Ac(α 2-8)}₁₋₃. Alternating α -2,8/ α -2,9-polysialyl chains, as present in some

bacterial polysaccharides, are also cleaved, but α -2,9-linked polysialyl chains are resistant. Poly-Kdn and {Neu5Gc(α 2–O5)}_n are not substrates for endo-N. A similar endo-sialidase associated with phage particles, namely, *E. coli* bacteriophage ϕ 92, has been reviewed in ref. [261].

5.2. Colorimetry

One of the oldest methods used to detect and to quantitate sialic acids is colorimetry [3,6,12,237,262]. When carried out on non-purified samples, the influence of contaminants interfering with the assays has to be taken into account. Greatest problems are encountered when using cells or tissue extracts, as the level of contamination is inevitably high. Moreover, factors such as non-identical reactions of different sialic acids in the same assay and the non-specificity of the reactions for the sialic acids are important.

Although several colorimetric methods have been developed in the past, only two main procedures are currently routinely applied, namely the orcinol/Fe³⁺/HCl assay, known as the "Bial" reaction, and the periodic acid/thiobarbituric acid assay. For microadaptations of these two different tests, see ref. [11].

In the first assay, the sample is mixed with orcinol, $FeCl_3$ and concentrated HCl and heated at 96°C. The formed purple to red-violet chromophore is extracted with isoamyl alcohol and its absorbance measured at 572 nm. Because of the use of HCl, the method can be used to quantitate the total amount of both free and glycosidically bound sialic acids. Due to the strongly acidic conditions, ester groups are released. As the assay does not discriminate between bound and free sialic acids, it is widely used to monitor the presence of sialic acids in either form during fractionation of biological material. It should be noted that other monosaccharides, especially pentoses, hexoses and uronic acids interfere with the assay, which is of importance when small amounts of sialic acid are present.

In the second assay, only suitable to quantitate free sialic acids, sialic acid is oxidized by periodate at 37°C under strongly acidic conditions. The oxidation leads to the formation of a prechromogen, a six carbon aldehyde, which then yields the chromogen β -formyl pyruvic acid by aldol cleavage between C4 and C5. The chromogen reacts with thiobarbituric acid to give a red chromophore, the absorbance of which is measured at 549 and 532 nm. In principle two approaches can be followed, called the Warren method and the Aminoff method. Differences between the methods lie in the acidity of the initial periodate oxidation and in the solvent used for the extraction of the pigment (cyclohexanone, Warren; acidic 1-butanol, Aminoff). It should be noted that substituents in the glycerol side chain severely influence the periodate oxidation. Therefore, in the case of ester substituents, a prior saponification is necessary (0.1 M NaOH, 37°C). Types of free sialic acid which do not yield the chromogen are negative in this test. Several compounds have been shown to interfere with the periodate/thiobarbituric acid assay, most especially 2-deoxyribose, 2-keto-3-deoxyaldonic acids other than Kdn, disaccharides such as lactose and maltose, and unsaturated fatty acids. The greatest errors arise in the quantitation of sialic acids from cellular extracts or homogenates containing membrane and nucleic acid material. Therefore, prior ion-exchange column chromatography and removal of lipids by ether extraction are of advantage. Special attention to the periodic acid/thiobarbituric acid assay of Kdn has been paid in ref. [243].

When both tests are used in combination, a differentiation between total and free sialic acid is possible, allowing the calculation of the amount of glycosidically bound sialic acid.

In a new approach for the direct determination of free and bound sialic acid, an acidic ninhydrin assay has been proposed [263]. Heating of solutions of sialic-acid-containing material with ninhydrin/acetic acid/37% HCl at 100°C yields a stable chromophore, the absorbance of which can be measured at 470 nm.

In view of the comments made in section 5.1 with respect to the release of sialic acid by sialidases, a quantification procedure for bound sialic acid based on the enzymatic analysis of pyruvate, formed after sialidase/aldolase treatment [6,264], should be handled with care.

Of the various fluorimetric assays available for sialic acid analysis, the method which allows the discrimination between sialic acids with or without *O*-acyl groups at C8 and/or C9 may be of interest [6,265]. After mild periodate oxidation, formaldehyde, derived from C9 in the case of non-substituted HO9 and HO8 is derivatized with acetylacetone in the presence of ammonium acetate, leading to the fluorigen 3,5-diacetyl-1,4-dihydro-2,6-dimethylpyridine (410 nm excitation, 510 nm emission). It is evident that all contaminants producing formaldehyde under the influence of periodate will interfere with this sialic acid analysis.

Finally, for the quantitative estimation of the *O*-acyl content of sialic-acid-containing samples (Hestrin assay), also a colorimetric assay is routinely used. The method is based on the reaction with hydroxylamine in alkaline medium yielding hydroxamates, which form with $Fe(ClO_4)_3$ red chromophores, the absorbance of which is measured at 520 nm [262].

5.3. Chromatography

5.3.1. Thin-layer chromatography

From the beginning of free sialic acid analysis, TLC has played a major role in screening and tentative assignment procedures, and over the years several solvent systems for both cellulose and silicagel plates have been reported [6,237,262]. One of the most popular TLC methods for the analysis of (substituted) *N*-acylneuraminic acids comprises the use of plastic HPTLC plates precoated with cellulose and 1-propanol/1-butanol/0.1 M HCl (2:1:1, v/v/v) as solvent system. It shows the best and most reproducible separation of different sialic acids, and is less sensitive to interfering substances when compared with other systems. Generally, the visualization of the different sialic acids is carried out by spraying with the orcinol/Fe³⁺/HCl reagent [237], yielding typical purple bands. For quantitative purposes densitometry is also used. It should be noted that due to differences in cellulose quality (impurities), even after pre-washing of the plates, the reproducibility of R_f values is relatively low. Therefore, analyses should be carried out in the presence of an appropriate reference sialic acid mixture on a separate lane. To give an impression of the separation capacity of cellulose plates, Table 7 summarizes the R_f values of a series

Compound	$R_{\rm f}$	Compound	$R_{\rm f}$
Neu5Ac	0.45	Neu2en5Ac	0.55
Neu4,5Ac ₂	0.60	Neu5Gc	0.35
Neu5,7Ac ₂	0.54	Neu4Ac5Gc	0.65
Neu5,9Ac ₂	0.63	Neu9Ac5Gc	0.55
Neu5,7,9Ac ₃	0.70	Neu7,9Ac ₂ 5Gc	0.70
Neu5,8,9Ac ₃	0.75	Neu5GcAc	0.51
Neu5,7,8,9Ac4	0.80	Neu2en5Gc	0.45
Neu5Ac9Lt	0.56		

 Table 7

 Thin-layer chromatographic migration rates (R_f) of sialic acids on 0.1 mm cellulose plates using 1-propanol/1butanol/0.1 M HCl (2:1:1, v/v/v) [6]

of N,O-acyl-neuraminic acids from one experiment [6]. In the case of radio-labelled sialic acids, the bands can also be traced by radio-TLC-scanning.

Application of the TLC method in a two-dimensional procedure with intermediate ammonia treatment gives information about the type of N-acylneuraminic acid of the constituting N, O-acylneuraminic acids. In this way a differentiation is possible for example between co-migrating Neu9Ac5Gc and Neu5, 7Ac₂ [6]. In principle, de-O-acetylations can also be carried out by sialate O-acetylesterases [266].

For the analysis of oligomers of α -2,8-linked Neu5Ac, Neu5Gc or Kdn, a TLC procedure on silicagel with the solvent system 1-propanol/25% ammonia/water (12:2:5, v/v/v) has shown good results [267]. In this way, mixtures of {Neu5Ac(α 2–8)}₂ to {Neu5Ac(α 2–8)}₁₄, {Neu5Gc(α 2–8)}₂ to {Neu5Ac(α 2–8)}₁₄, {Neu5Gc(α 2–8)}₂ to {Neu5Gc(α 2–8)}₁₀, or {Kdn(α 2–8)}₂ to {Kdn(α 2–8)}₇ are well separated. It should be noted that {Neu5Ac(α 2–8)}_n and {Neu5Gc(α 2–8)}_n can be visualized also with a resorcinol spray reagent, whereas for {Kdn(α 2–8)}_n the orcinol spray reagent is needed.

5.3.2. High-performance liquid chromatography

Initially, the separation of sialic acids was mainly carried out by cellulose chromatography at low temperature [6,237]. However, nowadays HPLC fractionations using different column materials, elution protocols and detection techniques have replaced this approach [6,11,268]. The application of HPLC has also introduced a rapid method for tentative assignments of sialic acids in complex mixtures, based on elution times of known standards, being more reliable when more than one HPLC procedure is followed. Moreover, a rapid method for quantification of released sialic acids has become available. Due to the relatively short HPLC runs, also fast transitions between members of the sialic acid family due to migration of substituents, introduction of substituents, cleavage of substituents, or other (enzymatic) modifications can easily be monitored.

First detailed reports on the separation of non-derivatized sialic acids deal with the application of Aminex A-28 or A-29 anion-exchange chromatography using 0.75 mM [269, 270] or 0.5 mM [268] Na₂SO₄ as eluting system and UV monitoring at 200 nm (nanomole range). In a different approach, fluorigenic derivatives of sialic acids, prepared by reaction

with 1,2-diamino-4,5-methylenedioxybenzene (DMB) in the presence of 2-mercaptoethanol and sodium hydrogensulfite, have been separated by C18 reversed-phase HPLC [271-273], using acetonitrile/methanol/water (9:7:84, v/v/v) as solvent system and fluorescence monitoring at 373 nm excitation and 448 nm emission wavelengths. An appropriate cutoff filter may be used instead. The fluorescence labelling makes a relatively specific and highly sensitive (femto- to picomole range) detection possible. However, using radiolabelled sialic acids, it was found that the derivatization reaction is not quantitative [268]. For an adapted protocol, see ref. [11]. Another interesting approach is the conversion of Neu5Ac/5Gc into chemiluminescent quinoxalinone derivatives using 4,5-diaminophthalhydrazide dihydrochloride (α -keto acid derivatization) [274]. These derivatives are analyzed by reversed-phase HPLC (femtomole range), whereby the chemiluminescence detection follows the reaction of the derivatives with hydrogen peroxide in the presence of potassium hexacyanoferrate(III) in alkaline solution. Other conversions of Neu5Ac, useful for HPLC separations, include the derivatization with 4,4'-dicarboxy-2,2'-biquinoline [275], 2-cyanoacetamide [276], periodatethiobarbituric acid [277], benzoic anhydride [278], 4'-hydrazino-2-stilbazole [279] and 1,2-diamino-4,5-dimethoxybenzene [280].

Taking advantage of the separation capacity of the anion-exchange resin CarboPac PA-1, fractionation of non-derivatized sialic acids at neutral pH, using sodium acetate as eluent and pulsed amperometric detection (PAD) following postcolumn addition of alkali, has shown excellent results in terms of the wide array of sialic acids that can be separated, the sensitivity of the detection method (picomole range), and the relative ease of use for preparative work (without PAD detection) [268]. The general problem of quantification in PAD analyses, due to differences in detector response attributed to differences in the number of free hydroxyl groups of the various components separated, holds also for sialic acids. So far, only for a limited number of sialic acids relative detector response factors have been calculated (e.g. Neu5Ac, 30 500; Neu5,9Ac₂, 14 500; Neu5Gc, 35 400). It is important to note that pH values >11, as usually applied in CarboPac-PAD analyses of oligosaccharides (for such a Neu5Ac/Neu5Gc separation, see ref. [281]), will lead to rapid de-O-acylation of the O-acylated sialic acids [282]. This phenomenon can even occur between the point of postcolumn alkali addition and the entry into the PAD detector [268].

In an evaluation of five different HPLC methods, it turned out that no single method is adequate to completely separate and quantitate complex mixtures of sialic acids [268], and the use of multi-dimensional HPLC is advised. As a clear illustration of this statement, Table 8 is included. This evaluation also compares a series of essential features of the five HPLC methods, namely, sensitivity, specificity of detection, separation by number of hydroxyl groups or substituents, separation of isomers, preparative use, avoiding of ester migration during purification, and avoiding of ester loss during purification. In this comparison [268,270,271,283,284] the HPLC systems I (CarboPac PA-1) and III (TSK-ODS 120T, DMB derivatives) gave the highest averaged scores in terms of applicability. A major advantage of HPLC system V (Aminex A-29) is the short running time, only 5–6 min, which makes this approach highly attractive for studying enzymatic conversions.

In order to obtain information about the structure of sialic acids, HPLC is a very useful

1 a Пp Ш¢ IV^d Ve Compound Neu2Me 0.17 0.13 0.23 0.40 0.85 0.46 Neu1,2Me₂ Neu5Ac 1.00 1.00 $1.00/1.00^{f}$ 1.00 1.00 0.76 1.68/1.71 f Neu4,5Ac₂ 0.39 1.22 1.41 Neu5,7Ac₂ 0.74 0.36 1.06 0.95 0.87 0.95 0.35 1.57/1.62^f Neu5,9Ac₂ 1.30 1.47 Neu5,7(8),9Ac3 0.74 0.23 1.90 1.17 1.76 Neu5,7,8,9Ac₄ 1.98 0.62 1.62 2.05 Neu5Ac2Me 0.73 1.02 0.34 0.73 Neu4.8an5Ac 2.15 0.58 1.39 2.04 Neu2en5Ac 2.21 0.66 1.54 1.68 Neu5Gc 1.50 $0.84/0.78^{f}$ 1.20 1.17 1.33 Neu4Ac5Gc 0.86 0.54 1.53 1.59 1.69 Neu9Ac5Gc 1.30^f 1.06 0.49 1.57 1.70 1.03/0.98^f Neu5Gc8Me 0.82 0.75 0.94 0.89 Neu9Ac5Gc8Me 1.70 1.21 0.67 0.36 Neu7,9Ac25Gc8Me 0.59 0.29 2.20 1.47 Kdn 0.73^{f} Acetic acid 0.08 1.70

Table 8 High-performance liquid chromatographic elution times of sialic acids in five different HPLC systems I to V, relative to Neu5Ac [268]

^a System I: Anion-exchange chromatography on CarboPac PA-1 with PAD detection at room temperature; 5 mM sodium acetate for 5 min, then a linear gradient to 50% 5 mM sodium acetate/50% 5 mM acetic acid in 30 min; mixing of the column effluent with 300 mM NaOH; running time 18–24 min [268].

^b System II: Amine adsorption/ion suppression chromatography on Micropak AX-5 with UV monitoring (200 nm) at room temperature; acetonitrile/water/0.25 M sodium dihydrogenphosphate (72:18:6, v/v/v); running time 21–24 min [283].

^c System III: Reversed-phase chromatography on TSK-ODS 120T and fluorescence monitoring (373 nm excitation and 448 nm emission) at room temperature; DMB derivatives; acetonitrile/methanol/water (9:7:84, v/v/v); running time 12–13 min [271].

^d System IV: Anion-exchange chromatography on Aminex HPX-72S with UV monitoring (200 nm) at 40°C; 0.1 M sodium sulfate; running time 15–17 min [284].

^e System V: Anion-exchange chromatography on Aminex A-29 with UV monitoring (200 nm) at room temperature; 0.75 mM sodium sulfate; running time 5–6 min [36,270].

^f Taken from [11], RP-18 column.

technique to be applied in combination with mild chemical or enzymatic degradation methods. For instance, HPLC before and after alkaline treatment of a mixture of O-acylated sialic acids can give information about the de-O-acylated sialic acids present, e.g. in terms of their N-acyl substituents. Also the linkage specificity of sialidases in releasing sialic acids from sialoglycoconjugates (see section 5.1) can be monitored by HPLC. Furthermore, the specificity of enzymes involved in sialic acid metabolism can be studied in this way. In this respect, interesting results have been obtained

with aldolase, cleaving sialic acids to N-acylmannosamine derivatives and pyruvate, and with sialate 9-O-acetylesterase, hydrolyzing O-acetyl groups from C9 of sialic acids. The aldolase degrades Neu5Ac faster than Neu5Gc; a slow degradation has been observed for O-acylated sialic acids, not affecting 4-O-acetylated sialic acids at all [5]. A typical example of the HPLC analysis of enzyme reactions, in which esterase and aldolase are involved, and including the non-enzymatic conversion of an O-acetyl group from C7 to C9, is presented in Fig. 5 [6,7,36]. Other examples are the determination of sialidase activity (sialyllactose as substrate, Neu5Ac as product, and Neu2en5Ac as inhibitor), CMP-Neu5Ac synthase activity (disappearance of Neu5Ac, appearance of CMP-Neu5Ac), CMP-Neu5Ac phosphodiesterase activity (appearance of Neu5Ac, disappearance of CMP-Neu5Ac) [36]. Recently, 5-N-acetyl-9-O-acetyl- $2-(N-\text{dansyl-4-aminophenylthio})-\alpha$ -neuraminic acid has been proposed as a highly sensitive fluorescent substrate for the HPLC measurement of sialate 9-O-acetylesterase (334 nm excitation, 564 nm emission) [285]. As a thioglycoside, the compound is very stable in acidic aqueous solution and towards enzymatic hydrolysis by sialidases. In this context, it is also worthwhile mentioning that a sensitive HPLC assay has been developed for the tracing of sialyltransferase activity, making use of the synthetic fluorigenic acceptor lactose 2-[(2-pyridyl)amino]ethyl glycoside [286]. Details for a HPLC separation of CMP-Neu5Ac, CMP-Neu5Gc and CMP-Kdn on a DC-613 cation-exchange column are reported in ref. [107].

In addition to the fractionation procedures described for free sialic acids, several approaches have been reported for the separation of sialyl-oligomers. These compounds with a degree of polymerization up to 16 have been fractionated with varying results using conventional gel-filtration, TLC, DEAE-Sephadex A-25, and HPLC methods (see section 5.1 for preparation; see section 5.3.1 for TLC). A survey of literature has been included in ref. [287]. In general, mixtures of sialo-oligomers from $\{\operatorname{Neu5Ac}(\alpha 2-8)\}_n, \{\operatorname{Neu5Ac}(\alpha 2-9)\}_n, \{\operatorname{Neu5Gc}(\alpha 2-8)\}_n, \text{ or } \{\operatorname{Kdn}(\alpha 2-8)\}_n \text{ can be}$ isolated on a preparative scale via convential DEAE-Sephadex A-25 [256,257] or DEAE-Toyopearl 650M [267] anion-exchange chromatography. HPLC procedures comprise anion-exchange and adsorption-partition chromatography. A mixture of {Neu5Ac($\alpha 2$ - $\{8\}_{2-16}$ has been efficiently separated on a Zorbax SAX anion-exchange column using 0.2-1 M NaCl in 10 mM phosphate buffer pH 3.5 [287]. Also adsorptionpartition chromatography on polystyrene DC-613 using mixtures of 0.02–0.025 M sodium phosphate buffer pH 7.4 and acetonitrile as solvent system, has shown good results [267]. On Mono Q anion-exchange columns, excellent results were obtained in the separation of $\{\operatorname{Neu5Ac}(\alpha 2-8)\}_{2-14}$, $\{\operatorname{Neu5Gc}(\alpha 2-8)\}_{2-10}$, or $\{\operatorname{Kdn}(\alpha 2-8)\}_{2-7}$ after conversion into alditols with $NaBH_4$ (or $NaBT_4$), and using a NaCl gradient in Tris-HCl buffer pH 8 as elution system [267,288]. In this context, several studies have focused on the determination of the chain lengths of sialo-oligomers and -polymers (for a review of methods currently employed in the analysis of polysialic acids, see ref. [289]), and recently a highly detailed adapted methodology for the analysis of α -2,8-linked sialooligomers and -polymers has appeared [290]. Using three variable assay procedures, providing overlapping information, details could be provided with respect to the degree of polymerization, the simultaneous identification of Neu5Ac, Neu5Gc and Kdn when present in a single preparation, and the ability to distinguish qualitatively between



Fig. 5. HPLC profiling (Aminex A-28, $0.75 \text{ mM Na}_2\text{SO}_4$) of the chemo-enzymatic conversion of Neu5,7Ac₂ into pyruvate and N-acetyl-D-mannosamine (ManNAc). (a) Intramolecular migration of the O-acetyl group from C7 to C9 under slightly alkaline conditions, yielding Neu5,9Ac₂, accompanied by some de-O-acetylation; (b) enzymic release of the O-acetyl group at C9 with the aid of sialate-9-O-acetylesterase, yielding Neu5Ac; (c) aldolase (lyase)-catalyzed degradation of Neu5Ac, yielding pyruvate and ManNAc.

reducing and non-reducing polymers. The developed approach may include mild periodate oxidation (degradation of non-reducing terminal unit) in combination with reduction (degraded glycerol side chain yielding C7-sialic acid; reducing unit if present affording the corresponding stereoisomeric alditols), whereas monomer analysis is carried out after sialidase or acid hydrolysis on CarboPac PA-1 with pulsed amperometric detection.

In the structural analysis of glycoprotein-derived N- and O-linked sialic-acid-containing carbohydrate chains, fractionation procedures based on HPLC play a major role. As this aspect is outside the scope of this chapter, no details are included. Typical examples, making use of anion-exchange chromatography (Mono Q, CarboPac), and normal (e.g. Lichrosorb-NH₂) or reversed-phase chromatography, are found in refs. [25,81,84,133,291–300]. In this context, it is also worth noting the recent use of high-performance capillary electrophoresis for the separation of glycoprotein-derived N-glycan chains [301] and O-glycan chains [302].

5.3.3. Gas-liquid chromatography combined with mass spectrometry

As discussed in section 5.1, methanolysis of free and glycosidically bound sialic acids gives rise to the formation of methyl ester β - and α -methyl glycosides. Using the conditions of the standard quantitative monosaccharide analysis of glycoconjugates, de-*N*-acylation/de-esterification takes place, which means that *N*,*O*-acylneuraminic acid residues are converted into neuraminic acid methyl ester methyl glycoside (β , ~96%; α , ~4%). For the characterization by GLC the sialic acid methyl ester methyl glycoside is derivatized via *N*-acetylation/trimethylsilylation or pertrifluoroacetylation [241]. It should be noted that during the *N*-acetylation step HO9 of Neu5Ac, when not substituted, is partially *O*-acetylated (~4%). Using milder methanolysis conditions [242], an *N*-acetylation step is not necessary, yielding a method to determine both Neu5Ac and Neu5Gc by GLC. In principle, the latter approach is also suitable to determine sialic acids bearing only *O*-alkyl groups. GLC analysis is generally carried out on SE-30 type column materials.

Starting from free sialic acids (purified or as a pool), mainly present in their β -anomeric forms, volatile sialic acid derivatives are generated using mild derivatization procedures such as esterification with diazomethane followed by trimethylsilylation [15,303] or pertrimethylsilylation [11,304]. With respect to silylation cocktails, it should be noted that *N*-methyl-*N*-trimethylsilyl-2,2,2-trifluoroacetamide/pyridine leads to the formation of *N*-trimethylsilyl derivatives, yielding two different peaks for each sialic acid [36,305]. Subsequent GLC analysis is generally carried out on SE-30 or OV-17 type column materials. Both types of derivatives are highly suitable for MS analysis, and GLC coupled with electron impact (EI) MS formed the basis for the development of a highly reliable mass spectrometric method for the identification of sialic acids [303], the method has also proved to be useful for the analysis of other naturally occurring sialic acids, of (partially) *O*-methylated sialic acid methyl ester methyl glycosides as obtained in methylation analyses, and of synthetic sialic acid(s) (derivatives) [6,11,15].

In the following the principles of the EI MS identification procedure will be explained. Typical derivatives are trimethylsilylated methyl ester or pertrimethylsilylated derivatives of *N*,*O*-acylneuraminic acids or of *N*-acyl-*O*-alkylneuraminic acids, acetylated *N*-acyl-



Fig. 6. Survey of the selected fragment ions A–H worked out for the following derivatives: trimethylsilylated methyl ester or pertrimethylsilylated derivatives of N,O-acylneuraminic acids or of N-acyl-O-alkylneuraminic acids, and trimethylsilylated/methylated N,N-acyl,methyl-neuraminic acid methyl ester methyl glycosides [15].

O-alkylneuraminic acid methyl ester methyl glycosides, trimethylsilylated/methylated *N*,*N*-acyl,methyl-neuraminic acid methyl ester methyl glycosides (methylation analysis), and acetylated/methylated *N*,*N*-acyl,methyl-neuraminic acid methyl ester methyl glycosides (methylation analysis). The determination of the type, number, and position of the *O*-acyl or *O*-alkyl groups as well as the type of the *N*-acyl group in neuraminic acids is facilitated by the highly specific EI mass spectra of the derivatized compounds. In Fig. 6, a schematic survey is depicted showing the selected fragment ions A–H, which furnish the information (abundances and *m*/*z* values of the ions) necessary to deduce the complete structure of the sialic acids. Fragments A and B indicate the molecular mass of the sialic acid derivatives and thereby the type and the number of substituents. Fragment A is formed from the molecular ion by the elimination of a methyl group originating from a trimethylsilyl substituent in trimethylsilylated (*O*-acylated/*O*-alkylated) *N*-acylneuraminic acid derivatives. When $R_{5'} = CH_3$ (methylation analysis), the eliminated methyl group can

also originate from the N,N-acyl, methyl group. Fragment B is formed by elimination of the C1 part of the molecule. Eliminations of OCOCH₃ in O-acylated sialic acid derivatives and of NH₂COCH₃ in N-acetylneuraminic acid derivatives, which in principle give rise to the same m/z value as fragment B in the case of $R_1 = CH_3$, can be neglected. For O-trimethylsilylated N,O-acylneuraminic acids (\beta-anomers) it holds that, when compared to their methyl esters, in their trimethylsilyl esters the intensity of fragment A decreases relative to B. Fragment C is formed by elimination of C8-C9, with localization of the charge on position 7. In general, cleavage occurs between two alkoxylated carbon atoms, or between an acetoxylated and an alkoxylated carbon atom, rather than between two acetoxylated carbon atoms. In accordance with the fragmentation rules for partially methylated alditol acetates [306], the charge is preferentially located on an ether oxygen instead of on an ester oxygen. Therefore, fragment C has only significant abundance if C7 bears an ether group. When an ester group is present at C7, this fragment ion is absent or hardly observable. Fragment D is formed from fragment C by consecutive eliminations of R_2OH and R_4OH . It is evident that the occurrence of this fragment ion is dependent on the presence of fragment C. Fragment E is formed by elimination of the whole side-chain C7-C8-C9 and the substituent at C5. This fragment ion is not observed if an O-acyl group is attached to C4, illustrating that the transition state in the McLafferty rearrangement is more favored when the substituent at C4 is an ether group rather than an ester group. For O-trimethylsilylated NO-acylneuraminic acids (β -anomers) it holds that, when compared to their methyl esters, in their trimethylsilyl esters the intensity of fragment E is much reduced but still present; instead, an additional fragment derived from fragment E by loss of Me₃SiOH is clearly present. Fragment F contains C8-C9. Based on the same fragmentation rules as mentioned above for fragments C and D, this ion can only readily be formed if an ether group is attached to C8. Fragment G consists of the C4-C5 part of the molecule. Fragment H, necessary to use for derivatives containing only O-alkyl substituents, is formed by elimination of the C9 part, followed by elimination of R_4OH and R_7OH . For instance, this fragment is useful to discriminate between an OSiMe₃ group at C8 or C9 in trimethylsilylated partially methylated N-acylneuraminic acids. Finally, for quadrupole analyzers, in the high mass range the fragment ions A, B and C often are of low intensity, especially when only small amounts of material are available.

In Table 9 a survey is presented of GLC retention times and of characteristic EI MS fragment ions for a series of naturally occurring sialic acids, analyzed as their trimethylsilylated methyl ester or as their pertrimethylsilylated derivatives [6,11,15,31,38,43,307]. Although the sialic acids predominantly occur in the β -anomeric form, the α -anomer could occasionally be detected separately from the β -anomer, in most cases as a small shoulder. As a typical example, in Figs. 7a,b the EI mass spectra of the trimethylsilylated methyl ester of β -Neu5Ac and of the pertrimethylsilylated derivative of β -Neu5Ac, respectively, are depicted. Additional spectra have been published in refs. [15,43,304,307]. For a detailed survey of fragment ions of other derivatives mentioned, including mass spectra, and those obtained from periodate-oxidized sialic acids (C7-Neu5Ac, C8-Neu5Ac, C7-Neu5Gc and C8-Neu5Ac), see refs. [15,308]; for EI MS data of Neu4,8an5Ac, see ref. [63]; for EI MS data of permethylated Kdn, see ref. [46].

In additional studies, the suitability of chemical ionization (CI) for the GLC-MS analysis of pertrimethylsilylated N,O-acylneuraminic acids has been investigated. Isobu-
									P								Dofeee co(a)
Sialic acid	K _{Ne}	eu5Ac							Fragme	nt (m/z)							Reference(s)
	TM	PT		<u>4</u>		3	(<u> </u>)]	E		F DT	(<u> </u>	
	· · · · · · · ·		1M	P1	IM	P1	IM	P1	IM	PI	IM	P1	IM	P1	IM	P1	
Neu5Ac	1.00	1.00	668	726	624	624	478	536	298	356	317	375	205	205	173	173	[11,15]
Neu4,5Ac ₂	1.18	1.05	638	696	594	594	448	506	298	356	-	-	205	205	143	143	[11,15]
Neu5,7Ac ₂	1.04		638		594				-		317		205		173		[15]
Neu5,8Ac ₂	1.05		638		594		478		298		317		-		173		[15]
Neu5,9Ac ₂	1.13	1.02	638	696	594	594	478	536	298	356	317	375	175	175	173	173	[11,15]
Neu4,5,9Ac3	1.31		608		564		448		298		-		175		143		[15]
Neu5,7,9Ac3	1.14		608	666	564	564	-	-		-	317	375	175	175	173	173	[6,15]
Neu5,8,9Ac3	1.19	1.04	608	666	564	564	478	536	298	356	317	375			173	173	[11,15]
Neu5,7,8,9Ac ₄	1.15		578		534						317		-		173		[11,15]
Neu5Ac9Lt	2.55		740	798	696	696	478	536	298	356	317	375	277	277	173	173	[15,307]
Neu4,5Ac ₂ 9Lt	3.01		710		666		448		298		-		277		143		[15]
Neu5Ac8Me		0.98		668		566		536		356		375		147		173	[11,31]
Neu5,9Ac ₂ 8Me		1.00		638		536		536		356		375		117		173	[11,31]
Neu2en5Ac	1.09	1.01	578	636	-	-	388	446	298	356	227	285	205	205	-	-	[11,15]
Neu2,7an5Ac				564		462		374		-		-		205		173	[6,38]
Neu5Gc	1.81	1.19	756	814	712	712	566	624	386	444	317	375	205	205	261	261	[11,15]
Neu4Ac5Gc	2.02		726	784	682	682	536	594	386	444	-	-	205	205	231	231	[6,15]
Neu7Ac5Gc	1.83		726		682		-		-		317		205		261		[15]
Neu9Ac5Gc	2.04	1.21	726	784	682	682	566	624	386	444	317	375	175	175	261	261	[11,15]
Neu7,9Ac ₂ 5Gc	2.01		696		652		-		-		317		175		261		[15]
Neu8,9Ac ₂ 5Gc	1.99		696		652		566		386		317				261		[15]

Table 9 GLC and characteristic EI MS fragment ions (70 eV) of (i) trimethylsilylated methyl ester (TM) and (ii) pertrimethylsilylated (PT) derivatives of naturally occurring sialic acids (β-anomers)^a

							Ta	able 9, <i>c</i>	ontinued	ł							
Sialic acid	R _{Ne}	eu5Ac							Fragme	nt (<i>m/z</i>)							Reference(s)
	TM	PT		٩	I	3	(2	1)	ł	E]	F	(3	
			TM	РТ	ТМ	РТ	ТМ	PT	ТМ	PT	TM	PT	TM	РТ	TM	РТ	
Neu7,8,9Ac ₃ 5Gc	1.93		666		622		_		-		317		_		261		[15]
Neu5Gc8Me		1.14		756		654		624		444		375		147		261	[11,31]
Neu9Ac5Gc8Me		1.17		726		624		624		444		375		117		261	[11,31]
Neu5GcAc		1.21		784		682		594		414		375		205		231	[43]
Neu2en5Gc				724		-		534		444		285		205		-	[6]

^a The R_{NeuSAc} values of the TM derivatives on 3.8% SE-30 at 215°C are given relative to the TM derivative of β -NeuSAc.

The R_{NeuSAc} values of the PT derivatives on CP-Sil 5 (capillary column), using the program 5 min/140°C; 2°C/min up to 220°C; 15 min/220°C, are given relative to the PT derivative of β -Neu5Ac.

For the preparation of TM derivatives, see [15]; for the preparation of PT derivatives, see [304]. For an explanation of the minus signs, see text.





Fig. 7. (A) EI mass spectrum (70 eV) of the trimethylsilylated methyl ester of β -Neu5Ac; (B) EI mass spectrum (70 eV) of the pertrimethylsilylated derivative of β -Neu5Ac.

tane [304], as well as methane and ammonia [11] were used as reactant gases. The CI mass spectra are characterized in the high mass range by $[M+H]^+$ pseudomolecular ions, and typical major fragment ions derived from $[M+H]^+$ by loss of R₂OH (fragment I), R₄OH (fragment I'), and R₂OH+R₄OH (fragment II). It was found that methane in particular gave CI spectra that also include several of the typical fragment ions observed in the EI spectra.

In addition to GLC-MS, HPLC-CI MS with Aminex A-29 as column material and ammonium formate in water/acetonitrile as solvent system has been explored for the analysis of underivatized *N*,*O*-acylneuraminic acids [6,309]. Although the positive-ion mass spectra allow the discrimination between different *N*-acylneuraminic acids (Neu5Ac, Neu5Gc) and the determination of the degree of *O*-acetylation (Neu5,9Ac₂, Neu5,7,9Ac₃, Neu5,7,8,9Ac₄), the position of the *O*-acetyl groups (Neu4,5Ac₂, Neu5,7Ac₂, Neu5,9Ac₂) could not be established. For the latter assignment, the combination with specific elution positions of standards on the HPLC column is advised. In the interpretation of the various fragment ions, the open-chain structure of the sialic acids has been generally used.

Neu5Ac and Neu5Gc have also been converted into phosphatidylethanolamine dipalmitoyl derivatives, and after separation by HPTLC and subsequent isolation, the sialic acid derivatives were analyzed by liquid secondary ion MS [281]. In both cases intense $[M-H]^-$ ions together with sodium attachment ions were detected. For the detection of Neu5Ac on human tumor mucin, after liberation with sialidase, electrospray MS has been used [310].

5.3.4. Fast atom bombardment mass spectrometry

Free sialic acids, isolated after cleavage from glycoconjugate starting material, have been investigated, without derivatization, by FAB MS using 5% aqueous acetic acid solutions for loading into glycerol on the FAB target [41]. The positive and negative FAB mass spectra of each sialic acid showed clear $[M+H]^+$ and $[M-H]^-$ pseudomolecular ions, respectively. Sialic acid mixture analysis (µg range) made the recognition of subgroups of sialic acids with the same molecular mass possible (e.g. Neu5Ac(Ac)₁, Neu5Ac(Ac)₂, Neu5Ac(Ac)₃). However, a differentiation between positional isomers was not possible.

Sialic acids were also studied after derivatization, which improves the sensitivity [41]. Direct peracylation failed to produce suitable derivatives, but reduction under acidic conditions followed by peracylation (perdeuteroacetylation or perpropionylation) gave good results. Generally, the sialic acids give rise to two major pseudomolecular ions, corresponding to the peracylated open chain form and an open-chain-derived lactone form, and a minor pseudomolecular ion corresponding to an open-chain-derived anhydroform (2,6 and/or 4,8). As the lactone peak is markedly reduced in the spectrum of Neu4,5Ac₂, HO4 seems to be mainly involved in the lactonization. In the case of sialic acid mixtures, a fast sialic acid subgroup analysis based on molecular masses is possible; again, a differentiation between positional isomers cannot be achieved. Careful analysis of the negative FAB spectra of reduced and perpropionylated sialic acids in mixtures demonstrated that these spectra could also be used for quantitative purposes. As worked out for mixtures of Neu5Ac, Neu5,9Ac₂ and Neu5Gc, an estimate of the relative amounts of these sialic acids can be given with an error of 10-15%, when the sum of the intensities

of the $[M + H]^+$ ions of the linear and the lactone forms of each component is compared, taking into account that the molar response of Neu5Gc is approximately 50% of that of Neu5Ac.

In order to generate sialic-acid-derived compounds, which can be used to differentiate between positional isomers, use has been made of the rather difficult periodate oxidation under mild or more rigorous conditions [53,308]. Under both conditions, the resulting aldehyde groups were derivatized with *p*-amino-benzoic acid ethyl ester, reductively introduced at acidic pH without loss of the native *O*-acetyl functions [41]. Sialic acids treated in this way were additionally reduced and peracylated, and then analyzed by FAB MS. Mixtures of products with different ring sizes (original, lactonized, anhydro) and/or open chain forms, depending on the substitution pattern, are often obtained. Of the mono-*O*-acetylated *N*-acylneuraminic acids Neu4,5Ac₂, Neu5,7Ac₂, Neu5,9Ac₂, Neu4Ac5Gc and Neu9Ac5Gc were investigated, however, no attention was paid to the behavior of Neu5,8Ac₂. Neu5,7(8),9Ac₃ and Neu7,9Ac₂5Gc were also included in these investigations.

Although not discussed in this chapter, FAB MS is widely used in the characterization of glycoprotein-derived N- and O-linked sialic-acid-containing carbohydrate chains. Typical information can be found in refs. [311,312].

5.3.5. ¹H NMR spectroscopy

Since the introduction of high-resolution ¹H NMR spectroscopy for the structural analysis of glycoprotein-derived glycans, a huge amount of NMR data have been generated, and highly detailed reviews on N-linked [76] and O-linked [77] carbohydrate chains have appeared. The continuous expansion in the amount of data has made it necessary to develop computerized search programs, and, in connection with the still growing Complex Carbohydrate Structural Database (CARBBANK), attention has been paid to the development of a NMR-spectroscopic data base of carbohydrate structures, called SUGABASE [313]. Sialic-acid-containing oligosaccharides/glycopeptides constitute a considerable majority of the glycoprotein glycans. In addition to the two reviews mentioned above, a specific review focusing on the NMR spectroscopy of sialic acids has also been published [16].

Free as well as glycosidically bound sialic acid give rise to highly characteristic ¹H NMR parameters. The ¹H NMR spectra are generally recorded in D_2O , and because of the pH dependency of the proton chemical shifts, the spectral data are standardized at pD 6–7. The choice of the pH is also of importance in view of the earlier discussed de-*O*-acylation, *O*-acyl migration, and autohydrolysis.

As a typical example of a free sialic acid, in Fig. 8 the 500 MHz ¹H NMR spectrum of Neu5Ac in D₂O at pD 7 is depicted. The spectrum shows a minor and a major set of protons, reflecting the subspectra of the α - and β -anomer of Neu5Ac, respectively ($\alpha:\beta=7:93$), and especially the H3e,3a signals, resonating outside the bulk signal, can be used for the differentiation between both anomers. The effect of pH on the proton chemical shifts is clearly illustrated by the positions of the Neu5Ac H3e,3a resonances. At pD 1.4, the H3e and H3a signals of β -Neu5Ac resonate at δ 2.313 and δ 1.880, respectively, whereas these values are δ 2.208 and δ 1.827 at pD 7. In the case of α -Neu5Ac, at pD 1.4, the resonances are found at δ 2.718 and δ 1.705,



Fig. 8. Resolution-enhanced 500 MHz ¹H NMR spectrum of Neu5Ac dissolved in D₂O recorded at pD 7 and 27°C.

respectively, and at pD 7.0 at δ 2.730 and δ 1.621, respectively. Over the years, a large number of (naturally occurring) sialic acids and related derivatives have been analyzed, and in Table 10 a survey of the chemical shift values of a selected group of compounds, including naturally occurring sialic acids and CMP-sialic acids, is presented [16,31,38,51,64,107,308,314,315]. Inspection of these data reveals the typical influences of the substituents on the chemical shift values of the skeleton H-atoms. Comparison of the H3a,3e resonances of Neu5Ac and Neu5Gc indicate more downfield positions for those of Neu5Gc ($\Delta \delta \approx +0.02$). Additional data for Neu5Ac, Neu5,7Ac₂, Neu5,9Ac₂, Neu5,7,9Ac₃, and Neu5,8,9Ac₃ in 0.1 M sodium phosphate/D₂O at 37°C and pD 7–7.5 are reported in ref. [23]. The ¹H NMR studies of free N.O-acylneuraminic acids have shown that the anomeric equilibrium of a 7-O-acetylated sialic acid differs strongly from that of the sialic acids not substituted at C7. In contrast to the normal equilibrium values of $\alpha:\beta \approx 7:93$, both Neu5,7Ac₂ and Neu5,7,9Ac₃ have an equilibrium ratio of about 23:77. Additional ¹H NMR data of methyl glycosides and methyl ester methyl glycosides of N_iO -acylneuraminic acids and some sialyllactoses can be found in ref. [16]. The ¹H NMR data of the (methyl ester) methyl glycoside of β -Kdn have been reported in ref. [46], and those of CMP-9amino-Neu5Ac and CMP-9NAc-Neu5Ac in ref. [316].

¹H NMR spectroscopy has shown to be an excellent method to monitor chemical and biochemical conversions of sialic acids, directly in the NMR-tube or by analysis of isolated reaction products. A typical example is the demonstration of the release of α -Neu5Ac as the primary product of bacterial and viral sialidase action on Neu5Ac(α 2-glycosides and oligosaccharides [317–320]. The initial formation of α -Neu5Ac, as traced by ¹H NMR spectroscopy, formed an excellent probe to investigate the kinetics of the mutarotation of Neu5Ac by means of 1 H NMR analysis in dependency of the pH [321]. At pD 5.4 the establishment of the equilibrium of mutarotation turned out to be rather slow, but at higher and lower pD values a more rapid establishment was observed, so that at pD 1.3 and pD 11.7 mutarotation was too fast to be measured. With the ability to generate α -Neu5Ac in situ, the aldolase-catalyzed degradation of Neu5Ac to pyruvate and N-acetylmannosamine (ManNAc) could also be investigated in more detail using ¹H NMR spectroscopy [322]. Using sialidase (pH optimum 5.4) and aldolase (pH optimum 7.2) from C. perfringens and N-acetyl-2-azido-2-deoxy- α -neuraminic acid as substrate at pH 5.4, only released α -Neu5Ac was found to be consumed by the aldolase, yielding specifically α -ManNAc followed by a fast mutarotation to α,β -ManNAc. These findings confirmed earlier work using Neu5Ac($\alpha 2$ -3)lactose as α -Neu5Ac-generating system and crystalline β -Neu5Ac [319]. In the reversed reaction α -ManNAc is the substrate [322]. For more details about the aldolase-catalyzed degradation, see section 9.3. In connection with these studies, it has to be noted that under comparable conditions the activities of sialidase and aldolase in D_2O are only about 50% of those in H_2O . With respect to O-acetyl migrations, also the earlier mentioned (see section 2) spontaneous conversion at physiological pH of Neu5,7Ac2 into Neu5,9Ac2 and of Neu5,7,9Ac3 into Neu5,8,9Ac₃ has been monitored by ¹H NMR spectroscopy [23]. Furthermore, the NMR approach has shown its value in the determination of substrate specificities of various sialidases using substrates with differently linked sialic acid residues [323-325].

Chemical shift Sialic acid pD Ref. H3a H3e H4 H5 H6 H7 H8 H9 H9′ 4Ac 5Ac 7Ac 8Ac. 9Ac 5Gc 8Me β-Neu5Ac 2 1.880 2.313 4.067 3.931 4.056 3.556 3.750 3.841 3.619 2.053 [16] --- α -Neu5Ac 2.718 n.d.^a 3.85 3.684 3.53 3.75 3.85 3.62 2.036 2 1.705 [16] — 3.899 β-Neu5Ac 1.827 2.208 4.024 3.984 3.514 3.753 3.835 3.608 2.050 [16] 7 _ _ α -Neu5Ac 1.621 2.730 n.d. n.d. n.d. n.d. n.d. 2.030 7 n.d. n.d. -[16] β-Neu4,5Ac₂ 7 1.951 2.249 5.274 4.15 4.15 3.570 3.775 3.844 3.619 2.065 1.992 [16] β-Neu5,7Ac₂ 2.236 3.950 3.767 4.246 5.045 3.911 3.629 3.444 1.976^b 2.144^b [16] 4 1.905 ----- α -Neu5.7Ac₂ 1.649 2.757 n.d. n.d. n.d. n.d. 1.947^b 2.128^b n.d. n.d. n.d. [16] 4 ____ ----_ β -Neu5,9Ac₂ 1.833 2.221 4.024 3.913 3.991 3.571 3.977 4.365 4.187 2.057 2.119 [16] 7 _ ----_ α -Neu5.9Ac₂ 1.624 2.720 n.d. n.d. n.d. n.d. n.d. 7 n.d. n.d. n.d. n.d. [16] _ -4.293 5.162 4.106 1.981^b 2.134^b β-Neu5,7,9Ac₃ 2.303 3.978 3.775 4.140 4.106 2.106 [16] 2 1.924 --------_ 1.956^b n.d. ^b α -Neu5,7,9Ac₃ 2.751 n.d. n.d. n.d. n.d. 2 1.686 n.d. n.d. n.d. n.d. [16] ---_ _ β-Neu5,8,9Ac₃ 2.189 3.903 3.780 3.838 5.114 4.528 4.287 2.057 2.089 2.105 7 1.838 3.978 _ _ _ [16] 1.862 2.250 4.006 3.912 3.830 3.866 5.115 4.545 2.059 2.091 β-Neu5,8,9Ac₁ 2 4.287 --2.107 [16] Neu2en5Ac 6 5.690 4.470 4.051 4.213 3.601 3.936 3.885 3.646 2.068 [16] -----_ _ _ Neu2,7an5Ac^c 7 2.167 2.007 3.953 3.919 4.543 4.434 3.537 3.592 3.755 2.035 [38] _ ---Neu4,8an5Ac^d 2.844 4.188 4.333 3.852 3.496 3.363 3.814 2.041 7 2.983 3.734 [64] _ 2.143 1.723 4.249 n.d. 3.439 3.268 3.690 2.041 [64] n.d. n.d. ____

Table 10 ¹H Chemical shifts for different types of sialic acids. Chemical shifts are given in ppm relative to internal acetone in D_2O (δ 2.225) at 300K, unless indicated otherwise

							Table I	0, contin	iueu								
Sialic acid	pD							Che	emical s	hift							Ref.
		H3a	H3e	H4	Н5	H6	H7	H8	H9	H9′	4Ac	5Ac	7Ac	8Ac, 8Me	9Ac	5Gc	
β-Neu5Ac2P	7	1.548	2.403	4.093	3.888	4.239	3.386	4.028	3.883	3.581	-	2.045		-	-	-	[16]
CMP-β-Neu5Ac	8	1.639	2.484	4.066	3.92	4.141	3.456	3.92	3.90	3.622	-	2.054	-	-	-	-	[16]
CMP-β-Neu5Ac9Ac ^e	7	1.64	2.48	n.d.	3.96	n.d.	3.51	4.07	n.d.	n.d.	-	2.05	-	-	2.08	-	[314]
C8-β-Neu5Ac	6	1.814	2.203	3.998	3.892	3.772	3.722	3.655 ^f	-	-	-	2.055		-	_	-	[308]
								3.586 ^f									
C7-β-Neu5Ac	6	1.821	2.238	3.992	3.710	3.800	3.671 ^g	-	-	_	-	2.044	_	-	-	-	[308]
							3.634 ^g										
C7-α-Neu5Ac	6	1.599	2.630	n.d.	n.d.	n.d.	n.d.		-	-	-	2.026	-	-		_	[308]
β-Neu5Gc	7	1.840	2.243	4.127	4.002	4.106	3.549	3.777	3.821	3.613	-	-	-	-	-	4.143	[16]
α-Neu5Gc	7	1.644	2.749	n.d.	n.d.	n.d.	n.d.	n. d .	n.d.	n.d.		-	-	-	-	4.12	[16]
β-Neu9Ac5Gc	7	1.842	2.234	4.14	4.006	4.109	3.570	3.970	4.365	4.183	-	-	-	-	2.115	4.144	[16]
α-Neu9Ac5Gc	7	1.649	2.751	n.d.	n.d.	n.d.	n.d.	n.d.	n.d.	n.d.	-	-	-		n.d.	4.123	[16]
β-Neu5Gc8Me	7	1.863	2.219	4.110	3.983	4.017	3.559	3.432	3.932	3.652	-	-	-	3.425		4.133	[31]
α-Neu5Gc8Me	7	1.643	2.551	n.d.	n.d.	n.d.	n.d.	n.d.	n.d.	n.d.	-			n.d.	-	4.125	[31]
CMP-β-Neu5Gc ^e	7	1.64	2.48	3.92	4.03	n.d.	3.42	3.60	3.87	3.60	-	-	-	_	-	4.11	[314]
C8-β-Neu5Gc	6	1.832	2.218	4.097	3.976	3.887	3.709	3.657 ^f	-	_	-		-	_	-	4.144	[308]

3.589^f

Table 10, c	ontinued
-------------	----------

Sialic acid	pD							Ch	emical s	hift							Ref.
		H3a	H3e	H4	Н5	H6	H7	H8	H9	H9′	4Ac	5Ac	7Ac	8Ac, 8Me	9Ac	5Gc	
C8-α-Neu5Gc	6	1.622	2.665	n.d.	n.d.	n.d.	n.d.	n.d.	_		_	-	_	-	-	4.129	[308]
C7-β-Neu5Gc	6	1.840	2.253	4.096	3.798	3.906	3.674 ^g	-	~	-	-	-	-	-	-	4.133	[308]
							3.632 ^g										
C7-α-Neu5Gc	6	n.d.	2.650	n.d.	n.d.	n.d.	n.d.	-		-	-	-	-	-	-	4.116	[308]
β-Kdn	7	1.837	2.265	4.015	3.603	3.995	3.881	3.772	3.883	3.675	-	-	-	-	-	-	[106]
β-Kdn9Ac ^h	7	1.768	2.158	3.94	3.575	3.94	3.91	3.91	4.376	4.238	-	-	-	-	2.124	-	[51]
CMP-β-Kdn ^h	7	1.568	2.379	3.968	3.538	4.046	3.863	3.896	3.731	3.618		-		-	-	-	[107]

^a n.d., not determined.

^b On one line values may have to be interchanged.

^c Sialic acid occurs in ${}^{5}C_{2}$ conformation.

^d Sialic acid occurs in ${}^{7}C_{4}$ conformation and is present in two tautomeric forms; chemical shifts are relative to HOD at δ 4.750; H3a=H3 and H3e=H3'.

^e Values are assigned relative to the HOD signal at δ 4.81 at 296 K.

 $^{\rm f}$ H8 and H8'.

^g H7 and H7'.

^h Personal communication Y. Inoue; values are assigned relative to

2,2-dimethyl-2-silapentane-5-sulphonate in D_2O (set to 0 ppm). Note that in CMP- β -Kdn the H8 and H9 signals have been interchanged when compared with ref. [315].

·····			
Structural element	H3a	H3e	
Neu5Ac(α2–3)Gal(β1-	1.78–1.81	2.75-2.78	
Neu4,5Ac ₂ (α 2–3)Gal(β 1-	1.93	2.77	
Neu5,9Ac ₂ (α 2–3)Gal(β 1-	1.80	2.76	
Neu5Gc($\alpha 2$ -3)Gal($\beta 1$ -	1.81-1.82	2.77-2.79	
$Kdn(\alpha 2-3)Gal(\beta 1-4)GlcNAc(\beta 1-$	1.75-1.76 ^a	2.71–2.72 ^a	
$Kdn(\alpha 2-3)Gal(\beta 1-3)GalNAc(\alpha 1-3)GalNAc-ol$	1.72ª	2.69 ^a	
Neu5Ac($\alpha 2$ -3)[Gal($\beta 1$ -4)]Gal($\beta 1$ -	1.82 ^a	2.73 ª	
Neu5Ac(α 2-3)[GalNAc(β 1-4)]Gal(β 1-	1.92-1.94	2.66-2.68	
Neu5Ac(α 2-6)Gal(β 1-	1.70-1.72	2.67	
Neu4,5Ac ₂ (α 2–6)Gal(β 1-	1.85	2.68	
Neu5,9Ac ₂ (α 2–6)Gal(β 1-	1.71	2.67	
Neu5Gc(α 2-6)Gal β 1-	1.73-1.74	2.69-2.70	
$Gal(\beta 1-3)[Neu5Ac(\alpha 2-6)]GlcNAc(\beta 1-2)Man(\alpha 1-6)]$	1.71-1.72	2.73-2.74	
Gal(β 1-3)[Neu5Ac(α 2-6)]GlcNAc(β 1-4)Man(α 1-	1.76-1.77	2.72-2.73	
Neu5Ac(α2–6)GalNAc(β1-	1.70-1.72	2.66	
Neu5Gc($\alpha 2$ -3)[GalNAc($\beta 1$ -4)]GalNAc($\beta 1$ -	1.85-1.86	2.55-2.56	
Kdn(α2–3)[GalNAc(β1–4)]GalNAc(β1-	1.79 ^a	2.48 ^a	
Neu5Ac(α 2–6)GalNAc-ol	1.69-1.71	2.72-2.74	
Neu5Gc(α2–6)GalNAc-ol	1.71-1.72	2.74-2.75	
Kdn(α2–6)GalNAc-ol	1.64-1.66	2.66-2.68	

¹H-Chemical shifts for the H3e and H3a signals of sialic acids as part of N- and O-linked glycoprotein glycans. Chemical shifts are given in ppm relative to internal acetone in D₂O (δ 2.225) at 300 K, or relative to internal 2,2-dimethyl-2-silapentane-5-sulphonate in D₂O set to 0 ppm (marked ^a)

Table 11

Another interesting pH phenomenon, for the first time demonstrated by ¹H NMR spectroscopy, is the complete replacement of H3a by a D-atom, when free Neu5Ac is kept in an alkaline D_2O solution at pD 9.0 [326]. In fact, H3a can be exchanged in the pH range 6.5–9.0, and the H–D exchange is reversible. In the ¹H NMR spectrum the H3a signal disappears, and the coupling patterns of H3e and H4 alter. At pD 12.4, H3e can also be replaced by a D-atom [327]. On the basis of this finding, the exchange experiments were also carried out in T₂O, yielding T-labelled Neu5Ac, which was converted enzymatically into T-labelled CMP-Neu5Ac [326]. In glycosidically linked Neu5Ac the H3 atoms are not exchangeable, rendering this specific labelling technique suitable for sialyltransferase experiments.

In the ¹H NMR structural-reporter-group concept, developed for the structural analysis of glycoprotein N- and O-glycans, advantage is taken of the fact that a number of the H-atoms of the constituting monosaccharides, occurring in special microenvironments, resonate outside the bulk-signal region [76,77,328]. In the case of glycosidically linked α -sialic acids the structural reporters are the H3a- and H3e-atoms, and the *N*-acetyl or *N*-glycolyl groups. The positions of the H3a,3e signals reflect

not only structural information with respect to the type of sialic acid present, but also with respect to the coupled monosaccharide in terms of type of linkage and type of monosaccharide. Furthermore, O-acyl substituents induce additional shifts for other H-atoms. In Table 11 a survey is presented of chemical shifts of H3e and H3a signals of sialic acid residues occurring in different linkage types as part of N- and O-glycans. For a further fine-tuning of the chemical shifts within the presented ranges. influenced by the different microenvironments wherein the sialic acid residues occur, see refs. [76,77] and the references cited in Table 2. It should be noted that the presence of a certain sialic acid in a certain linkage also influences the structural-reportergroup signals of other monosaccharide residues [76,77]. More detailed information with respect to the H3e,3a chemical shifts of sialic acids in sialo-oligomers and -polymers can be obtained from refs. [31,47,77,329,330]. A series of H3a,3e signals of sialic acids in milk- and glycolipid-derived oligosaccharides have been included in ref. [77]. More general NMR data of glycosidically linked sialic acids in glycolipids, milk and urinary oligosaccharides and (lipo)polysaccharides, if available, can be found in the references cited in Tables 3-5 (see also ref. [331]). For a series of sialocarbohydrates, it has been shown that H6 of Neu5Ac, easily traced from the TOCSY H3a,H6 correlation, also has potential value for discriminating between α -2,3- (δ 3.63 ± 0.011) and α -2,6- (δ 3.70 \pm 0.017) linked Neu5Ac in Neu5Ac(α 2–3/6)Gal(β 1–O)R/Neu5Ac(α 2– 6)GalNAc(α 1–O)Ser/Thr [331]. However, for branched oligosaccharides, this rule is not valid if the monosaccharide in the branching position is in the alditol form [332].

The H3a,3e/NAc structural reporters of Neu5Ac have proved to be suitable in studying sialylation reactions in terms of positional specificity and branching (N-linked carbohydrate chains) specificity, using different sialyltransferases and CMP-Neu5Ac as donor [333–335] (see also section 6.3).

In more biophysical studies, several aspects of sialic acids have been investigated by NMR spectroscopy. Although so far mainly α -forms of bound sialic acid have been detected, good differentiation systems for α - and β -forms are essential, and a number of empirical rules have been reported [336]. In a heteronuclear 2D-approach it could be demonstrated that the determination of the geminal C,H coupling constant ${}^{2}J(C2,H3a)$ offers a unique criterion for the anomeric assignment in sialic acid glycosides (α , -8 Hz; β , -3 to -4 Hz) [337]. Also the values of the vicinal C,H coupling constants ${}^{3}J(C1,H3a)$ can be applied for this differentiation (α , -6 Hz; β , -1 Hz) [55,336,338]. More details with respect to anomeric determinations have been reviewed in ref. [339].

¹³C NMR data of sialic acids and glycoprotein-derived sialocarbohydrates have been reviewed in ref. [16]. In several sialic-acid-related investigations, e.g. synthetic studies, ¹³C NMR spectroscopy forms part of the analysis techniques, and will not be reviewed in detail. For some additional data, more directly related with the glycoprotein glycan character of this chapter, see refs. [48,63,64,103,105,152,331,332,337,340–343].

6. Chemo-enzymatic highlights in sialic acid chemistry

During the last ten years, activity in sialic acid chemistry has grown exponentially. Synthetic as well as biosynthetic routes for the preparation of sialic acids, sialic acid derivatives, analogues, glycosides and sialoglycoconjugates have been explored. The main reason for this considerable interest in preparing sialic acids and sialic-acid-containing compounds lies in the fact that sialic acids were found to be among the most biologically important carbohydrate units in glycoconjugates. The progress in organic synthetic protocols and the availability of relevant enzymes in suitable amounts made it realistic to develop the sialic acid field from a preparative synthetic side. Initially, the main tools were to prepare suitable derivatives to study the properties of sialic acids, or to prepare substrates and inhibitors for sialidases, sialyltransferases or for sialic acid converting enzymes. Although these tools are still highly relevant, the preparation of sialo-oligosaccharides using strictly organic synthetic or enzymatic methods, or a mixture of both, are also receiving considerable attention. For relevant reviews on preparative (bio)synthetic aspects, see refs. [339,344–353].

6.1. Free sialic acids

Several protocols have been followed for the organic chemical synthesis of Neu5Ac. At first, the approaches were based on condensation reactions of (derivatives of) N-acetyl-D-mannosamine (ManNAc) or N-acetyl-D-glucosamine (GlcNAc) with (derivatives of) oxaloacetic acid (for a review, see ref. [3]), however, the yields were very low. In this context, a procedure was worked out, that allowed modifications at C1-C3 [354]. A total synthesis of Neu5Ac from non-carbohydrate precursors has been reported in ref. [355]. Using a protocol for indium-mediated allylations of aldehydes, Neu5Ac was prepared in good yields from a ManNAc precursor [356]. Furthermore, synthetic routes for Neu5Ac have been proposed, based upon 1-deoxy-1-nitro-sugar chemistry, that should also allow the preparation of Neu5Ac analogues, modified at several carbon atoms of the skeleton [357-359]. A separate route also yielded Neu5Ac [360]. A synthesis starting with the aldol condensation of D-glucose (Glc) and oxaloacetic acid, followed by adaptation of the substituent at C5 has been described in ref. [361]. More recently, another approach for the organic synthesis of Neu5Ac and Neu5Ac derivatives, based on the cis-selective Wittig reaction of benzoyl 2,3-O-isopropylidene-a-D-lyxo-pentodialdo-1,4-furanoside with [(3S)-3,4-(isopropylidenedioxy)butyl]-triphenylphosphonium iodide as a first step, has been reported in ref. [362]. Comments on the acetylation of Neu5Ac and its methyl ester have been published in ref. [363]. Starting from Neu5Ac, in some derivatization reactions 1.4- as well as 1.7-lactone formation has been observed [349]. For the organic synthesis of Kdn, several routes were employed [364-368], among them procedures starting from Neu5Ac or from D-mannose (Man).

Neu5Ac-aldolase-catalyzed condensations of ManNAc and pyruvate [3,33], initially only investigated to understand sialic acid metabolism, have been optimized for preparative purposes. In principle, ManNAc can be generated from the cheaper GlcNAc in an alkaline epimerization process, yielding an epimeric mixture of which only the monosaccharide with the D-manno-configuration is recognized by the aldolase [369,370]. However, ManNAc can also be generated from GlcNAc in a GlcNAc-epimerasecatalyzed isomerization [371]. A multigram-scale enzymatic synthesis based on the aldol condensation of ManNAc and pyruvate in the presence of phosphate, catalyzed by immobilized microbial Neu5Ac-aldolase, has been reported [372]. A similar approach, using the aldolase enclosed in a dialysis membrane instead of being immobilized, has also been described [373]. Although *in vivo*, the conversion of Neu5Ac into Neu5Gc occurs exclusively on the level of activated sialic acid (see section 8.4.1), Neu5Gc can be prepared *in vitro* by incubating a mixture of *N*-glycolyl-D-mannosamine/*N*-glycolyl-D-glucosamine and pyruvate with immobilized aldolase [374]. Interestingly, a series of other sugars also turned out to be accepted by the aldolase, and Man and 2-deoxy-D-glucose in particular are excellent substrates [372,375–377]. In the case of Man as starting material, relatively moderate amounts of Kdn have been prepared [254,376].

A series of free O-acetylated sialic acids, i.e., Neu4,5Ac₂, Neu5,9Ac₂, Neu4,5,9Ac₃, and Neu5,7,8,9Ac₄, together with the benzyl ester α -glycosides of Neu5,7Ac₂ and Neu5,7,9Ac₃, have been synthesized by organic synthetic routes using protecting group techniques [378,379]. Partially O-acetylated sialic acid derivatives have also been prepared using more simple synthetic routes. β -Neu5,9Ac₂1,2Me₂, β -Neu4,5,9Ac₃1,2Me₂, and β -Neu4,5,8,9Ac₄1,2Me₂ were obtained from β -Neu5Ac1,2Me₂ by using N-acetylimidazole [53]. To realize 9-O-acetylations, also other acetylating reagents were applied, such as trimethyl orthoacetate [380,381], acetyl chloride [378], and dimethylacetamide dimethyl acetal [382]. In a recent comprehensive study, in particularly the use of trimethyl orthoacetate and dimethylacetamide dimethyl acetal was explored using the 4-aminophenylthio, 4-nitrophenylthio, and 4-nitrophenyl glycosides of COOH-esterified α -Neu5Ac as acceptors, and depending on the acetylating reagent a range of partially O-acetylated derivatives could be generated [57].

One of the naturally occurring *O*-acetylated sialic acids, Neu5,9Ac₂, has also been synthesized in an enzymatic way [383,384] on a gram-scale [372]. After the enzymatic acetylation of O6 of ManNAc, using isopropenyl acetate and protease N as a catalyst, 2-*N*-acetyl-6-*O*-acetyl-D-mannosamine was condensed with pyruvate as catalyzed by the aldolase. These two enzymatic steps turned out to be highly regio- and stereoselective. Following another route, Neu5,9Ac₂ has been synthesized enzymatically by incubating Neu5Ac with trichloroethyl acetate in pyridine using porcine pancreas lipase as a catalyst [385]. An enzymatic synthesis of Neu5Ac9Lt has also been worked out [370,384].

For the study of biochemical pathways, several isotopically labelled sialic acids and sialic acid derivatives have been prepared, both by enzyme-catalyzed synthesis and by organic synthesis. A survey of labelled sialic acids is presented in Table 12 [107,344,386]. In enzymatic procedures, use is generally made of the aldolase-catalyzed condensation of N-acyl-D-mannosamines and (phosphoenol)pyruvate, suitably labelled in one or both of the two synthons. For the preparation of $N-[1-^{14}C]$ acetyl- and $N-[1-^{14}C]$ glycolylneuraminic acid, as well as O-[14C]acetylated sialic acids, surviving slices of submaxillary salivary glands incubated with [1-14C]acetate, followed by isolation of the glycoprotein fraction and mild acid hydrolysis, have been used. A number of these labelled sialic acids have been converted into their CMP-glycosides, and subsequently incorporated into glycoconjugates (see section 6.3). Of course, labelling of glycoconjugates can also be carried out by periodate oxidation/tritiated borohydride reduction, thereby converting sialic acids, if chemically possible, into their radiolabelled C7 and C8 analogues. Using the latter approach, fluorescent probes (dansylhydrazine, dansylethylenediamine, fluoresceinamine) and EPR spin labels can also be incorporated (see references cited in ref. [344]). The same holds for glycine [387].

Sialic acid	Reference	
N-[³ H]Acetylneuraminic acid	[344]	
N-[1- ¹⁴ C]Acetylneuraminic acid	[344]	
N-Acetyl-[3-3H]neuraminic acid	[344]	
N-Acetyl-[9-3H]neuraminic acid	[386]	
N-Acetyl-[1- ¹⁴ C]neuraminic acid	[344]	
N-Acetyl-[4- ¹⁴ C]neuraminic acid	[344]	
N-Acetyl-[2- ¹⁴ C,9- ³ H]neuraminic acid	[344]	
5-N-Acetyl-9-azido-9-deoxy-[1- ¹⁴ C]neuraminic acid	[344]	
5-N-Acetyl-4-O-methyl-[3- ³ H]neuraminic acid	[344]	
5-N-[1-14C]Acetyl-2-deoxy-2,3-didehydro-neuraminic acid	[344]	
N-[1- ¹⁴ C]Glycolylneuraminic acid	[344]	
N-Glycolyl-[1- ¹⁴ C]neuraminic acid	[344]	
N-Glycolyl-[2-14C,9-3H]neuraminic acid	[344]	
[¹⁴ C]-2-Keto-3-deoxynononic acid	[107]	

Table 12 Survey of radiolabelled sialic acids^a

^a For specific references, see [344].

Several interesting sialic acid variants and sialic acid derivatives have been synthesized, and a list is presented in Table 13 [316,344,347,349,357-359,368,370,376-380,384,388-446]. A number of these compounds have been surveyed in ref. [350]. Both organic synthetic and aldolase-catalyzed routes have been followed. The major part of these compounds were prepared to study sialic acid metabolism (aldolase, CMP-Neu5Ac synthase), sialic acid transfer (sialyltransferases), sialic acid release (sialidases), inhibition phenomena, or hemagglutinin-sialic acid interactions, and biological details are presented in sections 8-10. Compounds reported up to 1982 have been reviewed earlier [344]. Of special interest are the fluorescent and photoactivatable sialic acid derivatives [390,419], which can be applied, after conversion into their corresponding CMP-glycosides, to detect enzyme activities or to follow biological processes (see sections 6.2 and 8.2). In the context of sialic acid variants, the following compounds are also of interest. In view of the similarity in acidity of a tetrazole group and a carboxyl function, a variant of Neu5Ac has been prepared containing a CN₄H instead of a COOH group [447]. Also the synthesis of a series of Neu5Ac derivatives with specifically introduced tert-butyldimethylsilyl groups have been reported [400,448]. Furthermore, variants of 2d-2Hax-Neu5Ac and 2d-2Hea-NeuSAc, in which the carboxyl function has been replaced by a phosphono (PO_3H_2) group [449], and a phosphonic acid analogue of Neu2en5Ac [450], have been synthesized. In addition to 6-amino-2,6-dideoxy-sialic acids, as mentioned in Table 13, the preparation of 2-C-hydroxymethyl derivatives [451], and C6 and C7 analogues [452] have also been reported.

Compound	Abbreviation	Reference(s)
N-Acetylneuraminic acid	Neu5Ac	see text
5-N-Acetyl-2-deoxy-2,3-didehydro-neuraminic acid	Neu2en5Ac	[344]
2-Deoxy-2,3-didehydro-neuraminic acid	Neu2en	[350]
N-Glycolylneuraminic acid	Neu5Gc	see text
2-Deoxy-2,3-didehydro-5-N-glycolyl-neuraminic acid	Neu2en5Gc	[344]
5-Azido-neuraminic acid	Neu5N ₃	[388]
5-Azido-2-deoxy-2,3-didehydro-neuraminic acid	Neu2en5N ₃	[350]
2-Keto-3-deoxynononic acid	Kdn	see text
2,3-Didehydro-2,3-dideoxy-D-glycero-D-galacto-non-2-ulopyranosonic acid	Kdn2en	[389]
N-Acetyl-[1- ¹³ C]neuraminic acid	[1- ¹³ C]Neu5Ac	[344]
N-Acetyl-[3- ² H]neuraminic acid	[3- ² H]Neu5Ac	[344]
N-Acetyl-[6- ² H]neuraminic acid	[6- ² H]Neu5Ac	[359]
N-Aminoacetyl-neuraminic acid	Neu5AcNH ₂	[390]
N-Thioacetyl-neuraminic acid	Neu5AcSH	[390,391]
5-N-Acetyl-2,7-anhydro-neuraminic acid	Neu2,7an5Ac	[349]
5-N-Acetyl-2-deoxy-2-H _{ax} -neuraminic acid	2d-2H _{ax} -Neu5Ac	[392–394]
5-N-Acetyl-2-deoxy-2-H _{eq} -neuraminic acid	2d-2H _{eq} -Neu5Ac	[393–395]
5-N-Acetyl-2-deoxy-2-Hea-4-oxo-neuraminic acid	2d-2H _{eq} -40xo-Neu5Ac	[396]
5-N-Acetyl-2-deoxy-4-epi-neuraminic acid	2d-4epi-Neu5Ac	[397]
5-N-Acetyl-2-deoxy-7-epi-2-H _{eq} -neuraminic acid	2d-7epi-2H _{eq} -Neu5Ac	[394,398]
5-N-Acetyl-2-deoxy-8-epi-2-Heg-neuraminic acid	2d-8epi-2H _{eq} -Neu5Ac	[394,398]
5-N-Acetyl-2-deoxy-7,8-diepi-2-H _{eq} -neuraminic acid	2d-7,8epi ₂ -2H _{eq} -Neu5Ac	[394,398]

Table 13 List of sialic acids, prepared along organic chemical or aldolase-catalyzed routes for use in biochemical studies^a

Table 15, commacu		
Compound	Abbreviation	Reference(s)
5-N-Acetyl-3-fluoro-neuraminic acid	3F _{eq} -Neu5Ac	[344,392]
5-N-Acetyl-3-hydroxy-neuraminic acid	3OH _{eq} -Neu5Ac	[344,399]
5-N-Acetyl-4-O-acetyl-neuraminic acid	Neu4,5Ac ₂	[378,380]
5-N-Acetyl-4-O-acetyl-4-epi-\beta-neuraminic acid methyl ester methyl glycoside	4epi-βNeu4,5Ac ₂ 1,2Me ₂	[400]
5-N-Acetyl-4-deoxy-neuraminic acid	4d-Neu5Ac	[358,401– 403]
5-N-Acetyl-4-deoxy-4-iodo-neuraminic acid	4I-Neu5Ac	[404]
5-N-Acetyl-4-deoxy-4-(R)-C-methyl-neuraminic acid		[405]
5-N-Acetyl-4-deoxy-4-(S)-C-methyl-neuraminic acid		[405]
5-N-Acetyl-4-epi-neuraminic acid	4epi-Neu5Ac	[357,406]
5-N-Acetyl-4-epi-4-O-methyl-neuraminic acid ethyl ester	4epi-Neu5Ac1Et4Me	[344]
5-N-Acetyl-4-O-methyl-β-neuraminic acid (ethyl ester/ethyl glycoside)	Neu5Ac4Me	[344,407]
5-N-Acetyl-4-oxo-neuraminic acid (methyl β -glycoside or ethyl α -glycoside)	40xo-Neu5Ac	[408,409]
5-N-Acetyl-7-deoxy-neuraminic acid	7d-Neu5Ac	[377,410]
5-N-Acetyl-7-epi-neuraminic acid	7epi-Neu5Ac	[411,412]
5-N-Acetyl-7-O-methyl-neuraminic acid	Neu5Ac7Me	[377]
5-N-Acetyl-7-oxo-β-neuraminic acid methyl glycoside	70x0-βNeu5Ac2Me	[408]
5-N-Acetyl-8-deoxy-neuraminic acid	8d-Neu5Ac	[410]
5-N-Acetyl-8-epi-neuraminic acid	8epi-Neu5Ac	[411,412]
5-N-Acetyl-8-O-methyl-neuraminic acid	Neu5Ac8Me	[344]
5-N-Acetyl-8-oxo-neuraminic acid methyl α - and β -glycoside	80x0-Neu5Ac2Me	[394,408,413]
5-N-Acetyl-9-O-acetyl-neuraminic acid	Neu5,9Ac ₂	see text
5-N-Acetyl-9-S-acetyl-9-thio-neuraminic acid		[414]

Compound	Abbreviation	Reference(s)
5-N-Acetyl-9-amino-9-deoxy-neuraminic acid	9amino-Neu5Ac	[316,390, 415–417]
5-N-Acetyl-9-azido-9-deoxy-neuraminic acid (methyl α -glycoside)	9azido-Neu5Ac	[316,384,390, 417,418]
5-N-Acetyl-9-(4-azidobenzamido)-9-deoxy-neuraminic acid		[419]
5-N-Acetyl-9-(4-azidosalicylamido)-9-deoxy-neuraminic acid		[419]
5-N-Acetyl-9-benzamido-9-deoxy-neuraminic acid	9NBz-Neu5Ac	[316,390,415, 417]
5-N-Acetyl-9-(4-benzoylbenzamido)-9-deoxy-neuraminic acid		[419]
5-N-Acetyl-9-cyano-9-deoxy-α-neuraminic acid benzyl glycoside	9cyano-αNeu5Ac2Bn	[394]
5-N-Acetyl-9-O-(N-dansylglycyl)-neuraminic acid		[420]
5-N-Acetyl-9-deoxy-neuraminic acid (methyl α-glycoside)	9d-Neu5Ac	[394,410,421]
5-N-Acetyl-9-deoxy-9-(3-fluoresceinylthioureido)-neuraminic acid	9fluoresceinyl-Neu5Ac	[390,415,422, 423]
5-N-Acetyl-9-deoxy-9-fluoro-neuraminic acid	9F-Neu5Ac	[384,424– 426]
5-N-Acetyl-9-deoxy-9-iodo-neuraminic acid	9I-Neu5Ac	[344,427]
5-N-Acetyl-9-deoxy-9-thioacetamido-neuraminic acid		[390,391]
5-N-Acetyl-9-O-(dimethylphosphinyl)-neuraminic acid		[384]
5-N-Acetyl-9-O-glycyl-neuraminic acid methyl ester	Neu5Ac9Gly1Me	[344]
5-N-Acetyl-9-hexanoylamido-9-deoxy-neuraminic acid	9NHx-Neu5Ac	[316,390,417]
5-N-Acetyl-9-O-lactyl-neuraminic acid	Neu5Ac9Lt	[370,384]
5-N-Acetyl-9-O-methyl-neuraminic acid	Neu5Ac9Me	[347]
5-N-Acetyl-9-O-phosphoro-neuraminic acid	Neu5Ac9P	[344]

Table 13, continued

Compound	Abbreviation	Reference(s)
5-N-Acetyl-9-thio-neuraminic acid		[414]
5-N-Acetyl-4,9-di-O-acetyl-neuraminic acid	Neu4,5,9Ac ₃	[378]
5-N-Acetyl-2,7-dideoxy-2-H _{eq} -neuraminic acid	2,7d ₂ -2H _{eq} -Neu5Ac	[394,398]
5-N-Acetyl-2,8-dideoxy-2-H _{eq} -neuraminic acid	$2,8d_2-2H_{eq}-Neu5Ac$	[394,398]
5-N-Acetyl-4,7-dideoxy-neuraminic acid	4,7d ₂ -Neu5Ac	[410]
5-N-Acetyl-7,9-dideoxy-neuraminic acid	7,9d ₂ -Neu5Ac	[421]
5-N-Acetyl-7,8-diepi-neuraminic acid (methyl α-glycoside)	7,8epi ₂ -Neu5Ac	[394,411,412]
5-N-Acetyl-7,7-dimethoxy-β-neuraminic acid methyl glycoside		[408]
5-N-Acetyl-8,8-dimethoxy-neuraminic acid methyl α - and β -glycoside		[394,408,413]
5-N-Acetyl-7,8,9-tri-O-acetyl-neuraminic acid	Neu5,7,8,9Ac ₄	[379]
5-N-Acetyl-4,7,9-trideoxy-neuraminic acid	4,7,9d3-Neu5Ac	[421]
5-N-Acetyl-4-allylamino-2,3-didehydro-2,4-dideoxy-neuraminic acid		[428]
5-N-Acetyl-4-amino-2,3-didehydro-2,4-dideoxy-neuraminic acid	4amino-Neu2en5Ac	[428–430]
5-N-Acetyl-4-azido-2,3-didehydro-2,4-dideoxy-neuraminic acid	4azido-Neu2en5Ac	[430,431]
5-N-Acetyl-9-S-(4-azido-2-nitrophenyl)-2,3-didehydro-2,9-dideoxy-9-thio-neuraminic acid		[432]
5-N-Acetyl-2-deoxy-2,3-didehydro-4-epi-neuraminic acid (methyl ester)	4epi-Neu2en5Ac	[344,433,434]
5-N-Acetyl-2-deoxy-2,3-didehydro-4-oxo-neuraminic acid	40xo-Neu2en5Ac	[409,433,435]
5-N-Acetyl-2-deoxy-2,3-didehydro-6-thio-neuraminic acid		[436]
5-N-Acetyl-2-deoxy-2,3-didehydro-7-epi-neuraminic acid	7epi-Neu2en5Ac	[350]
5-N-Acetyl-2-deoxy-2,3-didehydro-8-epi-neuraminic acid	8epi-Neu2en5Ac	[437]
5-N-Acetyl-2-deoxy-2,3-didehydro-7,8-diepi-neuraminic acid	7,8epi ₂ -Neu2en5Ac	[350]
5-N-Acetyl-2-deoxy-2,3-didehydro-8,8-dimethoxy-neuraminic acid		[413]

Table 13, continued

Table	13,	continued
	~~,	

Compound	Abbreviation	Reference(s)
5-N-Acetyl-2,3-didehydro-2,4-dideoxy-4-dimethylamino-neuraminic acid		[428]
5-N-Acetyl-2,3-didehydro-2,4-dideoxy-4-guanidinyl-neuraminic acid	4guanidino-Neu2en5Ac	[428,429]
5-N-Acetyl-4-(N-hydroxy-N-allylamino)-2,3-didehydro-2,4-dideoxy-neuraminic acid		[428]
5-N-Acetyl-4-(2-hydroxyethylamino)-2,3-didehydro-2,4-dideoxy-neuraminic acid		[428]
5-N-acetyl-2,3-didehydro-2,4-dideoxy-neuraminic acid	4d-Neu2en5Ac	[438]
5-N-acetyl-2,3-didehydro-2,7-dideoxy-neuraminic acid	7d-Neu2en5Ac	[350]
5-N-acetyl-2,3-didehydro-2,8-dideoxy-neuraminic acid	8d-Neu2en5Ac	[350]
5-N-acetyl-2,3-didehydro-2,9-dideoxy-neuraminic acid	9d-Neu2en5Ac	[350]
5-N-acetyl-2,3-didehydro-2,4,7-trideoxy-neuraminic acid	4,7d ₂ -Neu2en5Ac	[350]
4-Acetamido-5-N-acetyl-2,3-didehydro-2,4-dideoxy-neuraminic acid		[428]
5-N-Acetyl-6-amino-2-deoxy-2-Hax-neuraminic acid		[439]
5-N-Acetyl-6-amino-2-deoxy-2-Hax-4-epi-neuraminic acid		[439]
5-N-Acetyl-6-amino-2-deoxy-2-Heq-4-epi-neuraminic acid		[439]
5-N-Acetyl-6-thio-neuraminic acid		[436]
5-N-Acetyl-5-epi-6-thio-neuraminic acid		[436]
5-N-Acetyl-4,5-diepi-6-thio-neuraminic acid		[436]
9-O-Acetyl-5-N-glycolyl-neuraminic acid	Neu9Ac5Gc	[380]
4-Acetamido-5-N-acetyl-4-deoxy-neuraminic acid	4NAc-Neu5Ac	[440]
4-Acetamido-3,4-dideoxy-D-glycero-D-galacto-non-2-ulopyranosonic acid	iso-Neu4Ac	[441]
5-Acetamido-2,6-anhydro-3,5-dideoxy-2,3-difluoro-D-arabino-L-talo-nononic acid	2d-2F _{ax} -3F _{eq} -Neu5Ac	[392]
5-Acetamido-2,6-anhydro-3,5-dideoxy-4-C-methyl-D-erythro-L-altro-nononic acid		[396]
5-Acetamido-2,6-anhydro-3,5-dideoxy-4-C-methyl-D-erythro-L-manno-nononic acid		[396]

Compound	Abbreviation	Reference(s)	
5-Acetamido-2,6-anhydro-4-C-methyl-3,4,5-trideoxy-D-erythro-L-altro-nononic acid		[396]	
5-Acetamido-2,6-anhydro-4-C-methyl-3,4,5-trideoxy-D-erythro-L-manno-nononic acid		[396]	
5-Acetamido-2,6-anhydro-4-C-methylene-3,4,5-trideoxy-D-glycero-D-galacto-nononic acid		[396]	
5-Acetamido-2,5-dideoxy-2,3-difluoro-D-erythro-L-gluco-non-2-ulopyranosonic acid		[442,443]	
5-C-Acetamidomethyl-5-deamino-neuraminic acid		[444]	
9-Acetamido-5-N-acetyl-9-deoxy-neuraminic acid	9NAc-Neu5Ac	[316,390,415, 417]	
5-N-Glycolyl-9-O-phosphoro-neuraminic acid	Neu5Gc9P	[344]	
5-Bromo-3,5-dideoxy-D-glycero-D-galacto-non-2-ulopyranosonic acid	5Br-Kdn	[446]	
3-Deoxy-D-glycero-D-gulo-non-2-ulopyranosonic acid	5epi-Kdn	[376]	
3-Deoxy-D-glycero-D-talo-non-2-ulopyranosonic acid	4epi-Kdn	[368]	
3,5-Dideoxy-5-fluoro-D-glycero-D-gulo-non-2-ulopyranosonic acid	5epi-5F-Kdn	[384]	
3,5-Dideoxy-D-glycero-D-galacto-non-2-ulopyranosonic acid	5d-Kdn	[376,384,445]	
3,7-Dideoxy-D-glycero-D-galacto-non-2-ulopyranosonic acid	7d-Kdn	[376]	
3,7,9-Trideoxy-7,9-difluoro-D-glycero-L-altro-non-2-ulopyranosonic acid	7epi-7,9F ₂ -Kdn	[384]	

Table 13, continued

^a For literature references before 1982 and additional lists of sialic acid(s) (derivatives), see ref. [344].

The organic synthesis of a long series of alkyl and aryl α -glycosides of *N*-acylneuraminic acids has been previously reported (for reviews, see refs. [339,344]). One of the famous condensation reactions (classical Koenigs–Knorr method) comprised the silver carbonate-promoted condensation of 5-*N*-acetyl-4,7,8,9-tetra-*O*-acetyl-2-chloro-2-deoxy- β -neuraminic acid or the corresponding methyl ester (Fig. 9, structure A) with the appropriate alcohol, followed by removal of the protecting groups. In order to improve the yields, much attention has been paid to better catalysts. In these glycosidation reactions typical side reactions are the formation of unsaturated sialic acid derivatives (elimination of HCl) and of β -glycosides (see section 6.3). For the preparation of simple β -glycosides, *N*-acylneuraminic acids are often heated with the appropriate alcohol in the presence of an acid catalyst, followed by saponification of the formed ester. However, complex alcohols give rise to problems (for a review, see ref. [344]). A mild and efficient Raney nickel-catalyzed deuteration procedure has been reported for Neu5Ac glycosides, with a rate of exchange at C8 > C9 > C7 \gg C4 [453].

Attention has also been paid to the synthesis of N-, S- and Se-glycosides, which are sialidase stable [339,454,455]. Specific S-glycosides are used as sialic acid donors in sialoglycoconjugate organic synthesis (see section 6.3). Early examples are the syntheses of the 4-nitrophenyl N- and S-glycosides of α -Neu5Ac [456]. Of special interest are the syntheses of 5-N-acetyl-2-azido-2-deoxy- α - and β -neuraminic acids [322,457,458]. The azides can readily be converted into the corresponding 2-amino derivatives, and used in e.g. N-acylation reactions [459]. For a preparation of the 6-thioanalogue of 2azido- α -Neu5Ac, see ref. [436]. In further investigations, a series of S-glycosides of α -Neu5Ac was synthesized (thiophenyl, 4-nitrothiophenyl, 4-aminothiophenyl, 2-mercaptopyridyl), starting from 5-N-acetyl-4,7,8,9-tetra-O-acetyl-2-chloro-2-deoxy- β -neuraminic acid methyl ester and using triethylbenzylammonium chloride as a phase transfer catalyst [454]. These compounds turned out to be effective sialidase inhibitors.

For the detection of sialidase activity both (naturally occurring) oligosaccharides and simple α -glycosides are used. In these assays, two approaches can be followed, namely, determination of the released (modified) sialic acid or identification of the released aglycon. In the case of α -glycosides, in which the released aglycon concentration is measured spectrophotometrically or detected on solid supports, substrates with synthetically introduced aglycons having specific chromogenic properties, are used. Among such substrates, released aglycons can be detected directly or after condensation with specific reagents. One of the oldest substrates is the 4-nitrophenyl glycoside of α -Neu5Ac [456], whereby released 4-nitrophenol is estimated by absorption at 400 nm. In an adapted synthetic version, the compound has been prepared by coupling of 5-N-acetyl-4,7,8,9-tetra-O-acetyl-2-chloro-2-deoxy-β-neuraminic acid methyl ester with sodium-nitrophenoxide in NN-dimethylformamide, and subsequent deprotection [460]. Another suitable substrate is the 3-methoxyphenyl glycoside of α -Neu5Ac, synthesized by coupling of 5-N-acetyl-4,7,8,9-tetra-O-acetyl-2-chloro-2deoxy- β -neuraminic acid with 3-methoxyphenol in the presence of silver carbonate, followed by de-O-acetylation. Liberated 3-methoxyphenol is determined after coupling



Fig. 9. Frequently used Neu5Ac donors in the organic synthesis of sialo-oligosaccharides.

with the diazonium salt of 4-amino-2,5-dimethoxy-4'-nitroazobenzene (red colored product) [461] or with 4-aminoantipyrine in the presence of the oxidizing agent potassium ferricyanide (colored quinone) [462]. The most popular fluorigenic substrate is the 4-methylumbelliferyl glycoside of α -Neu5Ac, which is prepared by different methods [344]. A convenient synthesis is the condensation of 5-N-acetyl-4,7,8,9-tetra-O-acetyl-2-chloro-2-deoxy- β -neuraminic acid methyl ester with the sodium salt of 4-methylumbelliferone in N,N-dimethylformamide, followed by deprotection [463]. Released 4-methylumbelliferone is measured at 360 nm (excitation)/440 nm (emission). Although the 4-methylumbelliferyl glycoside of α -Kdn has been synthesized starting from Neu5Ac [445], also a direct route using the glycosyl chloride of peracetylated B-Kdn methyl ester and the sodium salt of 4-methylumbelliferone has been explored [254]. In addition, several 4-methylumbelliferyl α -glycosides of sialic acid variants and substituted sialic acids, including partial O-acetylated ones, have been synthesized (e.g. refs. [252,350,399,445,464]. To develop a sensitive assay for the analysis of the linkage specificity of bacterial and viral sialidases, Neu5Ac($\alpha 2$ -3)- and Neu5Ac($\alpha 2$ -6)Gal($\beta 1$ -O)C₆H₄NO₂ were synthesized enzymatically by using α -2,3- and α -2,6-sialyltransferase, respectively, CMP-Neu5Ac (see section 6.3), and p-nitrophenyl-\beta-Galp [465]; after cleavage of Neu5Ac, p-nitrophenol can be released by additional treatment with β-galactosidase.

For the localization of sialidase on electropherograms or for histochemistry, the chromogenic 5-bromo-indol-3-yl glycoside of α -Neu5Ac has been synthesized by coupling of 5-*N*-acetyl-4,7,8,9-tetra-*O*-acetyl-2-chloro-2-deoxy- β -neuraminic acid methyl ester with 1-acetyl-5-bromo-3-hydroxyindole, and subsequent deprotection [466]. The unstable intermediate 5-bromo-indoxyl, released by sialidase, is readily transformed into insoluble blue-green 5,5'-dibromo-indigo, which marks the sites of enzyme activity. To facilitate the screening of bacterial colonies or plaques for sialidase activity, the 5-bromo-4-chloro-indol-3-yl glycoside variant has also been synthesized [467]. Using the same sialic acid synthon as starting product, the 4-azido-2-nitrophenyl S-glycoside of α -Neu5Ac has been prepared, which is a potential photoaffinity probe reagent for the screening of sialidases in tissues and the purification of sialic-acid-binding proteins [468]. The sialidase-resistant thioglycosyl linkage also makes the incorporation of ³⁵S possible.

In order to detect sialate 9-O-acetylesterase activity, a highly sensitive fluorescent substrate, 5-N-acetyl-9-O-acetyl-2-[4-(dansylamino)phenylthio]- α -neuraminic acid, has been synthesized (see also sections 5.3.2 and 9.1) [285]. The regioselective acetylation at O9 of the dansylated S-glycoside was carried out with trimethyl orthoacetate. Other useful fluorescent substrates for sialate O-acetylesterase assays comprise 5-N-acetyl-7,8,9-tri-O-acetyl-2-[4-(dansylamino)phenylthio]- α -neuraminic acid and 5-N-acetyl-2-[4-(dansylamino)phenylthio]- α -neuraminic acid [469], and the 4-[3-(fluoresceinyl)thioureido]phenyl S-glycoside of α -Neu5,9Ac₂ [57]. In all cases the fluorescent groups have been coupled to the glycosidic 4-aminophenylthio group of O-acetylated Neu5Ac derivatives.

In connection with the generation of a monoclonal antibody to free Neu5Ac for the purpose of establishing a simple and specific assay of Neu5Ac in serum and urine, a broad series of sialic acid α - and β -glycosides have been synthesized using substituted glycerol, substituted sphinganine and cholesterol as aglycons [470].



Fig. 10. One-pot synthesis of CMP-β-Neu5Ac from ManNAc and CMP [473]. PEP, phosphoenolpyruvate; PPi, pyrophosphate.

The synthesis of CMP-sialic acids is generally carried out enzymatically using CTP and CMP-sialic acid synthase as a catalyst [33,314,370,471,472]. A multigram-scale one-pot synthesis of CMP- β -Neu5Ac has been reported in ref. [473]. ManNAc, prepared by base-catalyzed epimerization of GlcNAc, was reacted with sodium pyruvate in the presence of Neu5Ac-aldolase to yield Neu5Ac (see section 6.1). For the formation of CMP-Neu5Ac, CTP was generated *in situ* from CMP by using adenylate kinase, pyruvate kinase, and phosphoenolpyruvate, and reacted with Neu5Ac in the presence of CMP-Neu5Ac synthase (Fig. 10). Instead of a one-pot synthesis, for practical reasons it is easier to generate and store crude solutions of Neu5Ac and CTP. For the use of GlcNAc in combination with GlcNAc-epimerase, see ref. [474]. Experiments with cloned CMP-Neu5Ac synthases from *E. coli* systems with Neu5Ac and Kdn showed a high specificity for Neu5Ac, thereby suggesting that in this case the 5-acetamido group is critical [384]. Chemical syntheses of CMP-Neu5Ac, applying the phosphoramidite method [475] or using sialyl phosphites [476], have also been described. Furthermore, a synthetic approach for the preparation of CMP-Neu5Gc based on the phosphite method has appeared [477].

In addition to CMP- β -Neu5Ac, CMP- β -Neu5,9Ac₂, CMP- β -Neu5Gc, and CMP- β -Kdn, a large series of artificial CMP-sialic acids have been prepared biochemically on microscale starting from the corresponding sialic acid (see references cited in Table 13) and CTP. Among them are CMP-9azido-Neu5Ac, CMP-9amino-Neu5Ac, CMP-9NAc-Neu5Ac and other C9-modified CMP-sialic acids, CMP-Neu5AcNH₂, CMP-Neu5Ac4Me, and CMP-4d-Neu5Ac [33,314,350,390,407,419,478]. The CMP-sialic acids have found a broad application in enzymatic sialylations using different sialyltransferases (see sections 6.3 and 8.3). Several of the artificial CMP-sialic acids turned out to be suitable donors for asialo- α_1 -acid glycoprotein as acceptor with Gal(β_1 -4)GlcNAc α -2,6-sialyltransferase from rat liver as a biocatalyst [316,402,417]. The transfer of CMP-9amino-Neu5Ac is of considerable interest, as α -linked 9amino-Neu5Ac in sialoglycoconjugates is not a substrate for bacterial, viral or mammalian sialidases tested so far. CMP-9amino-Neu5Ac and CMP-Neu5AcNH₂ have also been used as synthons

to prepare fluorescent and photoactivatable analogues [419]. Because of the defined acceptor specificity, sialyltransferases in combination with fluorescent or photoactivatable donor CMP-sialic acids are excellent tools for selective introduction of a fluorescent or photoactivatable substituent to a distinct glycoconjugate. The latter reference [419] also includes kinetic data and information concerning the fluorimetric sialyltransferase assay. Typical fluorescent products comprise the CMP-sialic acids of 5-N-acetyl-9-deoxy-9-(3-fluoresceinylthioureido)-neuraminic acid (CMP-9fluoresceinyl-Neu5Ac), 5-N-acetyl-9-(7-amino-4-methylcoumarinyl)acetamido-9-deoxy-neuraminic acid (CMP-9AMCA-Neu5Ac), 5-N-acetyl-9-deoxy-9-(fluoresceinylaminomonochlorotriazinyl)amino-neuraminic acid (CMP-9MTAF-Neu5Ac), and N-(3-fluoresceinvlthioureido-acetyl)neuraminic acid (CMP-Neu5fluoresceinyl). In the preparation of photoactivatable derivatives the NH_29 group of CMP-9amino-Neu5Ac has been substituted with a 4-azidobenzovl. a 4-azidosalicyl, a 4-benzoylbenzoyl or a 4-azido[T]benzoyl group. In a similar way, CMP-Neu5AcNH₂ has been labeled with a 4-azidobenzoyl group. Of special interest is the recently reported chemical synthesis of CMP-{Neu5Ac($\alpha 2$ -8)Neu5Ac} [479]; an attempt to prepare this compound biosynthetically with Neu5Ac(α 2–8)Neu5Ac and CMP-sialic acid synthase failed so far [472]. In addition to the preparation of regular CMP-sialic acids, synthetic approaches have been worked out for the organic synthesis of a S-(N-acetylneuraminyl)nucleoside analogue [480,481] and other CMP-sialic acid variants [482].

For the immobilization of sialic acids on Sepharose solid supports, which provides potentially useful affinity materials, see the references cited in ref. [344]. The preparation of an affinity adsorbent with immobilized sialic acid through a thioglycosidic linkage has been described in ref. [483]. Synthetic sialidase-stable α -Neu5,9Ac₂ p-aminophenylthio glycoside has been immobilized directly or by a six-carbon long spacer group to agarose for lectin isolations [382]. The allyl glycoside of α -Neu5Ac has been applied as a starting material for the synthesis of Neu5Ac-neoglycoproteins and pseudopolysaccharides. These polymers containing multivalent sialic acid are in principle useful for various applications related with recognition/binding/inhibition processes. Reductive ozonolysis of the allyl group $(O_3, \text{ then Me}_2S)$, followed by coupling of the formed aldehyde to protein carriers (ɛ-aminogroup of lysine) by sodium cyanoborohydride-mediated reductive amination, yielded neoglycoproteins with varying amounts of Neu5Ac [484,485]. Copolymerization of the allyl glycoside with acrylamide generated a water-soluble pseudopolysaccharide [484]. In order to create a longer spacer arm for copolymerization with acrylamide, the allyl glycoside was converted into a 3-(2-aminoethylthio)propyl glycoside by reaction with cysteamine hydrochloride, after which the amino function was N-acryloylated [486]. The same principle of conjugation or copolymerization via N-acryloyl groups was also used for the preparation of sialo-oligosaccharide-neoglycoproteins and copolymers of sialo-oligosaccharides and acrylamide [487]. Using the strategy of reductive amination, *p*-formylphenyl glycoside of α -Neu5Ac was also conjugated with proteins [488], and starting from the *p*-nitrophenyl O- and S-glycosides, *p*-*N*-acryloylamino analogues were synthesized, which could be copolymerized with acrylamide, yielding water-soluble pseudopolysaccharides with Neu5Ac and acrylamide in different ratios [489], or directly coupled with polylysine [490]. Using trimethyl orthoacetate, the Neu5Ac residue of the

S-glycoside-containing polymer was converted into Neu5,9Ac₂ [489]. Finally, a series of interesting Neu5Ac and Neu5,9Ac₂-based dendrimers have been synthesized [491].

6.3. Sialo-oligosaccharides

The organic synthesis of oligosaccharides having terminal α -linked sialic acid has proved to be highly complex. The specific difficulties arise from three factors inherent in the sialic acid molecule. First, the carboxylic acid function at the anomeric center (C2) electronically disfavors oxonium ion formation. Secondly, from a steric point of view, the carboxyl function restricts the glycoside formation. Thirdly, the presence of a neighboring methylene group in the ring (C3), instead of a substituted carbon atom, eliminates the possible assisting and/or directing effect of an adjacent substituent [346]. This means that side reactions can be relatively important, mainly the thermodynamically favored β -glycoside and 2,3-dehydro-derivative formation, and low yields are quite often obtained. Initially, the synthesized glycosidic linkages comprised mainly Neu5Ac(α 2– 6)Gal(β 1–, Neu5Ac(α 2–6)GlcNAc(β 1–, Neu5Ac(α 2–6)Glc(β 1–, Neu5Ac(α 2–3)GlcNAc(β 1–[344], and over the years these glycosidic linkages, together with Neu5Ac(α 2–6)GalNAc(α 1–, still receive most of the attention. In view of the desire to prepare biologically relevant carbohydrate chains, this is understandable.

In Fig. 9 a series of typical Neu5Ac donors, introduced by different research groups with the aim of increasing both the glycosidation yield and the α -stereoselectivity, is depicted [339,346–348]. For each class of donors, some further information is presented in the following paragraphs.

The oldest approach of synthesizing sialo-oligosaccharides is the one starting from 2-deoxy-2-halo- β -Neu5Ac derivatives (Fig. 9, structure A). Methyl 5-acetamido-4,7,8,9-tetra-*O*-acetyl-2-chloro-2,3,5-trideoxy-D-*glycero*- β -D-*galacto*-non-2-ulopyranosonate, X = Cl, Y = Me, turned out to be a particularly useful donor [492], and typical promoters are silver and mercury salts. Due to poor stereoselectivity, α , β -glycoside mixtures are generally obtained, and HCl-elimination from the donor is a major side reaction. The reaction with secondary hydroxy groups in particular gave rise to problems. In the case of the aim to prepare Neu5Gc-containing oligosaccharides, also the *N*-glycolyl group in the donor analogue is *O*-acetylated [493]. For the synthesis of Neu5Ac(α 2–9)Neu5Ac, see ref. [494].

In another approach, a series of 3-substituted Neu5Ac donors was prepared, starting from peracetylated [495] or perbenzylated [496,497] Neu2en5Ac methyl ester, thereby making use of the highly reactive 2,3-double bond to form adducts (Fig. 9, structures B–F) (see also ref. [498]). In the case of structure B as donor with silver triflate as a promoter, only β -glycosidic linkages were created, and among several products, the Neu5Ac(β 2–8)Neu5Ac linkage was synthesized [495]. Structure C yielded mainly β -glycosidic linkages. From structure D, only the bromo variant is effective, although α , β -glycoside mixtures are still formed. The bromo variant of structure D with silver triflate as a promoter has been applied in the synthesis of Neu5Ac(α 2–8)Neu5Ac and Neu5Ac(α 2–9)Neu5Ac linkages [499,500]. Structures E and F form another series of donors, and E with X=Br and Y=SPh (mercury salts as promoter) has been shown to give particularly high glycosidation yields and α -stereoselectivity.

A third type of donor involves the use of S-methyl or S-phenyl α -glysosides (Fig. 9, structure G) [498,501-503]. Initially developed to synthesize S-glycosides, making use of sodium salts of the peracetylated Neu5Ac methyl ester α - or β -thioglycosides and suitable protected bromides [504-506] (see also refs. [507,508]), this type of donors has shown to be highly attractive in O-glycosidation reactions. In these couplings, frequently used promoters are dimethyl(methylthio)sulfonium triflate or N-iodosuccinimide/triflic acid [347]. The choice of the solvent system is very important, as it greatly influences the stereoselectivity; e.g. acetonitrile gives mainly α -glycosidation. In addition to reports dealing with the synthesis of many monosialo-oligosaccharides, including those with a Neu5Ac($\alpha 2$ -2)Glc, a Neu5Ac($\alpha 2$ -3)GlcNAc, and a Neu5Ac($\alpha 2$ -3)GalNAc sequence [509], typical examples are the creation of Neu5Ac(2-9)Neu5Ac [510,511] and Neu5Ac($\alpha 2$ -8)Neu5Ac[512] linkages, as well as sialyl Le^x sequences (native and variants) (ref. [513] and references cited therein). In this context, it is also interesting to note that several syntheses of sialo-oligosaccharides include the use of a separately prepared disaccharide donor with a terminal α -linked sialic acid [514]. Although in general the donors contain a N-acetyl group at C5, other examples have been reported with the phthaloyl (benzeneselenenyl triflate as a promoter [515]) or the *tert*-butoxycarbonyl function as N-protecting group, e.g. in the case of the synthesis of Neu-containing glycoconjugates [516,517]. Following similar routes, Kdn-containing oligosaccharides have also been synthesized [518]. As a variation on this theme, the application of S-sialyl xanthates (SCSOEt) as donors in sialo-oligosaccharide synthesis has led to interesting results, including a high α -stereoselectivity [519–522]. Here, also the use of O-benzoyl protection instead of O-acetyl protection has been proposed [523,524]. Furthermore, the preparation of sialic acid S-glycosyl donors employing $S_{S'}$ -bis(1-phenyl-1*H*-tetrazol-5yl)dithiocarbonate should be mentioned [525]. Another efficient donor combines parts of the structures F and G, yielding an α -thioglycoside with a SPh substituent at C3 (Fig. 9, structure H) [526]. The high stereoselective α -sialylation was obtained using either methyl sulfenyl bromide/silver triflate or N-iodosuccinimide/triflic acid as promoters.

Sialyl phosphites with trimethylsilyl triflate as a promoter have additionally been shown to be of practical use (Fig. 9, structure I), affording good yields and α -stereoselectivity [527-529], and examples include the synthesis of sialyl Le^x sequences [528]. For a detailed study on the evaluation of different sialyl phosphites, see ref. [530].

For the organic synthesis of sialo-oligosaccharides with di- or trimeric Neu5Ac elements, specific glycosyl donors have been prepared directly from Neu5Ac($\alpha 2$ –8)Neu5Ac or Neu5Ac($\alpha 2$ –8)Neu5Ac($\alpha 2$

In terms of preparative chemistry, the use of CMP- β -Neu5Ac as a glycosyl donor and (cloned) α -2,3/6-sialyltransferases as biocatalysts have achieved a permanent position in the planning of synthetic routes for sialo-oligosaccharide chains; see e.g. refs. [314,342, 345,351,472,473,535–544]. Especially in the field of the preparation of sialyl Le^x frag-



Fig. 11. Chemo-enzymic synthesis of the sialyl Le^x sequence [541]. NMK, nucleoside-monophosphate kinase; PEP, phosphoenolpyruvate; PK, pyruvate kinase; PPase, inorganic phosphatase; Pi, phosphate; PPi, pyrophosphate; PYR, pyruvate; TADH, *Thermoanaerobium brockii* alcohol dehydrogenase.

ments, excellent results have been obtained. To illustrate the enzymatic conversions, a series of typical examples will be presented. A first comprehensive example comprises the enzymatic sialylation on a microscale of oligosaccharides containing Gal(β 1–3)GlcNAc, Gal(β 1–4)GlcNAc, Gal(β 1–4)Glc, and Gal(β 1–3)GalNAc sequences by using different purified mammalian sialyltransferases [342], creating Neu5Ac(α 2–6)Gal, Neu5Ac(α 2–6)GlcNAc linkages, as verified by ¹H and ¹³C NMR spectroscopy (see section 5.3.5). The generation of the Neu5Ac(α 2–6)GalNAc(α 1–0) sequence on a microscale has been described in ref. [536]. A second example (Fig. 11) is the chemo-enzymatic synthesis on a preparative scale of Neu5Ac(α 2–3)Gal(β 1–4)[Fuc(α 1–3)]GlcNAc(β 1–0)CH₂CH=CH₂ (and analogues) using β -1,4-galactosyltransferase and recombinant α -2,3-sialyltransferase and α -1,3-fucosyltransferase with *in situ* regeneration of UDP-Gal, CMP-Neu5Ac and GDP-Fuc [541]. A third example is the one-pot enzymatic synthesis of Neu5Ac(α 2–6)Gal(β 1–4)GlcNAc (and analogues) based on a β -galactosidase-catalyzed galactosylation, using lactose as a donor and GlcNAc as

an acceptor, and a pig liver α -2,6-sialyltransferase-catalyzed sialylation with in situ regeneration of CMP-Neu5Ac [542]. A fourth example is the enzymatic synthesis of Neu5Ac(α 2-6)Gal(β 1-4)GlcNAc(β 1-O)pent-4-ene, a precursor for the organic chemical synthesis of higher oligosaccharides [472]. The trisaccharide was synthesized starting from GlcNAc(B1-O)pent-4-ene, UDP-Gal (in situ generated from UDP-Glc catalyzed by UDP-Gal 4-epimerase), and Neu5Ac in a one-pot reaction employing β -1,4-galactosyltransferase and α -2,6-sialyltransferase in a complete cofactor regeneration system. The availability of specific sialyltransferases will certainly contribute to a further expansion of this area. In this context, the recent finding of a novel sialyltransferase which catalyzes the transfer of Kdn from CMP-Kdn to the non-reducing termini of oligo/polysialyl chains, thereby capping a further elongation of a {Neu5Gc($\alpha 2$ -8)}_n chain, is of interest [545]. In the framework of the finding that two NeuSAc(α 2–6)Gal(β 1– 4)GlcNAc units are the receptor determinants for the influenza virus hemagglutinin, these elements have been systematically anchored on a Gal residue in order to design structures capable of bimodal viral binding, and along chemo-enzymatic routes heptasaccharides with the general formula Neu5Ac($\alpha 2$ -6)Gal($\beta 1$ -4)GlcNAc($\beta 1$ -x)[Neu5Ac($\alpha 2$ -6)Gal($\beta 1$ -4)GlcNAc(β 1-y)]Gal(β 1-O)(CH₂)₅COOMe, where x and y are 2 and 3, 2 and 4, 2 and 6, 3 and 6, and 4 and 6, respectively, have been synthesized [546]. The concept of preparing compounds with a multivalent presentation of Neu5Ac(α 2--6)Gal(β 1-4)GlcNAc(β 1- fragments on a linear or branched (via lysine) peptide backbone has been nicely worked out in ref. [547]. After the organic synthesis of a large series of peptide backbones in which $GlcNAc(\beta 1-N)Asn$ units were incorporated, the oligosaccharide extensions were performed enzymatically by using β -1,4-galactosyltransferase and α -2,6-sialyltransferase.

Neu5,9Ac₂-containing oligosaccharides have been prepared along both organic chemical and enzymatic routes. In a reaction with trimethyl orthoacetate, Neu5Ac(α 2–6)-Gal(β 1–4)Glc could be readily converted into Neu5,9Ac₂(α 2–6)Gal(β 1–4)Glc [380]. Employing CMP-Neu5,9Ac₂ and immobilized porcine liver Gal(β 1–4)GlcNAc α -2,6-sialyltransferase, Neu5,9Ac₂(α 2–6)Gal(β 1–4)GlcNAc has been synthesized [478]. The Neu-5Ac4Me-, Neu5Ac9Me-, and 8epi-Neu5Ac-thioglycoside donors have been used to synthesize the corresponding sialoglycoconjugates [548], whereas also thioglycoside donors of 4d-, 7d-, 8d-, and 9d-Neu5Ac have been prepared [549].

In other enzymatic approaches, use has been made of a β -D-galactoside α -2,3-*trans*sialidase from *Trypanosoma cruzi* as biocatalyst (see also section 9.2.3). This enzyme catalyzes the reversible transfer of Neu5Ac from a donor-substrate of the sequence Neu5Ac(α 2–3)Gal(β 1–O)R₁ to virtually any Gal(β 1–O)R₂ acceptor substrate, affording a product Neu5Ac(α 2–3)Gal(β 1–O)R₂ [550–552] (Fig. 12). A frequently used donor substrate is Neu5Ac(α 2–3)Gal(β 1–4)Glc. In the case of Gal(β 1–3/4)GlcNAc(β 1–O)R sequences, the GlcNAc unit should not be substituted with a Fuc residue, as in Le^x or Le^a determinants [551]. The *trans*-sialidase reaction has been used in the chemo-enzymatic preparation of a water-soluble polyacrylamide, bearing multivalent Neu5Ac(α 2–3)Gal(β 1–4)GlcNAc elements [553]. To solve the problem of the poor enzymatic α -2,3-sialylation of Gal 2-(trimethylsilyl)ethyl β -glycoside using all known α -2,3-sialyltransferases and CMP-Neu5Ac, attention has been paid to the development of a sequence of enzymatic reactions, including cloned α -2,3-sialyltransferase and



Fig. 12. Reactions catalyzed by trans-sialidases.

CMP-Neu5Ac synthase, yielding an alternative active sialyl-donor-substrate *in situ* (e.g. Neu5Ac(α 2–3)lacto-*N*-tetraose), which can be used by *trans*-sialidase [554]. In this way it was possible to convert Gal 2-(trimethylsilyl)ethyl β -glycoside into Neu5Ac(α 2–3)Gal 2-(trimethylsilyl)ethyl β -glycoside, a sialodisaccharide that can readily be transformed into a disaccharide donor, of interest for additional organic syntheses. In another study 4-MU-Neu5Ac was tested as a donor with lactose as acceptor [555]. Starting from periodate treated/reductive aminated 4-MU-Neu5Ac derivatives, interesting possibilities for the inclusion of fluorescent or photolyzable groups were demonstrated.

Bacterial sialidases have also been explored in synthetic approaches. In a reverseenzyme reaction with *A. ureafaciens* sialidase, incubation of a concentrated solution of Neu5Ac and lactose yielded Neu5Ac($\alpha 2$ -6)Gal($\beta 1$ -4)Glc and Gal($\beta 1$ -4)[Neu5Ac($\alpha 2$ -6)]Glc [556]. Similar experiments were carried out with immobilized *V. cholerae* sialidase, using Neu5Ac *p*-nitrophenyl α -glycoside as a donor. In this way Neu5Ac($\alpha 2$ -*x*)Gal and Neu5Ac($\alpha 2$ -*x*)Glc linkages could be produced, in which the α -2,6-linkage dominated over the α -2,3-linkage [557]. Transglycosylation of a Neu5Ac unit using Neu5Ac($\alpha 2$ -8)Neu5Ac as a donor to Gal($\beta 1$ -4)GlcNAc and Gal($\beta 1$ -4)Glc was performed using sialidases of various origin [558]. Although the yields were low, a high regioselectivity was observed. The *C. perfringens, A. ureafaciens* and *V. cholerae* sialidases generated α -2,6-linkages, and the Newcastle disease virus sialidase α -2,3-linkages with the terminal Gal residue.

For detailed information with respect to the synthesis of C-glycosides of sialic acids, see refs. [559–563]. As an example, the synthesis of a multivalent material that consists of the C-glycoside of Neu5Ac, which is resistant to viral sialidase hydrolysis, should be mentioned [562].

7. Conformational aspects of sialic acids

Earlier studies have appeared on the X-ray crystallography of both crystalline β -Neu5Ac• H₂O and β-Neu5Ac1Me•1H₂O [56,564]. In an additional study, the crystal and molecular structure of α -Neu5Ac1,2Me₂ was also analyzed [565]. The C=O bond of the COOH function is approximately coplanar with the ring C–O bond in α -Neu5Ac1,2Me₂, whereas in both β -Neu5Ac and β -Neu5Ac1Me the C=O bond is found to be nearly eclipsed with the anomeric C-O bond. In all three derivatives, the N-acetyl group is essentially planar, adopting the Z-conformation of a peptide bond. For α -Neu5Ac1,2Me₂ and β -Neu5Ac1Me a hydrogen bond between the H-atom of HO7 and the carbonyl O-atom of AcNH5 was observed. The overall conformation of the glycerol side chain is the same for all three derivatives, as far as non-H atoms are concerned. In α -Neu5Ac1,2Me₂ a hydrogen bond between the H-atom of HO8 and the carbonyl O-atom of the COOMe group is detectable. One of the oldest NMR studies on the conformation of Neu5Ac is that focused on the spatial structure of α Neu5Ac2Me in D_2O [566,567]. On the basis of ¹H–¹H coupling constants in combination with ¹³C spin– lattice relaxation times (T_1) , a model could be constructed in which the amide H-atom of AcNH5 is hydrogen-bonded to O7, and the H-atom of HO8 is hydrogen-bonded to the ring-oxygen. A third hydrogen bond between the carbonyl O-atom of AcNH5

and the H-atom of HO4 was suggested on the basis of molecular model building. In this model, apparently, the anomeric center is not involved in any hydrogen bonding, leading to the same conformation for α - and β -anomers. Independent of the models discussed above, the results fit the observation made by ¹H NMR spectroscopy that HO7 and HO8 usually occur in a *trans*-orientation [53]. An NMR (H₂O-suppressed 1D TOCSY, ROESY, NOESY) study, carried out on Neu5Ac(α 2–3)Gal(β 1–4)GlcNAc and Neu5Ac(α 2–6)Gal(β 1–4)GlcNAc in 85% H₂O/15% (CD₃)₂CO, and aimed to detect hydroxyl and amido protons, indicated that in both compounds, thus irrespective of the type of linkage, the H-atom of HO8 of Neu5Ac is involved in a strong intramolecular hydrogen bond [568].

In view of the fact that 7epi-Neu5Ac and 7,8epi₂-Neu5Ac are substrates for CMP-sialic acid synthase (see section 6.1), but not 8epi-Neu5Ac [411], a conformational study on the side chain conformation of these sialic acids and Neu5Ac itself (all <10% α -anomer) was performed [569]. Use was made of ¹H and ¹³C NMR spectroscopy, including NOE measurements, combined with hard-sphere calculations. From a comparison of the four side chain conformations in aqueous solution at neutral pH, it was concluded that the absence of activation of 8epi-Neu5Ac may be due to diminished binding interactions between the hydrophobic and hydrophilic regions of this sialic acid and the activating enzyme, when compared with the other three sialic acids. In additional studies 7d-, 8d-, and 9d-Neu5Ac and 4,7d₂-Neu5Ac were also investigated [570].

Nowadays, much attention is paid to the conformational analysis of glycoprotein glycans, including combinations of NMR spectroscopy and molecular modelling programs, and with respect to sialic-acid-containing fragments the sequences Neu5Ac(α 2-6)Gal(β 1-4)GlcNAc β 1-, Neu5Ac α 2-3Gal β 1-4GlcNAc β 1-, and Neu5Ac α 2-3Gal β 1-4[Fuc(α 1-3)]-GlcNAc β 1- in particular are currently studied in detail by many research groups. As this aspect is outside the scope of this chapter, no details will be presented. In relation to glycoprotein glycans containing polysialyl elements, conformational studies on the polysaccharides of N. meningitidis B, $\{Neu5Ac(\alpha 2-8)\}_n$, and N. meningitidis C, {Neu5Ac($\alpha 2$ -9)}_n, are of interest [571–575]. Conformation and solution dynamics studies of the B- and C-polysaccharides, using NMR spectroscopy, showed an antiperiplanar orientation of H7 and H8 in the C-polysaccharide and a gauche orientation in the B-polysaccharide [571]. In a further NMR study on {Neu5Ac(α 2-8)}_n it was found that both linkages in {Neu5Ac($\alpha 2-8$)}₃ differed in conformation from each other and from the inner linkages of the polymer colominic acid. These conformational differences extend to both terminal disaccharides of oligosaccharides larger than {Neu5Ac($\alpha 2-8$)}₅. It could be demonstrated that at least 5 residues are required before a linkage with a similar orientation to that of the internal $\alpha 2$,8-linkages of colominic acid is generated [572]. In this context, it is interesting to note that $\{Neu5Ac(\alpha 2-8)\}_{10}$ at the minimum is required to bind to B-polysaccharide specific antibodies, which means that the immunologically functional moiety of the decamer is formed by the inner 6 residues. Finally, in more recent studies the helical character of {Neu5Ac($\alpha 2-8$)}_n in solution has been studied, and ordered helical structures [573] as well as an inherently flexible structure, containing a wide range of energetically favorable helices, one of which is an extended helix where $n \approx 9$ [574], has been proposed.

8. Biosynthesis of sialic acids

8.1. General

Cells from higher animals and various microorganisms produce sialic acids in a long pathway starting from glucose [5,8,33]. Only mammalian erythrocytes, which have lost their nucleus during maturation, can no longer synthesize sialic acids, although they are still sialylated on their surface. The presence of sialic acids on the outer cell membrane is considered to be a prerequisite for the viability of mammalian cells, at least in tissues or fluids of organisms. After the loss of cell surface sialic acids by sialidase, these monosaccharides are rapidly restored [576].

Since the last reviews on sialic acid metabolism [5,33], little progress has been made on the enzymatic and regulatory mechanisms as well as on the molecular biology involved in the biosynthesis of N-acetylneuraminic acid (Neu5Ac). However, further insight has been gained into the biosynthesis, degradation and role of modified sialic acids, especially N-acetyl hydroxylation, O-acetylation and O-methylation. An overview on the reactions involved, including catabolic enzymes, is shown in Fig. 13. The reactions, subcellular site, regulation, pathobiochemical role and molecular genetics of the enzymes activating sialic acids with CTP and transferring them onto nascent glycoconjugate molecules have also been intensively studied in animal and bacterial cells and much progress has been made. These new aspects of sialic acid metabolism will be discussed below.

The regulation of sialic acid biosynthesis and consequently the sialylation of glycoproteins and gangliosides of cells seem to be governed in a rather variable manner, resulting in characteristic qualitative and quantitative differences during cell differentiation, growth, functional changes, ageing and malignant transformation of cells. Details will be given in section 10 on the biological role of sialic acids.

In order to influence these processes, inhibitors of sialic acid biosynthesis or breakdown e.g. by sialidases are required. Various metal ions such as Zn^{2+} , Cu^{2+} and selenite were shown to inhibit enzymes of sialic acid biosynthesis in rat liver homogenates [578]. They impair the activity of the key regulatory enzyme UDP-*N*acetylglucosamine-2'-epimerase. In the same tissue, 3-*O*-methyl-*N*-acetyl-D-glucosamine efficiently inhibits *N*-acetylglucosamine and *N*-acetylmannosamine kinases involved in Neu5Ac formation [579]. Consequently, the incorporation of *N*-acetylhexosamines into sialic acids in human hepatoma cells was reduced. After intraperitoneal injection of *N*-propanoyl-D-hexosamines, *N*-propanoylneuraminic acid could be isolated from tissues at different relative amounts, showing that the enzymes for sialic acid biosynthesis and transfer can tolerate changes in the *N*-acetyl moiety of precursor acylhexosamines [580]. Other inhibitors specific for sialic acid enzymes will be mentioned as appropriate in the text.

8.2. Biosynthesis of CMP-sialic acids

The enzyme CMP-Neu5Ac synthase, which is also known as Neu5Ac cytidylyltransferase (EC 2.7.7.43), is ubiquitous in pro- and eukaryotic cells synthesizing sialic acids from glucose [5,33]. In contrast to other nucleotide-sugar-synthesizing enzymes, it is located



Fig. 13. Metabolism scheme of sialic acids. Anabolic (solid arrow) and catabolic (dashed arrow) reactions are indicated. For literature see the text. Enzymes: 1, CMP-sialate synthase (EC 2.7.7.43); 2, sialyltransferases (EC 2.4.99.1...); 3, CMP-Neu5Ac hydroxylase (EC 1.14.99.18); 4, acetyl-CoA:sialate 4-O-acetyltransferase (EC 2.3.1.44); 5, acetyl-CoA:sialate 7(9)-O-acetyltransferase (EC 2.3.1.45); 6, S-adenosyl-L-methionine:sialate 8-O-methyltransferase (proposed EC 2.1.1.78); 7, sialate 4- or 9-O-acetylesterases (EC 3.1.1.53); 8, sialidase (EC 3.2.1.18); 9, sialate-pyruvate lyase (aldolase; EC 4.1.3.3). Both Neu5Ac and Neu5Gc can be O-acetylated by the two O-acetyltransferases. There may also exist a sulfotransferase, since sulfated sialic acids have been found in e.g. echinoderms [13,577]. (*), Sialic-acid-accepting nascent glycoconjugate.

in the nucleus of mammalian cells, although it has not yet unequivocally been shown that CMP-Neu5Ac is really produced in this compartment [581]. It has been isolated from both animal and bacterial sources [5,33,582], and particularly large activities exist in frog [583] and trout liver [582], as well as in *Escherichia coli* [584,585]. The latter enzyme, which is expressed in *E. coli* K1 and K92 only at temperatures higher than 20°C, has been cloned and sequenced [586] and its molecular properties have been described. It exhibits 59% homology with the corresponding, cloned gene from *Neisseria meningitidis* [587, 588].

The availability of this enzyme is increasingly important, since CMP-Neu5Ac is required not only for sialylation of natural substances, but also for the combined chemical and enzymatic synthesis of sialylated complex carbohydrates in research and glycotechnology (see section 6.2 for details and ref. [589]). With the synthases from different sources it is possible to activate not only Neu5Ac, but also Neu5Gc, *O*-acetylated sialic acids and a variety of chemically prepared sialic acid derivatives (section 6.2). The transfer of non-natural sialic acids onto glycoconjugates may be useful for the study of sialic-acid-degrading enzymes and for various aspects of cell biology like cellular interactions and fluorescent labelling of cell surface glycoconjugates by CMP-9-fluoresceinyl-Neu5Ac [390,419]. The enzyme from trout also activates Kdn occurring as oligomers in this animal [107,122].

With regard to the reaction mechanism it has been shown with the *E. coli* synthase and ¹³C-labelled Neu5Ac in NMR studies that CMP- β -Neu5Ac is probably formed by a direct transfer of the anomeric oxygen of β -Neu5Ac to the α -phosphate of CTP [584]. Activation of Neu5Ac is only possible, if the hydroxyl groups at C4 and C8 are in the correct, natural conformation [350]. Correspondingly, 4epi- and 8epi-Neu5Ac cannot be linked to CMP by the rat liver cytidylyltransferase. Furthermore, the enzyme requires the presence of a hydroxyl group at C8, since 8d-Neu5Ac is inactive. This is in contrast to 4d-Neu5Ac and 9d-Neu5Ac. Substituents at C4, such as in Neu5Ac4Me or Neu4,5Ac₂, are tolerated. The strong competitive inhibition effect of β -Neu5Ac methyl glycoside shows that the HO2 group interacts with the enzyme via the lone electron pair of the oxygen atom and not by a hydrogen bridge.

Little information is available on the active site of this enzyme. In the recombinant enzyme from E. coli two cysteine residues were recognized at the positions 129 and 329, one of which seems to be at or close to the active center, as was shown by inhibition with thiol-specific reagents partly in the presence of CTP. However, exchange of these cysteine residues for other amino acids by site-directed mutagenesis did not change the enzyme activity significantly [590].

Several cytidylyltransferases were shown to be inhibited by CMP and CDP [582]. The observation that 2'-CMP, 3'-CMP and 2',3'-cyclic CMP do not inhibit shows that the phosphate residue at the 5' position of ribose is required for inhibition. These studies also demonstrate that the ribose moiety must be intact for inhibition and that chemical modifications of the cytidine residue tested do not lead to inhibitors. An inhibitory effect of the methyl β -glycoside and the 4-oxo-derivative of Neu5Ac on the cytidylyltransferase from rat liver was described [408].
8.3. Transfer of sialic acids

Specific patterns of oligosaccharides as components of glycoproteins and glycolipids cover the surface of individual cell types and may change during variation of the function or differentiation of these cells. This "make up" of cells, which is formed by the expression of a set of glycosyltransferases [335,591–594] gains enormous diversity by the transfer of sialic acids into mostly terminal position of the oligosaccharide chains and in different linkages (see Tables 2–6). This great variety of structures may be even more enlarged by the modification of sialic acids, i.e. *N*-acetyl-hydroxylation, *O*-acetylation, *O*-lactylation, *O*-methylation or *O*-sulfation (see Table 1 for natural sialic acids).

Many studies have been carried out in the past few years on the transfer of sialic acids onto oligosaccharides, polysaccharides, glycoproteins and glycolipids mediated by sialyltransferases of animal cells and bacteria. These have given new insight into the wide occurrence and subcellular distribution of these enzymes comprising a family of over 10 members, their substrate specificities yielding α -2,3-, α -2,6-, α -2,8- and α -2,9-linkages, regulation of gene expression, and especially primary structures. Since these topics have been summarized in refs. [593,595] and are mentioned in several chapters of volume 29a of New Comprehensive Biochemistry series, here only a selection of some aspects of these enzymes mainly from the most recent literature will be reported and a list of cloned sialyltransferases (Table 14) is included. The isolation of CMP-Neu5Ac:Gal(β 1-4)GlcNAc α -2,6- and Gal(β 1-3/4)GlcNAc α -2,3-sialyltransferases from rat liver (ST6Gal I and ST3Gal III, respectively) by affinity chromatography has been described [624].

The sialyltransferase assay is usually carried out with radioactive CMP-Neu5Ac and the sialic acid acceptor, followed by identification of the radioactive product. As this product is often membrane-bound and the radioactive cosubstrate is expensive, a non-radioactive assay was developed [625], in which CMP released from CMP-Neu5Ac by the transfer of Neu5Ac is determined with HPLC (see section 5.3.2). Since CMP-Neu5Ac can be analyzed in the same run, this test is also suited for the activity of the CMP-Neu5Ac synthase described above.

With regard to acceptor specificity, the oligosaccharide sequences accepting sialic acids by the action of sialyltransferases are summarized in Tables 2–6. Although sialyltransferases exhibit strong linkage specificity, in some cases they seem to be involved in the sialylation of not one but several glycoconjugates [595]. In mitochondrial outer membranes, there exist two sialyltransferases with galactoside α -2,3- and α -2,6specificities, which preferably sialylate O- and N-glycans, respectively, of glycoproteins [626]. CMP-Neu5Ac:Gal(β 1–4)GlcNAc-R- α -2,6-sialyltransferase (ST6Gal) was found to sialylate the disaccharide unit GalNAc(β 1–4)GlcNAc-R N-glycosidically linked to a variety of milk glycoproteins [627]. It is concluded from these studies that the same oligosaccharide moiety of a number of human glycoproteins is α -2,6-sialyltransferase activities specific for Gal(β 1–4)GlcNAc-R residues occur in both human liver and placenta [628].

In contrast to the acceptor specificity, the donor specificity of sialyltransferases is less pronounced, although differences exist. They can transfer not only Neu5Ac but also other sialic acids from their CMP-glycosides, e.g. both the natural Neu5Gc and *O*-acetylated species as well as synthetic Neu5Ac derivatives, which have been mentioned in sections 6.2 and 8.2. This is of great importance for cell biological studies.

In order to gain more insight into the structural requirements of the acceptor for rat liver sialyltransferases, the residues in the acceptors $Gal(\beta 1-3)GlcNAc$ and $Gal(\beta 1-3)GlcNAc$ 4)GlcNAc were modified mainly by deoxygenation of the pyranose rings to identify the key polar groups required for transfer [629]. Both the HO6 group of the β -Gal residue and the N-acetyl group of the GlcNAc unit are required for the activity of α -2,6-sialyltransferase (ST6Gal I). The α -2,3-sialyltransferase (ST3Gal III) requires the HO3, HO4 and HO6 groups of terminal β -Gal, and some influence from the subterminal sugars was noticed. Structural variants of the Gal residue can also be recognized by rat liver Gal(β 1–4)GlcNAc α -2,6-sialyltransferase. Besides Gal(β 1–4)GlcNAc(β 1– O)Me and GalNAc(β 1–4)GlcNAc(β 1–O)Me, also Man(β 1–4)GlcNAc(β 1–O)Me, Glc(β 1– 4)GlcNAc(β 1–O)Me and GlcNAc(β 1–4)GlcNAc(β 1–O)Me turned out to be acceptors, as proven by ${}^{1}H$ NMR spectroscopy [630]. In a more detailed systematic study the substrate specificity of α -2,3- and α -2,6-sialyltransferases was explored using a series of synthetic Hex(1-4)GlcNAc(β 1-2)Man(α 1-O)(CH₂)₇CH₃ acceptors with Hex = β -D-Gal, 4-deoxy-β-D-Gal, 4-O-methyl-β-D-Gal, 4-deoxy-4-fluoro-β-D-Gal, β-D-Glc, 3-O-methylβ-D-Gal, 3-deoxy-β-D-Gal, 3-deoxy-3-fluoro-β-D-Gal, 3-amino-3-deoxy-β-D-Gal, β-D-Gul, α -L-Alt and β -L-Gal [630a]. These reactions are relevant to the finding of unusual sialicacid-containing structures, namely Neu5Ac(α 2–6)Man(β 1–4)GlcNAc in the urine of patients with β -mannosidosis [153]. They now enable the synthesis of unusual sialic-acidcontaining carbohydrate sequences.

Sialyltransferases may be glycoproteins and the carbohydrate chain can modify enzyme activity. Thus, deglycosylation of rat liver Gal(β 1–4)GlcNAc α -2,6-sialyltransferase protein markedly decreases enzyme activity, showing that the trimannosyl-*N*,*N'*-diacetyl-chitobiose core with GlcNAc attached is particularly important for the expression of catalytic activity [631]. Glycosylation of α -2,6-sialyltransferase from rat hepatoma cells with *N*-acetyllactosamine- and oligomannose-type sugar chains was also reported [632].

The polypeptide part of an acceptor glycoprotein may also influence the activity of sialyltransferase, as was shown for desialylated human chorionic gonadotropin and bovine colostrum CMP-Neu5Ac:Gal(β 1-4)GlcNAc-R α -2,6-sialyltransferase [633]. When compared with the heterodimeric form of this hormone, the glycans of the α -subunit exhibited a higher kinetic efficiency (V/K_M). In contrast, the branch specificity of this enzyme, i.e. the preference for the Gal(β 1-4)GlcNAc(β 1-2)Man(α 1-3)Man branch for attachment of the first sialic acid molecule into the diantennary glycans of desialylated gonadotropin was not influenced.

Interestingly, evidence was obtained that α -2,3-sialyltransferase from cultured C6 glioma cells [634] must be phosphorylated for activity and it is postulated that enzyme activity is regulated by the action of Ca²⁺/calmodulin and phosphatase, respectively.

We are only beginning to understand the regulation of the expression of sialyltransferase activity. The higher level of sialylation of many tumor cells and increased sialyltransferase activities are well known (see chapter 3 of Vol. 30 and section 10.5 below]. Transformation of FR3T3 cells with the c-Ha-ras oncogene resulted in a marked increase of the expression of β -galactoside α -2,6-sialyltransferase activity and,

Abbreviations	Structures synthesized	Tissue	Ref.	Comments
Mammalian enzy	ymes			·····
ST3Gal I	Sia(α2–3) Gal(β1–3)GalNAc-R	porcine submand. gland	[596]	acts on O-linked chains and glycolipids
		mouse brain	[597]	
		chick embryo	[598]	
ST3Gal II	Sia($\alpha 2$-3) Gal($\beta 1$ -3)GalNAc-R	mouse and rat brain	[599,600]	similar to ST3Gal I, but seems to act better on glycolipids than ST3Gal I; kinetics see ref. [600]
ST3Gal III	Sia(α23) Gal(β13/4)GlcNAc-R	rat liver	[601]	acts preferentially on Gal(\beta1-3)GlcNAc
		human placenta	[602]	
		mouse brain	[603]	
ST3Gal IV	Sia(α2–3)Gal(β1–3)GalNAc-R and Sia(α2–3)Gal(β1–4)GlcNAc-R	human melanoma, placenta, mouse brain	[603–605]	also called STZ or SAT-3, acts preferentially on O-glycans [Gal(β1–3)GalNAc] and on N-glycans [Gal(β1–4)GlcNAc], also on glycolipids
ST6Gal I	Sia(α2-6)Gal(β1-4)GlcNAc-R	rat liver	[606]	acts on N-glycans, possibly also on lacto-neo-glycolipids
		human placenta	[607]	
		mouse liver	[608]	
		chick embryo	[609]	
ST6GalNAc I	Sia(α2–6)GalNAc-R	chick embryo	[610]	
		mouse submand. gland	N. Kurosawa, pers.commun.	
ST6GalNAc II	Sia(α2–6) [Gal(β1–3)]GalNAc-R	chick testis human epithelial cells	[611] [612]	requires Gal(β 1–3)GalNAc as substrate, does not act on GalNAc-R
		mouse	[612a]	
ST8Sia I	Sia(α2-8)Sia-R (glycolipid)	human melanoma	[613-615]	synthesizes $GD3 \gg GT1a$ and $GQ1b$
		mouse brain	[616]	

 Table 14

 Sialyltransferases with known gene structures^a

continued on next page

Abbreviations	Structures synthesized	Tissue	Ref.	Comments	
ST8Sia II	Sia(α2–8) Sia-R	rat brain	[617]	also known as STX, specificity see ref. [618]	
		human fetal brain	[619]	human homologue to STX [617], partial cDNA-clone	
		mouse brain	[618]	murine homologue to STX [617], acts on N-glycans	
ST8Sia III	Sia(α2–8) Sia-R	mouse brain	[616]	acts on N-glycans, synthesis of GD3 and GT3	
ST8Sia IV	Sia(a2-8)Sia-R (NCAM)	Chinese hamster fibroblasts	[620]	synthesis of polysialic acid on NCAM	
		mouse lung	[621]	acts on α -2,3-linked Sia of N-glycans	
ST8Sia V	Sia(α2–8)Sia-R	mouse brain	[621a]	acts on glycoprotein	
Bacterial enzym	es				
PST	Sia(a2-8)Sia-R on Sia-polymers	<i>Neisseria meningitidis</i> group B	[622]		
PST	Sia(a2-8)Sia-R on Sia-polymers	Escherichia coli Kl	[622]		
PST	Sia(α2-8) Sia-R and Sia(α2-9) Sia-R on Sia-polymers	Escherichia coli K92	[623]	synthesis of alternating α -2,8- and α -2,9-linkages	

^a With regard to nomenclature, ST stands for sialyltransferase, 3, 6 and 8 for the linkage formed, Gal or Sia for the monosaccharide carrying the acceptor hydroxyl group, and Roman numbers (I–V) for the chronological order of published primary structures. The species designation should be given as a single letter at the beginning (e.g. pSTGal I for the porcine, mSTGal I for the murine and cSTGal I for the chicken enzyme). This nomenclature follows suggestions by Tsuji et al. [623a]. PST stands for bacterial polysialyltransferases. In the structures synthesized, Sia is used, since the sialyltransferases may transfer not only Neu5Ac, but also other natural as well as synthetic sialic acids (see the text). As an example, the systematic name of ST3Gal I is CMP-Neu5Ac:Gal(β 1–3)GalNAc α -2,3-sialyltransferase (EC 2.4.99.4). However, this name would also be used for ST3Gal II and ST3Gal IV.

Table 14, continued

Inhibitor		Ref(s).			
	ST6GalI ^b	ST3GalI ^c	Lactosyl-ceramide α -2,3-sialyl-transferase ^d		
5'-CMP	0.090	0.064	n.t. ^e	[642]	
5'-CDP ^f	0.050	0.050	n.t.	[642]	
5'-CTP	0.046	0.060	n.t.	[642]	
Cytidine	0.13	22	n.t.	[642]	
2-Thiocytidine	0.15	n.i. ^g	n.t.	[642]	
CMP-dialdehyde ^h	3.3	5.7	0.085	[642,643]	
UDP-dialdehyde	n.t.	n.t.	0.22	[643]	
6'-deoxy-lactosaminide ⁱ	0.76			[644]	
6'-thio-lactosaminide	3.78			[644]	
6'-disulfide-lactosaminide	2.00			[644]	
Protein (20 kDa; calf brain)	80% ^j			[642,645]	
Protein (14.8 and 22.4 kDa; rat brain) 50% ^k				[642,646]	
 ^a For sialyltransferase nomenclature see Table 14. ^b From rat liver. 		^h Obtained by periodate oxidation of the ribose moiety.			
^c From porcine submandibular gland.	Lactosaminide stands for the methyl glycoside of				
^a From embryonic chicken brain.		N-acetyilacto	samine.	- (1 [(40])	
⁶ n.t., not tested. ^f This compound inhibits β galactoride	¹ This inhibition was observed at 20 µg/ml [642]; ovine submandibular gland CMP-Neu5Ac: GalNAc g_protein g_2 6-sialyltransferase was inhibited to a				
α_{2} 6 significant for a s					
K: of 0.025 mM [647].	uun with a	similar extent	[645].	moned to a	
^g n.i., no significant inhibition.		* Inhibitor concentration 20 µg/ml.			

 Table 15

 Natural and synthetic inhibitors of sialyltransferases^a

correspondingly, of the α -2,6-sialylation of cell surface glycoconjugates [635]. Sialylation may also be increased by hormones, e.g. interferon [636], although in this publication the activity of sialyltransferase was not described. However, in the rat intestine hydrocortisone leads to an increase of sialyltransferase activity during tissue maturation [637]. Similar observations were made with rat fibroblast FR3T3 cells [638]. Cytokines were found to induce the expression of β -galactoside α -2,6-sialyltransferase in human endothelial cells, which mediates sialylation of adhesion molecules and CD22 ligands [639]. Based on mRNA measurements using *in situ* hybridization histochemistry, it was found that the expression of β -galactoside α -2,6-sialyltransferase in rat retina increases in the early post-natal days but to different degrees in the different retinal cell types [640]. Five different sialyltransferases were described to be involved in the synthesis of specific terminal sialoside structures of human fetal and adult tissues, and each gene of these enzymes seems to be regulated independently [619]. During the development of trout eggs α -2,6-, α -2,8-sialyl- and α -2,8-polysialyltransferase activities required for the synthesis of polysialoglycoproteins were expressed at various levels [641].

Little is known about inhibitors of sialyltransferases, although such compounds are very interesting for the modulation of sialylation, e.g. to study tumor biology and cancer therapy. Various sialic acids and their derivatives, as well as other synthetic or natural substances were tested as inhibitors (refs. [642,642a], Table 15). The K_i values of periodatetreated UDP and CMP, resulting in two aldehyde groups from the ribose ring each, which inhibit CMP-Neu5Ac:lactosylceramide α -2,3-sialyltransferase in microsomes of embryonic chicken brain in an irreversible, competitive manner [643], are also presented. While the natural 5'-CMP nucleotide is a potent inhibitor of rat liver and porcine submandibular gland sialyltransferases, the 3'-CMP isomer has no inhibitory effect. Strikingly, cytidine itself and its 2-thio-derivative also markedly inhibit the α -2.6- and α -2.3-sialvltransferases tested. Other substituents on cytidine both on the ribose and base moiety abolish this inhibitory effect. Oxidative opening of the ribose group of CMP significantly reduces the inhibitory potency of CMP on the liver and submandibulary gland enzymes, but yields a good inhibitor for the chicken brain sialyltransferase. The interesting compound 5-fluoro-2',3'-isopropylidene-5'-O-(4-N-acetyl-2,4-dideoxy-3,6,7,8-tetra-O-acetyl-1-methoxycarbonyl-D-glycero- α -D-galacto-octapyranosyl)uridine (KI-8110), a CMP-Neu5Ac analogue, has been synthesized and inhibits the sialylation of cells in culture and pulmonary metastasis of mouse colon adenocarcinoma [648,649]. Although this compound does not inhibit the pure sialyltransferases from rat liver and porcine submandibular gland, it seems to inhibit the translocation of CMP-Neu5Ac through the Golgi membrane and thus sialylation of nascent glycoconjugates, as was observed with mouse and trout microsomes [642] as well as with Golgi vesicles from human liver and colorectal cancer cells [650]. As consequence of reduced sialylation, the growth of Sindbis viruses in cell culture was much hindered (M. Odenthal-Schnittler et al., unpublished observations).

While these inhibitors are on the donor site of sialyltransferase substrates, the effect of inhibiting substances on the acceptor site of Neu5Ac has also been reported [644]. For example, the methyl glycosides of 6'-deoxy- and 6'-thio-*N*-acetyllactosamine, which inhibit rat liver α -2,6-sialyltransferase in the millimolar range, are shown in Table 15. In all these cases, the inhibition does not follow pure Michaelis–Menten kinetics, but is of mixed competitive/non-competitive type. Natural sialyltransferase inhibitors have also been found, which are considered to play a role in the regulation of the synthesis of sialoglycoproteins and gangliosides. Whereas heparin and other polyanions are weak inhibitors, proteins from rat and calf brain have been isolated which inhibit α -2,6-sialyltransferases from ovine submandibular gland and rat liver by about 80% at a concentration of 20 µg/mol [641,645]. Two proteins of molecular mass 14.8 and 22.4 kDa isolated from rat brain inhibit both CMP-Neu5Ac:lactosylceramide α -2,6-sialyltransferase [646] and rat liver α -2,6-sialyltransferase [646]. This inhibitor also occurs in chicken and bovine brain as well as in other tissues of these animals and of rat [646].

Many enzymatic and histochemical (see refs. [651,652] and chapter 5 of Vol. 29a, "Glycoproteins") studies have shown that sialylation occurs in the trans-Golgi network [653] as a terminal glycosylation step after the transfer of CMP-sialic acid from the cytosol through the Golgi membrane by a specific translocator (see also Fig. 15 below). This protein was shown to transfer both CMP-Neu5Ac and CMP-Neu5Gc equally well [654]. A detailed study on the glycosylation including sialylation of oligosaccharides in rat liver Golgi preparations was reported in refs. [655,656]. In rat liver hepatoma cells, the glycosylation of α -2,6-sialyltransferase is involved in the maturation of the enzyme to a higher molecular mass form [632]. The enzyme follows the secretory pathway as a membrane protein and is retained at a late Golgi stage. Due to rapid degradation in a post-Golgi compartment, the enzyme, exhibiting a half-life time of 3 h, is secreted from the cell in only small amounts. For the retention of the sialyltransferase in the trans-Golgi network, signals in the enzyme's polypeptide chain were suggested to be responsible for the specific binding to Golgi membranes. It was shown with mutant and chimeric α -2,6-sialyltransferase that the cytoplasmatic tail and signal anchor region alone are not sufficient for Golgi retention [657]. However, when two lysine residues were placed next to the signal anchor on the luminal side of the enzyme, more efficient Golgi retention was observed. This suggests that the signal anchor region is not sufficient for Golgi retention and that it can be replaced by any transmembrane region which allows correct spacing and folding of sequences flanking the membrane.

So far, the acceptor specificity of nearly 20 eu- and prokaryotic sialyltransferases has been elucidated and 15 different cDNA clones of these enzymes have been obtained (Table 14 and references therein). This list demonstrates the rapid progress in this field and also includes the first cloning of mammalian polysialyltransferase (polysialyltransferase-1 from hamster ovary cells) [620]. The same group has recently published the molecular analysis of the biosynthetic pathway of the α -2,8-polysialic acid capsule by *Neisseria meningitidis* serogroup B [658].

The sialyltransferase clones contain a stretch of about 50 amino acids with significant homology and this so-called sialyl motif is possibly involved in substrate binding [601]. Using polymerase chain reactions with degenerate primers deduced from this motif, additional cDNA clones have been obtained [617], for which, however, no sialyltransferase activity can yet be assigned. Sialyltransferases, like other glycosyltransferases, are structurally highly related, consisting of a short NH₂-terminal cytoplasmic domain, a signal-membrane anchor domain, a proteolytically sensitive stem region, and a large COOH-terminal catalytic domain. For example, the deduced amino acid sequence of the mouse brain Gal(β 1–3)GalNAc α -2,3-sialyltransferase shows 80% identity with that of the porcine submandibular gland Gal(β 1–3)GalNAc α -2,3-sialyltransferase [597].

8.4. Enzymatic modification of sialic acid

8.4.1. Biosynthesis and functions of N-glycolylneuraminic acid

Neu5Gc is present in essentially all animal groups of the deuterostomate lineage, from the echinodermata up to the mammals, as component of all types of glycoconjugates [5,8,33, 659,660]. Remarkably, in healthy human tissues it is missing, but it is expressed in small quantities in some tumors [661]. Although Neu5Gc has never been found in bacteria, it is a component of membrane glycoconjugates of *Trypanosoma cruzi* [662].

The wide occurrence of this sialic acid and its tissue-specific and developmentally regulated expression suggest that it has important biological functions. However, very few established functions have yet been allocated to Neu5Gc. Our knowledge of this field is summarized in Table 16. It shows that in addition to structural aspects, the presence of Neu5Gc modulates the general functions of the main, precursor sialic acid, Neu5Ac.

Phenomenon	Reference(s)
Influence on physicochemical properties of glycoconjugates, e.g., increasing hydrophilicity	[5]
Slower hydrolysis rate by sialidases when compared with Neu5Ac, thus possibly hindering spreading and virulence of bacteria	[5,33,245]
Slower degradation rate by sialate-pyruvate lyase	[5,33,245]
Slower rate of demasking of subterminal galactose residues, thus prolonging the biological effect of sialic acids	[663]
Modulation of sialic acid-receptor interactions, e.g., increase of the binding of murine B lymphocyte CD22 adhesin and of <i>Escherichia coli</i> K99 lectin; decrease of the binding of mouse macrophage sialoadhesin and Myelin-Associated Glycoprotein (MAG)	[664–667]
Specific antigenic epitopes, such as Hanganutziu-Deicher antigens	[668–670]
Dog and cat blood group determinants	[671–673]
Tumor-associated antigen in man and chicken	[661,670]
Differentiation marker	[307,674]
The N-glycolyl group can be an acceptor site of O-acetylation and O-methylation, yielding Neu5GcAc and Neu5GcMe (Table 1) in rat thrombocytes and starfish, respectively	[43,44]
The N-glycolyl group can be a site of glycosylation, leading to	
(a) Neu5Gc8Me(α 2–O5)Neu5Gc8Me- chains in the starfish Asterias rubens	[675]
(b) Branching points of glycans in echinoderm glycoconjugates	[5,13,31,676]

Table 16 Biological significance of N-glycolylneuraminic acid

This is best illustrated with sialidases, the action of which is suppressed by the presence of a *N*-glycolyl group in sialic acids. This may afford some protection for host cells from attack by pathogen-derived sialidases, since the removal of sialic acids facilitates the further degradation of cells by other glycosidases and proteases [245]. Neu5Gc may therefore effectively act against the virulence of certain infectious bacteria. It is also conceivable that Neu5Gc may mask subterminal galactose residues of oligosaccharide chains thus preventing their recognition and phagocytosis by macrophages more potently than does Neu5Ac [663]. Other cell recognition phenomena, such as binding of sialic acids to sialoadhesin of mouse macrophages [664] or to the enterotoxic *E. coli* strain K99, which infects young pigs [665], is modulated by Neu5Gc.

Since the Neu5Gc epitope on glycoconjugates is antigenic to man and leads to Hanganutziu–Deicher antibodies [668,677], the clinical application of recombinant glycoproteins, e.g. erythropoietin (EPO), containing small amounts of Neu5Gc due to



Fig. 14. Redox-protein compounds involved in the formation of Neu5Gc by CMP-Neu5Ac hydroxylase [188, 680,681].

their production in cells such as CHO cells, requires special attention [669]. Formation of such antibodies would lead to rapid removal of the drug from the blood stream and the immuno-complex accumulated in renal glomeruli may cause nephritis or other symptoms known as serum sickness. To the authors' knowledge, however, no such problems with recombinant glycoproteins have been reported.

While it was already detected in 1968/69 that Neu5Gc is derived from Neu5Ac by hydroxylation of the N-acetyl moiety [678,679], it was not until 1988 that CMP-Neu5Ac was recognized as the substrate for this hydroxylase [188]. This fact, together with the solubility of the enzyme following tissue homogenization stimulated purification and characterization of the enzyme from porcine submandibular gland [680] and mouse liver [681,682]. The enzyme from both sources is a monomer with a molecular mass of 65 kDa. The CMP-Neu5Ac hydroxylase was also studied in the gonads of the starfish Asterias rubens, which belongs to the phylogenetically oldest animals synthesizing Neu5Gc [683]. It has many characteristics in common with the mammalian enzymes. All hydroxylases studied so far show high affinity for CMP-Neu5Ac [660], i.e. apparent $K_{\rm M}$ values in the low micromolar range (e.g. $1.3 \,\mu M$ for the mouse liver enzyme [684]). They require a number of cofactors for activity including molecular oxygen, reduced pyridine nucleotides, and, as was shown for the mammalian enzymes, cytochrome b_5 , and cytochrome b₅-reductase. These form an electron transport chain resulting in the incorporation of one oxygen into the methyl residue of the N-acetyl group, as shown in Fig. 14. Some experimental evidence was presented [685] that at the beginning of this concerted action the binding of CMP-Neu5Ac to the hydroxylase changes the conformation of the enzyme in a way leading to the recognition and binding of the enzyme by cytochrome b_5 and the formation of a ternary complex. After the transfer of electrons from NAD(P)H to the enzyme through cytochrome b_5 , CMP-Neu5Ac is converted to CMP-Neu5Gc which was shown not to bind to the enzyme and is therefore released with concomitant dissociation of the ternary complex. The observation that the addition of iron salts can stimulate enzyme activity and iron-binding substances are potent inhibitors, together with electron-spin-resonance studies of the purified hydroxylase, strongly point to the presence of an iron-sulfur center in the active site, which is possibly of the Rieske type [686]. The CMP-Neu5Ac hydroxylases from mouse liver [687] and pig submandibular gland [688] have both been cloned. The gene structures show high homology.

In addition to the CMP-Neu5Ac-specific hydroxylase, pig submandibular glands were reported to contain a second hydroxylase, which is specific for free Neu5Ac and also



Fig. 15. Biosynthesis of CMP-*N*-glycolylneuraminic acid with the aid of CMP-*N*-acetylneuraminic acid monooxygenase in the cytosol, translocation of CMP-Neu5Gc into the Golgi vesicle and transfer of the Neu5Gc moiety onto nascent glycoconjugates [188,654,684]. From ref. [690] by permission of Oxford University Press, Oxford.

differs from the former hydroxylase with respect to its sensitivity to certain inhibitors [689].

The activity of CMP-Neu5Ac hydroxylase and accordingly the concentration of CMP-Neu5Gc in the cytosol probably play the most important role in regulating the level of sialylation with Neu5Gc. This is assumed because neither the Golgi CMP-sialic acid antiporter nor the sialyltransferases exhibit a pronounced preference for CMP-Neu5Ac or CMP-Neu5Gc (Fig. 15). The activity of the hydroxylase may thus be tuned so that the ratio of Neu5Gc and Neu5Ac required in the resulting glycoconjugates is generated in the form of CMP-glycosides in the cytosol. The multiplicity of functions and constitutive expression of the cytochrome b_5 system, together with the lack of effect of any metabolites

on the hydroxylase activity [681], suggest that the rate of production of CMP-Neu5Ac is regulated at the level of the expression of the monooxygenase. The tissue-specific and developmental factors affecting the production of this protein on the gene level as well as a possible influence of the oxygen pressure still remain to be elucidated. No experimental explanation has so far been obtained for the formation of Neu5Gc in human tumors, which may be caused by the anomalous expression of a CMP-Neu5Ac hydroxylase gene, which seems to be dormant or repressed in normal tissues. Significantly, in T-cell lymphomas of baboons the level of Neu5Gc, as component of gangliosides, was much higher than in those from the corresponding normal tissues [691]. The fusion of human B lymphocytes with mouse myeloma cells leads to high expression of Neu5Gc, showing activation of the enzymes involved in Neu5Gc biosynthesis due to the somatic cell fusion process [692]. It was furthermore reported that the insertion of retroviruses into Chinese hamster ovary cells almost completely replaces Neu5Ac by Neu5Gc [693].

8.4.2. Biosynthesis and functions of O-acetylated sialic acids

O-Acetylated sialic acids are found in all types of glycoconjugates and in oligo- and polysaccharides of many animal species from the echinoderms onwards and also in some bacterial species (refs. [5,7,8,11,13,27,690,694]; see also sections 2 and 3, and Table 1). This modification is either at position 4 of the pyranose ring of both Neu5Ac and Neu5Gc, as was found in horse, donkey, guinea pig and echidna (Tachyglossus aculeatus), or in the glycerol side chain of sialic acids. In the latter case, the O-acetyl group is most frequently located at C9, but it may be accompanied by acetylation at C8 and/or C7, leading to di- and tri-O-acetylated species. This side-chain O-acetylation is very frequent in the animal kingdom and is also found in tissues and fluids of man. However, in the expression of O-acetylated sialic acids ethnic differences do exist in colonic mucin of Sino-Japanese and non-Sino-Japanese races. O-Acetylation was more frequent in the latter ethnic groups studied (British, Icelanders, South African blacks and Bahrainis) than in Chinese and Japanese. It is speculated that loss of O-acetylation may be due to a single mutation of the gene regulating O-acetylation and that this may be related to selection pressure resulting from differential enteric colonization by bacterial flora and from resulting infectious diseases [695].

The occurrence of differently *O*-acetylated sialic acids suggests an important biological role of these modifications. We are only just beginning to understand these effects, which influence many biological and pathobiochemical systems, as listed in Table 17. As with Neu5Gc, *O*-acetylation may influence the physicochemical properties of sialylated glycoconjugates, which is especially important on cell surfaces. Correspondingly, an increased protective effect of the mucin containing a large quantity of *O*-acetylated sialic acids and lining the endothelia of human colon is discussed [245]. This protection against an aqueous environment containing a large quantity of different bacteria may be due firstly to the greater hydrophobicity imparted by the *O*-acetyl groups and secondly to a greater resistance of *O*-acetylated sialic acids to microbial sialidases when compared with Neu5Ac.

Sialic acid O-acetylation not only prevents (in the case of 4-O-acetylation) or much reduces (in the case of side-chain O-acetylation) the activity of viral, bacterial and animal (including trypanosomal) sialidases, but also impairs the activity of acylneuraminate-

Phenomenon	Reference(s)
Influence on physicochemical properties of glycoconjugates, e.g. increasing hydrophobicity	[5]
Hindering biologically active sialoglycoconjugates from degradation, by inhibition of sialidase, <i>trans</i> -sialidase, endoglycosidase and lyase action	[5,33,660,696]
Hindering cell degradation, e.g., of erythrocytes	[5,690,697]
Hindering binding of influenza A and B viruses, reovirus, <i>Plasmodium falciparum</i> , mouse macrophage sialoadhesin, B cell adhesion molecule CD22	[664,698–700]
Hindering activation of the alternate complement pathway, probably by preventing sialic acid binding to factor H	[701,702]
Hindering infection of cattle by Trypanosoma brucei	[703]
Decreasing antigenicity	[5,668,704]
Providing epitopes for the binding of influenza C, Corona and encephalomyelitis viruses	[705–707]
Increase of bacterial virulence	[708]
Providing epitopes for recognition by antibodies	[5,709]
Providing epitopes for recognition by lectins	[709–711]
Representing differentiation antigens and influencing morphogenesis	[5,668,712–716]
Representing tumor-associated antigens	[5,668,717-723]
Regulating environmental adaptation	[724]

Table 17						
Effects	of	O-acetyl	groups	in	sialic	acids

pyruvate lyase, the next enzyme involved in the catabolism of sialic acids [5,33] (see Fig. 13). This effect may facilitate recycling of sialic acids released from glycoconjugates. *O*-Acetyl groups have a further hindering effect on the binding by several proteins, as was observed with the hemagglutinins of influenza A and B viruses, which only bind to non-*O*-acetylated Neu5Ac and Neu5Gc [699], with the malaria parasite *Plasmodium falciparum*, where the attachment of *O*-acetylated mouse erythrocytes was weaker than to non-*O*-acetylated cells [700], and with sialoadhesin from mouse macrophages [664]. In contrast to this masking of recognition sites, 9-*O*-acetylated sialic acids are specifically recognized by influenza C virus [705,725], as well as encephalomyelitis and corona viruses [706], leading to binding of these viruses to cells with concomitant infection. In the case of influenza C virus, this may occur in human nasal epithelia, which contain *O*-acetylated gangliosides and glycoproteins [J. Tack et al., unpublished]. In human nasal mucin 5–10% of the Neu5Ac residues were found to be *O*-acetylated [726]. A polyacrylamide glycopolymer containing Neu5,9Ac₂ was synthesized (see section 6.2) as inhibitor of influenza C virus hemagglutination, which is useful as a ligand for binding studies and as a potential anti-influenza C virus drug [489].

Influenza C viruses can be used for the detection of glycoproteins and gangliosides with 9-O-acetylated sialic acids on thin-layer chromatograms, and blots of SDS-polyacrylamide gels as well as in histochemical sections (see ref. [727] and unpublished data). This staining procedure, described in section 4, is possible with the aid of the hemagglutinin and the sialate-9-O-acetylesterase activities (section 9.1) of influenza C virus, which is part of the viral spike hemagglutinin-esterase-fusion (HEF) glycoprotein [234,728,729]. In Fig. 16 an example for the localization of Neu5,9Ac₂ in the mucin-producing Goblet cells of the mucosa of human colon is shown. With these techniques the frequent occurrence of Neu5,9Ac₂-containing glycoconjugates especially in epithelia and in endothelia of blood vessels of rat, cow and man was demonstrated [234,235,727,730,730a]. Sialic acids O-acetylated at C4 cannot be stained by influenza C virus.

In addition to these functions, O-acetylated sialic acids are involved in differentiation and tumorigenesis as evidenced by their differential expression during these processes (for some examples see Table 17). In the adenoma-carcinoma sequence in colorectum, analysis of sialic acid O-acetylation may assist in diagnosis of tumor disease [723]. 9-O-Acetyl-GD3 was found to be a surface marker for basal cell carcinomas, which may lead to novel, immunological therapeutic interventions [718], as is also expected for Neu5,9Ac₂-containing glycoconjugates of melanoma cells. The great importance of O-acetylation during ontogenesis was demonstrated by transfection of sialate-9-O-acetylesterase into mouse embryos, which led to developmental abnormalities in these transgenic mice [714]. There is also evidence that the expression of O-acetyltransferase activity in human T lymphocytes is regulated by maturation events taking place in the periphery [716].

Although both the sialate 4- and 9(7)-O-acetyltransferases (EC 2.3.1.44 and 45) were discovered 27 years ago in bovine and equine submandibular glands, respectively [731,732], the enzymes have not yet been purified and characterized in detail. Furthermore, nothing is known about the regulation of their expression at the gene level. In contrast to the CMP-Neu5Ac hydroxylase, the O-acetyltransferases are bound to subcellular membranes, as was studied with bovine and equine submandibular glands [733] and with rat liver [284,734]. Using detergents, the O-acetyltransferase from bovine submandibular glands was partially solubilized and enriched about 600-fold [7]. This labile enzyme exhibits a pH optimum of 7.2 and transfers acetyl groups from acetyl-CoA only onto α -glycosidically bound sialic acids of glycoconjugates. This corresponds with the finding that O-acetylation of the side chain at C7 and C9 of Golgibound sialic acids occurs after the translocation of acetyl-CoA into this compartment [8].

The complete pattern of sialic acid side-chain O-acetylation may require several sialate O-acetyltransferases each with a specificity for only one position. On the other hand, the primary insertion place for the O-acetyl function may be the HO7 group alone from where the ester group migrates even under physiological conditions to the 9 position, presumably via C8 [7,23], leaving HO7 ready for a new transfer. Thus, for complete side-



Fig. 16. Staining with influenza C virus of Neu5,9Ac₂-containing mucin in goblet cells (arrow) of the mucosa of human colon tissue. Thin-sections were overlayed with virus suspension and virus bound to O-acetylated sialic acids visualized using FITC-conjugated anti-influenza antibodies [730]. (top) Goblet cells containing native mucin are stained yellow-green. (bottom) After removal of Neu5,9Ac₂ by sialidase, no staining occurs and the cells appear dark. This effect can also be obtained by saponification of the ester groups. Magnification about 400-fold.

It is also possible that several sialic acid O-acetyltransferases preferring glycoproteins or gangliosides or at least specific oligosaccharide chains exist. Enzymes were reported to be specific for Neu5Ac in N-glycans of rat liver [735] and for gangliosides in human melanoma cells [694,736], respectively. An O-acetyltransferase incorporating acetyl residues into C7 and C9 of polysialic acid was demonstrated to occur in K1-positive *Escherichia coli* [159]. From the results reported above, an O-acetyltransferase specific for the O-glycan of mucin in bovine submandibular glands could also be postulated.

8.4.3. Biosynthesis of 9-O-lactylated sialic acid

5-*N*-Acetyl-9-*O*-lactylneuraminic acid (Neu5Ac9Lt) has been found in man, cow and trout (see refs. [5,8,13,582,737], see also Table 1). Not much is known with regard to the origin of this compound. However, investigation with a particulate fraction from horse liver suggested that this modification occurs enzymatically, although the exact mechanism and type of lactyl donor is not known [29].

8.4.4. Biosynthesis of 8-O-methylated sialic acids

Sialic acid *O*-methylation is a modification that seems to be restricted to the echinodermata, especially in certain starfish [5,8,13]. *O*-Methyl groups have only been found in the 8 position of Neu5Gc and in much lower quantities on Neu5Ac [675]. 9-*O*-Acetylated derivatives of 8-*O*-methylated sialic acids were also detected in small amounts (Table 1). The existence of a sialate 8-*O*-methyltransferase required for the biosynthesis of this sialic acid was first demonstrated to occur in the starfish *Asterias rubens* [675,738]. The enzyme has now been further characterized after solubilization by detergents from a crude preparation of starfish gonads as a membrane-associated protein that transfers methyl groups from *S*-adenosyl-L-methionine preferably onto glycosidically linked Neu5Gc residues. Neither free Neu5Gc nor CMP-Neu5Gc are substrates for this enzyme. Horse erythrocytes or glycopeptides from pig submandibular gland mucin immobilized on Sepharose, containing almost exclusively Neu5Gc residues, are excellent substrates for the sialate *O*-methyltransferase [739]. The systematic name *S*-adenosyl-L-methionine:sialate-8-*O*-methyltransferase (EC 2.1.1.78) has been proposed.

8.4.5. Biosynthesis of 5-N-acetyl-2-deoxy-2,3-didehydro-neuraminic acid

This Neu5Ac derivative lacking a glycosidic hydroxyl group at C2 has been found as free sialic acid in small quantities in blood serum, saliva and urine of man [5,8,13] and in tissue extracts of starfish [36] (Table 1). It is unknown, whether this compound, which is a potent competitive inhibitor of sialidases (see below), has a biological function. Its formation under physiological conditions has been described in section 2.

8.4.6. Biosynthesis of 5-N-acetyl-2,7-anhydro-neuraminic acid

The occurrence of this neuraminic acid derivative (Table 1) has first been described in cerumen of men [37], although its origin in this material remained unknown. Some hints to the possible enzymatic formation (by microorganisms?) come from the discovery that Neu2,7an5Ac is a reaction product of a leech (*Macrobdella decora*) sialidase ("sialidase L") with sialoglycoconjugates [61] (see also section 2).

8.4.7. Occurrence of 2-keto-3-deoxynononic acid (Kdn)

This substance (for the chemistry see section 3, Tables 1 and 3) was found in the eggs of rainbow trout [740], and the egg jelly coat from the amphibians Pleurodeles waltlii [48], Axolotl mexicanum [105], and Xenopus laevis [741]. In trout, it is part of gangliosides [122] and glycoproteins, where it can terminate long sialic acid chains or form poly-Kdn chains [267]. To our knowledge, nothing is known about the biosynthesis of this sugar, including the possibility of deamination of neuraminic acid at C5. However, the linkage of Kdn to CMP by the cytidylyltransferase from the testis of rainbow trout has been reported [107,122]. This enzyme can also activate Neu5Ac and Neu5Gc. A Golgi-membrane-bound Kdn-transferase from the ovaries of the rainbow trout Oncorhynchus mykiss has been characterized [545]. This enzyme terminates the growth of α -2,8-polysially chains in a series of actions of four glycosyltransferases involved in the glycosylation of ovary glycoproteins. Three of these transferases are sialyltransferases, the first transferring Neu5Gc to C6 of serine- or threonine-bound GalNAc, the second linking another Neu5Gc residue to C8 of this first Neu5Gc and the third enzyme being an α -2,8-polysialyltransferase for chain elongation, before termination of its growth by the 2,8-Kdn-transferase [545]. Remarkably, a developmentally regulated α -2,8-poly Kdn was also discovered in various tissues of mammals [J. Roth, personal communication].

9. Catabolism of sialic acids

A well-balanced system of enzymes involved in the degradation of sialoglycoconjugates is required for the normal functions of cells and tissues. A lack of catabolic enzymes, often due to genetic errors, may lead to severe diseases, as may a surplus of such enzymes, e.g. due to infections by microorganisms. These problems will be addressed in section 10.5.

Three enzyme systems are required for the catabolism of sialic acids: *O*-acetylesterases, sialidases and lyases. Furthermore, a sialic acid transporter has been described (section 10.5), which carries liberated sialic acids from lysosomes into the cytosol, where they are either degraded by the lyase or recycled after activation with CTP (section 8.3). Sialic acid permeases provide bacteria with sialic acids for nutritional purposes (section 9.4).

9.1. Sialate-O-acetylesterases

One of the main functions of sialic acid *O*-acetyl groups is their inhibitory effect on the action of both sialidases and sialic acid-specific lyases [5,33]. While a 4-*O*-acetyl group completely hinders the action of these enzymes (with the exception of a slow release by viral sialidases [252]), such ester groups at the sialic acid side chain appreciably hinder hydrolysis of the glycosidic bond of these sugars and their further breakdown by lyases. The existence of esterases acting on *O*-acetylated sialic acids prior to sialidase is therefore a prerequisite or at least supports the rapid turnover of *O*-acetylated sialoglycoconjugates. The possibility of the existence of such enzymes was raised by the observation that

horse glycoproteins, which are heavily 4-O-acetylated, seem to have a normal turnover, and sialidase studied in horse liver cannot act on 4-O-acetylated sialic acids [742]. Consequently, the existence of such esterases in horse liver was assumed and two enzymes were found, one hydrolyzing only 9-O-acetyl groups and the other mainly 4-O-acetyl residues from sialic acids (ref. [266], and unpublished work). Furthermore, sialate-9-O-acetylesterases (EC 3.1.1.53) have been isolated and characterized from influenza C virus [705,743], bovine brain [744] and rat liver [745,746]. They also occur in human erythrocytes [747] as well as in enteric bacteria [748]. These enzymes specifically release 9-O-acetyl groups from free and glycosidically bound sialic acids. 9-O-Lactyl groups, O-acetyl groups of Neu5,7Ac₂ and methyl-esterified carboxyl groups of sialic acid are not hydrolyzed. Although they also attack O-acetyl groups of non-physiological substances, e.g. naphthyl acetate [266,749], they belong to the few esterases which have been recognized to have a physiological function [266]. Amide groups including the synthetically added 9-N-acetyl moiety of 9NAc-Neu5Ac (Table 13) are resistant to the action of these esterases. However, the latter Neu5Ac derivative binds to influenza C virus hemagglutinin and it does this in a stable manner, as it cannot be removed by the viral esterase [750]. The staining of glycoconjugates containing O-acetylated sialic acids with influenza C virus and the problems involved due to esterase activity were discussed in sections 4 and 8.4.2.

The mammalian sialate esterases are considered to be involved in the turnover processes of glycoconjugates in lysosomes, thus facilitating the action of sialidases as was discussed for the rat liver enzyme [735,745]. Tissue esterases may also participate in the recycling of sialoglycoconjugates, as the O-acetyl groups of the ganglioside GD3 in human melanoma cells turn over faster than the underlying sialic acids of this molecule [751]. It should be noted that the de-N-acetylated sialic acid (Table 1), as a component of GD3 ganglioside, can also occur in the same malignant cells [14], as was shown by immunological means. Thus, the unstable neuraminic acid molecule can exist in nature, however, only in glycosidic linkage (section 1). The viral enzyme may be of pathophysiological significance and be involved in the binding to mucins and to endothelial cell surfaces prior to endocytosis of the virus particles [266,728]. With regard to the epidemiology of viral infections, the viral esterase might contribute to the unmasking of new receptor sites to facilitate superinfection by type A and B influenza viruses [752]. Sialate esterase from human enteric bacteria acts on O-acetylated sialic acids of colon mucin and thus facilitates the further degradation of the mucin barrier by bacterial sialidase and other glycosidases [748].

The activity of sialate-O-acetylesterase can easily be followed using free Neu5,9Ac₂ either isolated from bovine submandibular gland mucin [5,13] or O-acetylated chemically (see section 6) and HPLC determination (see section 5.3.2). Alternatively, 5-N-acetyl-9-O-acetyl-2-[4-(dansylamino)phenylthio]- α -neuraminic acid [285] acts as a specific and highly sensitive, fluorescent esterase substrate (see sections 5.3.2 and 6.2). With the corresponding 7,8,9-tri-O-acetylated derivative it was shown that influenza C virus esterase can remove all three O-acetyl groups [469]. Since this esterase acts only on the O-acetyl residue at C9, de-O-acetylation of the whole sialic acid side chain can occur only after migration of the other O-acetyl groups to the primary alcohol group at C9. These experiments provide further evidence for the existence of the cascade of

alternative enzymatic de-O-acetylation and non-enzymatic O-acetyl migration proposed earlier (refs. [5,7,33], and sections 2 and 5.3.5).

9.2. Sialidases

These enzymes (EC 3.2.1.18) are essential tools in sialic acid metabolism, mainly in catabolic reactions, usually hydrolyzing the O-glycosidic linkages between the terminal sialic acids and the subterminal monosaccharides of free and glycoconjugatebound oligosaccharides. Sialidase, as well as its substrate, is common in metazoan animals of the deuterostomate lineage from echinoderms to mammals [33,244,246,753]. Diverse viruses and microorganisms, such as fungi, protozoa and bacteria also produce sialidases [33,244,246,753–756], although they mostly lack sialic acids. The most recent review on the biochemistry and function of the large sialidase family discusses great variety in cellular location, molecular mass and substrate specificity [246]. Therefore, only a few, more general aspects of sialidases, the molecular biology, the pathophysiological significance and inhibitors.

9.2.1. Types of sialidases

Sialidases are exo- or endo-sialidases, hydrolyzing either terminal sialic acid residues of complex carbohydrates or internal sialic acid glycosidic bonds of oligo- or polysialyl chains. Most microbial and animal sialidases belong to the first group [33,244,246,753–755], in the following simply referred to as "sialidases", while an endo-sialidase has been found in the coliphage E hydrolyzing the α -2,8-linkages of colominic acid [757].

Based on the nature of the chemical reaction, sialidases can tentatively be grouped into eight types, irrespective of the nature of the complex carbohydrate attacked and the subcellular location of the enzyme.

(1) Most microbial and animal sialidases readily hydrolyze terminal α -2,3-, α -2,6and α -2,8-glycosidic linkages, although at different rates. In the majority of cases, the α -2,3-linkage is cleaved most rapidly [244,246,753–755]. Due to steric hindrance the side-positioned (branched) terminal Neu5Ac as in GM1, however, is resistant to the action of most sialidases with the exception of the *Arthrobacter ureafaciens* sialidase. The sialidases from *C. perfringens* and *V. cholerae* cleave the glycosidic bond of their substrates between C2 of Neu5Ac and the glycosidic oxygen, as demonstrated by release experiments carried out in H₂¹⁸O] buffer solutions and monitoring by GLC [758]. Sialic acids are released in the α -anomeric form after which they slowly mutarotate to the more stable β -form (see section 5.3.5). Although Neu2en5Ac is a competitive inhibitor for most members of this sialidase group, it has not been detected to be an intermediate of enzyme reaction, as was the case for some viral sialidases (see below).

(2) With regard to the primary reaction product, *Salmonella typhimurium* sialidase behaves quite differently, since it was reported not to retain the configuration of sialic acid during the hydrolysis of NeuSAc *p*-nitrophenyl α -glycoside. As primary reaction product the β -isomer was detected by optical rotation measurements [759,760].

(3) Close investigation of the reaction mechanism of influenza virus sialidases [60,320] has shown that Neu2en5Ac appears as a transition intermediate, resulting in the

appearance of small amounts of this substance in free form during prolonged incubation of sialic acids with influenza B virus sialidase and sialyllactose, as was identified by GLC/MS analysis [60]. Thus, this sialidase can "synthesize its own inhibitor". During the reaction, the chair conformation of Neu5Ac is distorted to the boat conformation by the concerted action of mainly aspartate, arginine, glutamate and tyrosine residues of the active center. The aspartate serves as proton donor for the release of the glycosidic partner of sialic acid. After the formation of the oxocarbonium intermediate, the addition of a hydroxyl group leads to the formation of α -Neu5Ac, or, in a side reaction by deprotonation, to Neu2en5Ac. The site of enzyme catalysis was shown by X-ray crystallographic studies of the inhibitor–influenza A N9-sialidase complex [761]. Here, twelve amino acid residues directly interact with Neu2en5Ac. Additionally, a further six conserved amino acids exist lining the active site pocket. On the basis of these studies, a mechanism of influenza virus sialidase action was suggested [762].

(4) As mentioned in sections 2 and 8.4.5, a sialidase from the leech *Macrobdella decora* yields Neu2,7an5Ac as reaction product [61]. The purified enzyme interacts only with the α -2,3-glycosidically bound sialic acid of sialyllactose and various sialoglycoconjugates. This marked linkage specificity is unique among the sialidases. The origin of Neu2,7an5Ac in cerumen [37] is unknown, but it is hypothesized that it results from the action of a similar, probably microbial sialidase. The small amounts of this sialic acid found in human urine [763] may have the same origin.

(5) The bacteriophage E which infects strains of *E. coli* displaying the α -2,8-linked polysialic acid colominic acid possesses an endo-sialidase, which was first described in ref. [764]. The reports on such endo-sialidases associated with K1 coli phages have been summarized [765]. These enzymes recognizing α -2,8-linkages of both Neu5Ac and Neu5Gc are believed to be involved in the attachment of the phages to the bacteria.

(6) Shortly after the discovery of Kdn, now also included in the sialic acid family (sections 1–3, Tables 1–3), an enzyme activity was discovered in the loach *Misgurnus fossilis* hydrolyzing the terminal α -glycosidic linkage of this monosaccharide [39,766]. Although conventional sialidases cannot hydrolyze Kdn bonds, the *Misgurnus* Kdn-ase can also act on Neu5Ac bound to 4-methylumbelliferone (MU) or GM3 ganglioside. Kdn-ase found in several tissues, particularly in the ovary of rainbow trout, also has sialidase activity [254]. In contrast, a Kdn-ase from *Sphingobacterium multivorum* only hydrolyzes α -2,3-, α -2,6- and α -2,8-bonds of Kdn glycosides, but does not cleave Neu5Ac or Neu5Gc from a variety of sialoglycoconjugates tested [255]. The use of these hydrolases for analytical purposes is mentioned in section 5.1.

(7) All these rather different sialidase types specifically recognize α -glycosidic bonds. However, a hydrolase acting on the β -glycosidic bond of CMP-Neu5Ac is CMP-sialate hydrolase (EC 3.1.4.40) which occurs in plasma membranes [767]. It is assumed to be involved in the regulation of the cellular level of CMP-sialic acids. Evidence for the existence of a Golgi-associated CMP-sialate hydrolase was also obtained in rat and mouse liver [654].

(8) While all these enzymes hydrolytically cleave α -glycosidic bonds of sialic acids, another type of sialidase has been discovered, which can also hydrolyze these linkages but preferably forms sialic acid linkages under physiological conditions in the presence of suitable acceptors. These are the *trans*-sialidases, occurring in parasites, and a

special section (9.2.3) is devoted to these enzymes. Since in principle all enzymatic reactions are reversible, transglycosidation of a Neu5Ac unit is also possible by the use of conventional, bacterial sialidases under appropriate conditions. For examples of corresponding glycoside syntheses see section 6.3.

9.2.2. Primary structures of sialidases

Comparison of the enzymatic properties of sialidases from a variety of microbial and animal sources revealed that these enzymes are highly diverse with respect to subcellular location, molecular mass, number of subunits, isoelectric point, temperature optimum, influence of Ca^{2+} on activity, substrate specificity and specific activity [244,246,753,754]. The only common properties are the α -anomeric configuration of the sialic acid glycosidic linkage and the acidic pH optimum (pH 5.0–6.1), though the trypanosomal *trans*-sialidases exhibit maximum activity at neutral pH [755]. As these properties apparently do not reveal much relationship between the different microbial sialidases, the primary structures of the enzymes were further investigated at the DNA level, from which it became clear that, although their enzymatic properties show many differences, the pro-and eukaryotic sialidases so far investigated belong to one superfamily [660,768].

In the first primary structures of microbial sialidases, obtained by cloning and sequencing of the respective genes from *Clostridium perfringens* [769], *Vibrio cholerae* [770], *Clostridium sordellii* [771] and *Salmonella typhimurium* [772], an amino acid sequence motif was detected, which is repeated four-fold in each protein: S-X-D-X-G-X-T-W [773]. This motif, named the Asp-box, was found in all 16 sialidases of animals, trypanosomes, and bacteria, which have so far been sequenced (see refs. [660,768] and Table 18). In viral sialidases, however, the motif was rarely detectable (e.g. only the sialidase from N9 influenza A virus strain exhibits the complete motif [786] and has probably undergone mutational alterations).

Further identical motifs, or single amino acids at certain positions, became evident by an alignment of the pro- and eukaryotic sialidase sequences known [660]. Gaps had to be introduced as a consequence of the differences in protein size. The central regions of these proteins are especially homologous and exhibit most of the conserved amino acids. A further motif, the F-R-I-P region, which is located N-terminally from the first Asp-box, is highly conserved in clostridial sialidases, but was found to be degenerated to X-R-X-P, when further bacterial and animal sialidase sequences were included in the alignment. Altogether, sixteen amino acids are found to be conserved.

The function of the conserved and repeated Asp-box is not yet fully understood. In the N9 influenza A virus sialidase [786], the single corresponding motif is located as part of a β -pleated sheet polypeptide at the connections between the four protein subunits. Immunological studies revealed that the Asp-box I of *Trypanosoma cruzi trans*-sialidase (see below) is inaccessible to antibodies, which has been taken to indicate that it is not part of the external catalytic domain [788]. Site-directed mutagenesis experiments of the "small"(43 kDa) sialidase isoenzyme of *C. perfringens* resulted in only small alterations of enzyme activity by changing some of the Asp-box amino acids [789], while the exchange of other highly conserved amino acids drastically reduced enzyme activity and increased the K_M value, e.g. by replacement of the N-terminally located, conserved arginine-37, probably belonging to the active center, with lysine.

Organism	Reference(s)
Rattus rattus (rat)	[774]
Cricetulus griseus (hamster)	[775]
Macrobdella decora (leech) ^b	[61]
Trypanosoma cruzi	[776]
Trypanosoma rangeli ^c	[777]
Actinomyces viscosus	[778,779]
Bacteroides fragilis	[780]
Clostridium perfringens ("small" isoenzyme) ^d	[769]
Clostridium perfringens ("large" isoenzyme)	[782]
Clostridium septicum	[781]
Clostridium sordellii	[771]
Micromonospora viridifaciens	[783]
Salmonella typhimurium	[772]
Streptococcus pneumoniae	[784]
Vibrio cholerae	[770,785]
Influenza virus (A/whale/Maine/1/84; H13n9)°	[786]
Bacteriophage E ^f	[787]

Table 18 Cloned sialidases and *trans*-sialidases with Asp-boxes^a

^a The gene structures can be found in the EMBL Data Library.

^b The reaction product is Neu2,7an5Ac.

^c Sialidase-like protein.

^d This enzyme has also been cloned [769a].

^e No Asp-boxes were found so far in the other viral sialidase sequences of the EMBL Data Library.

^f Endo-sialidase.

More insight into the possible role and spatial location within the protein molecule of the Asp-boxes and the other conserved amino acids came from the crystal structure of the sialidase from *S. typhimurium* [790]. The enzyme has the shape of a six-bladed propeller, each blade consisting of four-stranded antiparallel β -sheets (Fig. 17). The axis of this propeller passes through the active site. The Asp-boxes are located at topologically equivalent positions on the outside of the blades which is not in favor of a role of these sequence motifs in enzyme catalysis. Cocrystallization of the enzyme with the inhibitor Neu2en5Ac enabled the identification of the amino acids involved in sialic acid binding and catalysis. Accordingly, the active site consists of an arginine triad, a hydrophobic pocket and key tyrosine and glutamic acid residues. The interaction of these residues with different parts of the Neu5Ac molecule is shown in Fig. 18 for three bacterial sialidases. Identical amino acids are assumed to be involved in catalysis by the low-molecular-mass sialidase from *C. perfringens*, as was shown by the finding of conserved amino acids at positions identical or similar to the *S. typhimurium* enzyme, site-directed mutagenesis



Fig. 17. Three-dimensional structure of Salmonella typhimurium LT2 sialidase with bound inhibitor Neu2en5Ac obtained from X-ray crystallography. From ref. [790] by permission of National Academy of Sciences, New York.

and hydropathy studies of the amino acid sequence [Kleineidam et al., unpublished]. The same is valid for the *V. cholerae* sialidase, although the positions of the catalytic amino acids are different, due to the large size of this enzyme [791].

Since the tertiary structure of bacterial sialidases is similar to that found in viral sialidases (e.g. refs. [756,792–794], for a review see ref. [246]) and residues of the Asp-box have also been found, viral and bacterial sialidases can be considered to belong to one enzyme family. The few mammalian sialidases sequenced so far, i.e. from rat muscle [774] or hamster ovary cells [775], as well as trypanosomal sialidases (see in section 9.2.3), also show many structural features in common with viral and bacterial sialidases and are correspondingly members of this family. A DNA sequence of human origin showing features similar to microbial sialidase genes is also known [795]. However,



Fig. 18. Amino acids essential for bacterial sialidase action. The positions of the amino acid residues interacting with different parts of the Neu5Ac molecule or forming a hydrophobic pocket (by Leu, Trp and Met; indicated by a dotted line at the left side of the sialic acid molecule) are shown. Vc, *Vibrio cholerae* sialidase [791]; St, *Salmonella typhimurium* sialidase [790]; and Cp, *Clostridium perfringens* sialidase [R.G. Kleineidam, personal communication].

the expression of an active enzyme from this nucleotide sequence has not yet been reported. The human sialidase gene is located on chromosome 10 [796].

Interestingly, the leech sialidase resulting in Neu2,7an5Ac also belongs to this family, since the consensus repeat S-X-D-X-G-X-T-W was found in tryptic peptides of this enzyme [61].

The following are considerations with regard to the evolutionary origin of sialidases. Those enzymes whose gene structures are known, have been found in mammals or in microorganisms frequently living in a symbiotic or pathogenic association. This fact and the similarity of the structures and properties of mammalian and microbial sialidases together with the observation that the acquisition of sialidase activity may be of benefit for the microorganisms (see below) has led to the assumption that the sialidase gene may have been acquired by microorganisms from the host [797]. This may have occurred during infections by the uptake of DNA from decomposed host cells or during the life of microorganisms in intestine. It is therefore necessary to investigate the sialidase genes from lower animals, particularly of echinodermata, which are considered to be the "inventors" of both the sialidase and sialic acids. The sialidase from the starfish *Asterias rubens* was isolated to apparent homogeneity [798] and experiments for the elucidation of its gene structure are intended.

Evidence for a horizontal sialidase gene transfer between bacteria has been obtained by comparison of the similarities of bacterial sialidases so far sequenced [246,660,768,799]. It was found that some of the sialidases are related in accordance with the phylogenetic distances of their producers, e.g. *Micromonospora viridifaciens* and *Actinomyces viscosus*

sialidase, or the "large" (72 kDa) isoenzyme of *C. perfringens* and the *C. septicum* sialidase. On the other hand, the three "small" sialidases (42–44 kDa) produced by *S. typhimurium*, *C. perfringens* and *C. sordellii*, exhibit a higher similarity than is expected from the phylogenetic relationship of the bacterial species. These Gram-positive (*Clostridia*) or Gram-negative (*Salmonella*) bacteria are quite distinct from an evolutionary point of view, but are found at the same ecological location, e.g. in wounds or intestine of vertebrates. Here, an exchange of genes even between unrelated microorganisms might be possible via phages, transposons or plasmids by typical sequence motifs up- and downstream from sialidase genes of *S. typhimurium* and *M. viridifaciens* [772,783], and from the observation that the "small" sialidase gene is located near a phage attachment site on the chromosome of *C. perfringens* [800].

Further evidence for such a horizontal gene transfer comes from the investigation of the percentages of the bases G + C, which in the sialidase genes are frequently atypical for the chromosomal DNA of the respective bacterial species. From these data the hypothesis of a gene transfer, from mammals to microorganisms, is supported, since on basis of the mol% G + C, the origin of bacterial sialidase is expected in an organism exhibiting genes which contain about 45 mol% G + C, a value not uncommon in higher animals [660].

9.2.3. Trans-sialidases

As all chemical reactions, the sialidase reaction is theoretically reversible. This has been exploited, by choosing suitable conditions and bacterial or viral sialidases, for the α -2,3- or α -2,6-sialylation of oligosaccharides as final step in the course of chemical synthesis (see section 6.3). During the last years, several sialidases have been discovered in some protozoa which behave like normal sialidases if only water is present, but preferably transfer sialyl residues from one glycan chain to the terminal galactose of another non-sialylated oligosaccharide or glycoconjugate forming α -2,3-linkages.

The first hint for the existence of an unusual sialic acid transfer reaction was given in 1983, when sialic acids (Neu5Ac and Neu5Gc) were detected in Trypanosoma *cruzi* [662,801], which were found not to be synthesized by the parasites themselves. Since the molar ratio of these trypanosomal sialic acids corresponded to that of the incubation medium, their acquisition from this source, or in the case of an infection, from the host glycoconjugates, was assumed. Furthermore, an involvement of the sialidase activity found in these trypanosomes [802] and later localized on their cell surface, was suspected. These observations finally led to the discovery of enzymes called "trans-sialidases", first in the American species T. cruzi [803-805] and later in the African species T. brucei [755] and T. congolense [806]. A closer investigation revealed the presence of this enzyme in the whole T. brucei group, i.e. several strains of T. brucei brucei, T. brucei rhodesiense and T. brucei gambiense [806]. It also occurs in Endotrypanum promastigotes, an intraerythrocytic flagellate distantly related to trypanosomes [807]. In contrast, many other members of the kinetoplastida lack both sialidase and trans-sialidase activities, or express only sialidase, such as T. vivax [808] and T. rangeli [809]. This rapidly expanding field has been reviewed in refs. [246,810]. It is possible to discriminate between morphologically indistinguishable trypanosomatids by measurement of trans-sialidase and sialidase activities [811]. Furthermore, the occurrence of sialic acids on major cell-surface epitopes correlates with the expression of *trans*-sialidase [812].

With regard to substrate specificity, *trans*-sialidases exhibit a preference for sialyl α -2,3linkages in common with most of the usual sialidases. While for the T. cruzi enzyme only α -2,3- but not α -2,6-linked sialic acids are donors in the transfer reaction [550], the T. brucei trans-sialidase can also transfer α -2,6-linked sialic acids from sialyllactose to other glycans, however at an eight times lower rate than from sialyl($\alpha 2$ -3)lactose [755]. While both Neu5Ac and Neu5Gc are transferred equally well, Neu5,9Ac₂ from the corresponding sialyllactose derivative serves as sialyl donor only at a much reduced rate [696]. This may be the reason why N'dama cattle exhibiting a high degree of O-acetylation of erythrocyte sialic acids are more trypanotolerant than Zebu cattle, whose erythrocyte sialic acids are much less O-acetylated [703]. Only β-galactose residues of a variety of oligosaccharides and glycoconjugates are acceptors of these transferases. Sialyl($\alpha 2$ -3)lactose is the best donor for the *trans*-sialidase from *T. brucei* and lactose the best acceptor when compared with serum glycoproteins, mucins and gangliosides [755]. MUNeu5Ac also serves as a sialyl donor. A novel substrate for the assay of trans-sialidase activities is MUB-D-galactopyranoside as sialic acid acceptor which was tested with the T. brucei enzyme [806]. The main advantage of this accurate and sensitive assay is the avoidance of radioactive lactose as substrate.

Due to its wide substrate specificity but very limited linkage specificity, the *trans*sialidase from *T. cruzi* has been used for the sialylation of oligosaccharides, yielding Neu5Ac(α 2-3)Gal sequences, in glycotechnological experiments [554], discussed in detail in section 6.3.

Remarkably, the *trans*-sialidases from *T. cruzi* and *T. brucei* are not inhibited by Neu2en5Ac [550,809,813]. As this is in contrast to most other sialidases described in the groups 1–3 of the previous section, Neu2en5Ac is probably not an intermediate of the *trans*-sialidase reaction. Furthermore, N-(4-nitrophenyl)oxamic acid and related N-acylanilines do not inhibit *T. cruzi* or *T. brucei trans*-sialidases, which is also in contrast to the influence of this substance on the activity of sialidases from *V. cholerae* and other bacteria [813].

The term "trans-sialidase" used for the transglycosidase described in this section is not an adequate denomination of this enzyme, which also can react as a conventional sialidase, thus representing a hybrid glycosidase/glycosyltransferase. The reactions are shown in Fig. 12 of section 6.3. The product of the transsialidase reaction corresponds to that formed by sialyltransferases, however, instead of CMP-Neu5Ac, α -sialyl residues bound to β -galactose of glycan chains or to aglycones such as 4-methylumbelliferone (MU) are the sialyl donors. Therefore, the systematic name for the glycosyltransferase function of this enzyme, acylneuraminosyl-(or sialyl-): β -D-galactoside- α -2,3-sialyltransferase (EC 2.4.99.7?), has been proposed. This name takes into account that both Neu5Ac and Neu5Gc can be transferred.

Presently, several laboratories are intensively studying the gene structures of trypanosomal *trans*-sialidases, in order to understand the differences between this unique enzyme and conventional sialidases. In accordance with the latter sialidases, the *trans*sialidase genes of *T. cruzi* [768,776,814–818] also contain several Asp-boxes, first found in bacterial sialidases (see section 9.2.2 and Table 18). In addition, the *T. cruzi trans*- sialidase contains conserved amino acids, found to be involved in the catalysis of *S. typhimurium* sialidase [790] at similar positions [768,819]. It was shown at least for one of these amino acids, namely tyrosine-342, that it is involved in enzymatic catalysis [819]. Enzyme activity was lost after exchange of this amino acid by a histidine residue. Remarkably, the latter, inactive mutant is a member of a group of proteins with similar structures, expressed by several homologous genes, of which only some exhibit *trans*-sialidase activity [815,816]. The others are enzymatically inactive and their function is as yet unknown. However, these studies show that all animal, microbial and probably viral exo-sialidases and *trans*-sialidases studied belong to one gene family [660,768].

As first observed by Pereira et al. [776], the T. cruzi trans-sialidase consists of an N-terminal half, containing the enzymatic function and a C-terminal portion, composed mainly of a tandem series of twelve amino acid repeats. Four Asp-boxes exist within the N-terminus of the enzyme. However, only three of these conserved motifs occur in positions of the amino acid sequence similar to those found in bacterial sialidases, while the distance between the third and the fourth box is much larger in the parasite enzyme than in the bacterial proteins. It may be assumed that this extended sequence together with some other structural features is a domain functioning in the trans-sialidase mechanism [768,815]. The carboxy-terminal repetitive motif is most probably not involved in catalytic activity [815,820] and a recombinant T. cruzi trans-sialidase lacking the amino acid repeats was shown to retain the enzymatic activity [821]. Remarkably, the American parasite T. rangeli secretes a conventional sialidase [809] lacking transsialidase activity [822], the primary structure of which is related (70% identity of the amino acid sequences) to that of the trans-sialidase of T. cruzi [777]. Sequences encoding the tandemly repeated motif characteristic of the members of the *trans*-sialidase family were not detected in the T. rangeli genomic DNA.

T. cruzi trans-sialidase first named SAPA ("Shed-Acute-Phase-Antigen") and found in human serum in Chagas disease, is also unique from an immunological aspect, since it contains two immunologically distinct domains and seems to be a naturally chimeric protein having functionally independent enzymatic and antigenic domains [823]. (It is thus reminiscent of *V. cholerae* sialidase which consists of a catalytic domain and a non-catalytic sequence similar to those of legume lectins [768,791].)

Antibodies against both regions have been raised, and the one recognizing the catalytic region inhibits enzyme activity [788,823,824]. The SAPA epitopes were shown to be located on the tandemly repeated 12-amino-acid units and to induce an early and strong antibody response in acute and congenital human infections, as well as in mouse infections. In patients, antibodies against the immunodominant, non-enzymatic domain are formed much earlier (8 days) after infection than against the enzymatic domain (about 30 days) [823,824]. The availability of antibodies may be suitable for diagnosis and therapy of Chagas disease which affects several million people in Central and South America. The structural and functional properties of *Trypanosoma trans*-sialidases have been reviewed [825,826].

9.2.4. Pathophysiological significance of sialidases and trans-sialidases

9.2.4.1. Eukaryotic (trans-)sialidases. It is well known that sialidases are involved in the breakdown of sialoglycoconjugates, which may be of physiological or pathological

significance. These enzymes are physiologically important in the lysosomes, their main localization in animal cells, where they release sialic acids from glycoconjugates, thus initiating the degradation of these molecules by other glycosidases or proteases [5,8,33, 244]. The significance of mammalian sialidases is indicated by diseases, called sialidosis [827]. In these genetic disorders, sialidase activity is reduced or lacking, resulting in reduced turnover of sialylated glycoconjugates, excretion of sialo-oligosaccharides in the urine as well as mental and physical impairments (see below).

Since sialidases not only occur in lysosomes, but also in other compartments such as the cytosol or plasma membranes [33,244,246,660], other cellular functions of these enzymes are assumed. This is easily understandable, since sialic acids are not only structural but also functional constituents of many biologically active molecules of the cell, for example receptors, ion channels and growth-regulating molecules (refs. [246,660] and section 10). Sialidases may regulate the functional activity of these and other molecules. Plasma-membrane-bound sialidase for instance is involved in the degradation of cell membrane gangliosides, which may not only represent a catabolic function, but also a regulative effect on the biological functioning of these molecules [828]. A striking example of such a role is the myelin-associated sialidase of rat oligodendroglial cells which was found to adhere to immobilized GM1 and to be inhibited by this ganglioside [829]. This interaction can also be inhibited by the sialidase –GM1 interaction plays a role in the formation and stabilization of the multilamellar structure of the myelin sheath.

Another system of eukaryotic sialidases, whose significance we are just beginning to understand, is the trypanosomal *trans*-sialidase described above. It is evident that the main function of this enzyme is the acquisition of sialic acids by the parasites at expense of the host which is presumably of benefit for the trypanosomes. In the case of *T. brucei* the accepting glycoprotein is the GPI-anchored "procyclic acidic repetitive protein" (PARP/procyclin) [755] and in the case of *T. cruzi* the Ssp-3 epitope of trypomastigotes [823]. In epimastigote forms glycoproteins with molecular masses of 38–43 kDa were described to be the main sialic acid acceptor, and the structures of the oligosaccharides of these glycoproteins, linked O-glycosidically via GlcNAc to serine or threonine, were elucidated [830].

Some understanding of this chemical modification of cell surface molecules may come from consideration of the period of sialylation during the life cycles of the African and American trypanosomes. In *T. brucei*, sialylation only occurs in the insect stage, where sialic acids are probably derived from mammalian blood sucked in by the tsetse fly [806]. In contrast, in *T. cruzi* sialic acid is acquired by the parasite not in the insect (triatomine bugs) stage, but only during its life in the mammalian host. Since sialylation may be of advantage to the trypanosomes, it is hypothesized that the dense, charged glycocalix formed by sialylation of *T. brucei* may protect *T. brucei* from the attack of proteases, glycosidases, antibodies or complement from the fly's digestive tract or from the ingested blood meal [806]. It is also conceivable that the parasites require sialic acids for maturation and adhesion to the fly's salivary glands before transfer to the mammalian hosts. Furthermore, after infection, sialylation may be of benefit for the survival of the parasites in the primary infection site ("chancre") of the host's skin, before the non-sialylated variable surface antigenic glycoproteins (VSG) are expressed. These antigens then represent another potential site for the immunological protection of the parasites when sialic acids are no longer available.

Since T. cruzi does not express such variable antigens on the cell surface, it is assumed that it starts to express *trans*-sialidase in the mammalian host to capture sialic acids for defense against the host's immune and complement systems and against phagocytosis [818,831,832]. The role of sialic acid in the resistance of T. cruzi trypomastigotes to complement has clearly been shown [833]. Other experiments demonstrated that the acquisition of sialic acids increases the infectivity of T. cruzi and is involved in invasion of host cells by these parasites [823,831,834], and accordingly the action of sialidase reduces their virulence. In mice the virulence-promoting activity of *trans*-sialidase was achieved with tiny doses of $1-2\mu g$ enzyme/kg animal [835]. (An increase of pathogenicity, i.e. evasion of host defence systems, is also achieved by sialylation of some pathogenic bacteria, such as group B Streptococcus [836] and Neisseria gonorrhea [837] (see also Table 3 and sections 3 and 10.2). Correspondingly, monoclonal antibodies against the Ssp-3 epitope, the target of trans-sialidase activity and required for invasion after sialylation inhibit infection of the host cell [823,838]. Such antibodies or other transsialidase inhibitors (see below), as well as prevention of the expression of this enzyme on the gene level, may become potent therapeutic tools in Chagas and Nagana (sleeping) diseases.

The role of conventional sialidases secreted by non-sialylated trypanosomes lacking an insect vector (see above), such as *T. rangeli* [809,822] or *T. vivax* [806], is unknown. It is likely that they hydrolyze sialic acids from host glycoconjugates for nutritional purposes as do bacteria (see below). It is, however, also possible that these trypanosomes demask recognition sites on host cells, e.g. subterminal galactose residues, for their attachment mediated by lectins or other receptors on the parasite's surface. This hypothesis clearly warrants further investigation.

Infection of host cells by *T. cruzi* parasites is a complicated mechanism which is not yet fully understood. However, sialylation of both cell types seems to be necessary for this process. For the attachment and penetration of *T. cruzi* into host cells the latter must be sialylated, as was shown with CHO cell mutants expressing no or varying amounts of sialic acids. O-Linked sialylated glycans of host cells were shown to be particularly important for infection [838]. Various trypanosomal surface glycoproteins are considered to be involved in adhesion, i.e. the developmentally regulated MTS-gp82 [839], a small mucin-like sialoglycoprotein [840], or enzymatically inactive members of the (*trans*-)sialidase family, which may be capable of binding sialic acid without cleaving it, and in this way may function as mammalian-stage-specific surface receptors [841]. On the parasite surface, the relative balance between exposed sialic acid and galactose/*N*-acetylgalactosamine residues may determine the parasite's capacity to invade host cells including macrophages [842]. A stage-specific 82 kDa adhesion molecule of *T. cruzi* metacyclic trypomastigotes may play a role in host cell invasion [843].

After binding, *T. cruzi* enters host cells via an acidic vacuole, which fuses with lysosomes [818,844–846]. The parasitophorous vacuole formed in this way is later disrupted in order to release the trypanosomes into the cytosol. *Trans*-sialidase activity seems to be required for the latter step, as it desialylates the highly sialylated lysosomal

membrane, which then becomes more susceptible to a parasite hemolysin [844,846]. The crucial role of sialic acid in host-cell-parasite interaction and the reversible sialylation of cell surface molecules mediated by sialidase and *trans*-sialidase activities, which in some respects resembles the infection of mammalian cells by influenza viruses, has also been pointed out in other studies [826,847].

In a species of *Acanthamoeba*, another protozoon, the sialidase of the trophozoites and cysts also seems to play a pathophysiological role, as it was shown to be involved in colonization and damage of the sialic-acid-rich corneal epithelium [848]. Spreading of the microorganisms may have been facilitated by a decrease in the viscosity of the protective mucus layer occurring after desialylation.

9.2.4.2. Bacterial sialidases. In bacteria, sialidases also seem to be of great significance in pathogenesis, although only few well-defined effects are known, in contrast to the wide occurrence of this enzyme. Strikingly, bacterial species pathogenic for higher animals and man often produce sialidases so that it is suspected that they enhance virulence. It was also discussed above that bacteria may have acquired the sialidase gene from the mammalian host, which may have been of evolutionary advantage.

Certainly, not only one function can be attributed to bacterial sialidases, since their cellular localization varies much with the species. They may be intracellular, periplasmic, membrane-bound or excreted, and they vary in their properties such as complexity, molecular mass and substrate specificity (refs. [33,244,246,660,768], and section 9.2). However, they belong to one gene family.

One of the main functions of bacterial sialidases is nutritive, providing an energy source for the microorganisms [245,246]. This has long been assumed and a model first proposed for the *C. perfringens* enzyme shows that sialic acids released from host tissues by secreted sialidase are taken up by the bacteria with the aid of a permease, intracellularly cleaved by acylneuraminate-pyruvate lyase to pyruvate and N-acylmannosamines (section 9.3) and then further metabolized [5,33,245]. After the loss of sialic acids, host cell membrane glycoconjugates are more vulnerable to the action of other glycosidases and proteases secreted by bacteria, since the presence of terminal sialic acids on glycoconjugates is known to hinder the action of these enzymes [5]. Thus, sialidase can be considered as pacemaker in the breakdown of host cells by a lytic cocktail secreted by many infectious bacteria.

Suitable models are required to verify this assumption. For example, the advantage of the availability of free sialic acid for bacterial survival has been demonstrated in two model systems [849]. The growth of two isogenic strains of *Bacteroides fragilis* in CHO cell culture monolayers *in vitro* or in rat granuloma pouches *in vivo* was compared. While one strain expressed sialidase, the activity of this enzyme was deleted in the other group. At the beginning of the culture, both strains grew equally well, however, after 48 to 72 h, the sialidase-producing strain was enriched. The latter bacteria grew more rapidly than those lacking sialidase, because after the consumption of glucose some 16 to 24 h after infection, other carbon sources, including liberated sialic acid, were required for further bacterial growth. This experiment supports the assumption that sialidase may facilitate spreading and thus virulence of bacteria in infected tissues. Similar conclusions were drawn from studies on arteritis caused by *Erysipelothrix*

rhusiopathiae [850], in which a close correlation between bacterial invasion, desialylation and cell infiltration in the common iliac artery was observed.

Several clostridial species secrete large amounts of this enzyme into their environment, which, together with other lytic enzymes, is believed to facilitate their rapid and life-threatening spreading and desialylation of vital structures in the whole organism [851, 852]. The sialidase production of *C. perfringens* is so great that it can be detected not only in wound fluids, but also in blood serum. Antibodies were raised against sialidases from *C. perfringens*, *C. septicum* and *C. sordellii*, the main causative agents for clostridial myonecrosis or gas oedema, which allow an early, sensitive and specific diagnosis of this infection in wound fluids or blood serum. This enables a prompt, specific and therefore life-saving treatment of this severe disease.

The action of sialidase also contributes to the liquification of mucin barriers lining endothelia, as mentioned above for *Acanthamoeba* infection of the eye. This property is most relevant in tracheobronchial, intestinal and urogenital tracts [245,853]. Often other hydrolases, "mucinases", support the viscosity-decreasing effect initiated by mere desialylation of mucins, as was observed with intestinal [854,855], vaginal [856] and other [857] bacteria, as well as with the flagellates *Trichomonas vaginalis* and *T. foetus* [858]. Microbial sialidase is considered a risk factor for intra-uterine infection and preterm birth [858].

O-Acetylation of mucin-bound and membrane-bound sialic acids makes them more resistant towards sialidase action [8,33,245,853,854]. This may be one of the reasons why intestinal mucins, especially of the colon, are often *O*-acetylated (ref. [245], and section 8.4.2). The high level of *O*-acetylation of sialic acids observed in the endothelia of blood vessels, e.g. in liver, detected by histochemical methods using influenza C virus hemagglutinin, is assumed to have a similar function [234,235,730].

A pathogenic role facilitating colonization and invasion of the host during the development of pneumonia is also attributed to the sialidase from *Streptococcus pneumoniae* [859]. This enzyme has also been sequenced and found to contain four copies of the sequence S-X-D-X-G-X-T-W typical for sialidases [784]. The antigenic potential of this enzyme as well as other pneumococcal proteins makes them potentially suitable candidates for the vaccination against lung inflammation (refs. [784,860], and [K. Fischer, personal communication]). During the inflammation of bovine lung, sialidase from *Pasteurella haemolytica* may play a supportive role [861].

In all these cases, the production of sialidase may be physiologically important for the microorganisms causing an enhancement of their virulence. A further, well-defined example for such a role is the *Vibrio cholerae* sialidase, which is an accessory virulence factor, causing a partial desialylation of higher sialylated gangliosides in intestinal endothelia, resulting in GM1 [770]. Cholera toxin subsequently binds to this ganglioside and is taken up by intestinal cells, which are stimulated to secrete cell fluid required for growth of the vibrions in the intestinal lumen. The efficiency of the sialidase in this process is increased by the capability of the enzyme to attach to cell surfaces via lectinlike domains [768,791].

Since sialic acids are involved in so many important physiological processes in animals and man (section 10), the flooding of a tissue or the whole organism with bacterial sialidases is potentially disastrous, leading to acute and chronic diseases.

As a consequence of sialidase production by microorganisms and viruses, chronic inflammatory or immunological diseases such as glomerulonephritis or asthma may result [245,795]. Sialidase-producing bacteria in disease and phenomena resulting from the action of sialidases in bacterial infections have been summarized [245].

9.2.4.3. Viral sialidases. Sialidase is a surface-bound enzyme of ortho- and paramyxoviruses as reviewed [246,862]. The enzymes of influenza A and B viruses have been intensively studied with regard to their structure, substrate specificity, immunological properties and function during viral infection. These sialidases play a crucial role during penetration of the virus into cells and probably also in their release from infected cells, especially of respiratory endothelia, after the attachment to these cells via the sialicacid-specific receptor hemagglutinin [246,863,864]. The first step of this process is the binding of viruses to cell surfaces, which is followed by loosening of these linkages as a prerequisite for endocytosis of virus particles. Thus, the host sialic acid is exploited by the infecting virus. Sialidase may also be required by the virus to reach the endothelial surface and not to be bound to the highly sialylated, sticky, protective mucus layer, which would lead to elimination of the virus from the respiratory tract.

9.2.5. Sialidase and trans-sialidase inhibitors

The involvement of sialidases in the severe pathogenesis of many viruses and other microorganisms and the aggravation of these infectious diseases including late and often irreversible, e.g. immunological effects prompted pharmaceutical strategies for the inhibition of these enzymes. Some sialidase inhibitors will therefore be described in the following. For the terminology and abbreviations of natural and synthetic sialic acids see Tables 1 and 13 and section 6.1.

A survey of sialidase inhibitors was given earlier [5,33,865] and some recent inhibitors of higher efficiency ($K_i < 10^{-4}$ M) are listed in Table 19. Neu2en5Ac is a competitive inhibitor which is still widely used, since it inhibits most viral, bacterial and mammalian sialidases with a K_i of around 5µM. This substance is therefore very useful to prove the specificity of sialidase action when studying the role of sialic acids in biological experiments. The replacement of the *N*-acetyl moiety in Neu2en5Ac by an *N*-trifluoroacetyl residue leads to an about 2-fold better inhibitor of *V. cholerae* sialidase with a K_i of 1.3–1.8 × 10⁻⁶ M [389,866]. In the case of 9d-Neu5Ac a K_i of 1 × 10⁻⁵ M was measured [869]. Using *V. cholerae* sialidase, it was shown, with various Neu2en5Ac derivatives structurally varied at C4, that the highest inhibitory effect was obtained with compounds having an axial hydroxyl residue, as in the parent substance Neu2en5Ac [430], whereas the 4epi- and 4d-derivatives were less inhibitory. These studies also revealed that 4amino-Neu2en5Ac inhibits *V. cholerae* sialidase only slightly, which is in contrast to influenza virus sialidases. The transition-state analogue 4d-Neu2en5Ac inhibits influenza virus sialidase with a K_i of 8 × 10⁻⁵ M) [441].

Most interest is currently being focussed on the inhibition of viral sialidases, since their role in the infection mechanism has long been known. The fact that Neu2en5Ac is a transition-state analogue for the sialosyl residue during the viral enzyme reaction described in section 9.2.1 [60,320], served as a point of reference in the design of new inhibitors. The observation that the 4-O-acetyl group in the ring of Neu4,5Ac₂

Sialidase inhibitors ^a					
Name	Inhibition (M)	Sialidases tested	Reference(s)		
Neu2en5Ac	$K_i = 1 \times 10^{-6} - 2 \times 10^{-5}$	viral, bacterial, mammalian	[428,865]		
5-Trifluoroacetyl-Neu2en	$K_i = 8 \times 10^{-7} - 2 \times 10^{-6}$	viral, bacterial	[389,866]		
4amino-Neu2en5Ac	$K_i = 4 \times 10^{-8} - 7 \times 10^{-4}$	viral, bacterial, mammalian	[428,867,868]		
4guanidino-Neu2en5Ac	$K_i = 1 \times 10^{-9} - 1 \times 10^{-2}$	viral, bacterial, mammalian	[428,867,868]		
3F _{eq} -Neu5Ac	$K_i = 2 \times 10^{-6} - 8 \times 10^{-6}$	viral, bacterial, mammalian	[442]		
2d-6amino-2H _{ax} -Neu5Ac	$K_i = 5.4 \times 10^{-5}$	bacterial	[438,452]		
2d-4epi-6amino-2H _{ax} -Neu5Ac	$K_i = 2.9 \times 10^{-5}$	bacterial	[438,452]		
Phosphonate analogue of Neu5Ac	$K_i = 5.5 \times 10^{-5}$	bacterial	[449]		
7d-Neu2en5Ac	$K_i = 9.0 \times 10^{-5}$	bacterial	[869]		
8d-Neu2en5Ac	$K_i = 5.0 \times 10^{-5}$	bacterial	[869]		
9d-Neu2en5Ac	$K_i = 1.1 \times 10^{-5}$	bacterial	[869]		
8epi-Neu2en5Ac	$K_i = 3.8 \times 10^{-5}$	bacterial	[870]		
Neu5Ac α (2-S-6)Glc(β 1-1)ceramide	$K_i = 2.8 \times 10^{-6} - 4.0 \times 10^{-4}$	viral, bacterial	[871]		
Neu5Ac α (2-S-6)Gal(β 1-4)Glc(β 1-1)ceramide	$K_i = 1.5 \times 10^{-5}$	viral	[871]		
Siastatin A and B	$K_i = 1.7 \times 10^{-5} - 4.3 \times 10^{-5}$	bacterial	[865,872]		
N-(1,2Dihydroxypropyl)-siastatin B	$IC_{50} = 1.0 \times 10^{-5} - 2.7 \times 10^{-5}$	bacterial	[873]		
N-(1,2Dihydroxypropyl)-4-deoxy-siastatin B	$IC_{50} = 5 \times 10^{-6} - 5 \times 10^{-5}$	bacterial	[873]		
3epi-Siastatin B	$IC_{50} = 1 \times 10^{-5} - 7.4 \times 10^{-5}$	viral	[874]		
5,7,8,4'-Tetrahydroxyflavone (Isoscutellarein)	$K_i = 4.1 \times 10^{-5}$	viral	[875]		
5,7,4'-Trihydroxy-8-methoxy-flavone	$1C_{50} = 5.5 \times 10^{-5}$	viral	[875]		

Table 19

^a For the abbreviation of natural and synthetic sialic acids see Tables 1 and 13, respectively.

completely hinders the action of mammalian and bacterial sialidases [5,33] but allows slow hydrolysis of the glycosidic bond by viral enzymes [252,876], as well as the finding by X-ray crystallography studies that in the active center of viral sialidases only acidic amino acids (i.e. aspartic acid) interact with HO4 of sialic acid, led to the synthesis of much more potent inhibitors of influenza viruses. These are 4amino-Neu2en5Ac and 4guanidino-Neu2en5Ac [428,429,867,868,877], which exhibit K_i values in the nanomolar range, corresponding to 2-4 orders of magnitude lower than Neu2en5Ac. The guanidino derivative inhibits more effectively than the amino compound. 4-Guanidino-Neu2en5Ac was shown to inhibit influenza virus A sialidase in a slow-binding fashion, and a K_i value of 3×10^{-11} M was observed for the tightly bound form under steady-state conditions [878]. This strong binding to viral sialidases and not to those from bacteria or mammals, is due to the existence of a pocket in the binding site of viral sialidases near to the HO4 group of sialic acid [867,879,880]. This pocket is considered to be the reason why Neu4,5Ac₂ [252] and Neu5Ac4Me [407] can be released by viral sialidase, but not by the bacterial enzyme. Since the other sialidases do not seem to have this pocket (for bacterial sialidases see refs. [790,881]) and are therefore inhibited to a much lesser extent, the new Neu2en5Ac derivatives appear suitable for the prophylaxis and/or therapy of influenza A and B virus infections [867]. It has been shown for 4guanidino-Neu2en5Ac that it inhibits the growth of a wide range of influenza A and B viruses in in vitro systems, including human respiratory epithelium, at lower concentrations than classical inhibitors of virus replication such as amantadine or ribavirin [868,882-884]. The same effect was observed in mice and ferrets when the new compounds were administered as a nasal spray [867,880,882], thus raising hopes that these non-toxic substances can be used therapeutically during epidemics of influenza in man. These viral sialidase inhibitors are an excellent example of how drugs may be designed on the basis of a crystal structure of a target protein and on other biochemical parameters [867]. Potent inhibitors of bacterial sialidases or trypanosomal trans-sialidases are also desirable as pharmaceuticals to assist the therapy of inflammatory diseases caused by these microorganisms.

Some natural substances exist which inhibit viral sialidases in the $10\,\mu$ M range, such as siastatin A and B and panosialin [865] as well as flavonoids [875] (Table 19). Two epimers of siastatin B, 3-epi-siastatin B and 3,4-diepi-siastatin B, have been synthesized which are good inhibitors of sialidases from various influenza virus strains, as well as from Sendai and Newcastle disease viruses [874]. They effectively inhibit the growth of these viruses in cell cultures. Sialidases from *Streptococcus* sp. and *C. perfringens* are inhibited to an extent comparable to Neu2en5Ac by the *N*-(1,2-dihydroxypropyl) derivative of siastatin B and its 4-deoxy analogue [873]. Isoscutellarein (5,7,8,4'-tetrahydroxyflavone) isolated from the leafs of *Scutellaria baicalensis* is a good inhibitor of influenza virus sialidases ($K_i = 4 \times 10^{-5}$ M). It is not toxic to mice and inhibits replication of influenza viruses in Madin–Darby bovine kidney cells. It was shown to be a better inhibitor than amantadine. The 8-methoxy derivative of isoscutellarein is also a potent inhibitor [885].

Thioglycoside-analogues of gangliosides [871] (for structures see Table 19) are good inhibitors of viral sialidases, but hinder the bacterial enzymes to a considerably lower degree ($K_i = 0.1 \text{ mM}$). Various sialyl($\alpha 2$ -6)galactosides in which one of the hydrogen atoms at C6 of Gal has been replaced by a methyl group, yielding R and S isomers, and in some of which thioglycosidic bonds occur, act on both viral and bacterial

sialidases [886]. Although their inhibition constants are only in the millimolar range, these compounds are expected to have important implications in the rational design of sialidase inhibitors. The finding that these glycosides are the first common inhibitors of both hemagglutinin and sialidase of influenza viruses may prove to be of great advantage. The *o*-(diffuoromethyl)phenyl-glycoside of α -Neu5Ac irreversibly inhibits *C. perfringens* sialidase [887]. This mechanism-based inhibitor acts by a fluoromethylene-quinone formed spontaneously from the liberated difluoromethylphenol. Analogues of free Neu5Ac, such as HO4 epimers of 6-amino-6-deoxy-sialic acid are competitive inhibitors of medium potency for both viral and bacterial sialidases [438]. 3F-Neu5Ac strongly and competitively inhibits bacterial, viral and mouse spleen sialidases [442]. Other strategies for the production of sialidase inhibitors are the Neu5Ac isomer with a 6-acetylamino group occurring in the furanose ring form, which could be interpreted as a 6-acetylamino derivative of Kdn or 4epi-Kdn [888]. Promising inhibitors also represent 4azido-Neu5Ac-, 4amino-Neu5Ac- and 4NAc-Neu5Ac-containing sialosides and thiosialosides [889].

Another principle for tailor-made sialidase inhibitors is the design of low-molecularmass compounds that mimic the binding function of a macromolecular antibody. Such studies are based on the crystal structure of an influenza virus N9-sialidase (antigen)–NC41 (antibody) complex, showing the direct contact of four amino acid residues on the antibody binding surface with the active-site loop 368-370 of the antigen [890]. Correspondingly, a constrained cyclic peptide composed of 5 amino acids was synthesized, which mimics the receptor-bound conformation of these amino acids and inhibits the sialidase with a K_i of 0.1 mM. Although this is about 3 orders of magnitude less than the parent protein, because the antibody forms contacts with about 17 amino acid residues of the antigen sialidase, it nevertheless opens a possible new approach for the design of more potent inhibitors of this important viral enzyme.

The rapid development of research on trypanosomal trans-sialidases, their involvement in the pathogenic mechanism and the morbidity of men and agricultural animals caused by these parasites, has evoked considerable interest in inhibitors of these enzymes. One potential hope is the availability of antibodies, mentioned in sections 9.2.3 and 9.2.4.1. There it was also outlined that classical sialidase inhibitors like Neu2en5Ac and N-(4-nitrophenyl)oxamic acid are not effective on the trans-sialidases studied. A nonimmunological, non-competitive inhibitor of T. cruzi sialidase activity, named cruzin, was first found in blood plasma from patients with Chagas' disease and later identified as high density lipoprotein (HDL) [834,891]. This lipoprotein does not inhibit sialidases from T. rangeli, various bacteria and influenza viruses. Remarkably, in high concentrations it enhances infection of culture cells by T. cruzi trypomastigotes, while V. cholerae sialidase reduces it, which points to the significance of sialylated structures on the parasites during infection. All these effects could be explained by the fact that HDL is not an inhibitor of T. cruzi sialidase, but as a glycoprotein provided sialic acids for the trans-sialidase action of this enzyme, thus mimicking an inhibition of enzymatic sialic acid release in the test system used [815].

9.3. Sialate-pyruvate lyase

As shown in Fig. 13, sialic acids liberated by sialidase are degraded by the aldolase systematically named sialate (acylneuraminate)-pyruvate lyase (EC 4.1.3.3), resulting in



Fig. 19. Model of sialic acid structural features required for the interaction with sialate lyase and binding of inhibitors. Based on data from ref. [892].

acylmannosamines and pyruvate. This reaction and the occurrence of the soluble enzyme in both bacteria and higher animals as well as the influence of sialic acid substituents and enzyme inhibitors has been reviewed [5,8,33]. The biological role of this lyase in bacteria is believed to be nutritive, by splitting the sialic acid 9-carbon chain as prerequisite for further consumption in the energy-producing metabolism. In animals, the main function of the cytosolic enzyme seems to be in the degradation of sialic acids in order to regulate sialic acid metabolism by prevention of recycling of this sugar.

With regard to substrate specificity, the lyase from pig kidney, the first mammalian sialic-acid-specific aldolase purified [891a], shows similar behavior to the enzyme from *Clostridium perfringens* studied earlier [5,33]. The relative cleavage rates for Neu5Ac, Neu5Gc and Neu5,9Ac₂ are 100%, 70% and 33%, respectively. Neu4,5Ac₂ and Neu2en5Ac are inactive. An investigation into the kinetic behavior of various epi- and deoxy-analogues of Neu5Ac with the lyase from *C. perfringens* revealed that modifications at C8 and C4 strongly reduce cleavage rates and affinity of the substrates to the enzyme, or are even inhibitory [409,892,893]. 4Epi- and 4d-Neu5Ac were most inhibitory, with K_i values in the millimolar range. Computer-assisted drawings of these sialic acids (CPK models) indicate that the region most important for the binding of sialic acids to the enzyme is an equatorial zone stretching from C8 via the ring oxygen atom to C4 of the monosaccharide. The substituents at C5 and C9 may be varied to a greater extent without significantly disturbing enzyme action. The structural features of sialic acid required for the interaction with the lyase [892] are depicted in Fig. 19.

With regard to enzyme catalysis, for the *C. perfringens* lyase a Schiff-base mechanism involving a lysine residue was elucidated to be part of the catalysis of this enzyme [5, 33]. The participation of a histidine residue in the cleavage reaction was also shown to be likely. The same amino acids probably are also parts of the active center of the aldolase purified from porcine kidney [891a], as was shown by borohydride reduction in the presence of Neu5Ac and the influence of reagents such as Rose Bengal which interacts with histidine. Heavy metal ions and other substances were also inhibitory, pointing to the essential role of cysteine residues.

¹H NMR spectroscopic investigations have shown that only the α -anomer of Neu5Ac is consumed by the *C. perfringens* lyase, yielding α -ManNAc, as described in detail in section 5.3.5. From this observation, the chemical modification experiments, and models drawn in refs. [33,892], a reaction scheme for the heterolytic fragmentation of cyclic sialic acids into pyruvic acid and *N*-acylmannosamines is delineated (Fig. 20, overleaf).

Since the enzyme reaction is reversible, conditions for the synthesis of Neu5Ac and natural or synthetic derivatives in high yield as well as of Kdn with the aid of bacterial lyase were elaborated (see section 6.1). The recombinant and overexpressed sialate-pyruvate lyase from *E. coli* is now in wide use as a specific chiral catalyst which mediates highly enantioselective aldol condensation reactions leading to a variety of sialic acids.

The enzyme first purified from *E. coli* [894] was cloned, the nucleotide sequence of its gene elucidated [895–897] and later crystallized [898]. The lyase gene encodes a polypeptide of 297 amino acids [897]. The three-dimensional structure of this aldolase from *E. coli* has been investigated by X-ray crystallography and shown to be a tetramer, each subunit representing an eight-stranded α/β -barrel [899] (Fig. 21). The active center was tentatively identified as a pocket located at the carboxy-terminal end of this barrel. Lysine-165 lies within this pocket and is probably the reactive residue forming the Schiffbase intermediate with the substrate described above (see Fig. 20). Several additional amino acids were recognized to line this pocket.

It is tempting to speculate that, as with sialidases, an exchange of the gene for the sialate lyase between mammals and microorganisms may have occurred. This assumption is based on the similarity of many properties of both bacterial and mammalian lyases studied so far.

It has been mentioned in section 5.3.2 and refs. [11,12] that the commercially available sialate lyase is an excellent tool for the analysis, e.g. by HPLC or GLC, of a substance suspected to be a sialic acid. The enzyme has also proved to be useful in clinical determinations of these monosaccharides [900,901]. For the analysis of glycoconjugate-bound sialic acids, e.g. in the bloodstream, a combination with sialidase treatment is necessary.

9.4. Sialic acid permease

Although only few experiments have been carried out, sialic acids do not seem to be able to penetrate bacterial and animal cell membranes at significant amounts. Only minimal amounts of radioactivity from Neu5Ac are taken up by surviving tissue slices from porcine submandibular gland [679]. Furthermore, Neu5Gc and Neu2en5Ac administered orally and intravenously into mouse and rat are excreted rapidly, mainly in the urine [902]. While the unsaturated sialic acid was excreted completely, less than 10% of the radioactivity from Neu5Gc was found in the tissues. It is assumed that most of this portion was cleaved by the sialate-pyruvate lyase before its uptake by cells.

Since sialic acids released from host glycoconjugates by sialidases are of advantage for the growth of bacteria, as described above, it is conceivable that these microorganisms developed a mechanism for the uptake of these anionic sugars [245]. First hints for the existence of a sialate permease were obtained with *C. perfringens* when studying the induction of both sialidase and sialate lyase with free or glycopeptide-bound




Fig. 20. Proposed reaction scheme of sialate-pyruvate lyase from Clostridium perfringens. Based on data from refs. [33,892].



Fig. 21. Three-dimensional structure of the sialate-pyruvate lyase from *Escherichia coli*. (top) Viewed down the β -barrel axis from the carboxy-terminal end of this aldolase. The putative catalytic residue Lys-165 is shown in ball-and-stick representation. (bottom) Putative active site of Neu5Ac lyase showing the side chains of nine of the residues forming the surface of the pocket. Carbon atoms are white, oxygens black and nitrogens grey. From ref. [899] by permission of Current Biology Ltd., London.

Neu5Ac [903]. A $K_{\rm M}$ value of 0.3 mM for Neu5Ac was measured for this hypothetical transporter protein. This question has not yet been solved in *C. perfringens*, although recent experiments have also shown that sialic acid uptake is not due to mere diffusion (C. Traving et al., unpublished). However, in *E. coli* evidence for a sialic acid permease was obtained, based on molecular biological studies [585,904].

10. Physiological and pathobiochemical significance of sialic acids

The growing understanding of the involvement of sialic acids in many biological and pathological processes has considerably enhanced interest and research in this field. It started in 1946 when McFarlane Burnet and colleagues [905] recognized the reversible binding of influenza viruses to mammalian erythrocytes via cell surface sialic acids. Several reviews on the biological aspects of sialic acid biochemistry have appeared [5,8,245,660,663,690,853,906–909].

From these articles it is clear that there are general functions of sialic acids resulting from their physico-chemical properties such as electronegativity, hydrophilicity and relatively large size, coupled with their exposed position on cells and secreted macromolecules. However, there is a growing number of examples where sialic acids exert specific functions, mostly in molecular and cellular adhesion phenomena. The influence of natural and artificial modifications of the structure of Neu5Ac on enzymatic and some biological phenomena has been mentioned at relevant places throughout this article, showing that the configuration of Neu5Ac and related molecules is of great importance for the metabolism and biology of these monosaccharides, including those of the whole glycoconjugate molecules, cells and tissues, to which sialic acids are bound.

When trying to summarize the numerous roles of sialic acids, three groups of main effects can be described. Although there is considerable overlap, this classification has proved to be very useful and has been supported by many recent examples. Due to lack of space, only a few characteristic examples will be given here and new trends will be emphasized.

10.1. General physico-chemical effects

Firstly, due to their high acidity (pK value around pH 2), sialic acids are fully dissociated under physiological conditions and act as anions for the binding of inorganic and organic cations, e.g. on cell surfaces, thus facilitating the cellular uptake of substances with physiological or pharmacological significance. Cell aggregation may be prevented by the repulsive effects of the negatively charged sialic acids, as was observed for example when studying the adhesion of cultured cells to their substratum [910]. These cells adhered better to a layer of desialylated gangliosides than to the native, negatively charged, species, or after desialylation of the cells. The prevention of erythrocyte aggregation by surface sialic acids is a further, excellent example of their repulsive effect [911,912]. After the removal of sialic acids by sialidase, erythrocytes were found to aggregate readily, either without the influence of other compounds or under the influence of immunoglobulins and fibrinogen, respectively. Correspondingly, erythrocytes which are undersialylated in diabetes, as was analyzed with their glycophorin components, tend to increase aggregation and adhesion to blood vessel walls. This may be of importance in the development of vascular disease during diabetes [913]. The high degree of sialylation existing in the epithelial sialomucin "episialin", which is expressed not only on several secretory epithelial cell types but also in many carcinoma cell lines, is also believed to have antiadhesion function. This effect may facilitate metastasis of tumor cells [914]. Such a role can also be attributed to the sialic acids found on the surface of Streptococci of the serological group B. In contrast to the wild-type strain, an asialo mutant obtained by the transposon method showed stronger cell adherence, probably due to the more hydrophobic surface resulting from the lack of sialic acid [915].

In contrast to this repulsive effect of sialic acids, cell adhesion may be facilitated via positively charged substances or Ca²⁺ bridges. Furthermore, binding of Ca²⁺ to ganglioside sialic acids is of great importance in the function of nervous tissues [916]. These glycoconjugates are localized in clusters on neuronal and especially synaptic membranes in the vicinity of a membrane-bound calcium pump, thus facilitating the supply of these ions for neuronal cells. These calcium–ganglioside interactions may modulate neuronal functions, not only for the short-term process of synaptic transmission of information, but also for long-term events of neuronal adaptations including storage of information. The highly charged anionic sialic acid residues, α -2,3- and α -2,6-glycosidically bound to the trabecular meshwork of the human eye, are considered to be involved in the regulation of the aqueous outflow and the control of intraocular pressure [917].

Secondly, sialic acids may modulate the conformation of glycoproteins and gangliosides, thus influencing their arrangement, physical properties and biological functions in cell membranes, as well as the solubility, thermal stability and other properties of soluble glycoproteins including sialylated enzymes and hormones [5,8,907]. This is, indeed, a very large field and we still are only beginning to understand the influences of glycoconjugate conformations in cell biology. To obtain such information on individual molecules existing on living cells is particularly difficult to obtain with the techniques available. However, with α -dodecyl-Neu5Ac anchored to the surface of a phospholipidbased membrane fragment, it was shown by ¹H and ¹³C NMR studies that the sialic acid pyranose ring is extended into the aqueous phase with the carboxyl group remaining close to the membrane surface, where it may interact with a hydrogen acceptor [918]. For glycoproteins it has been postulated that the carbohydrate moiety mimics the effect of a molecular chaperon [919].

The anti-proteolytic effect, long known [906] to be exerted by glycan chains on glycoproteins in general and sialic acids in particular, should also be mentioned here. This may be due to conformational changes mediated by sialic acids, to their negative charge, or to mere shielding of e.g. peptide bonds from proteases by the bulky carbohydrate moiety. The high grade of sialylation of lysosomal cell membranes [920] may be a protective barrier against proteases in this cellular compartment. Glycosylation of influenza virus hemagglutinin governs hydrolysis of certain peptide bonds by host proteases in order to enhance the virulence of these viruses [863,921]. It is suggested that the removal of sialic acid from the carbohydrate moiety of the hemagglutinin of influenza B virus by sialidase is essential for the cleavage of this glycoprotein

by cellular protease [922]. Mucins, which are highly glycosylated and often heavily sialylated [5,13,659,906] are excellent anti-proteolytic substances of high viscosity, protecting endothelia particularly in the intestine or respiratory tract from the attack by proteases and pathogenic substances [5,8,245,660,853]. The high viscosity of mucins is due, to an extent, to their sialic acid and often also sulfate content, or, in lower animals, to the presence of other anionic components such as uronic acids, sulfate or phosphate residues [5,853,906,923]. This property enables mucins to act as very efficient biological lubricants, thus for example facilitating the transport of foodstuffs through the intestine, assisting in the movement of the eye bulbus, facilitating the mechanical processes of reproduction, or serving as protection for fish and many lower animals, the body surfaces of which must be lubricated for movement in aqueous or muddy environments. The structure, biosynthesis and molecular biology of mucins expressing cell- and tissue specificity have been reviewed [722,924].

The next two large groups of sialic acid functions are two-fold: sialic acids are either masks of biological recognition systems or they are directly recognized during the interaction of molecules and cells [663,925]. Since the masking functions of these sugars were recognized earlier, they will be described first.

10.2. Sialic acids masking biological recognition sites

It is well established that the main reason why sialic acids occur chiefly on membrane surfaces as well as on circulating and secreted molecules is because of the protective role these highly acidic and relatively bulky sugar molecules play. In this way, cells and molecules are protected against damaging influences from the environment, such as proteases and immunological or phagocytotic agents. This may be the reason why so many microorganisms such as trypanosomes (see section 9.2.4.1), bacteria (see below) and even pathogenic fungi [926–928] also developed mechanisms to sialylate cell surface components. Remarkably, in the fungi *Sporothrix schenkii* [926] and *Fonsecaea pedrosi* [928], both of which are pathogenic to man, Neu5Ac and Neu5Gc were found. The protective role of sialic acids was shown in the case of *S. schenkii*, whose phagocytosis by macrophages was increased after sialidase treatment [929].

In addition to this more general function, sialylation can mask specific biological recognition sites and, by reversible sialylation, can control biological events such as the action of hormones, ion channels, growth and ageing, some examples of which will be discussed in the following.

One of the most frequent biological recognition sites of glycoconjugates, which are often masked by sialic acids, are galactose residues. These monosaccharides, after their exposure by sialidase or reduced sialyltransferase activity, interact with a variety of galactose-specific receptors and are thus involved in many physiological and pathological events (for reviews see refs. [5,8,663,925,930–934]). The galactose-specific lectin of animals was found to be a phylogenetically very old protein, with homologues occurring in organisms ranging from the sponge *Geodia cydonium* to man [935,936]. One of the best known galactose-specific receptors is the hepatic asialoglycoprotein receptor, which was the first well-characterized animal lectin [933,937]. The properties, molecular biology and biosynthesis of this Ca^{2+} -dependent receptor have been intensively studied (e.g.

ref. [938]). It is proposed to function as a "vacuum cleaner" for the removal of serum glycoconjugates, which had become "non-self" and have exposed galactose residues, resulting from the loss of sialic acids. However, this lectin also mediates the hepatic binding and uptake of hepatitis B virus particles by the same mechanism [939]. This clearance of asialoglycoconjugates is also a problem during the therapeutic administration of recombinant glycoproteins, including those with hormonal or enzymatic activity, or the follicle-stimulating hormone isolated from urine [940–945]. Only the fully sialylated glycoproteins exhibit a long life in serum and thus extended biological activity, in contrast to those possessing terminal mannose or galactose residues. In the latter cases, rapid removal mediated by carbohydrate-specific receptors takes place.

In contrast to this positive effect of sialic acids on the circulation time and the biological activity of these glycoprotein hormones, the same sugar negatively influences the hormonal effect of e.g. thyrotropin [940], erythropoietin [944] and prolactin [946] in vitro. It was shown in these studies that asialo-erythropoietin binds better to the hormone receptor than the sialylated form. Deglycosylated human lutropin and chorionic gonadotropin also bind better to the receptors than the native form, but their hormonal activities, such as cAMP formation and steroidogenesis are lost [947]. The presence of sialic acid residues is a functional requirement of the hormone molecules and may modulate the efficiency of signal transduction. It was shown in these studies that a sialic acid residue from a chorionic gonadotropin glycan either binds to the hormone receptor directly or to a neighboring lectin-like molecule. This interaction is essential for signal transduction mediated by the hormone. A comparable receptor which is responsible for the sequestration of sialic-acid-depleted blood cells, such as erythrocytes [948], lymphocytes [949] and thrombocytes [43] exists on the surface of liver and spleen macrophages, and also on the peritoneal phagocytes of some animals [925,950,951]. This receptor was isolated from rat peritoneal macrophages [952]. It mediates binding of desialylated blood cells via demasked galactose and N-acetylgalactosamine residues by macrophages and requires Ca²⁺ for this function. Immunoglobulin or complement factors do not influence this process. Although sialidase-treated erythrocytes and thrombocytes are phagocytozed after adhesion [43,948], lymphocytes are released from the macrophages within 24 h after binding [949], probably due to the resynthesis of their sialic acid moieties. The binding of the three cell types to phagocytes is shown in Fig. 22.

This lectinophagocytosis not only acts on blood cells damaged by sialidase, which may have resulted from viral or bacterial infections (see above), but also seems to be responsible for the removal of aged human erythrocytes [953]. It was shown with lectins and by resialylation with 9-fluoresceinyl-Neu5Ac that old blood cells expose more terminal galactose residues compared with young erythrocytes. Loss of sialic acid residues from glycophorin during the life-time of human erythrocytes seems to be responsible for the clearance of senescent red blood cells from circulation [954]. Injection of sialidase into mice induces lectin-mediated thrombocytopenia, which stimulates thrombopoiesis [955].

As a consequence of programmed cell death ("apoptosis"), changes of surface carbohydrates including a loss of sialic acids and exposure of galactose residues occurs leading to phagocytosis of apoptotic cells [956]. In neonatal rat liver cell cultures, the asialoglycoprotein receptor has been implicated in the ingestion by healthy hepatocytes



Fig. 22. Binding of sialidase-treated rat erythrocytes, lymphocytes and thrombocytes to homologous peritoneal macrophages by the galactose-specific receptor. (a) Scanning electron microscopy of erythrocyte-macrophage interaction (from ref. [925] by permission of Kodansha Publishers, Tokyo). After prolonged incubation, the erythrocytes were ingested. (b) Scanning electron microscopy of cultured, sialidase-treated lymphocytes (L) bound by two macrophages (M). Lymphocytes were not phagocytozed but separated from the macrophages within one day [949]. (c) Scanning electron microscopy of sialidase-treated thrombocytes to a macrophage adherent to a Petri dish. From ref. [925] by permission of Kodansha Publishers, Tokyo.



Fig. 22c.

of neighboring cells undergoing apoptosis [957]. After desialylation of rooster sperm glycocalyx, the fertility of the spermatozoa was reduced due to an increased rate of sequestration in the hen's reproductive tract when compared with untreated cells [958]. This effect is believed to be due to demasking of antigenic or other recognition sites on the spermatozoa. However, treatment of human motile spermatozoa with sialidase results in better attachment of these cells to the zona pellucida of oocytes [959]. It is concluded from this observation that the release of sialic acid from the sperm plasma membrane could be one of the capacitation events necessary for unmasking certain sperm surface antigens or carbohydrate ligands implicated in zona pellucida recognition. The surface of human spermatozoa is highly sialylated, containing predominantly Neu5Ac(α 2–6)Gal/GalNAc-glycoconjugate sequences [959,960]. It was also shown with a sialylated wild-type and a mutant *Streptococcus* strain that sialic acids hinder phagocytosis of the bacteria and thus increase their virulence [915].

The galactose receptor also enables rat peritoneal macrophages to take up either free [961] or gold-particle-bound asialoglycoproteins [925,950,962]. Interestingly, for activity this receptor requires the presence of sialic acids on the macrophages to which it is bound, as was shown by sialidase treatment [925]. No explanation for this phenomenon is available so far. It may be due to the requirement of sialic acid for stabilization of the conformation of the receptor, to prevent self-aggregation, if it is a glycoprotein, or to hinder interaction with neighboring desialylated molecules in the macrophage membrane. The interaction of galactose-exposing particles with rat Kupffer cells has also been characterized [963].

From the reversible binding of lymphocytes mentioned above, as well as from the inhibition of binding of erythrocytes by enzymatic resialylation [964], a model has been proposed for the reversible binding of cells via sialic acid and galactose, which is



Fig. 23. Model showing how sialidase and sialyltransferase may regulate cellular interactions, i.e. association and dissociation mediated by a galactose-recognizing system. Symbols: solid square, galactose; open oval, sialic acid; open castle, galactose-recognizing receptor; dashed sausage, hypothetical recognition site (cluster of three galactose residues) by the lectin. From ref. [965] by permission of Gustav Fischer Verlag, Jena.

regulated by the activity of sialidase and sialyltransferase [925,950] (Fig. 23). This may generally operate in cell communication, especially in growth, differentiation, ageing, malignant transformation and metastasis.

The protection of host cells by sialic acids from the interaction resulting in colonization and possibly infection with bacteria is illustrated by the adhesion of *Bacteroides intermedius* to sialidase-treated human buccal epithelial cells and erythrocytes [966]. A corresponding phenomenon was described above in the sialidase section 9.2.4 for another oral bacterium, *Actinomyces viscosus*, which is also capable, by sialidase secretion, to demask its own attachment sites on host cells. *Pseudomonas aeruginosa*, a dangerous, infectious agent in patients suffering from cystic fibrosis or other diseases, employs pili to adhere to asialo-GM1 [967,968]. As sialic acid interferes with binding, GM1 itself is not a ligand for this bacterium.

Sialic acid residues on cell surfaces are important as modulators of immune and complement reactivity. Most importantly, animal cells and some microorganisms are shielded from the influence of these defence systems, since sialic acids can mask corresponding recognition sites. This has many physiological and pathological implications, as it may prevent autoimmune diseases, but can also increase the virulence of microbial infections or of cancer. A classical example for this protective effect is the interaction of the non-immune channel fish *Ictalurus punctatus* [969] with various Gram-negative bacteria. Very little bactericidal activity is exerted by the alternative complement pathway against the fish pathogens containing sialic acid, in contrast to a very strong bactericidal response against the non-pathogens lacking sialic acid. To show that only the sialic acid coat protected the bacteria against the complement system and thus rendered them virulent, the bacteria were treated with sialidase. This destroyed their virulence, due to an enhancement of the bactericidal response of the fish.

The virulence of Gonococci (*Neisseria gonorrhea*) is also much increased by sialylation of the bacteria [184,185,970]. Terminal Gal(β 1–4)GlcNAc structures of the bacterial 4.5 kDa lipopolysaccharide are sialylated during the life-time of the bacteria in the host organism. In this way, Gonococci in patients become resistant to complement-mediated killing by serum and to phagocytosis [971]. This resistance is lost in subcultures of the bacteria, but can be acquired again in the host. It is also absent in mutants lacking the necessary lipopolysaccharides. However, after infection with such Gonococci, variants appeared in the patient again expressing the acceptor for sialic acids, which shows a fascinating capability of the bacteria to adapt to the life in the host [970]. Closer investigation has revealed that sialic acids also mask porin epitopes on the Gonococci, which results in the reduction of the binding and bactericidal activity of antibodies recognizing this peptide moiety [972]. Furthermore, binding to human neutrophils and killing of these bacteria is prevented after sialylation [973].

Curiously, this chemical modification of Gonococci living in the host is possible by a sialyltransferase expressed by the bacterium itself and the sialic acid donor substrate, CMP-Neu5Ac, produced by the host. The properties of the sialyltransferase extracted from the bacteria have been reported [974]. It would be very interesting to compare the primary structure of this enzyme with those of host sialyltransferases. This mechanism is different from that used by trypanosomes for sialylation, although in both cases sialic acids are acquired from the host.

One of the major functions of sialic acids on mammalian cells is their masking effect on immunological recognition sites, as has been reviewed [5,8,663,795]. Loss of sialic acids, which may occur by exogenous sialidases from infections by microorganisms and viruses (see section 9.2.4), can thus lead to autoimmune diseases. One example is glomerulonephritis of human kidney, probably resulting from streptococcal infections. We are only at the beginning to understand such pathobiochemical mechanisms. It was also shown that cultured epithelial cells from guinea pig after treatment with bacterial sialidase bind more IgM than the controls [795,975]. These cells and those from lung epithelium taken from influenza-virus-infected animals were found to be hyposialylated. In serum, the level of IgG autoantibodies binding to the sialidase-treated cells increased [975]. It is assumed that these autoantibodies participate in cellular disfunctions and modify bronchoreactivity occurring during the infection of the respiratory tract. Asthma and chronic pulmonary disease may be the consequence of such desialylation. In these processes, the parasympathetic tone of the airways and the affinity of muscarinic agonists is impaired and the interaction of acetylcholine and histamine with their receptors also seems to be disturbed [795,976].

10.3. Sialic acids representing biological recognition sites

Whereas for a long time the main function of terminal sialic acid residues was considered as a mask for recognition sites on cell surfaces, e.g. galactose residues or antigens as outlined above, it is only in the last few years that receptors have been described which connect the structural diversity of sialylated glycoconjugates to specific functions in cellular interactions.

10.3.1. Sialic acid receptors of microorganisms, plants and lower animals

Many microorganisms, plants and animals express proteins which bind to sialic acids as components of glycoconjugates. Such sialic-acid-recognizing proteins or lectins from microorganisms, plants and invertebrates (mainly snails, scorpions and crabs) have been summarized in Tables 20 and 21, and have been discussed elsewhere in this series. They show more or less specific binding to different sialic acids or even specific sialic acid linkages and oligosaccharide sequences, as reviewed [709]. A careful investigation of the binding determinants of the sialic-acid-specific lectin from e.g. the slug Limax flavus has revealed that the α -anomer, the N-acetyl group, and an intact side chain of sialic acid are necessary for high affinity interaction. An axial position of the hydroxyl group at C4, hydroxylation of the N-acetyl group or substitution by a formyl residue, as well as 9-O-acetylation or periodate cleavage of the glycerol chain of Neu5Ac considerably reduced binding to the lectin [224]. Since these organisms do not express sialic acids themselves, it is unlikely that these lectins play a role as sialic-acid-binding proteins in cell recognition. However, they may be involved in defence against sialic-acid-containing microorganisms and animals. Some of the sialic-acid-binding lectins have proved to be useful tools in the analysis and histochemistry of glycoconjugates. Most frequently used are the agglutinins from wheat germ (WGA), Limax flavus (LFA), Sambucus nigra (SNA) and Maackia amurensis (MAA) (see section 4). Initial insight into the putative sialic-acidbinding sites of these unique lectins can be gained from a comparison of the primary structure of MAA with other plant lectins [1001]. A conserved asparagine residue was found to be exchanged by aspartate (Asp-135), which is proposed to be involved in the interaction between MAA and sialo-oligosaccharides.

It has long been known that pathogens, i.e. viruses, mycoplasma, bacteria and protozoa, take advantage of cell surface sialic acids to adhere to their respective host cells (for summaries see refs. [660,1002,1003] and chapter 13 of the present volume). Of greatest importance in this connection is the role of sialic acids in the attachment to and infection of mammalian cells by influenza viruses (see section 9.2.4.3).

From 500 MHz ¹H NMR spectroscopic [1004] and from X-ray crystallographic studies [1005,1006], models for the position of sialic acid in the binding pocket and the interaction with certain amino acids of influenza A virus hemagglutinins were delineated (Fig. 24) [394]. The use of sialic acid analogues modified in the glycerol side chain and in the N-acetyl and in the carboxyl groups of Neu5Ac, gave support to these models and revealed that the hydroxyl group at C9 does not interact with the hemagglutinin [394]. The N-acetyl moiety is critical for the interaction of sialic acids via hydrophobic bonds. These studies were extended by investigation of the crystal structures of this influenza virus hemagglutinin with different high-affinity analogues

Pathogen	Specificity	Ref(s).
Viruses		
Influenza A and B	Neu5Ac (some strains prefer Neu5Ac(α 2–3)Gal or Neu5Ac(α 2–6)Gal, dependent on host specificity)	[977]
Influenza C	Neu5,9Ac ₂	[978]
Corona virus	Neu5,9Ac ₂	[979]
Sendai virus	Neu5Ac	[980]
Polyoma virus	Neu5Ac(α 2–3)Gal(β 1–3)GalNAc	[981]
Rotavirus group C	Neu5Ac	[982]
Mycoplasma		
Mycoplasma pneumoniae	Neu5Ac(α 2–3)Gal on polylactosamine chains	[983]
Bacteria		
Streptococcus sanguis	O-linked sialylated tetrasaccharides	[984]
Escherichia coli K99	Neu5Gc-containing glycolipids	[667,985]
<i>Escherichia coli</i> , S-fimbriae (newborn human meningitis)	Neu5Ac(α2-3)Gal(β1-3)GalNAc Neu5Gc-GM3, GD3, GD1b	[986] [987–990]
Bordetella bronchiseptica	Neu5Ac	[991]
Pseudomonas aeruginosa	Neu5Ac	[992]
Helicobacter (Campylobacter) pylori	Neu5Ac(α 2-3)Lactose > Neu5Ac(α 2-6)Lactose	[993]
Streptococcus suis	Neu5Ac(α 2-3)Gal(β 1-4)GlcNAc(β 1-3)Gal	[994]
Protozoa		
<i>Plasmodium falciparum</i> (Malaria, MSA-1)	Neu5Ac	[995]
Trypanosoma cruzi (Chagas disease)	Neu5Ac	[831]
Tritrichomonas species	Neu5Ac (α -2,3- and α -2,6-linkages)	[226,996]
Toxins		
Vibrio cholerae toxin	GM1	[997]
Petussis toxin	Neu5Ac	[998,999]
Tetanus toxin	sialoglycolipids	[1000]

Table 20 Pathogenic microorganisms and toxins binding to sialic acids on host cells

of sialic acid in order to design possibly therapeutic inhibitors of viral attachment to host cells, which may prevent membrane fusion and circumvent evasion of inhibition by antigenic variation of the viruses [1006]. A series of poly(acrylic acid-co-acrylamides) and dextrans bearing pendant glycine-4-amidobenzyl α -Neu5Ac groups were synthesized for anti-influenza chemotherapy. Some of these compounds suppress virus replication in embryonated eggs [1007]. X-ray crystallographic studies of influenza virus sialidase– antibody complexes gave interesting insight into the structure of such complexes between two cross-reacting antibodies both recognizing sialidase [1008].

Source	Specificity		
A: Plants			
Wheat germ Triticum vulgare	Neu5Ac < GlcNAc		
Elderberry Sambucus nigra	Neu5Ac(α 2–6)Gal/GalNAc		
Maackia amurensis	Neu5Ac(α 2–3)Gal(β 1–4)GlcNAc		
B: Invertebrates			
Snail Dolabella	Neu5Ac		
Slug Limax flavus	Neu5Ac > Neu5Gc		
Snail Cepaea hortensis	Neu5Ac > Neu5Gc		
Snail Achatina fulica	Neu5Ac(α 2–3)Gal > Neu5Ac(α 2–6)Gal		
Snail <i>Pila globosa</i>	Neu5Gc		
Oyster Crassostrea gigas	Neu5Ac		
Horseshoe crab Limulus polyphemus	Neu5Ac		
Lobster Homarus americanus	Neu5Ac, Neu5Gc		
Horseshoe crab Tachypleus tridentatus	Neu5Ac, Neu5Gc		
Scorpion Androctonus australis	sialyllactose		
Horseshoe crab Carcinoscorpius rotunda	Neu5Ac(α 2-6)Gal > Neu5Ac(α 2-3)Gal		
Scorpion Centruroides sculpturatus	Neu5Ac, Neu5Gc		
Prawn Macrobrachium rosenbergii	Neu5Ac		
Scorpion Masticoproctus giganteus	Neu5Ac		
Spider Aphonopelma cepaeahortensis	sialoglycoproteins		
Scorpion Heterometrus granulomanus	Neu5Ac(α 2–3)Lactose		
Prawn Peneaus monodon	Neu5Ac		
Scorpion Paruroctonus mesaenis	sialoglycoproteins		

 Table 21

 Sialic-acid-binding lectins from plants and invertebrates (see ref. [709] and references therein)

The interest in sialic-acid-specific adhesion of bacteria is increasing, since it often is a critical step in infectious diseases. Examples are the inflammation of gastric mucosa by *Helicobacter pylori* after adhesion to sialoglycoproteins of the cell surface [993,1009] and meningitis of infants as well as urinary tract infections by *E. coli* [985–989]. The binding of the latter bacteria via their S fimbriae to epithelial, e.g. buccal cells can be inhibited by sialylated mucins from human milk. Correspondingly, in human newborn babies sialylated substances have an anti-infective effect. Colostrum was shown to be more effective in this respect than mature milk [1010]. The S fimbrial adhesin of *E. coli* represents a protein complex and the genes encoding its subunits have been sequenced [988]. The introduction of specific mutations into the subunit gene *sfa*S revealed that a region of six amino acids of the adhesin which includes two lysine and one arginine residues is involved in the interaction of S fimbriae with sialic acid [990].

It should also be mentioned that several bacterial toxins are known which bind to gangliosides in a sialic-acid-dependent manner, e.g. cholera, pertussis and tetanus toxins



Fig. 24. Model for the position of sialic acid in the binding pocket of the hemagglutinin of influenza virus, based on X-ray crystallographic studies. Some of the hydrogen bonds proposed in this model are shown by dashed lines. From ref. [394] by permission of European Journal of Biochemistry, Zürich.

(ref. [660] and Table 20). The role of host sialic acids in the binding of protozoa is also of growing interest, and the possible involvement of this monosaccharide in the attachment of trypanosomes to host cells during infection has already been mentioned (section 9.2.4). The role of sialic acid in the invasion of erythrocytes by malaria parasites has been demonstrated and a 175 kDa protein of *Plasmodium falciparum* has been identified as an erythrocyte-binding protein with sialic acid specificity [1011]. It preferentially binds to Neu5Ac(α 2–3)Gal sequences on the O-linked tetrasaccharide of glycophorin. Tritrichomonads (*Tritrichomonas suis, foetus* and *mobiliensis*) also exhibit a sialic-acid-dependent adhesion to host cells and mucus [226,996]. A corresponding lectin consisting of three subunits was isolated from *Tritrichomonas mobiliensis*, which recognizes free as well as α -2,3- and α -2,6-linked Neu5Ac [226]. A model of binding was delineated from the observation showing that optimal interaction with the receptor requires a free

carboxylic group, the *N*-acetyl moiety and a free HO8 group of sialic acid. Remarkably, *O*-acetylation at C9 or C4 does not influence adhesion.

10.3.2. Sialic acid receptors of vertebrates

Sialic-acid-recognizing receptors have been discovered in a variety of vertebrates, mostly mammalian tissues, summarized in Tables 22–24. This line of research was much stimulated by the discovery of a family of cell adhesion molecules now generally called selectins [1033]. Due to its great importance, this topic has been reviewed frequently [660,828,1034–1039]. It will, therefore, not be discussed in detail here.

The selectin family consists of three members, L-selectin, found on lymphocytes and other white blood cells, E-selectin expressed on endothelial cells activated by cytokines, and P-selectin occurring on activated blood platelets and endothelia (Table 23). They possess a similar primary structure, representing type I trans-membrane glycoproteins containing an amino-terminal carbohydrate recognition domain, a single epidermal growth-factor-like domain, a variable number of short consensus repeats and a relatively short carboxy-terminal cytoplasmic domain. The selectins also share common aspects in their function. They all play crucial roles in the initial event of white blood cell adhesion to specific endothelia, so-called rolling. Before firm adhesion, cells floating in the blood stream begin to slow down by rolling along the endothelial lining of the vessel. This is mediated by selectins interacting with sialic-acid-containing ligands. The specificity of this interaction is accomplished by the expression pattern of the receptors and their appropriate ligands. The selectins bind with low affinity to many small, sialylated, fucosylated and in some cases sulfated carbohydrates, of which the prototypes are the tetrasaccharides sialyl-Lewis^x { sLe^x ; Neu5Ac($\alpha 2-3$)Gal($\beta 1-4$)[Fuc($\alpha 1-3$)]GlcNAc β -R} and the isomeric sialyl-Lewis^a { sLe^{a} ;Neu5Ac($\alpha 2-3$)Gal($\beta 1-3$)[Fuc($\alpha 1-4$)]GlcNAc β -R}. At least two of the selectins (Table 23) recognize sLe^x and/or sLe^a, which are expressed at high levels on leukocytes [1040] and some tumor cells, particularly of the colon [1041-1044] as components of glycoproteins or gangliosides. Sialyl-Lewis^x has also been shown to be a marker of dysplasia in the colonic adenoma-carcinoma sequence [1045]. It has therefore been proposed that selectins are involved in malignant growth and metastatic events. Remarkably, 6'-sulfated sLe^x (Table 23) has been identified as a major ligand (GlyCAM-1) involved in the initial attachment of lymphocytes via L-selectin to high endothelial venules of lymph nodes [1046]. The synthesis of this "addressin", as L-selectin ligands are called now, O-sulfated at C6 of galactose, has been reported [1047]. Using an L-selectin chimera, it could be shown that ligands for L-selectin also occur on the myelin sheaths of neurons of the central nervous system [1048]. Progress has been made in understanding how these molecules function at the atomic level, by determination of the three-dimensional structure of a portion of the E-selectin molecule [1049,1050]. A spin label study has shown that the interaction of selectin on lymphocytes with sialic acid severely restricts the rotational mobility of the cell surface proteins and lipids. Additionally, the cytoplasmic viscosity increases appreciably [1051].

Presently, much interest is focussed on the role of the selectins during inflammation and in transplantation medicine, since it was recognized that selectins are involved in the rolling and adhesion of granulocytes to endothelia under the influence of inflammatory cytokines [1034,1037,1039,1052]. This is a multistep process in which other adhesion

Source	Specificity	Reference(s)
Frog egg	sialylated glycoproteins	[1012]
Rat uterus	Neu5Ac	[1013]
Rat brain	Neu5Ac, Neu5Gc	[1014]
Rat brain myelin	gangliosides, preferentially GT1b, GQ1b, GD1b	[1015]
Human endometrium	Neu5Ac, Neu5Gc	[1016]
Human placenta (IgG)	O-acetylated sialic acids	[1017]
Blood (factor H of alternate complement pathway)	sialylated glycoconjugates, other polyanionic molecules	[1018]
Murine macrophages (sialoadhesin)	Neu5Ac(α 2–3)Gal(β 1–3)GalNAc on glycoproteins and glycolipids	[1019]
Bovine heart (calcyclin)	Neu5Ac, Neu5Gc	[1020]
Human placenta (sarcolectin)	Neu5Ac, Neu5Gc	[1021]
B lymphocytes (CD22)	Neu5Ac(α 2–6)Gal(β 1–4)GlcNAc	[1022,1023]

Table 22 Vertebrate sialic-acid-binding proteins^a

^a For selectins and sialoadhesins see Tables 23 and 24.

Table 23 Distribution and binding specificity of selectins (for references see the text)					
Selectin	Cell type	Ligand determinant			
E-selectin	activated endothelia	sialyl-Le ^x , Neu5Ac(α 2-3)Gal(β 1-4)[Fuc(α 1-3)]GlcNAc sialyl-Le ^a , Neu5Ac(α 2-3)Gal(β 1-3)[Fuc(α 1-4)]GlcNAc			
L-selectin	leukocytes	sialylated, sulfated and fucosylated O-glycans like Neu5Ac(α 2-3)(SO ₄ -6)Gal(β 1-4)[Fuc(α 1-3)]GlcNAc			
P-selectin	activated platelets and endothelia	sialyl-Le ^x and sialyl-Le ^a			

molecules of the integrin superfamily [1053] are also involved. Such reactions may occur under the influence of microbial infections [1054] or other inflammatory events, such as in skin and neuronal tissue [1055] or in lung injury [1056]. The application of sLe^x oligosaccharide has thus been shown to have a protective effect in such inflammatory reactions, e.g. lung inflammation due to oxygen radical formation by cobra venom. From this research, a therapeutic effect of sLe^x analogues in the treatment of many diseases including reperfusion injury observed after organ transplantation, heart attack and stroke is expected, as shown in animal models (ref. [1057] and J.C. Paulson, personal communication). Consequently, much work is dedicated to the synthesis of selectin ligands on a commercial scale (ref. [1058] and section 6.3). Another strategy to prevent

Statoadnesins, a family of immunoglobulin-like adnesion molecules binding to statylated glycans								
Adhesion molecule	Occurrence	Number of Ig-domains	Ligand determinant	Target cells	References			
Sialoadhesin	macrophage subpopulations	17	Sia(α2–3)Gal(β1–3)GalNAc Sia(α2–3)Gal(β1–3/4)GlcNAc	myeloid cells	[1019] [664,1024,1025]			
Myelin-associated glycoprotein (MAG)	myelin of oligodendrocytes and Schwann cells	5	Sia($\alpha 2$ -3)Gal($\beta 1$ -3)GalNAc	neurons, oligodendrocytes	[664,1026]			
CD22	B cells	7	Sia(α 26)Gal(β 14)GlcNAc	lymphocytes	[1023,1027–1030]			
CD33	myelomonocytic cells	2	Sia(α2–3)Gal(β1–3)GalNAc Sia(α2–3)Gal(β1–3/4)GlcNAc	myelomonocytic cells	[1031] [1032]			

 Table 24

 Sialoadhesins, a family of immunoglobulin-like adhesion molecules binding to sialylated glycans

selectin-mediated adhesion phenomena in pathological states is the neutralization of the biological activity of these receptors by antibodies [1059,1060].

Other well defined sialic-acid-dependent adhesion receptors are sialoadhesin [660,664, 1019,1024] found on specific subsets of macrophages in bone marrow and lymphatic tissues, such as lymph nodes and spleen, as well as CD22 [660,1025,1061,1062], a B-cell-specific protein, CD33, occurring on myelomonocytic cells [1031,1032] and MAG, the myelin-associated glycoprotein of neuronal tissues [1025,1063] (Table 24). All these receptors belong to the immunoglobulin superfamily [1025]. The murine macrophage sialoadhesin has a molecular mass of 185 kDa as well as 17 immunoglobulin-like domains [1024] and binds to the sequence Neu5Ac($\alpha 2$ -3)Gal($\beta 1$ -3)GalNAc or Neu5Ac($\alpha 2$ -3)Gal($\beta 1$ -3/4)GlcNAc, i.e. both N- and O-glycans on glycoproteins and glycolipids of cell surfaces, preferentially of the granulocytic lineage [1064]. Remarkably, Neu5Gc and Neu5,9Ac₂ are not recognized by the receptor [664]. With regard to its function, it may play a role in hematopoiesis, since it was found to be enriched on bone marrow macrophages at the contact sites with developing myeloid cells. A role in the trafficking of leucocytes in lymphatic organs is also assumed. The receptor preferentially binds to inflammatory and circulating neutrophils.

CD22 occurs on B cells, where it mediates their binding to T cells and to neutrophils, monocytes and erythrocytes [1065]. The interaction with T cells is assumed to be involved in early B cell activation and may also modulate signalling through the surface IgMcell receptor complex. In contrast to sialoadhesin, CD22 binds specifically to sialy $(\alpha 2 - \alpha)$ 6)Gal(\beta1-4)GlcNAc structures of N-glycans [1023,1025]. Again, Neu5Gc and Neu5,9Ac₂ modulate binding of this ligand. Whereas Neu5Gc increases binding to murine CD22, 9-O-acetylation of Neu5Ac is inhibitory [664]. It can only be assumed so far that these sialic acid modifications, both occurring in mice, have a biological role in the activity of the two receptors. Recognition of Neu5Ac by human CD22 is also masked by 9-O-acetylation [235]. A further regulation of CD22 function is α -2,6-sialylation of the glycan chains of the receptor molecule itself [1066], which inhibits receptor activity. In this way, the ligand binding sites of CD22 may be blocked and the interaction of the receptor with ligands on adjacent cells prevented. In resting B cells, which only express low levels of α -2,6-sialyltransferase activity. CD22 binding sites may be free, thus enabling interaction with neighboring cells. Lymphocyte activation, however, results in an increase in the activity of this sialyltransferase, causing sialylation of the receptor and a decrease in cellular interaction. Regulation of the sialylation of CD22 may therefore mediate adhesion and trafficking at discrete stages of B cell differentiation. In this regulatory system the expression of sialylated ligands in non-lymphoid cells may be important. It was shown that inflammatory cytokines stimulate the expression of both ligands for CD22 and β -galactoside α -2,6-sialyltransferase in endothelial cells of human umbilical vein [1028,1029]. Thus, CD22 could direct interactions between mature B cells and endothelial cells during inflammatory stages. These events can be influenced, i.e. lymphocyte binding inhibited, by α -2,6-sialylated glycoproteins of the blood serum. Interestingly, the two serum molecules much involved in immune and inflammatory responses, haptoglobin and IgM, are selectively recognized by lymphocyte CD22 [1029]. Using a soluble CD22-immunoglobulin fusion product or expression of CD22 on the surface of Chinese hamster ovary cells, it was shown that this lectin has a higher 368

apparent affinity for multiply sialylated substances over monosialylated, unbranched glycans [1030]. This observation provides a mechanism for strong CD22-dependent cell adhesion despite the relatively low K_d for protein–carbohydrate binding.

The myelin-associated glycoprotein (MAG), involved in the proper myelination of axons [1063], has recently been shown to be a sialic-acid-dependent receptor, recognizing α -2,3-glycosidically bound Neu5Ac, preferably on O-glycans [1025]. It shares sequence similarity with both sialoadhesin and CD22, showing that it belongs to the same immunoglobulin superfamily. The term "sialoadhesin family" was proposed for this group of related sialic-acid-binding proteins [1025]. Since MAG can activate the MAG-associated tyrosine kinase [1067], it is possible that sialylated glycoconjugate ligands play a role in signal transduction during myelination. Furthermore, myelin-associated sialidase adhering to the ganglioside GM1 may be involved in the formation and stabilization of the multilamellar structure of the myelin sheath [829].

Other sialic-acid-binding activities have been found in mammalian tissues (Table 22), these being reviewed in ref. [660] and in chapter 14 of the present volume. Recently, a sialic-acid-binding protein was purified from human endometrium, which can bind to human spermatozoa, unless they have been sialidase-treated [1016]. Glycophorin A of mouse erythrocytes is recognized by a homologous peritoneal macrophage receptor specific for sialylated carbohydrates [1068]. It should, furthermore, be mentioned that various antibodies have been described, which recognize epitopes containing mono- or oligomeric sialic acids (glycoproteins, gangliosides and bacterial capsular polysaccharides) on a variety of cells (refs. [289,709,1011,1069-1071], and various chapters of this series). Monoclonal antibodies specific for α -2,8-linked Kdn sequences suitable for histochemistry are available now [1072]. The specificity of this interaction is higher than is generally the case for lectins, since the total carbohydrate structures bearing the sialic acids have a greater influence on the binding of antibodies. Interest is, therefore, increasing in the expression of sialyltransferase activity, this being a critical regulatory step in the formation of cell-surface differentiation antigens, which may be necessary for the function of these cells, so far most dramatically shown in lymphocytes [1070,1073,1074]. In the immune-reactivity of sialic acid epitopes, O-acetylation and N-acetyl hydroxylation have a strong modulatory role [668,669,1075].

10.4. Do sialic acids have "specific" functions?

The question arises as to whether these monosaccharides indeed have specific functions. Certainly, sialic acids are involved in numerous specific biological events, and many of them have been discussed in this and other articles. However, on closer inspection, it seems that sialylation exerts its influence in the diverse biological systems by the factors discussed, i.e. charge, conformation, masking and interaction with receptors. With regard to conformation, sialic acids may contribute to the specific structure of a given glycoconjugate molecule and thus may influence its biological activity. The highest degree of specificity can be observed in sialic-acid–receptor interactions, which require structurally well-defined features on both sites, not only related to the sialyl group but also to the penultimate sugars, thus resembling the formation of an enzyme–substrate complex. The interaction of influenza C virus with 9-O-acetylated sialic acid may be such an

example. The specificity in these phenomena can be related mostly to the non-sialylated part of the molecule or the cell, for example to the biological specificity of a glycoprotein hormone, a bacterial toxin, a neuronal reaction, the activity of an ion channel, or the nature of the receptor and the kind of signal it transfers into the cell after ligand binding. In these systems, the sialic acid moiety plays an essential role, since its presence is a prerequisite for the receipt and transfer of biological messages. Furthermore, sialic acid can control and modulate these biological events. The exposed position of this highly negatively charged, hydrophilic molecule on extracellular glycoconjugates makes it suitable for this purpose.

The physiological role of sialic acids could therefore be grouped on the basis of the long list of biological phenomena influenced by these carbohydrates. This will not be made here, since another grouping, into three main topics, was done in sections 10.1-10.3. However, a few interesting and unusual systems, influenced by sialic acids, as well as open questions for future research, will be mentioned.

In reproductive biology, sialic acid polymers are receiving more and more attention. Polysialic acid with $\alpha(2-8)$ -linkages was found to be expressed on mouse embryos before and after implantation and the neuronal cell adhesion molecule (NCAM), which bears a polysialic acid chain, seems to be involved in cellular interactions in the early mammalian embryo [101,1076]. A similar function is attributed to polysialic acid observed during a short period of the larval stage of *Drosophila melanogaster* [1077]. It has to be noted that this is the first report on the occurrence of sialic acid in insects.

The participation of sialic acids in different linkages and grade of polymerization during differentiation of cells and tissues was excellently demonstrated in a histochemical study [1078]. It was also reported there that polysialic acids typical for developing tissues are expressed in some malignant tumors. Sialic acid is involved in changes of the glycoconjugate patterns during the development of murine molar tooth germs [1079]. Corresponding to the variable expression of sialic acids during ontogenesis, the expression of sialyltransferases varies with the state of development, e.g. during the maturation of human myeloid cells [1080] or of the epithelium of small intestine upon weaning [1081]. Sialic acid modification is also involved in differentiation. In several rat and cow organs, the expression of Neu5Gc was found to be developmentally regulated [307,674]. Similar observations have been made for the appearance of Neu5,9Ac₂ in chicken erythrocytes (ref. [712]; see also section 8.4.2).

Many studies have revealed that sialic acids are of significance in the growth and functioning of neuronal cells. This can be mainly attributed to gangliosides, the most important carriers of sialic acids in these tissues [1082,1083]. How these molecules function is the object of intensive research, however, they are mainly involved in transmembrane signalling and the regulation of mediators. Interestingly, the synthetic glycoconjugate sialyl cholesterol also promotes neurite outgrowth in a mouse neuroblastoma cell line [1084]. Since this substance is transported to the nucleus, its localization may play an as yet unknown key role in neuritogenesis. Polysialic acid of NCAM affects cellular interactions not only during the development of the nervous system and skeletal muscle but also during the development of the ventricular conduction system in embryonic hearts of man and various animals [1085]. In adult rat brain, polysialic acid was found to be associated with sodium channels, and was shown to be required for the conductivity of the channel and thus for the function of this nervous tissue [1086].

Interestingly, polysialic acid is a rather labile substance, which easily degrades spontaneously even under physiological conditions, due to an anomalously high pK value of the internal sialic acid residues [258]. This intramolecular self-cleavage may influence neuronal adhesion, embryogenesis and bacterial virulence.

10.5. Medical significance of sialic acids

Since sialic acids are involved in so many biological phenomena, it is easily understandable that disturbances in the metabolism of these sugars, either due to genetic errors or at the post-translational level, may impair the physiological functions of an organism and thus lead to disease. Such pathophysiological influences of sialic acids and sialidases as e.g. virulence factor of microbial diseases have been mentioned throughout this chapter. Four diseases in which metabolic errors of sialic acids have been clearly recognized will be briefly discussed: galactosialidosis, Salla disease, Alzheimer disease and malignant tumors. (Sialidosis and sialuria are described by Michalski in chapter 1b of *Glycoproteins and Disease*, Vol. 30 of this series).

Galactosialidosis, first described as sialidosis [1087], is a rare genetic disorder and occurs in at least three clinical phenotypes: an early infantile form where abnormalities (sialidase deficiency and an increased amount of sialo-oligosaccharides) can be detected in utero; a late infantile form characterized by skeletal abnormalities, macular cherry-red spot and mild mental retardation; and a juvenile-adult form with variable age of onset of skeletal abnormalities, corneal clouding, macular cherry-red spot, neurological manifestations and mental retardation [1088-1091]. All three forms are characterized by accumulation of sialo- and galactoglycoconjugates and the excretion of sialyloligosaccharides. The primary defect of this genetically well-studied disease is in the capability of a protective protein, a serine carboxypeptidase which has also deamidase activity (see chapter 1a by Jourdian of Glycoproteins and Disease, Vol. 30 of this series), to associate with β-galactosidase and sialidase to protect them from intralysosomal proteolysis. Fibroblasts from these patients show a marked reduction in β -galactosidase and virtually no sialidase activity. The genes encoding β -galactosidase, sialidase and the protective protein, are located on the human chromosomes 3, 10 and 20, respectively [1092,1093]. With regard to the function of the protease normally stabilizing the ternary enzyme complex, various mutations have been found which affect the stability of the carboxypeptidase/deamidase central sheet [1094]. The stoichiometry of this complex from human lysosomes has been described [1095].

Lysosomal accumulation of free sialic acid occurs in two phenotypically distinct inherited metabolic disorders, Salla disease and infantile sialic acid storage disease [1096]. Salla disease is an autosomal recessive lysosomal storage disorder and was first observed in patients of Finnish ancestry, but also occurs outside Finland. The clinical symptoms are a slow progressive psychomotor retardation, impaired speech, ataxia and a prolonged course. Sialic acid accumulates in the lysosomes due to a defective efflux into the cytosol. The genetic defect affects the function of the specific transport protein for sialic acid and other acidic monosaccharides in the lysosomal membrane [1097]. The Salla disease locus was found on the long arm of chromosome 6, and the critical region for this locus is in the range of 190 kb [1098].

Both in Alzheimer's disease and older Down's syndrome subjects, a decrease in serum sialyltransferase activity was observed [1099,1100]. This was found to affect only the α -2,3-sialyltransferase, also leading to a decrease of α -2,3-linked sialic acid in serum glycoproteins. It can only be assumed at present that this observation mirrors reduced activity of sialyltransferase and decreased sialoglycoconjugate biosynthesis in neuronal tissue subjected to these degenerative diseases. Serum sialyltransferase may thus be an early biochemical marker of neurodegeneration.

Cell surface carbohydrates are recognized to play crucial roles in malignant growth and metastasis, as is reported in detail in volume 30 of this series, "Glycoproteins and Disease". Here, some recent observations on the special influence of sialic acids in these processes will be presented to complete the main aspects of sialic acid biology. There are several reports in the literature showing that the amount and type of sialylation of membrane components influence the metastatic potential of tumor cells. A higher amount of membrane-bound sialic acid as component of glycoproteins and gangliosides promotes invasion and interferes with intercellular adhesion. Sialic acid is also involved in tumor cell attachment to endothelial cells via sLe^x- or sLe^a-binding selectins (section 10.3.2), and in adhesion of tumor cells to substances of the extracellular matrix [1044,1101]. These oligosaccharides are therefore considered as markers of some tumors, e.g. of human colon [1102] or thyroid gland carcinomas [1103]. In the latter case, clinicopathological studies indicate that sLe^a antigens are related to biologically aggressive thyroid tumors. In human colon, sLe^x was found to be expressed only in cultivated cells from premalignant polyposis, but not in intermediate or carcinoma cells [1102]. The higher degree of sialylation in human colorectal tumor tissue seems to be due mainly to an increase of α -2,6sialyltransferase activity sialylating N-glycan and N-acetyllactosamine sequences [1104-1106]. When studying the activity of several glycosyltransferases in human colorectal adenoma cells during progression to cancer, α -2,3-sialyltransferase activity exhibited a peak in the intermediate premalignant stage, while α -2,6-sialyltransferase appeared to be turned off in the final stage, i.e. the adenomacarcinoma cells [1102]. An increase of serum sialic acids was shown to be relevant for the diagnosis and prognosis of tumor diseases and leukemia [1101,1104,1107].

The role of sialic acids in the formation of metastases was also shown with nude mice and human colorectal cancer cell lines [1108]. The better differentiated cells had higher levels of sialyltransferase activity and sialic acids which correlated with their enhanced ability to form liver metastases compared to the poorly differentiated, less sialylated cell lines. The role of sialic acid in metastasis was further demonstrated by inhibition of enzymatic sialyltransfer with a CMP-sialic acid derivative (refs. [648,649] and section 8.3), which resulted in reduced liver metastasis. An increase of α -2,6-sialyltransferase activity was also observed in cultured Ehrlich ascites tumor cells [1109] and in human gastric epithelium carcinoma [1110]. The increase of sialylation of an antigenic epitope of the latter tumor cells enhanced their metastatic potential. The expression of *ras* oncogenes in NIH 3T3 fibroblasts increases the sialic acid density and concomitantly the invasiveness of these cells and reduces cell surface galactosylation [1111]. Correspondingly, α -1,3-galactosyltransferase activity was decreased and

those of α -2,3- and α -2,6-sialyltransferase activities were increased. Furthermore, the number and/or length of polylactosaminoglycan chains and the branching of N-glycans was also enhanced. This is another example, in which multiple changes in the expression of glycoconjugates occur on malignant transformation. It was shown with rat 3Y1 cell lines that oncogenesis can be accompanied not only by increased sialylation but simultaneously by decreased sialidase activity, thus supporting hypersialylation and metastatic potential of the cells [1112]. There was an inverse correlation between lysosomal-type sialidase activity and invasion. How the oncogenes mentioned operate in these processes is presently being studied. All these studies and experiments with murine lymphoid tumor cells [1113] suggest that changes either in the amount, the type or linkage of sialic acid in tumor cell glycoconjugates can affect tumor growth and metastasis.

11. Conclusion

In the field of the primary structural analysis and synthesis of sialic acids and sialic-acidcontaining glycans, the methods developed have reached a high level of sophistication. The conformational analysis of sialylated carbohydrate chains of glycoconjugates in their natural environment needs specific attention in order to better understand the physical and biological functions of these glycosylated molecules. Although deep insight into the metabolism of the different sialic acids has been obtained, it is challenging to further unravel the secrets of expression and transfer of these monosaccharides on the enzyme and gene levels, as well as receiving information on their regulation including hormonal control. This is particularly valid for the new group in the sialic acid family, Kdn and its derivatives. The biological roles of sialic acids are least understood, although solving these open questions seems to gain momentum. Thus, sialobiology, sialopathology and also sialopharmacology will remain most fascinating topics of future research on sialic acids.

References

- [1] Blix, G. (1936) Hoppe-Seyler's Z. Physiol. Chem. 240, 43-54.
- [2] Klenk, E. (1941) Hoppe-Seyler's Z. Physiol. Chem. 268, 50-58.
- [3] Tuppy, H. and Gottschalk, A. (1972) In: A. Gottschalk (Ed.), Glycoproteins Their Composition, Structure and Function, Part A. Elsevier, Amsterdam, pp. 403-449.
- [4] Schauer, R. (Ed.) (1982) Sialic acids Chemistry, Metabolism and Function. Springer, Wien.
- [5] Schauer, R. (1982) Adv. Carbohydr. Chem. Biochem. 40, 131-234.
- [6] Schauer, R. (1987) Methods Enzymol. 138, 132-161.
- [7] Schauer, R. (1987) Methods Enzymol. 138, 611-626.
- [8] Varki, A. (1992) Glycobiology 2, 25-40.
- [9] Troy, F.A. (1992) Glycobiology 2, 5-23.
- [10] Manzi, A.E. and Varki, A. (1993) In: M. Fukuda and A. Kobata (Eds.), Glycobiology A Practical Approach. IRL Press, Oxford, pp. 27–77.
- [11] Reuter, G. and Schauer, R. (1994) Methods Enzymol. 230, 168-199.
- [12] Reuter, G. and Schauer, R. (1994) Methods Carbohydr. Chem. 10, 29-39.
- [13] Corfield, A.P. and Schauer, R. (1982) In: R. Schauer (Ed.), Sialic acids Chemistry, Metabolism and Function. Springer, Wien, pp. 5-50.

- [14] Sjoberg, E.R., Chammas, R., Ozawa, H., Kawashima, I., Khoo, K.-H., Morris, H.R., Dell, A., Tai, T. and Varki, A. (1995) J. Biol. Chem. 270, 2921–2930.
- [15] Kamerling, J.P. and Vliegenthart, J.F.G. (1982) In: R. Schauer (Ed.), Sialic acids Chemistry, Metabolism and Function. Springer, Wien, pp. 95-125.
- [16] Vliegenthart, J.F.G., Dorland, L., van Halbeek, H. and Haverkamp, J. (1982) In: R. Schauer (Ed.), Sialic acids - Chemistry, Metabolism and Function. Springer, Wien, pp. 127–172.
- [17] Kamerling, J.P., Vliegenthart, J.F.G., Schauer, R., Strecker, G. and Montreuil, J. (1975) Eur. J. Biochem. 56, 253–258.
- [18] Haverkamp, J., van Halbeek, H., Dorland, L., Vliegenthart, J.F.G., Pfeil, R. and Schauer, R. (1982) Eur. J. Biochem. 122, 305–311.
- [19] Kamerling, J.P., Dorland, L., van Halbeek, H., Vliegenthart, J.F.G., Messer, M. and Schauer, R. (1982) Carbohydr. Res. 100, 331–340.
- [20] Damm, J.B.L., Voshol, H., Hård, K., Kamerling, J.P. and Vliegenthart, J.F.G. (1989) Eur. J. Biochem. 180, 101–110.
- Hanaoka, K., Pritchett, T.J., Takasaki, S., Kochibe, N., Sabesan, S., Paulson, J.C. and Kobata, A. (1989)
 J. Biol. Chem. 264, 9842–9849.
- [22] Reuter, G., Pfeil, R., Stoll, S., Schauer, R., Kamerling, J.P., Versluis, C. and Vliegenthart, J.F.G. (1983) Eur. J. Biochem. 134, 139–143.
- [23] Kamerling, J.P., Schauer, R., Shukla, A.K., van Halbeek, H. and Vliegenthart, J.F.G. (1987) Eur. J. Biochem. 162, 601–607.
- [24] Jennings, H.J. (1983) Adv. Carbohydr. Chem. Biochem. 41, 155-208.
- [25] Bergwerff, A.A., van Oostrum, J., Asselbergs, F.A.M., Bürgi, R., Hokke, C.H., Kamerling, J.P. and Vliegenthart, J.F.G. (1993) Eur. J. Biochem. 212, 639–656.
- [26] Kogan, G., Jann, B. and Jann, K. (1992) FEMS Microbiol. Lett. 91, 135-140.
- [27] Gamian, A., Romanowska, E., Ulrich, J. and Defaye, J. (1992) Carbohydr. Res. 236, 195-208.
- [28] Schauer, R., Haverkamp, J., Wember, M., Vliegenthart, J.F.G. and Kamerling, J.P. (1976) Eur. J. Biochem. 62, 237–242.
- [29] Kleineidam, R.G., Hofmann, O., Reuter, G. and Schauer, R. (1993) Glycoconjugate J. 10, 116-119.
- [30] Reuter, G., Pfeil, R., Kamerling, J.P., Vliegenthart, J.F.G. and Schauer, R. (1980) Biochim. Biophys. Acta 630, 306-310.
- [31] Bergwerff, A.A., Hulleman, S.H.D., Kamerling, J.P., Vliegenthart, J.F.G., Shaw, L., Reuter, G. and Schauer, R. (1992) Biochimie 74, 25–38.
- [32] Slomiany, A., Kojima, K., Banas-Gruszka, Z. and Slomiany, B.L. (1981) Biochem. Biophys. Res. Commun. 100, 778-784.
- [33] Corfield, A.P. and Schauer, R. (1982) In: R. Schauer (Ed.), Sialic acids Chemistry, Metabolism and Function. Springer, Wien, pp. 195–261.
- [34] Beau, J.-M., Schauer, R., Haverkamp, J., Kamerling, J.P., Dorland, L. and Vliegenthart, J.F.G. (1984) Eur. J. Biochem. 140, 203–208.
- [35] Shukla, A.K., Schröder, C., Nöhle, U. and Schauer, R. (1987) Carbohydr. Res. 168, 199-209.
- [36] Schauer, R., Schröder, C. and Shukla, A.K. (1984) Adv. Exp. Med. Biol. 174, 75-86.
- [37] Suzuki, M., Suzuki, A., Yamakawa, T. and Matsunaga, E. (1985) J. Biochem. (Tokyo) 97, 509-515.
- [38] Li, Y.-T., Nakagawa, H., Ross, S.A., Hansson, G. and Li, S.-C. (1990) J. Biol. Chem. 265, 21629– 21633.
- [39] Li, Y.-T., Yuziuk, J.A., Li, S.-C., Nematalla, A., Hasegawa, A., Kimura, M. and Nakagawa, H. (1994) Arch. Biochem. Biophys. 310, 243–246.
- [40] Damm, J.B.L., Bergwerff, A.A., Hård, K., Kamerling, J.P. and Vliegenhart, J.F.G. (1989) Recl. Trav. Chim. Pays-Bas 108, 351–359.
- [41] Manzi, A.E., Dell, A., Parastoo, A. and Varki, A. (1990) J. Biol. Chem. 265, 8094-8107.
- [42] Kochetkov, N.K., Smirnova, G.P. and Chekareva, N.V. (1976) Biochim. Biophys. Acta 424, 274-283.
- [43] Kluge, A., Reuter, G., Lee, H., Ruch-Heeger, B. and Schauer, R. (1992) Eur. J. Cell Biol. 59, 12-20.
- [44] Higuchi, R., Inukai, K., Jhou, J.X., Honda, M., Komori, T., Tsuji, S. and Nagai, Y. (1993) Liebigs Ann. Chem. 359-366.

- [45] Nöhle, U., Shukla, A.K., Schröder, C., Reuter, G., Schauer, R., Kamerling, J.P. and Vliegenthart, J.F.G. (1985) Eur. J. Biochem. 152, 459–463.
- [46] Nadano, D., Iwasaki, M., Endo, S., Kitajima, K., Inoue, S. and Inoue, Y. (1986) J. Biol. Chem. 261, 11550–11557.
- [47] Kanamori, A., Inoue, S., Iwasaki, M., Kitajima, K., Kawai, G., Yokoyama, S. and Inoue, Y. (1990) J. Biol. Chem. 265, 21811–21819.
- [48] Strecker, G., Wieruszeski, J.-M., Michalski, J.-C., Alonso, C., Boilly, B. and Montreuil, J. (1992) FEBS Lett. 298, 39–43.
- [49] Knirel, Y.A., Kocharova, N.A., Shashkov, A.S., Kochetkov, N.K., Mamontova, V.A. and Solov'eva, T.F. (1989) Carbohydr. Res. 188, 145–155.
- [50] Ziak, M., Yu, B.X., Zuo, X.L., Zuber, C., Kanamori, A., Kitajima, K., Inoue, S., Inoue, Y. and Roth, J. (1996) Proc. Natl. Acad. Sci. USA 93, 2759–2763.
- [51] Iwasaki, M., Inoue, S. and Troy, F.A. (1990) J. Biol. Chem. 265, 2596-2602.
- [52] Lutz, P., Lochinger, W. and Taigel, G. (1968) Chem. Ber. 101, 1089-1094.
- [53] Haverkamp, J., Schauer, R., Wember, M., Kamerling, J.P. and Vliegenthart, J.F.G. (1975) Hoppe-Seyler's Z. Physiol. Chem. 356, 1575–1583.
- [54] Yu, R.K. and Ledeen, R. (1969) J. Biol. Chem. 244, 1306-1313.
- [55] Haverkamp, J., Spoormaker, T., Dorland, L., Vliegenthart, J.F.G. and Schauer, R. (1979) J. Am. Chem. Soc. 101, 4851–4853.
- [56] Flippen, J.L. (1973) Acta Cryst. B 29, 1881-1886.
- [57] Reinhard, B. and Faillard, H. (1994) Liebigs Ann. Chem. 193-203.
- [58] Haverkamp, J., Schauer, R., Wember, M., Farriaux, J.-P., Kamerling, J.P., Versluis, C. and Vliegenthart, J.F.G. (1976) Hoppe-Seyler's Z. Physiol. Chem. 357, 1699–1705.
- [59] Miyoshi, I., Higashi, H., Hirabayashi, Y., Kato, S. and Naiki, M. (1986) Mol. Immunol. 23, 631-638.
- [60] Burmeister, W.P., Henrissat, B., Bosso, C., Cusack, S. and Ruigrok, R.W.H. (1993) Structure 1, 19-26.
- [61] Chou, M.-Y., Li, S.-C., Kiso, M., Hasegawa, A. and Li, Y.-T. (1994) J. Biol. Chem. 269, 18821--18826.
- [62] Lifely, M.R. and Cottee, F.H. (1982) Carbohydr. Res. 107, 187-197.
- [63] Pozsgay, V., Jennings, H. and Kasper, D.L. (1987) Eur. J. Biochem. 162, 445-450.
- [64] Saito, K., Sugai, K., Fujikura, K., Yamada, N., Goto, M., Ban, C., Hayasaka, E., Sugiyama, N. and Tomita, K. (1989) Carbohydr. Res. 185, 307–314.
- [65] Montreuil, J., Biserte, G., Strecker, G., Spik, G., Fontaine, G. and Farriaux, J.-P. (1968) Clin. Chim. Acta 21, 61–69.
- [66] Renlund, M., Chester, M.A., Lundblad, A., Aula, P., Raivio, K.O., Autio, S. and Koskela, S.-L. (1979) Eur. J. Biochem. 101, 245–250.
- [67] Stevenson, R.E., Lubinsky, M., Taylor, H.A., Wenger, D.A., Schroer, R.J. and Olmstead, P.M. (1983) Pediatrics 72, 441-449.
- [68] Palo, J., Rauvala, H., Finne, J., Haltia, M. and Palmgren, K. (1985) Clin. Chim. Acta 145, 237-242.
- [69] Krauss, J.H., Reuter, G., Schauer, R., Weckesser, J. and Mayer, H. (1988) Arch. Microbiol. 150, 584– 589.
- [70] Green, E.D., Adelt, G., Baenziger, J.U., Wilson, S. and van Halbeek, H. (1988) J. Biol. Chem. 263, 18253-18268.
- [71] Bendiak, B., Harris-Brandts, M., Michnick, S.W., Carver, J.P. and Cumming, D.A. (1989) Biochemistry 28, 6491–6499.
- [72] Bernard, N., Engler, R., Strecker, G., Montreuil, J., van Halbeek, H. and Vliegenthart, J.F.G. (1984) Glycoconjugate J. 1, 123–140.
- [73] Cumming, D.A., Hellerqvist, C.G., Harris-Brandts, M., Michnick, S.W., Carver, J.P. and Bendiak, B. (1989) Biochemistry 28, 6500-6512.
- [74] Hanisch, F.G., Uhlenbruck, G., Peter-Katalinic, J. and Egge, H. (1988) Carbohydr. Res. 178, 29-47.
- [75] Takasaki, S., Yamashita, K., Suzuki, K., Iwanaga, S. and Kobata, A. (1979) J. Biol. Chem. 254, 8548– 8553.
- [76] Vliegenthart, J.F.G., Dorland, L. and van Halbeek, H. (1983) Adv. Carbohydr. Chem. Biochem. 41, 209–374.
- [77] Kamerling, J.P. and Vliegenthart, J.F.G. (1992) Biol. Magn. Reson. 10, 1-194.

- [78] Rivera-Sagedro, A., Bergwerff, A.A., Kamerling, J.P. and Vliegenthart, J.F.G. (1997) in preparation.
- [79] Inoue, S., Iwasaki, M., Ishii, K., Kitajima, K. and Inoue, Y. (1989) J. Biol. Chem. 264, 18520-18526.
- [80] Iwasaki, M., Seko, A., Kitajima, K., Inoue, Y. and Inoue, S. (1992) J. Biol. Chem. 267, 24287-24296.
- [81] Hård, K., van Zadelhoff, G., Moonen, P., Kamerling, J.P. and Vliegenthart, J.F.G. (1992) Eur. J. Biochem. 209 (1992) 895-915.
- [82] Pfeiffer, G., Stirm, S., Geyer, R., Strube, K.-H., Bergwerff, A.A., Kamerling, J.P. and Vliegenthart, J.F.G. (1992) Glycobiology 2, 411–418.
- [83] Van Kuik, J.A., Hård, K. and Vliegenthart, J.F.G. (1992) Carbohydr. Res. 235, 53-68.
- [84] Hokke, C.H., Bergwerff, A.A., van Dedem, G.W.K., Kamerling, J.P. and Vliegenthart, J.F.G. (1995) Eur. J. Biochem. 228, 981–1008.
- [85] De Waard, P., Koorevaar, A., Kamerling, J.P. and Vliegenthart, J.F.G. (1991) J. Biol. Chem. 266, 4237– 4243.
- [86] Slomiany, B.L., Murty, V.L.N. and Slomiany, A. (1980) J. Biol. Chem. 255, 9719-9723.
- [87] Damm, J.B.L., Hård, K., Kamerling, J.P., van Dedem, G.W.K. and Vliegenthart, J.F.G. (1990) Eur. J. Biochem. 189, 175–183.
- [88] Coddeville, B., Stratil, A., Wieruszeski, J.-M., Strecker, G., Montreuil, J. and Spik, G. (1989) Eur. J. Biochem. 186, 583–590.
- [89] Taguchi, T., Seko, A., Kitajima, K., Inoue, S., Iwamatsu, T., Khoo, K.-H., Morris, H.R., Dell, A. and Inoue, Y. (1993) J. Biol. Chem. 268, 2353–2362.
- [90] Taguchi, T., Seko, A., Kitajima, K., Muto, Y., Inoue, S., Khoo, K.-H., Morris, H.R., Dell, A. and Inoue, Y. (1994) J. Biol. Chem. 269, 8762–8771.
- [91] Fukuda, M., Lauffenburger, M., Sasaki, H., Rogers, M.E. and Dell, A. (1987) J. Biol. Chem. 262, 11952-11957.
- [92] Pfeiffer, G., Dabrowski, U., Dabrowski, J., Stirm, S., Strube, K.-H. and Geyer, R. (1992) Eur. J. Biochem. 205, 961–978.
- [93] Weisshaar, G., Hiyama, J., Renwick, A.G.C. and Nimtz, M. (1991) Eur. J. Biochem. 195, 257-268.
- [94] Bergwerff, A.A., van Oostrum, J., Kamerling, J.P. and Vliegenthart, J.F.G. (1995) Eur. J. Biochem. 228, 1009–1019.
- [95] Iwasaki, M. and Inoue, S. (1985) Glycoconjugate J. 2, 209-228.
- [96] Slomiany, A. and Slomiany, B.L. (1978) J. Biol. Chem. 253, 7301-7306.
- [97] Van der Meer, A., Kamerling, J.P., Vliegenthart, J.F.G., Schmid, K. and Schauer, R. (1983) Biochim. Biophys. Acta 757, 371–376.
- [98] Fukuda, M.N., Dell, A., Oates, J.E. and Fukuda, M. (1985) J. Biol. Chem. 260, 6623-6631.
- [99] Finne, J. (1985) Trends Biochem. Sci. 10, 129-132.
- [100] Sato, C., Kitajima, K., Tazawa, I., Inoue, Y., Inoue, S. and Troy, F.A. (1993) J. Biol. Chem. 268, 23675-23684.
- [101] Kitazume, S., Kitajima, K., Inoue, S., Troy, F.A., Cho, J.-W., Lennarz, W.J. and Inoue, Y. (1994) J. Biol. Chem. 269, 22712–22718.
- [102] Tezuka, T., Taguchi, T., Kanamari, A., Muto, Y., Kitajima, K., Inoue, Y. and Inoue, S. (1994) Biochemistry 33, 6495-6502.
- [103] Strecker, G., Wieruszeski, J.-M., Michalski, J.-C. and Montreuil, J. (1992) Biochem. J. 287, 905-909.
- [104] Maes, E., Wieruszeski, J.-M., Plancke, Y. and Strecker, G. (1995) FEBS Lett. 358, 205-210.
- [105] Strecker, G., Wieruszeski, J.-M., Michalski, J.-C., Alonso, C., Leroy, Y., Boilly, B. and Montreuil, J. (1992) Eur. J. Biochem. 207, 995–1002.
- [106] Kimura, M., Hama, Y., Sumi, T., Asakawa, M., Rao, B.N.N., Horne, A.P., Li, S.-C., Li, Y.-T. and Nakagawa, H. (1994) J. Biol. Chem. 269, 32138–32143.
- [106a] Nakagawa, H., Sumi, T., Kimura, M., Hama, Y., Li, S.-C. and Li, Y.-T. (1995) Glycoconjugate J. 12, 516.
- [107] Terada, T., Kitazume, S., Kitajima, K., Inoue, S., Ito, F., Troy, F.A. and Inoue, Y. (1993) J. Biol. Chem. 268, 2640–2648.
- [108] Yu, R.K. and Saito, M. (1989) In: R.U. Margolis and R.K. Margolis (Eds.), Neurobiology of Glycoconjugates. Plenum Publishing Corporation, New York, pp. 1–42.
- [109] Kubo, H., Irie, A., Inagaki, F. and Hoshi, M. (1990) J. Biochem. (Tokyo) 108, 185-192.

- [110] Ren, S.-L., Scarsdale, J.N., Ariga, T., Zhang, Y.-J., Klein, R.A., Hartmann, R., Kushi, Y., Egge, H. and Yu, R.K. (1992) J. Biol. Chem. 267, 12632–12638.
- [111] Waki, H., Murata, A., Kon, K., Maruyama, K., Kimura, S., Ogura, H. and Ando, S. (1993) J. Biochem. (Tokyo) 113, 502–507.
- [112] Kazuya, I.-P., Hidari, J., Irie, F., Suzuki, M., Kon, K., Ando, S. and Hirabayashi, Y. (1993) Biochem. J. 296, 259–263.
- [113] Waki, H., Masuzawa, A., Kon, K. and Ando, S. (1993) J. Biochem. (Tokyo) 114, 459-462.
- [114] Nakao, T., Kon, K., Ando, S. and Hirabayashi, Y. (1991) Biochim. Biophys. Acta 1086, 305-309.
- [115] Muralikrishna, G., Reuter, G., Peter-Katalinic, J., Egge, H., Hanisch, F.-G., Siebert, H.-C. and Schauer, R. (1992) Carbohydr. Res. 236, 321–326.
- [116] Miller-Podraza, H., Månsson, J.-E. and Svennerholm, L. (1991) FEBS Lett. 288, 212-214.
- [117] Nakamura, K., Suzuki, M., Taya, C., Inagaki, F., Yamakawa, T. and Suzuki, A. (1991) J. Biochem. (Tokyo) 110, 832-841.
- [118] Nakamura, K., Inagaki, F. and Tamai, Y. (1988) J. Biol. Chem. 263, 9896-9900.
- [119] Hirabayashi, Y., Nakao, T., Irie, F., Whittaker, V.P., Kon, K. and Ando, S. (1992) J. Biol. Chem. 267, 12973–12978.
- [120] Furukawa, K., Yamaguchi, H., Oettgen, H.F., Old, L.J. and Lloyd, K.O. (1988) J. Biol. Chem. 263, 18507–18512.
- [121] Fredman, P., Månsson, J.-E., Wikstrand, C.J., Vrionis, F.D., Rynmark, B.-M., Bigner, D.D. and Svennerholm, L. (1989) J. Biol. Chem. 264, 12122–12125.
- [122] Song, Y., Kitajima, K., Inoue, S. and Inoue, Y. (1991) J. Biol. Chem. 266, 21929-21935.
- [123] Song, Y., Kitajima, K., Inoue, S., Muto, Y., Kasama, T., Handa, S. and Inoue, Y. (1993) Biochemistry 32, 9221–9229.
- [124] Song, Y., Kitajima, K., Inoue, S., Khoo, K.-H., Morris, H.R., Dell, A. and Inoue, Y. (1995) Glycobiology 5, 207–218.
- [125] Kobata, A. (1977) In: M.I. Horowitz and W. Pigman (Eds.), The Glycoconjugates, Vol. 1. Academic Press, New York, pp. 423-440.
- [126] Grönberg, G., Lipniunas, P., Lundgren, T., Erlansson, K., Lindh, F. and Nilsson, B. (1989) Carbohydr. Res. 191, 261–278.
- [127] Platzer, N., Davoust, D., Lhermitte, M., Bauvy, C., Meyer, D.M. and Derappe, C. (1989) Carbohydr. Res. 191, 191–207.
- [128] Strecker, G., Wieruszeski, J.-M., Michalski, J.-C. and Montreuil, J. (1989) Glycoconjugate J. 6, 67-83.
- [129] Kitagawa, H., Takaoka, M., Nakada, H., Fukui, S., Funakoshi, I., Kawasaki, T., Tate, S., Inagaki, F. and Yamashina, I. (1991) J. Biochem. (Tokyo) 110, 598–604.
- [130] Grönberg, G., Lipniunas, P., Lundgren, T., Lindh, F. and Nilsson, B. (1990) Arch. Biochem. Biophys. 278, 297–311.
- [131] Wieruszeski, J.-M., Chekkor, A., Bouquelet, S., Montreuil, J., Strecker, G., Peter-Katalinic, J. and Egge, H. (1985) Carbohydr. Res. 137, 127–138.
- [132] Kitagawa, H., Nakada, H., Numata, Y., Korosaka, A., Fukui, S., Funakoshi, I., Kawasaki, T., Shimada, K., Inagaki, F. and Yamashina, I. (1988) J. Biochem. (Tokyo) 104, 591–594.
- [133] Wang, W.T., Erlansson, K., Lindh, F., Lundgren, T. and Zopf, D. (1990) Anal. Biochem. 190, 182-187.
- [134] Huang, R.T.C. (1971) Hoppe-Seyler's Z. Physiol. Chem. 352, 1645-1652.
- [135] Prieto, P.A. and Smith, D.F. (1984) Arch. Biochem. Biophys. 229, 650-656.
- [136] Grönberg, G., Lipniunas, P., Lundgren, T., Lindh, F. and Nilsson, B. (1992) Arch. Biochem. Biophys. 296, 597-610.
- [137] Yamashita, K., Tachibana, Y. and Kobata, A. (1976) Arch. Biochem. Biophys. 174, 582-591.
- [138] Fievre, S., Wieruszeski, J.-M., Michalski, J.-C., Lemoine, J., Montreuil, J. and Strecker, G. (1991) Biochem. Biophys. Res. Commun. 177, 720–725.
- [139] Kitagawa, H., Nakada, H., Kurosaka, A., Hiraiwa, N., Numata, Y., Fukui, S., Funakoshi, I., Kawasaki, T., Yamashina, I., Shimada, I. and Inagaki, F. (1989) Biochemistry 28, 8891–8897.
- [140] Kitagawa, H., Nakada, H., Fukui, S., Funakoshi, I., Kawasaki, T., Yamashina, I., Tate, S. and Inagaki, F. (1993) J. Biochem. (Tokyo) 114, 504–508.

- [141] Kitagawa, H., Nakada, H., Fukui, S., Funakoshi, I., Kawasaki, T., Yamashina, I., Tate, S. and Inagaki, F. (1991) Biochemistry 30, 2869–2876.
- [142] Wang, W.T., Lundgren, T., Lindh, F., Nilsson, B., Grönberg, G., Brown, J.P., Mentzer-Dibert, H. and Zopf, D. (1992) Arch. Biochem. Biophys. 292, 433–441.
- [143] Yamashita, K., Tachibana, Y. and Kobata, A. (1977) Arch. Biochem. Biophys. 182, 546-555.
- [144] Tarrago, M.T., Tucker, K.H., van Halbeek, H. and Smith, D.F. (1988) Arch. Biochem. Biophys. 267, 353–362.
- [145] Smith, D.F., Prieto, P.A., McCrumb, D.K. and Wang, W.-C. (1987) J. Biol. Chem. 262, 12040-12047.
- [146] Kitagawa, H., Nakada, H., Numata, Y., Kurosaka, A., Fukui, S., Funakoshi, I., Kawasaki, T., Shimada, I., Inagaki, F. and Yamashina, I. (1990) J. Biol. Chem. 265, 4859–4862.
- [147] Sabharwal, H., Nilsson, B., Grönberg, G., Chester, M.A., Dakour, J., Sjöblad, S. and Lundblad, A. (1988) Arch. Biochem. Biophys. 265, 390–406.
- [148] Parkkinen, J. and Finne, J. (1987) Methods Enzymol. 138, 289-300.
- [149] Weisshaar, G., Baumann, W., Friebolin, H., Brunner, H., Mann, H., Sieberth, H.-G. and Opferkuch, H.-J. (1987) Biol. Chem. Hoppe-Seyler 368, 647–658.
- [150] Weisshaar, G., Baumann, W., Friebolin, H., Brunner, H., Mann, H., Sieberth, H.-G. and Opferkuch, H.-J. (1987) Biol. Chem. Hoppe-Seyler 368, 1545–1558.
- [151] Berman, E. (1983) Carbohydr. Res. 118, 9-20.
- [152] Breg, J., Kroon-Batenburg, L.M.J., Strecker, G., Montreuil, J. and Vliegenthart, J.F.G. (1989) Eur. J. Biochem. 178, 727–739.
- [153] Van Pelt, J., Dorland, L., Duran, M., Hokke, C.H., Kamerling, J.P. and Vliegenthart, J.F.G. (1990) J. Biol. Chem. 265, 19685–19689.
- [154] Ryan, L.C., Carubelli, R., Caputto, R. and Trucco, R.E. (1965) Biochim. Biophys. Acta 101, 252-258.
- [155] Lemonnier, M. and Bourrillon, R. (1976) Carbohydr. Res. 51, 99-106.
- [156] Rodríguez-Aparicio, L.B., Reglero, A., Ortiz, A.I. and Luengo, J.M. (1988) Appl. Microbiol. Biotechnol. 27, 474–483.
- [157] Adlam, C., Knight, J., Mugridge, A., Williams, J. and Lindon, J. (1987) FEMS Microbiol. Lett. 42, 23-25.
- [158] Devi, S.J.N., Schneerson, R., Egan, W., Vann, W.F., Robbins, J.B. and Shiloach, J. (1991) Infect. Immun. 59, 732–736.
- [159] Higa, H.H. and Varki, A. (1988) J. Biol. Chem. 263, 8872-8878.
- [160] González-Clemente, C., Luengo, J.M., Rodríguez-Aparicio, L.B., Ferrero, M.A. and Reglero, A. (1990) Biol. Chem. Hoppe-Seyler 371, 1101–1106.
- [161] Dutton, G.G.S., Parolis, H. and Parolis, L.A.S. (1987) Carbohydr. Res. 170, 193-206.
- [162] Wessels, M.R., Pozsgay, V., Kasper, D.L. and Jennings, H.J. (1987) J. Biol. Chem. 262, 8262-8267.
- [163] DiFabio, J.L., Michon, F., Brisson, J.-R., Jennings, H.J., Wessels, M.R., Benedí, V.-J. and Kasper, D.L. (1989) Can. J. Chem. 67, 877–882.
- [164] Wessels, M.R., DiFabio, J.L., Benedi, V.-J., Kasper, D.L., Michon, F., Brisson, J.-R., Jelinková, J. and Jennings, H.J. (1991) J. Biol. Chem. 266, 6714–6719.
- [165] Kogan, G., Uhrín, D., Brisson, J.-R., Paoletti, L.C., Kasper, D.L., von Hunolstein, C., Orefici, G. and Jennings, H.J. (1994) J. Carbohydr. Chem. 13, 1071–1078.
- [165a] Kogan, G., Brisson, J.-R., Kasper, D.L., von Hunolstein, C., Orefici, G. and Jennings, H.J. (1995) Carbohydr. Res. 277, 1–9.
- [166] Gamian, A., Kenne, L., Mieszala, M., Ulrich, J. and Defaye, J. (1994) Eur. J. Biochem. 225, 1211– 1220.
- [167] Torgov, V.I., Shashkov, A.S., Jann, B. and Jann, K. (1995) Carbohydr. Res. 272, 73-90.
- [168] Vinogradov, E.V., Knirel, Y.A., Shashkov, A.S., Paramonov, N.A., Kochetkov, N.K., Stanislavsky, E.S. and Kholodkova, E.V. (1994) Carbohydr. Res. 259, 59–65.
- [169] Vinogradov, E.V., Holst, O., Thomas-Oates, J.E., Broady, K.W. and Brade, H. (1992) Eur. J. Biochem. 210, 491–498.
- [170] Gamian, A., Romanowska, E., Dabrowski, U. and Dabrowski, J. (1991) Biochemistry 30, 5032-5038.
- [171] Pavliak, V., Brisson, J.-R., Michon, F., Uhrin, D. and Jennings, H.J. (1993) J. Biol. Chem. 268, 14146– 14152.

- [172] Yamasaki, R., Griffiss, J.M., Quinn, K.P. and Mandrell, R.E. (1993) J. Bacteriol. 175 (1993) 4565– 4568.
- [173] Kim, J.J., Phillips, N.J., Gibson, B.W., Griffiss, J.M. and Yamasaki, R. (1994) Infect. Immun. 62, 1566– 1575.
- [174] Melaugh, W., Phillips, N.J., Campagnari, A.A., Tullius, M.V. and Gibson, B.W. (1994) Biochemistry 33, 13070–13078.
- [175] Aspinall, G.O., McDonald, A.G., Raju, T.S., Pang, H., Moran, A.P. and Penner, J.L. (1993) Eur. J. Biochem. 213, 1017–1027.
- [176] Aspinall, G.O., McDonald, A.G., Raju, T.S., Pang, H., Kurjanczyk, L.A., Penner, J.L. and Moran, A.P. (1993) Eur. J. Biochem. 213, 1029–1037.
- [177] Krauss, J.H., Himmelspach, K., Reuter, G., Schauer, R. and Mayer, H. (1992) Eur. J. Biochem. 204, 217–223.
- [178] Defives, C., Bouslamti, R., Derieux, J.C., Kol, O. and Fournet, B. (1989) FEMS Microbiol. Lett. 57, 203-208.
- [179] Gamian, A. and Kenne, L. (1993) J. Bacteriol. 175, 1508-1513.
- [180] Basu, S., Schlecht, S., Wagner, M. and Mayer, H.L. (1994) FEMS Immunol. Med. Microbiol. 9, 189– 198.
- [181] Utley, S.R., Bhat, U.R., Byrd, W. and Kadis, S. (1992) FEMS Microbiol. Lett. 92, 211-216.
- [182] Gibson, B.W., Melaugh, W., Phillips, N.J., Apicella, M.A., Campagnari, A.A. and McLeod Griffiss, J. (1993) J. Bacteriol. 175, 2702–2712.
- [183] Mandrell, R.E., McLaughlin, R., Kwaik, Y.A., Lesse, A., Yamasaki, R., Gibson, B., Spinola, S.M. and Apicella, M.A. (1992) Infect. Immun. 60, 1322–1328.
- [184] Smith, H., Cole, J.A. and Parsons, N.J. (1992) FEMS Microbiol. Lett. 100, 287-292.
- [185] Wetzler, L.M., Barry, K., Blake, M.S. and Gotschlich, E.C. (1992) Infect. Immun. 60, 39-43.
- [186] Stahl, N., Baldwin, M.A., Hecker, R., Pan, K.-M., Burlingame, A.L. and Prusiner, S.B. (1992) Biochemistry 31, 5043-5053.
- [187] Nöhle, U. and Schauer, R. (1984) Hoppe-Seyler's Z. Physiol. Chem. 365, 1457-1467.
- [188] Shaw, L. and Schauer, R. (1988) Biol. Chem. Hoppe-Seyler 369, 477-486.
- [189] Kamerling, J.P. (1994) Pure Appl. Chem. 66, 2235-2238.
- [190] Schachter, H. and Brockhausen, I. (1992) In: H.J. Allen and E.C. Kisailus (Eds.), Glycoconjugates Composition, Structure, and Function. Marcel Dekker, New York, pp. 263–332.
- [191] Dorland, L., Haverkamp, J., Vliegenthart, J.F.G., Strecker, G., Michalski, J.-C., Fournet, B., Spik, G. and Montreuil, J. (1978) Eur. J. Biochem. 87, 323–329.
- [192] Lecat, D., Lemonnier, M., Derappe, C., Lhermitte, M., van Halbeek, H., Dorland, L. and Vliegenthart, J.F.G. (1984) Eur. J. Biochem. 140, 415–420.
- [193] Kuriyama, M., Ariga, T., Ando, S., Suzuki, M., Yamada, T., Miyatake, T. and Igata, A. (1985)
 J. Biochem. (Tokyo) 98, 1049–1054.
- [194] Van Pelt, J., Kamerling, J.P., Vliegenthart, J.F.G., Verheijen, F.W. and Galjaard, H. (1988) Biochim. Biophys. Acta 965, 36–45.
- [195] Van Pelt, J., Kamerling, J.P., Vliegenthart, J.F.G., Hoogeveen, A.T. and Galjaard, H. (1988) Clin. Chim. Acta 174, 325–336.
- [196] Van Pelt, J., van Kuik, J.A., Kamerling, J.P., Vliegenthart, J.F.G., van Diggelen, O.P. and Galjaard, H. (1988) Eur. J. Biochem. 177, 327–338.
- [197] Van Pelt, J., van Bilsen, D.G.J.L., Kamerling, J.P. and Vliegenthart, J.F.G. (1988) Eur. J. Biochem. 174, 183–187.
- [198] Van Pelt, J., Hård, K., Kamerling, J.P., Vliegenthart, J.F.G., Reuser, A.J.J. and Galjaard, H. (1989) Biol. Chem. Hoppe-Seyler 370, 191–203.
- [199] Van Pelt, J., Kamerling, J.P., Bakker, H.D. and Vliegenthart, J.F.G. (1991) J. Inher. Metab. Dis. 14, 730-740.
- [200] Lifely, M.R., Gilbert, A.S. and Moreno, C. (1981) Carbohydr. Res. 94, 193-203.
- [201] Lifely, M.R., Gilbert, A.S. and Moreno, C. (1984) Carbohydr. Res. 134, 229-243.
- [202] Lifely, M.R., Lindon, J.C., Williams, J.M. and Moreno, C. (1985) Carbohydr. Res. 143, 191-205.
- [203] Bassi, R., Riboni, L., Sonnino, S. and Tettamanti, G. (1989) Carbohydr. Res. 141-146.

- [204] Fronza, G., Kirschner, G., Acquotti, D. and Sonnino, S. (1989) Carbohydr. Res. 195, 51-58.
- [205] Ando, S., Yu, R.K., Scarsdale, J.N., Kusunoki, S. and Prestegard, J.H. (1989) J. Biol. Chem. 264, 3478-3483.
- [206] Terabayashi, T., Ogawa, T. and Kawanishi, Y. (1990) J. Biochem. (Tokyo) 107, 868-871.
- [207] Zilliken, F. and O'Brien, P.J. (1960) Biochem. Prep. 7, 1-5.
- [208] Uchida, Y., Tsukada, Y. and Sugimori, T. (1973) Agric. Biol. Chem. 37, 2105-2110.
- [209] Juneja, L.R., Koketsu, M., Nishimoto, K., Kim, M., Yamamoto, T. and Itoh, T. (1991) Carbohydr. Res. 214, 179–186.
- [210] Koketsu, M., Juneja, L.R., Kawanami, H., Kim, M. and Yamamoto, T. (1992) Glycoconjugate J. 9 (1992) 70-74.
- [211] Roy, R. and Pon, R.A. (1990) Glycoconjugate J. 7, 3-12.
- [212] Roy, R. and Boratynski, J. (1990) Biotechnol. Lett. 12, 347-352.
- [213] Schauer, R. and Corfield, A.P. (1982) In: R. Schauer (Ed.), Sialic acids Chemistry, Metabolism and Function. Springer, Wien, pp. 51–57.
- [214] Knirel, Y.A. and Kochetkov, N.K. (1987) FEMS Microbiol. Rev. 46, 381-385.
- [214a] Edebrink, P., Jansson, P.-E., Bøgwald, J. and Hoffmann, J. (1996) Carbohydr. Res. 287, 225-246.
- [215] Kenne, L., Lindberg, B., Schweda, E., Gustafsson, B. and Holme, T. (1988) Carbohydr. Res. 180, 285-294.
- [216] Nazarenko, E.L., Shashkov, A.S., Knirel, Y.A., Ivanova, E.P. and Ovodov, Y.S. (1990) Bioorg. Khim. 16, 1426–1429.
- [217] Vinogradov, E.V., Shashkov, A.S., Knirel, Y.A., Kochetkov, N.K., Dabrowski, J., Grosskurth, H., Stanislavsky, E.S. and Kholodkova, E.V. (1992) Carbohydr. Res. 231, 1–11.
- [218] Beynon, L.M., Richards, J.C. and Perry, M.B. (1994) Carbohydr. Res. 256, 303-317.
- [219] Knirel, Y.A., Rietschel, E.T., Marre, R. and Zähringer, U. (1994) Eur. J. Biochem. 221, 239-245.
- [220] Zähringer, U., Knirel, Y.A., Lindner, B., Helbig, J.H., Sonesson, A., Marre, R. and Rietschel, E.T. (1995) In: J. Levin, C.R. Alving, R.S. Munford and H. Redl (Eds.), Progress in Clinical and Biological Research, Vol. 392, Bacterial endotoxins – Lipopolysaccharides from genes to therapy. Wiley and Liss, New York, pp. 113–139.
- [220a] Knirel, Y.A., Helbig, J.H. and Zähringer, U. (1996) Carbohydr. Res. 283, 129-139.
- [221] Culling, C.F.A. and Reid, P.E. (1982) In: R. Schauer (Ed.), Sialic acids Chemistry, Metabolism and Function. Springer, Wien, pp. 173-193.
- [222] Mandal, C. and Mandal, C. (1990) Experientia 46, 433-441.
- [223] Mercy, S.P.D. and Ravindranath, M.H. (1993) Eur. J. Biochem. 215, 697-704.
- [224] Knibbs, R.N., Osborne, S.E., Glick, G.D. and Goldstein, I.J. (1993) J. Biol. Chem. 268, 18524-18531.
- [225] Tsuboi, I. and Yanagi, K. (1993) Comp. Biochem. Physiol. 104B, 19-26.
- [226] Babál, P., Pindak, F.F., Wells, D.J. and Gardner, W.A. (1994) Biochem. J. 299, 341-346.
- [227] Powell, L.D. and Varki, A. (1994) J. Biol. Chem. 269, 10628-10636.
- [228] Haselbeck, A. and Hösel, W. (1990) Glycoconjugate J. 7, 63-74.
- [229] Haselbeck, A., Schickaneder, E., von der Eltz, H. and Hösel, W. (1990) Anal. Biochem. 191, 25-30.
- [230] Zimmer, G., Reuter, G. and Schauer, R. (1992) Eur. J. Biochem. 204, 209-215.
- [231] Zimmer, G., Suguri, T., Reuter, G., Yu, R.K., Schauer, R. and Herrler, G. (1994) Glycobiology 4, 343-349.
- [232] Manuguerra, J.-C., DuBois, C. and Hannoun, C. (1991) Anal. Biochem. 194, 425-432.
- [233] Muchmore, E.A. and Varki, A. (1987) Science 236, 1293-1295.
- [234] Klein, A., Krishna, M., Varki, N.M. and Varki, A. (1994) Proc. Natl. Acad. Sci. USA 91, 7782-7786.
- [235] Sjoberg, E.R., Powell, L.D., Klein, A. and Varki, A. (1994) J. Cell Biol. 126, 549-562.
- [236] Aubeck, R., Eppelsheim, C., Bräuchle, C. and Hampp, N. (1993) Analyst 118, 1389-1392.
- [237] Schauer, R. (1978) Methods Enzymol. 50, 64-89.
- [238] Varki, A. and Diaz, S. (1984) Anal. Biochem. 137, 236-247.
- [239] Laganà, A., Marino, A., Fago, G. and Martinez, B.P. (1993) Anal. Biochem. 215, 266-272.
- [240] Kamerling, J.P., Gerwig, G.J., Vliegenthart, J.F.G. and Clamp, J.R. (1975) Biochem. J. 151, 491-495.
- [241] Kamerling, J.P. and Vliegenthart, J.F.G. (1989) In: A.M. Lawson (Ed.), Clinical Biochemistry -

Principles, Methods, Applications, Vol. 1, Mass spectrometry. Walter de Gruyter, Berlin, pp. 175-263.

- [242] Yu, R.K. and Ledeen, R.W. (1970) J. Lipid Res. 11, 506-516.
- [243] Kitajima, K., Inoue, S., Kitazume, S. and Inoue, Y. (1992) Anal. Biochem. 205, 244-250.
- [244] Corfield, A.P., Michalski, J.-C. and Schauer, R. (1981) In: G. Tettamanti, P. Durand and S. Di Donato (Eds.), Sialidases and Sialidosis, Perspectives in Inherited Metabolic Diseases, Vol. 4. Edi Ermes, Milan, pp. 3–70.
- [245] Corfield, T. (1992) Glycobiology 2, 509-521.
- [246] Saito, M. and Yu, R.K. (1995) In: A. Rosenberg (Ed.), Biology of the Sialic Acids. Plenum Press, New York, pp. 261–313.
- [247] Hoyer, L.L., Roggentin, P., Schauer, R. and Vimr, E.R. (1991) J. Biochem. (Tokyo) 110, 462-467.
- [248] Tanaka, H., Ito, F. and Iwasaki, T. (1994) J. Biochem. (Tokyo) 115, 318-321.
- [249] Corfield, A.P., Higa, H., Paulson, J.C. and Schauer, R. (1983) Biochim. Biophys. Acta 744, 121-126.
- [250] Varki, A. and Diaz, S. (1983) J. Biol. Chem. 258, 12465-12471.
- [251] Corfield, A.P., Sander-Wewer, M., Veh, R.W., Wember, M. and Schauer, R. (1986) Biol. Chem. Hoppe-Seyler 367, 433–439.
- [252] Kleineidam, R.G., Furuhata, K., Ogura, H. and Schauer, R. (1990) Biol. Chem. Hoppe-Seyler 371, 715-719.
- [253] Kitazume, S., Kitajima, K., Inoue, S., Troy, F.A., Lennarz, W.J. and Inoue, Y. (1994) Biochem. Biophys. Res. Commun. 205, 893–898.
- [254] Angata, T., Kitajima, K., Inoue, S., Chang, J., Warner, T.G., Troy, F.A. and Inoue, Y. (1994) Glycobiology 4, 517–523.
- [255] Kitajima, K., Kuroyanagi, H., Inoue, S., Ye, J., Troy, F.A. and Inoue, Y. (1994) J. Biol. Chem. 269, 21415–21419.
- [256] Jennings, H.J., Roy, R. and Michon, F. (1985) J. Immunol. 134, 2651-2657.
- [257] Lifely, M.R., Nowicka, U.T. and Moreno, C. (1986) Carbohydr. Res. 156, 123-135.
- [258] Manzi, A.E., Higa, H.H., Diaz, S. and Varki, A. (1994) J. Biol. Chem. 269, 23617-23624.
- [259] Hallenbeck, P.C., Vimr, E.R., Yu, F., Bassler, B. and Troy, F.A. (1987) J. Biol. Chem. 262, 3553-3561.
- [260] Troy, F.A., Hallenbeck, P.C., McCoy, R.D. and Vimr, E.R. (1987) Methods Enzymol. 138, 169-185.
- [261] Kwiatkowski, B. and Stirm, S. (1987) Methods Enzymol. 138, 786-792.
- [262] Schauer, R. and Corfield, A.P. (1982) In: R. Schauer (Ed.), Sialic acids Chemistry, Metabolism and Function. Springer, Wien, pp. 77–94.
- [263] Yao, K., Ubuka, T., Masuoka, N., Kinuta, M. and Ikeda, T. (1989) Anal. Biochem. 179, 332-335.
- [264] Sugahara, K., Sugimoto, K., Nomura, O. and Usui, T. (1980) Clin. Chim. Acta 108, 493-498.
- [265] Shukla, A.K. and Schauer, R. (1982) Hoppe-Seyler's Z. Physiol. Chem. 363, 255-262.
- [266] Schauer, R., Reuter, G. and Stoll, S. (1988) Biochimie 70, 1511-1519.
- [267] Kitazume, S., Kitajima, K., Inoue, S. and Inoue, Y. (1992) Anal. Biochem. 202, 25-34.
- [268] Manzi, A.E., Diaz, S. and Varki, A. (1990) Anal. Biochem. 188, 20-32.
- [269] Shukla, A.K. and Schauer, R. (1982) J. Chromatogr. 244, 81-89.
- [270] Shukla, A.K. and Schauer, R. (1986) Anal. Biochem. 158, 158-164.
- [271] Hara, S., Yamaguchi, M., Takemori, Y., Furahata, K., Ogura, H. and Nakamura, M. (1989) Anal. Biochem. 179, 162–166.
- [272] Hokke, C.H., Bergwerff, A.A., van Dedem, G.W.K., van Oostrum, J., Kamerling, J.P. and Vliegenthart, J.F.G. (1990) FEBS Lett. 275, 9–14.
- [273] Hayakawa, K., De Felice, C., Watanabe, T., Tanaka, T., Iinuma, K., Nihei, K., Higuchi, S., Ezoe, T., Hibi, I. and Kurosawa, K. (1993) J. Chromatogr. 620, 25–31.
- [274] Ishida, J., Nakahara, T. and Yamaguchi, M. (1992) Biomed. Chromatogr. 6, 135-140.
- [275] Shukla, A.K., Scholz, N., Reimerdes, E.H. and Schauer, R. (1982) Anal. Biochem. 123, 78-82.
- [276] Honda, S. and Suzuki, S. (1984) Anal. Biochem. 142, 167-174.
- [277] Powell, L.D. and Hart, G.W. (1986) Anal. Biochem. 157, 179-185.
- [278] Unland, F. and Müthing, J. (1992) Biomed. Chromatogr. 6, 155-159.
- [279] Kobayashi, K., Akiyama, Y., Kawaguchi, K., Tanabe, S. and Imanari, T. (1985) Anal. Sci. 1, 81.

- [280] Hara, S., Yamaguchi, M., Takemori, Y., Nakamura, M. and Ohkura, Y. (1986) J. Chromatogr. 377, 111-119.
- [281] Budd, T.J., Dolman, C.D., Lawson, A.M., Chai, W., Saxton, J. and Hemming, F.W. (1992) Glycoconjugate J. 9, 274–278.
- [282] Shukla, A.K. and Schauer, R. (1982) Fresenius Z. Anal. Chem. 311, 377.
- [283] Diaz, S. and Varki, A. (1985) Anal. Biochem. 150, 32-46.
- [284] Diaz, S., Higa, H.H., Hayes, B.K. and Varki, A. (1989) J. Biol. Chem. 264, 19416-19426.
- [285] Reinhard, B., Becker, A., Rothermel, J. and Faillard, H. (1992) Biol. Chem. Hoppe-Seyler 373, 63-68.
- [286] Sato, T., Omichi, K. and Ikenaka, T. (1988) J. Biochem. (Tokyo) 104, 18-21.
- [287] Ferrero, M.A., Luengo, J.M. and Reglero, A. (1991) Biochem. J. 280, 575-579.
- [288] Hallenbeck, P.C., Yu, F. and Troy, F.A. (1987) Anal. Biochem. 161, 181-186.
- [289] Ye, J., Kitajima, K., Inoue, Y., Inoue, S. and Troy, F.A. (1994) Methods Enzymol. 230, 460-484.
- [290] Ashwell, G., Berlin, W.K. and Gabriel, O. (1994) Anal. Biochem. 222, 495-502.
- [291] Townsend, R.R., Hardy, M.R., Hindsgaul, O. and Lee, Y.C. (1988) Anal. Biochem. 174, 459-470.
- [292] Townsend, R.R., Hardy, M.R., Cumming, D.A., Carver, J.P. and Bendiak, B. (1989) Anal. Biochem. 182, 1-8.
- [293] Hermentin, P., Witzel, R., Vliegenthart, J.F.G., Kamerling, J.P., Nimtz, M. and Conradt, H.S. (1992) Anal. Biochem. 203, 281–289.
- [294] Hayase, T., Sheykhanazari, M., Bhavanandan, V.P., Savage, A.V. and Lee, Y.C. (1993) Anal. Biochem. 211, 72–80.
- [295] Lee, Y.C. and Rice, K.G. (1993) In: M. Fukuda and A. Kobata (Eds.), Glycobiology A Practical Approach. IRL Press, Oxford, pp. 127–163.
- [296] Piller, F. and Piller, V. (1993) In: M. Fukuda and A. Kobata (Eds.), Glycobiology A Practical Approach. IRL Press, Oxford, pp. 291–328.
- [297] Hounsell, E.F. (Ed.) (1993) Glycoprotein Analysis in Biomedicine. Humana Press, Towota.
- [298] Toomre, D.K. and Varki, A. (1994) Glycobiology 4, 653-663.
- [299] Hase, S. (1994) Methods Enzymol. 230, 225-237.
- [300] Ruan, S. and Lloyd, K.O. (1994) Glycoconjugate J. 11, 249-256.
- [301] Hermentin, P., Doenges, R., Witzel, R., Hokke, C.H., Vliegenthart, J.F.G., Kamerling, J.P., Conradt, H.S., Nimtz, M. and Brazel, D. (1994) Anal. Biochem. 221, 29–41.
- [302] Kakehi, K., Susami, A., Taga, A., Suzuki, S. and Honda, S. (1994) J. Chromatogr. 680, 209-215.
- [303] Kamerling, J.P., Vliegenthart, J.F.G., Versluis, C. and Schauer, R. (1975) Carbohydr. Res. 41, 7-17.
- [304] Reuter, G. and Schauer, R. (1986) Anal. Biochem. 157, 39-46.
- [305] Schröder, C. and Schauer, R. (1982) Fresenius Z. Anal. Chem. 311, 385.
- [306] Lönngren, J. and Svensson, S. (1974) Adv. Carbohydr. Chem. Biochem. 29, 41-106.
- [307] Schauer, R., Stoll, S. and Reuter, G. (1991) Carbohydr. Res. 213, 353-359.
- [308] Reuter, G., Schauer, R., Szeiki, C., Kamerling, J.P. and Vliegenthart, J.F.G. (1989) Glycoconjugate J. 6, 35–44.
- [309] Shukla, A.K., Schauer, R., Schade, U., Moll, H. and Rietschel, E. Th. (1985) J. Chromatogr. 337, 231-238.
- [310] Devine, P.L., Clark, B.A., Birrell, G.W., Layton, G.T., Ward, B.G., Alewood, P.F. and McKenzie, I.F.C. (1991) Cancer Res. 51, 5826–5836.
- [311] Dell, A. (1987) Adv. Carbohydr. Chem. Biochem. 45, 19-72.
- [312] Dell, A., Khoo, K.-H., Panico, M., McDowell, R.A., Etienne, A.T., Reason, A.J. and Morris, H.R. (1993) In: M. Fukuda and A. Kobata (Eds.), Glycobiology – A Practical Approach. IRL Press, Oxford, pp. 187–222.
- [313] Van Kuik, J.A. and Vliegenthart, J.F.G. (1994) Carbohydr. Europe 10, 31-32.
- [314] Higa, H.H. and Paulson, J.C. (1985) J. Biol. Chem. 260, 8838-8849.
- [315] Augé, C. and Gautheron, C. (1988) Tetrahedron Lett. 29, 789-790.
- [316] Gross, H.J., Bünsch, A., Paulson, J.C. and Brossmer, R. (1987) Eur. J. Biochem. 168, 595-602.
- [317] Friebolin, H., Brossmer, R., Keilich, G., Ziegler, D. and Supp, M. (1980) Hoppe-Seyler's Z. Physiol. Chem. 361, 697-702.

- [318] Friebolin, H., Baumann, W., Brossmer, R., Keilich, G., Supp, M., Ziegler, D. and Von Nicolai, H. (1981) Biochem. Int. 4, 321–326.
- [319] Deijl, C.M. and Vliegenthart, J.F.G. (1983) Biochem. Biophys. Res. Commun. 111, 668-674.
- [320] Chong, A.K.J., Pegg, M.S., Taylor, N.R., von Itzstein, M. (1992) Eur. J. Biochem. 207, 335-343.
- [321] Friebolin, H., Kunzelmann, P., Supp, M., Brossmer, R., Keilich, G. and Ziegler, D. (1981) Tetrahedron Lett. 22, 1383–1386.
- [322] Baumann, W., Freidenreich, J., Weisshaar, G., Brossmer, R. and Friebolin, H. (1989) Biol. Chem. Hoppe-Seyler 370, 141-149.
- [323] Friebolin, H., Baumann, W., Keilich, G., Ziegler, D., Brossmer, R. and Von Nicolai, H. (1981) Hoppe-Seyler's Z. Physiol. Chem. 362, 1455–1463.
- [324] Paulson, J.C., Weinstein, J., Dorland, L., van Halbeek, H. and Vliegenthart, J.F.G. (1982) J. Biol. Chem. 257, 12734–12738.
- [325] Friebolin, H., Baumann, W., Hauck, M., Kurz, D., Wajda, R., Weisshaar, G., Keilich, G., Ziegler, D., Brossmer, R. and Von Nicolai, H. (1984) Hoppe-Seyler's Z. Physiol. Chem. 365, 1309-1321.
- [326] Dorland, L., Haverkamp, J., Schauer, R., Veldink, G.A. and Vliegenthart, J.F.G. (1982) Biochem. Biophys. Res. Commun. 104, 1114–1119.
- [327] Friebolin, H., Schmidt, H. and Supp, M. (1981) Tetrahedron Lett. 22, 5171-5174.
- [328] Hård, K. and Vliegenthart, J.F.G. (1993) In: M. Fukuda and A. Kobata (Eds.), Glycobiology A Practical Approach. IRL Press, Oxford, pp. 223–242.
- [329] Nomoto, H., Iwasaki, M., Endo, T., Inoue, S., Inoue, Y. and Matsumura, G. (1982) Arch. Biochem. Biophys. 218, 335–341.
- [330] Kitajima, K., Nomoto, H., Inoue, Y., Iwasaki, M. and Inoue, S. (1984) Biochemistry 23, 310-316.
- [331] Machytka, D., Klein, R.A. and Egge, H. (1994) Carbohydr. Res. 254, 289-294.
- [332] Strecker, G., Wieruszeski, J.-M., Cuvillier, O., Michalski, J.-C. and Montreuil, J. (1992) Biochimie 74, 39–52.
- [333] Joziasse, D.H., Schiphorst, W.E.C.M., van den Eijnden, D.H., van Kuik, J.A., van Halbeek, H. and Vliegenthart, J.F.G. (1985) J. Biol. Chem. 260, 714-719.
- [334] Nemansky, M. and van den Eijnden, D.H. (1993) Glycoconjugate J. 10, 99-108.
- [335] Van den Eijnden, D.H. and Joziasse, D.H. (1993) Curr. Opinion Struct. Biol. 3, 711-721.
- [336] Hori, H., Nakajima, T., Nishida, Y., Ohrui, H. and Meguro, H. (1988) Tetrahedron Lett. 29, 6317-6320.
- [337] Prytulla, S., Lambert, J., Lauterwein, J., Klessinger, M. and Thiem, J. (1990) Magn. Reson. Chem. 28, 888-901.
- [338] Prytulla, S., Lauterwein, J., Klessinger, M. and Thiem, J. (1991) Carbohydr. Res. 215, 345-349.
- [339] Okamoto, K. and Goto, T. (1990) Tetrahedron 46, 5835-5857.
- [340] Berman, E. (1986) Carbohydr. Res. 152, 33-46.
- [341] Berman, E. and Bendel, P. (1986) FEBS Lett. 204, 257-260.
- [342] Sabesan, S. and Paulson, J.C. (1986) J. Am. Chem. Soc. 108, 2068-2080.
- [343] Berman, E., Dabrowski, U. and Dabrowski, J. (1988) Carbohydr. Res. 176, 1-15.
- [344] Vliegenthart, J.F.G. and Kamerling, J.P. (1982) In: R. Schauer (Ed.), Sialic acids Chemistry, Metabolism and Function. Springer, Wien, pp. 59–76.
- [345] Toone, E.J., Simon, E.S., Bednarski, M.D. and Whitesides, G.M. (1989) Tetrahedron 45, 5365-5422.
- [346] DeNinno, M.P. (1991) Synthesis 583-593.
- [347] Hasegawa, A. and Kiso, M. (1992) In: H. Ogura, A. Hasegawa and T. Suami (Eds.), Carbohydrates Synthetic Methods and Applications in Medicinal Chemistry. Kodansha, Tokyo, pp. 243–266.
- [348] Ikeda, K. and Achiwa, K. (1992) In: H. Ogura, A. Hasegawa and T. Suami (Eds.), Carbohydrates Synthetic Methods and Applications in Medicinal Chemistry. Kodansha, Tokyo, pp. 267–281.
- [349] Ogura, H. (1992) In: H. Ogura, A. Hasegawa and T. Suami (Eds.), Carbohydrates Synthetic Methods and Applications in Medicinal Chemistry. Kodansha, Tokyo, pp. 282–303.
- [350] Zbiral, E. (1992) In: H. Ogura, A. Hasegawa and T. Suami (Eds.), Carbohydrates Synthetic Methods and Applications in Medicinal Chemistry. Kodansha, Tokyo, pp. 304–339.
- [351] Ichikawa, Y., Look, G.C. and Wong, C.-H. (1992) Anal. Biochem. 202, 215-238.
- [352] Ogura, H. (1994) Yakugaku Zasshi 114, 277-303.

- [353] Wong, C.-H. and Whitesides, G.M. (1994) Enzymes in Synthetic Organic Chemistry, Tetrahedron Organic Chemistry, Vol. 12. Elsevier Science, Oxford.
- [354] Benzing-Nguyen, L. and Perry, M.B. (1978) J. Org. Chem. 43, 551-554.
- [355] Danishefsky, S.J., DeNinno, M.P. and Chen, S. (1988) J. Am. Chem. Soc. 110, 3929-3940.
- [356] Gordon, D.M. and Whitesides, G.M. (1993) J. Org. Chem. 58, 7937-7938.
- [357] Baumberger, F. and Vasella, A. (1986) Helv. Chim. Acta 69, 1205-1215.
- [358] Baumberger, F. and Vasella, A. (1986) Helv. Chim. Acta 69, 1535-1541.
- [359] Julina, R., Müller, I., Vasella, A. and Wyler, R. (1987) Carbohydr. Res. 164, 415-432.
- [360] Csuk, R., Hugener, M. and Vasella, A. (1988) Helv. Chim. Acta 71, 609-618.
- [361] Yamamoto, T., Teshima, T., Inami, K. and Shiba, T. (1992) Tetrahedron Lett. 33, 325-328.
- [362] Bodenmüller, A. and Schmidt, R.R. (1994) Liebigs Ann. Chem. 541-548.
- [363] Marra, A. and Sinaÿ, P. (1989) Carbohydr. Res. 190, 317-322.
- [364] Shirai, R., Nakamura, M., Hara, S., Takayanagi, H. and Ogura, H. (1988) Tetrahedron Lett. 29, 4449– 4452.
- [365] Shirai, R. and Ogura, H. (1989) Tetrahedron Lett. 30, 2263-2264.
- [366] Chan, T.-H. and Li, C.-J. (1992) J. Chem. Soc. Chem. Commun. 747-748.
- [367] Dondoni, A. and Marra, A. (1994) Carbohydr. Lett. 1, 43-46.
- [368] Dondoni, A., Marra, A. and Merino, P. (1994) J. Am. Chem. Soc. 116, 3324-3336.
- [369] Augé, C., David, S. and Gautheron, C. (1984) Tetrahedron Lett. 25, 4663-4664.
- [370] David, S. and Augé, C. (1987) Pure Appl. Chem. 59, 1501-1508.
- [371] Kragl, U., Gygax, D., Ghisalba, O. and Wandrey, C. (1991) Angew. Chem. Int. Ed. Engl. 30, 827-828.
- [372] Kim, M.-J., Hennen, W.J., Sweers, H.M. and Wong, C.-H. (1988) J. Am. Chem. Soc. 110, 6481-6486.
- [373] Bednarski, M.D., Chenault, H.K., Simon, E.S. and Whitesides, G.M. (1987) J. Am. Chem. Soc. 109, 1283–1285.
- [374] Lubineau, A., Augé, C., Gautheron-Le Narvor, C. and Ginet, J.-C. (1994) BioMed. Chem. 2, 669-674.
- [375] Augé, C. and Gautheron, C. (1987) J. Chem. Soc. Chem. Commun. 859-860.
- [376] Augé, C., Gautheron, C., David, S., Malleron, A., Cavayé, B. and Bouxom, B. (1990) Tetrahedron 46, 201-214.
- [377] Halcomb, R.L., Fitz, W. and Wong, C.-H. (1994) Tetrahedron: Asymmetry 5, 2437-2442.
- [378] Hasegawa, A., Murase, T., Ogawa, M., Ishida, H. and Kiso, M. (1990) J. Carbohydr. Chem. 9, 415-428.
- [379] Hasegawa, A., Murase, T., Ogawa, M., Ishida, H. and Kiso, M. (1990) J. Carbohydr. Chem. 9, 429-439.
- [380] Ogura, H., Furuhata, K., Sato, S., Anazawa, K., Itoh, M. and Shitori, Y. (1987) Carbohydr. Res. 167, 77-86.
- [381] Furuhata, K. and Ogura, H. (1989) Chem. Pharm. Bull. 37, 2037-2040.
- [382] Reinhard, B., Götz, C. and Faillard, H. (1992) Biol. Chem. Hoppe-Seyler 373, 1243-1248.
- [383] Augé, C., David, S., Gautheron, C. and Veyrières, A. (1985) Tetrahedron Lett. 26, 2439–2440.
- [384] Liu, J.L.-C., Shen, G.-J., Ichikawa, Y., Rutan, J.F., Zapata, G., Vann, W.F. and Wong, C.-H. (1992) J. Am. Chem. Soc. 114, 3901–3910.
- [385] Forstner, M., Freytag, K. and Paschke, E. (1989) Carbohydr. Res. 193, 294-295.
- [386] Shigeta, S., Winter, H.C. and Goldstein, I.J. (1994) Carbohydr. Res. 264, 111-121.
- [387] Murray, M.C., Bhavanandan, V.P., Davidson, E.A. and Reinhold, V. (1989) Carbohydr. Res. 186, 255– 265.
- [388] Kragl, U., Gödde, A., Wandrey, C., Kinzy, W., Cappon, J.J. and Lugtenburg, J. (1993) Tetrahedron: Asymmetry 4, 1193–1202.
- [389] Schreiner, E., Zbiral, E., Kleineidam, R.G. and Schauer, R. (1991) Carbohydr. Res. 216, 61-66.
- [390] Brossmer, R. and Gross, H.J. (1994) Methods Enzymol. 247, 153-176.
- [391] Isecke, R. and Brossmer, R. (1994) Tetrahedron 50, 7445-7460.
- [392] Petrie, C.R., Sharma, M., Simmons, O.D. and Korytnyk, W. (1989) Carbohydr. Res. 186, 326-334.
- [393] Schmid, W., Christian, R. and Zbiral, E. (1988) Tetrahedron Lett. 29, 3643-3646.
- [394] Kelm, S., Paulson, J.C., Rose, U., Brossmer, R., Schmid, W., Bandgar, B.P., Schreiner, E., Hartmann, M. and Zbiral, E. (1992) Eur. J. Biochem. 205, 147–153.
- [395] Hanessian, S. and Girard, C. (1994) Synlett 863-864.
- [396] Hartmann, M. and Zbiral, E. (1991) Liebigs Ann. Chem. 795-801.

- [397] Kessler, J., Heck, J., Tanenbaum, S.W. and Flashner, M. (1982) J. Biol. Chem. 257, 5056-5060.
- [398] Bandgar, B.P., Hartmann, M., Schmid, W. and Zbiral, E. (1990) Liebigs Ann. Chem. 1185-1195.
- [399] Okamoto, K., Hasegawa, T., Toyomaki, Y., Yamakawa, M. and Okukado, N. (1992) Chem. Pharm. Bull. 40, 2728–2734.
- [400] Brandstetter, H.H. and Zbiral, E. (1983) Liebigs Ann. Chem. 2055-2065.
- [401] Hagedorn, H.W. and Brossmer, R. (1986) Helv. Chim. Acta 69, 2127-2132.
- [402] Gross, H.J. and Brossmer, R. (1987) Glycoconjugate J. 4, 145-156.
- [403] Estenne, G., Saroli, A. and Doutheau, A. (1991) J. Carbohydr. Chem. 10, 181-195.
- [404] Luger, P., Zaki, C., Hagedorn, H.-W. and Brossmer, R. (1987) Carbohydr. Res. 164, 49-58.
- [405] Hartmann, M. and Zbiral, E. (1990) Tetrahedron Lett. 31, 2875-2878.
- [406] Gross, H.J., Kovac, A., Rose, U., Watzlawick, H. and Brossmer, R. (1988) Biochemistry 27, 4279– 4283.
- [407] Beau, J.-M. and Schauer, R. (1980) Eur. J. Biochem. 106, 531-540.
- [408] Hartmann, M., Christian, R. and Zbiral, E. (1990) Liebigs Ann. Chem. 83-91.
- [409] Gross, H.J. and Brossmer, R. (1988) FEBS Lett. 232, 145-147.
- [410] Zbiral, E., Brandstetter, H.H. and Schreiner, E.P. (1988) Monatsh. Chem. 119, 127-141.
- [411] Zbiral, E. and Brandstetter, H.H. (1985) Monatsh. Chem. 116, 87-98.
- [412] Salunkhe, M., Hartmann, M., Schmid, W. and Zbiral, E. (1988) Liebigs Ann. Chem. 187-189.
- [413] Hartmann, M. and Zbiral, E. (1991) Monatsh. Chem. 122, 995-1003.
- [414] Isecke, R. and Brossmer, R. (1995) Carbohydr. Res. 274, 303-311.
- [415] Sticher, U., Gross, H.J. and Brossmer, R. (1991) Glycoconjugate J. 8, 45-54.
- [416] Gross, H.J. and Brossmer, R. (1988) Glycoconjugate J. 5, 411-417.
- [417] Gross, H.J., Rose, U., Krause, J.M., Paulson, J.C., Schmid, K., Feeney, R.E. and Brossmer, R. (1989) Biochemistry 28, 7386-7392.
- [418] Brossmer, R., Rose, U., Kasper, D., Smith, T.L., Grasmuk, H., Unger, F.M. (1980) Biochem. Biophys. Res. Commun. 96, 1282–1289.
- [419] Brossmer, R. and Gross, H.J. (1994) Methods Enzymol. 247, 177-193.
- [420] Fitz, W. and Wong, C.-H. (1994) J. Org. Chem. 59, 8279-8280.
- [421] Schreiner, E., Christian, R. and Zbiral, E. (1990) Liebigs Ann. Chem. 93-97.
- [422] Gross, H.J. and Brossmer, R. (1988) Eur. J. Biochem. 177, 583-589.
- [423] Gross, H.J., Sticher, U. and Brossmer, R. (1990) Anal. Biochem. 186, 127-134.
- [424] Conradt, H.S., Bünsch, A. and Brossmer, R. (1984) FEBS Lett. 170, 295-300.
- [425] Sharma, M., Petrie, C.R. and Korytnyk, W. (1988) Carbohydr. Res. 175, 25-34.
- [426] Gross, H.J., Bünsch, A. and Brossmer, R. (1984) Hoppe-Seyler's Z. Physiol. Chem. 365, 994–995.
- [427] Petrie, C.R. and Korytnyk, W. (1981) Anal. Biochem. 131, 153-159.
- [428] Holzer, C.T., von Itzstein, M., Jin, B., Pegg, M.S., Stewart, W.P. and Wu, W.-Y. (1993) Glycoconjugate J. 10, 40–44.
- [429] Von Itzstein, M., Wu, W.-Y. and Jin, B. (1994) Carbohydr. Res. 259, 301-305.
- [430] Schreiner, E., Zbiral, E., Kleineidam, R.G. and Schauer, R. (1991) Liebigs Ann. Chem. 129-134.
- [431] Von Itzstein, M., Jin, B., Wu, W.-Y. and Chandler, M. (1993) Carbohydr. Res. 244, 181-185.
- [432] Warner, T.G. (1987) Biochem. Biophys. Res. Commun. 148, 1323-1329.
- [433] Kumar, V., Tanenbaum, S.W. and Flashner, M. (1982) Carbohydr. Res. 103, 281-285.
- [434] Kumar, V., Kessler, J., Scott, M.E., Patwardhan, B.H., Tanenbaum, S.W. and Flashner, M. (1981) Carbohydr. Res. 94, 123–130.
- [435] Flashner, M., Kessler, J. and Tanenbaum, S.W. (1983) Arch. Biochem. Biophys. 221, 188-196.
- [436] Mack, H. and Brossmer, R. (1987) Tetrahedron Lett. 28, 191-194.
- [437] Zbiral, E., Brandstetter, H.H., Christian, R. and Schauer, R. (1987) Liebigs Ann. Chem. 781-786.
- [438] Driguez, P.-A., Barrere, B., Quash, G. and Doutheau, A. (1994) Carbohydr. Res. 262, 297-310.
- [439] Baumberger, F., Vasella, A. and Schauer, R. (1988) Helv. Chim. Acta 71, 429-445.
- [440] Zbiral, E., Schreiner, E. and Christian, R. (1989) Carbohydr. Res. 194, C15-C18.
- [441] Dondoni, A., Boscarato, A. and Marra, A. (1994) Tetrahedron: Asymmetry 5, 2209-2212.
- [442] Hagiwara, T., Kijima-Suda, I., Ido, T., Ohrui, H. and Tomita, K. (1994) Carbohydr. Res. 263, 167-172.
- [443] Nakajima, T., Hori, H., Ohrui, H., Meguro, H. and Ido, T. (1988) Agric. Biol. Chem. 52, 1209-1215.

- [444] Koppert, K. and Brossmer, R. (1992) Tetrahedron Lett. 33, 8031-8034.
- [445] Schreiner, E. and Zbiral, E. (1990) Liebigs Ann. Chem. 581-586.
- [446] David, S., Malleron, A. and Cavayé, B. (1994) Carbohydr. Res. 260, 233-241.
- [447] Schmid, W., Zbiral, E., Christian, R. and Schulz, G. (1986) Liebigs Ann. Chem. 2104-2111.
- [448] Brandstetter, H.H., Zbiral, E. and Schulz, G. (1982) Liebigs Ann. Chem. 1-13.
- [449] Wallimann, K. and Vasella, A. (1990) Helv. Chim. Acta 73, 1359-1372.
- [450] Vasella, A. and Wyler, R. (1991) Helv. Chim. Acta 74, 451-463.
- [451] Bernet, B., Murty, A.R.C.B. and Vasella, A. (1990) Helv. Chim. Acta 73, 940-958.
- [452] Glänzer, B.I., Györgydeák, Z., Bernet, B. and Vasella, A. (1991) Helv. Chim. Acta 74, 343-369.
- [453] Thomson, D.S. and Prestegard, J.H. (1990) Carbohydr. Res. 196, 206-210.
- [454] Rothermel, J. and Faillard, H. (1989) Biol. Chem. Hoppe-Seyler 370, 1077-1084.
- [455] Rothermel, J. and Faillard, H. (1990) Carbohydr. Res. 208, 251-254.
- [456] Privalova, I.M. and Khorlin, A.Ya. (1969) Izv. Akad. Nauk. SSSR, Ser. Khim. 12, 2785–2792.
- [457] Supp, M., Rose, U. and Brossmer, R. (1980) Hoppe-Seyler's Z. Physiol. Chem. 361, 338.
- [458] Tropper, F.D., Andersson, F.O., Braun, S. and Roy, R. (1992) Synthesis 618-620.
- [459] Rothermel, J., Weber, B. and Faillard, H. (1992) Liebigs Ann. Chem. 799-802.
- [460] Eschenfelder, V. and Brossmer, R. (1987) Carbohydr. Res. 162, 294-297.
- [461] Tuppy, H. and Palese, P. (1969) FEBS Lett. 3, 72-75.
- [462] Santer, U.V., Yee-Foon, J. and Glick, M.C. (1978) Biochim. Biophys. Acta 523, 435-442.
- [463] Myers, R.W., Lee, R.T., Lee, Y.C., Thomas, G.H., Reynolds, L.W. and Uchida, Y. (1980) Anal. Biochem. 101, 166–174.
- [464] Baumberger, F., Vasella, A. and Schauer, R. (1986) Helv. Chim. Acta 69, 1927-1935.
- [465] Kodama, H., Baum, L.G. and Paulson, J.C. (1991) Carbohydr. Res. 218, 111-119.
- [466] Eschenfelder, V. and Brossmer, R. (1987) Glycoconjugate J. 4, 171-178.
- [467] Fujii, I., Iwabuchi, Y., Teshima, T., Shiba, T. and Kikuchi, M. (1993) BioMed. Chem. 1, 147-149.
- [468] Warner, T.G. and Lee, L.A. (1988) Carbohydr. Res. 176, 211-218.
- [469] Roth, A. and Faillard, H. (1993) Liebigs Ann. Chem. 485-489.
- [470] Kawamura, A., Kijima-Suda, I., Sugimoto, M., Itoh, M., Takada, K., Tomita, K., Ogawa, T. and Nagai, Y. (1990) Biochim. Biophys. Acta 1033, 201–206.
- [471] Heidlas, J.E., Williams, K.W. and Whitesides, G.M. (1992) Acc. Chem. Res. 25, 307-314.
- [472] Stangier, P., Treder, W. and Thiem, J. (1993) Glycoconjugate J. 10, 26-33.
- [473] Simon, E.S., Bednarski, M.D. and Whitesides, G.M. (1988) J. Am. Chem. Soc. 110, 7159-7163.
- [474] Kittelmann, M., Klein, T., Kragl, U., Wandrey, C. and Ghisalba, O. (1992) Ann. NY Acad. Sci. 672, 444–450.
- [475] Makino, S., Ueno, Y., Ishikawa, M., Hayakawa, Y. and Hata, T. (1993) Tetrahedron Lett. 34, 2775– 2778.
- [476] Martin, T.J. and Schmidt, R.R. (1993) Tetrahedron Lett. 34, 1765-1768.
- [477] Yoshino, T. and Schmidt, R.R. (1995) Carbohydr. Lett. 1, 329-334.
- [478] Augé, C., Fernandez-Fernandez, R. and Gautheron, C. (1990) Carbohydr. Res. 200, 257-268.
- [479] Kajihara, Y., Koseki, K., Ebata, T., Kodama, H., Matsushita, H. and Hashimoto, H. (1994) Carbohydr. Res. 264, C1–C5.
- [480] Kanie, O., Nakamura, J., Kiso, M. and Hasegawa, A. (1987) J. Carbohydr. Chem. 6, 105-115.
- [481] Smalec, B. and von Itzstein, M. (1995) Carbohydr. Res. 266, 269-272.
- [482] Ikeda, K., Nagao, Y. and Achiwa, K. (1992) Carbohydr. Res. 224, 123--131.
- [483] Suzuki, J., Murakami, K. and Nishimura, Y. (1993) J. Carbohydr. Chem. 12, 201-208.
- [484] Roy, R., Laferrière, C.A., Gamian, A. and Jennings, H.J. (1987) J. Carbohydr. Chem. 6, 161-165.
- [485] Roy, R. and Laferrière, C.A. (1990) Can. J. Chem. 68, 2045-2054.
- [486] Roy, R. and Laferrière, C.A. (1988) Carbohydr. Res. 177, C1-C4.
- [487] Roy, R. and Laferrière, C.A. (1990) J. Chem. Soc. Chem. Commun. 1709-1711.
- [488] Roy, R., Tropper, F.D., Romanowska, A., Letellier, M., Cousineau, L., Meunier, S.J. and Boratynski, J. (1991) Glycoconjugate J. 8, 75–81.
- [489] Roy, R., Andersson, F.O., Harms, G., Kelm, S. and Schauer, R. (1992) Angew. Chem. Int. Ed. Engl. 31, 1478–1481.
- [490] Roy, R., Pon, R.A., Tropper, F.D. and Andersson, F.O. (1993) J. Chem. Soc. Chem. Commun. 264-265.
- [491] Roy, R., Zanini, D., Meunier, S.J. and Romanowska, A. (1994) ACS Symp. Ser. 560, 104-119.
- [492] Kuhn, R., Lutz, P. and MacDonald, D.L. (1966) Chem. Ber. 99, 611-617.
- [493] Numata, M., Sugimoto, M. Shibayama, S. and Ogawa, T. (1988) Carbohydr. Res. 174, 73-85.
- [494] Shimizu, C. and Achiwa, K. (1987) Carbohydr. Res. 166, 314-316.
- [495] Okamoto, K., Kondo, T. and Goto, T. (1986) Chem. Lett. 1449-1452.
- [496] Ito, Y. and Ogawa, T. (1987) Tetrahedron Lett. 28, 6221-6224.
- [497] Ito, Y. and Ogawa, T. (1988) Tetrahedron Lett. 29, 3987-3990.
- [498] Kirchner, E., Thiem, F., Dernick, R., Heukeshoven, J. and Thiem, J. (1988) J. Carbohydr. Chem. 7, 453–486.
- [499] Okamoto, K., Kondo, T. and Goto, T. (1986) Tetrahedron Lett. 27, 5229-5232.
- [500] Okamoto, K., Kondo, T. and Goto, T. (1988) Tetrahedron 44, 1291-1298.
- [501] Kanie, O., Kiso, M. and Hasegawa, A. (1988) J. Carbohydr. Chem. 7, 501-506.
- [502] Murase, T., Ishida, H., Kiso, M. and Hasegawa, A. (1988) Carbohydr. Res. 184, C1-C4.
- [503] Ito, Y. and Ogawa, T. (1988) Tetrahedron Lett. 29, 1061-1064.
- [504] Hasegawa, A., Nakamura, J. and Kiso, M. (1986) J. Carbohydr. Chem. 5, 11-19.
- [505] Hasegawa, A., Nakamura, J. and Kiso, M. (1986) J. Carbohydr. Chem. 5, 21-31.
- [506] Hasegawa, A., Morita, M., Ito, Y., Ishida, H. and Kiso, M. (1990) J. Carbohydr. Chem. 9, 369-392.
- [507] Bennett, S., von Itzstein, M. and Kiefel, M.J. (1994) Carbohydr. Res. 259, 293-299.
- [508] Hasegawa, A., Ogawa, H., Ishida, H. and Kiso, M. (1992) Carbohydr. Res. 224, 175-184.
- [509] Hasegawa, A., Ogawa, M., Kojima, Y. and Kiso, M. (1992) J. Carbohydr. Chem. 11, 333-341.
- [510] Ito, Y., Numata, M., Sugimoto, M. and Ogawa, T. (1989) J. Am. Chem. Soc. 111, 8508-8510.
- [511] Hasegawa, A., Ogawa, M., Ishida, H. and Kiso, M. (1990) J. Carbohydr. Chem. 9, 393-414.
- [512] Ishida, H., Ohta, Y., Tsukada, Y., Kiso, M. and Hasegawa, A. (1993) Carbohydr. Res. 246, 75-88.
- [513] Hasegawa, A., Fushimi, K., Ishida, H. and Kiso, M. (1993) J. Carbohydr. Chem. 12, 1203-1216.
- [514] Hotta, K., Ishida, H., Kiso, M. and Hasegawa, A. (1994) J. Carbohydr. Chem. 13, 175-191.
- [515] Ito, Y., Ogawa, T., Numata, M. and Sugimoto, M. (1990) Carbohydr. Res. 202, 165-175.
- [516] Fujita, S., Numata, M., Sugimoto, M., Tomita, K. and Ogawa, T. (1992) Carbohydr. Res. 228. 347-370.
- [517] Fujita, S., Numata, M., Sugimoto, M., Tomita, K. and Ogawa, T. (1994) Carbohydr. Res. 263, 181-196.
- [518] Terada, T., Kiso, M. and Hasegawa, A. (1993) J. Carbohydr. Chem. 12, 425-440.
- [519] Marra, A. and Sinaÿ, P. (1989) Carbohydr. Res. 187, 35-42.
- [520] Marra, A. and Sinaÿ, P. (1990) Carbohydr. Res. 195, 303-308.
- [521] Lönn, H. and Stenvall, K. (1992) Tetrahedron Lett. 33, 115-116.
- [522] Tropper, F.D., Andersson, F.O., Cao, S. and Roy, R. (1992) J. Carbohydr. Chem. 11, 741-750.
- [523] Birberg, W. and Lönn, H. (1991) Tetrahedron Lett. 32, 7453-7456.
- [524] Birberg, W. and Lönn, H. (1991) Tetrahedron Lett. 32, 7457-7458.
- [525] Takeda, K., Tsuboyama, K., Torii, K., Furuhata, K., Sato, N. and Ogura, H. (1990) Carbohydr. Res. 203, 57–63.
- [526] Ercegovic, T. and Magnusson, G. (1994) J. Chem. Soc. Chem. Commun. 831-832.
- [527] Martin, T.J. and Schmidt, R.R. (1992) Tetrahedron Lett. 33, 6123-6126.
- [528] Kondo, H., Ichikawa, Y. and Wong, C.-H. (1992) J. Am. Chem. Soc. 114, 8748-8750.
- [529] Kondo, H., Aoki, S., Ichikawa, Y., Halcomb, R.L., Ritzen, H. and Wong, C.-H. (1994) J. Org. Chem. 59, 864–877.
- [530] Martin, T.J., Brescello, R., Toepfer, A. and Schmidt, R.R. (1993) Glycoconjugate J. 10, 16-25.
- [531] Matsuzaki, Y., Nunomura, S., Ito, Y., Sugimoto, M., Nakahara, Y. and Ogawa, T. (1993) Carbohydr. Res. 242, C1-C6.
- [532] Ishida, H.-K., Ishida, H., Kiso, M. and Hasegawa, A. (1994) Tetrahedron: Asymmetry 5, 2493-2512.
- [533] Ishida, H., Ishida, H., Kiso, M. and Hasegawa, A. (1994) J. Carbohydr. Chem. 13, 655-664.
- [534] Abbas, S.Z., Sugiyama, S., Diakur, J., Pon, R.A. and Roy, R. (1990) J. Carbohydr. Chem. 9, 891-901.
- [535] Thiem, J. and Treder, W. (1986) Angew. Chem. Int. Ed. Engl. 25, 1096.
- [536] De Heij, H.T., Kloosterman, M., Koppen, P.L., van Boom, J.H. and van den Eijnden, D.H. (1988) J. Carbohydr. Chem. 7, 209-219.
- [537] Augé, C., Gautheron, C. and Pora, H. (1989) Carbohydr. Res. 193, 288-293.

- [538] Unverzagt, C., Kunz, H. and Paulson, J.C. (1990) J. Am. Chem. Soc. 112, 9308-9309.
- [539] Lubineau, A., Augé, C. and Francois, P. (1992) Carbohydr. Res. 228, 137-144.
- [540] Ichikawa, Y., Liu, J., Shen, G.-J. and Wong, C.-H. (1991) J. Am. Chem. Soc. 113, 6300-6302.
- [541] Ichikawa, Y., Lin, Y.-C., Dumas, D.P., Shen, G.-J., Garcia-Junceda, E., Williams, M.A., Bayer, R., Ketcham, C., Walker, L.E., Paulson, J.C. and Wong, C.-H. (1992) J. Am. Chem. Soc. 114, 9283– 9298.
- [542] Herrmann, G., Ichikawa, Y., Wandrey, C., Gaeta, F.C.A., Paulson, J.C. and Wong, C.-H. (1993) Tetrahedron Lett. 34, 3091–3094.
- [543] Kashem, M.A., Wlasichuk, K.B., Gregson, J.M. and Venot, A.P. (1993) Carbohydr. Res. 250, 129-144.
- [544] Pohlentz, G., Trimborn, M. and Egge, H. (1994) Glycobiology 4, 625-631.
- [545] Angata, T., Kitazume, S., Terada, T., Kitajima, K., Inoue, S., Troy, F.A. and Inoue, Y. (1994) Glycoconjugate J. 11, 493–499.
- [546] Sabesan, S., Duus, J.Ø., Neira, S., Domaille, P., Kelm, S., Paulson, J.C. and Bock, K. (1992) J. Am. Chem. Soc. 114, 8363–8375.
- [547] Unverzagt, C., Kelm, S. and Paulson, J.C. (1994) Carbohydr. Res. 285-301.
- [548] Hasegawa, A., Adachi, K., Yoshida, M. and Kiso, M. (1992) J. Carbohydr. Chem. 11, 95-116.
- [549] Hasegawa, A., Adachi, K., Yoshida, M. and Kiso, M. (1992) Carbohydr Res. 230, 257-272.
- [550] Vandekerckhove, F., Schenkman, S., Pontes de Carvalho, L., Tomlinson, S., Kiso, M., Yoshida, M., Hasegawa, A. and Nussenzweig, V. (1992) Glycobiology 2, 541–548.
- [551] Scudder, P., Doom, J.P., Chuenkova, M., Manger, I.D. and Pereira, M.E.A. (1993) J. Biol. Chem. 268, 9886–9891.
- [552] Ferrero-Garcia, M.A., Trombetta, S.E., Sanchez, D.O., Reglero, A., Frasch, A.C.C. and Parodi, A.J. (1993) Eur. J. Biochem. 213, 765–771.
- [553] Nishimura, S.-I., Lee, K.B., Matsuoka, K. and Lee, Y.C. (1994) Biochem. Biophys. Res. Commun. 199, 249–254.
- [554] Ito, Y. and Paulson, J.C. (1993) J. Am. Chem. Soc. 115, 7862-7863.
- [555] Lee, K.B. and Lee, Y.C. (1994) Anal. Biochem. 216, 356-364.
- [556] Maru, I., Ohta, Y., Okamoto, K., Suzuki, S., Kakehi, K. and Tsukada, Y. (1992) Biosci. Biotech. Biochem. 56, 1557–1561.
- [557] Thiem, J. and Sauerbrei, B. (1991) Angew. Chem. Int. Ed. Engl. 30, 1503-1505.
- [558] Ajisaka, K., Fujimoto, H. and Isomura, M. (1994) Carbohydr. Res. 259, 103-115.
- [559] Nagy, J.O. and Bednarski, M.D. (1991) Tetrahedron Lett. 32, 3953-3955.
- [560] Paulsen, H. and Matschulat, P. (1991) Liebigs Ann. Chem. 487-495.
- [561] Wallimann, K. and Vasella, A. (1991) Helv. Chim. Acta 74, 1520-1532.
- [562] Nagy, J.O., Wang, P., Gilbert, J.H., Schaefer, M.E., Hill, T.G., Callstrom, M.R. and Bednarski, M.D. (1992) J. Med. Chem. 35, 4501-4502.
- [563] Spevak, W., Nagy, J.O., Charych, D.H., Schaefer, M.E., Gilbert, J.H. and Bednarski, M.D. (1993) J. Am. Chem. Soc. 115, 1146–1147.
- [564] O'Connell, A.M. (1973) Acta Cryst. B 29, 2320-2328.
- [565] Kooijman, H., Kroon-Batenburg, L.M.J., Kroon, J., Breg, J.N. and de Boer, J.L. (1990) Acta Cryst. C 46, 407–410.
- [566] Czarniecki, M.F. and Thornton, E.R. (1976) J. Am. Chem. Soc. 98, 1023-1025.
- [567] Czarniecki, M.F. and Thornton, E.R. (1977) J. Am. Chem. Soc. 99, 8273-8279.
- [568] Poppe, L. and van Halbeek, H. (1991) J. Am. Chem. Soc. 113, 363-365.
- [569] Christian, R., Schulz, G., Brandstetter, H.H. and Zbiral, E. (1987) Carbohydr. Res. 162, 1-11.
- [570] Christian, R., Schreiner, E., Zbiral, E. and Schulz, G. (1989) Carbohydr. Res. 194, 49-61.
- [571] Lindon, J.C., Vinter, J.G., Lifely, M.R. and Moreno, C. (1984) Carbohydr. Res. 133, 59-74.
- [572] Michon, F., Brisson, J.-R. and Jennings, H.J. (1987) Biochemistry 26, 8399-8405.
- [573] Yamasaki, R. and Bacon, B. (1991) Biochemistry 30, 851-857.
- [574] Brisson, J.-R., Baumann, H., Imberty, A., Pérez, S. and Jennings, H.J. (1992) Biochemistry 31, 4996– 5004.
- [575] Baumann, H., Brisson, J.-R., Michon, F., Pon, R. and Jennings, H.J. (1993) Biochemistry 32, 4007– 4013.

- [576] Hughes, R.C., Sanford, B. and Jeanloz, R.W. (1972) Proc. Natl. Acad. Sci. USA 69, 942-945.
- [577] Kitazume, S., Kitajima, K., Inoue, S., Haslam, S.M., Morris, H.R., Dell, A., Lennarz, W.J. and Inoue, Y. (1996) J. Biol. Chem. 271, 6694–6701.
- [578] Zeitler, R., Banzer, J.-P., Bauer, C. and Reutter, W. (1992) BioMetals 5, 103-109.
- [579] Zeitler, R., Giannis, A., Danneschewski, S., Henk, E., Henk, T., Bauer, C., Reutter, W. and Sandhoff, K. (1992) Eur. J. Biochem. 204, 1165–1168.
- [580] Kayser, H., Zeitler, R., Kannicht, C., Grunow, D., Nuck, R. and Reutter, W. (1992) J. Biol. Chem. 267, 16934–16938.
- [581] Pels Rijcken, W.R., Overdijk, B., van den Eijnden, D.H. and Ferwerda, W. (1993) Biochem. J. 293, 207-213.
- [582] Schmelter, T., Ivanov, S., Wember, M., Stangier, P., Thiem, J. and Schauer, R. (1993) Biol. Chem. Hoppe-Seyler 374, 337–342.
- [583] Schauer, R., Haverkamp, J. and Ehrlich, K. (1980) Hoppe-Seyler's Z. Physiol. Chem. 361, 641-648.
- [584] Ambrose, M.G., Freese, S.J., Reinhold, M.S., Warner, T.G. and Vann, W.F. (1992) Biochemistry 31, 775–780.
- [585] Vimr, E.R. (1992) J. Bacteriol. 174, 6191-6197.
- [586] Vann, W.F., Silver, R.P., Abeijon, C., Chang, K., Aaronson, W., Sutton, A., Finn, C.W., Lindner, W. and Kotsatos, M. (1987) J. Biol. Chem. 262, 17556–17562.
- [587] Edwards, U. and Frosch, M. (1992) FEMS Microbiol. Lett. 96, 161-166.
- [588] Ganguli, S., Zapata, G., Wallis, T., Reid, C., Boulnois, G., Vann, W.F. and Roberts, I.S. (1994) J. Bacteriol. 176, 4583-4589.
- [589] Ichikawa, Y., Wang, R. and Wong, C.-H. (1994) Methods Enzymol. 247, 107-127.
- [590] Zapata, G., Roller, P.P., Crowley, J. and Vann, W.F. (1993) Biochem. J. 295, 485-491.
- [591] Paulson, J.C. and Colley, K.J. (1989) J. Biol. Chem. 264, 17615-17618.
- [592] Kleene, R. and Berger, E.G. (1993) Biochim. Biophys. Acta 1154, 283-325.
- [593] Basu, S.C. (1991) Glycobiology 1, 469-475.
- [594] Clausen, H., Bennett, E.P. and Dabelsteen, E. (1992) APMIS Suppl. 27, 100, 9-17.
- [595] Basu, S., Basu, M. and Basu, S.S. (1995) In: A. Rosenberg (Ed.), Biology of the Sialic Acids. Plenum Publ. Corp., New York, pp. 69–93.
- [596] Gillespie, W.M., Kelm, S. and Paulson, J.C. (1992) J. Biol. Chem. 267, 21004-21010.
- [597] Lee, Y.-C., Kurosawa, N., Hamamoto, T., Nakaoka, T. and Tsuji, S. (1993) Eur. J. Biochem. 216, 377–385.
- [598] Kurosawa, N., Hamamoto, T., Inoue, M. and Tsuji, S. (1995) Biochim. Biophys. Acta 1244, 216-222.
- [599] Lee, Y.-C., Kojima, N., Wada, E., Kurosawa, N., Nakaoka, T., Hamamoto, T. and Tsuji, S. (1994) J. Biol. Chem. 269, 10028–10033.
- [600] Kojima, N., Lee, Y.-C., Hamamoto, T., Kurosawa, N. and Tsuji, S. (1994) Biochemistry 33, 5772–5776.
- [601] Wen, D.X., Livingston, B.D., Medzihradszky, K.F., Kelm, S., Burlingame, A.L. and Paulson, J.C. (1992)
 J. Biol. Chem. 267, 21011–21019.
- [602] Kitagawa, H. and Paulson, J.C. (1993) Biochem. Biophys. Res. Commun. 194, 375-382.
- [603] Kono, M., Ohyama, Y., Lee, Y.-C., Hamamoto, T., Kojima, N. and Tsuji, S. (1997) Glycobiology 7, 469–479.
- [604] Sasaki, K., Watanabe, E., Kawashima, K., Sekine, S., Dohi, T., Oshima, M., Hanai, N., Nishi, T. and Hasegawa, M. (1993) J. Biol. Chem. 268, 22782–22787.
- [605] Kitagawa, H. and Paulson, J.C. (1994) J. Biol. Chem. 269, 1394-1401.
- [606] Weinstein, J., Lee, E.U., McEntee, K., Lai, P.-H., Paulson, J.C. (1987) J. Biol. Chem. 262, 17735– 17743.
- [607] Grundmann, U., Nerlich, C., Rein, T., and Zettlmeissl, G. (1990) Nucl. Acid Res. 18, 667.
- [608] Hamamoto, T., Kawasaki, M., Kurosawa, N., Nakaoka, T., Lee, Y.-C. and Tsuji, S. (1993) Bioorg. Med. Chem. 1, 141–145.
- [609] Kurosawa, N., Kawasaki, M., Hamamoto, T., Nakaoka, T., Lee, Y.-C., Arita, M. and Tsuji, S. (1994) Eur. J. Biochem. 219, 375–381.
- [610] Kurosawa, N., Hamamoto, T., Lee, Y.-C., Nakaoka, T., Kojima, N. and Tsuji, S. (1994) J. Biol. Chem. 269, 1402–1409.

- [611] Kurosawa, N., Kojima, N., Inoue, M., Hamamoto, T. and Tsuji, S. (1994) J. Biol. Chem. 269, 19048– 19053.
- [612] Soutiropoulou, G., Anisowicz, A. and Sager, R. (1994) GenBank(TM)/EMBL Data Library, accession number U14550.
- [612a] Kurosawa, N., Inoue, M., Yoshida, Y. and Tsuji, S. (1996) J. Biol. Chem. 271, 15109-15116.
- [613] Sasaki, K., Kurata, K., Kojima, N., Kurosawa, N., Ohta, S., Hanai, N., Tsuji, S. and Nishi, T. (1994)
 J. Biol. Chem. 269, 15950–15956.
- [614] Nara, K., Watanabe, Y., Maruyama, K., Kasahara, K., Nagai, Y. and Sanai, Y. (1994) Proc. Natl. Acad. Sci. USA 91, 7952–7956.
- [615] Haraguchi, M., Yamashiro, S., Yamamoto, A., Furukawa, K., Takamiya, K., Lloyd, K.O., Shiku, H. and Furukawa, K. (1994) Proc. Natl. Acad. Sci. USA 91, 10455–10459.
- [616] Yoshida, Y., Kojima, N., Kurosawa, N., Hamamoto, T. and Tsuji, S. (1995) J. Biol. Chem. 270, 14628– 14633.
- [617] Livingston, B.D. and Paulson, J.C. (1993) J. Biol. Chem. 268, 11504-11507.
- [618] Kojima, N., Yoshida, Y., Kurosawa, N., Lee, Y.-C. and Tsuji, S. (1995) FEBS Lett. 360, 1-4.
- [619] Kitagawa, H. and Paulson, J.C. (1994) J. Biol. Chem. 269, 17872-17878.
- [620] Eckhardt, M., Mühlenhoff, M., Bethe, A., Koopman, J., Frosch, M. and Gerardy-Schahn, R. (1995) Nature 373, 715–718.
- [621] Yoshida, Y., Kojima, N. and Tsuji, S. (1995) J. Biochem. 118, 658-664.
- [621a] Kono, M., Yoshida, Y., Kojima, N. and Tsuji, S. (1996) J. Biol. Chem. 271, 29366-29371.
- [622] Weisgerber, C., Hansen, A. and Frosch, M. (1991) Glycobiology 1, 357-365.
- [623] Steenbergen, S.M., Wrona, T.J. and Vimr, E.R. (1992) J. Bacteriol. 174, 1099-1108.
- [623a] Tsuji, S., Datta, A.K. and Paulson, J.C. (1996) Glycobiology 6, V-VII.
- [624] Sabesan, S., Paulson, J.C. and Weinstein, J. (1994) Methods Enzymol. 247, 237-243.
- [625] Spiegel, L.B., Hadjimichael, J. and Rossomando, E.F. (1992) J. Chromatogr. 573, 23-27.
- [626] Gasnier, F., Baubichon-Cortay, H., Louisot, P. and Gateau-Roesch, O. (1991) J. Biochem. 110, 702– 707.
- [627] Nemansky, M. and van den Eijnden, D.H. (1992) Biochem. J. 287, 311-316.
- [628] Nemansky, M., Schiphorst, W.E.C.M., Koeleman, C.A.M. and van den Eijnden, D.H. (1992) FEBS Lett. 312, 31-36.
- [629] Wlasichuk, K.B., Kashem, M.A., Nikrad, P.V., Bird, P., Jiang, C. and Venot, A.P. (1993) J. Biol. Chem. 268, 13971–13977.
- [630] Hokke, C.H., van der Ven, J.G.M., Kamerling, J.P. and Vliegenthart, J.F.G. (1993) Glycoconjugate J. 10, 82–90.
- [630a] Van Dorst, J.A.L.M., Tikkanen, J.M., Krezdorn, C.H., Streift, M.B., Berger, E.G., van Kuik, J.A., Kamerling, J.P. and Vliegenthart, J.F.G. (1996) Eur. J. Biochem. 242, 674–681.
- [631] Fast, D.G., Jamieson, J.C. and McCaffrey, G. (1993) Biochim. Biophys. Acta 1202, 325-330.
- [632] Bosshart, H. and Berger, E.G. (1992) Eur. J. Biochem. 208, 341-349.
- [633] Nemansky, M., Edzes, H.T., Wijnands, R.A. and van den Eijnden, D.H. (1992) Glycobiology 2, 109– 117.
- [634] Reboul, P., George, P., Geoffroy, J., Louisot, P. and Broquet, P. (1992) Biochem. Biophys. Res. Commun. 186, 1575–1581.
- [635] Le Marer, N., Laudet, V., Svensson, E.C., Cazlaris, H., Van Hille, B., Lagrou, C., Stehelin, D., Montreuil, J., Verbert, A. and Delannoy, P. (1992) Glycobiology 2, 49-56.
- [636] Yogeeswaran, G., Fujinami, R., Kiessling, R. and Welsh, R.M. (1982) Virology 121, 363-371.
- [637] Kolínská, J., Ivanov, S. and Chelibonova-Lorer, H. (1988) FEBS Lett. 242, 57-60.
- [638] Vandamme, V., Pierce, A., Verbert, A. and Delannoy, P. (1993) Eur. J. Biochem. 211, 135-140.
- [639] Hanasaki, K., Varki, A., Stamenkovic, I. and Bevilacqua, M.P. (1994) J. Biol. Chem. 269, 10637-10643.
- [640] Uehara, F., Ohba, N., Nakashima, Y., Yanagita, T., Ozawa, M. and Muramatsu, T. (1993) Exp. Eye Res. 56, 89–93.
- [641] Kitazume, S., Kitajima, K., Inoue, S., Inoue, Y. and Troy II, F.A. (1994) J. Biol. Chem. 269, 10330– 10340.

390

- [642] Schmelter, T. (1991) Doctoral thesis, University of Kiel.
- [642a] Kleineidam, R.G., Schmelter, T., Schwarz, R.T. and Schauer, R. (1997) Glycoconjugate J. 14, 57-66.
- [643] Cambron, L.D. and Leskawa, K.C. (1993) Biochem. Biophys. Res. Commun. 193, 585-590.
- [644] Kajihara, Y., Hashimoto, H., Kodama, H., Wakabayashi, T. and Sato, K.-i. (1993) J. Carbohydr. Chem. 12, 991–995.
- [645] Van den Eijnden, D.H. and Schiphorst, W.E.C.M. (1988) In: R. Schauer and T. Yamakawa (Eds.), Sialic Acids – Proc. Jap.–German Symp. Verlag Wissenschaft und Bildung, Kiel, pp. 128–129.
- [646] Albarracin, I., Lassaga, F.E. and Caputto, R. (1988) Biochem. J. 254, 559-565.
- [647] Paulson, J.C., Rearick, J.I. and Hill, R.L. (1977) J. Biol. Chem. 252, 2363-2371.
- [648] Kijima-Suda, I., Miyamoto, Y., Toyoshima, S., Itoh, M. and Osawa, T. (1986) Cancer Res. 46, 858– 862.
- [649] Kijima-Suda, I., Miyazawa, T., Itoh, M., Toyoshima, S. and Osawa, T. (1988) Cancer Res. 48, 3728– 3732.
- [650] Harvey, B.E. and Thomas, P. (1993) Biochem. Biophys. Res. Commun. 190, 571-575.
- [651] Berger, E.G., Thurnher, M. and Müller, U. (1987) Exp. Cell Res. 173, 267-273.
- [652] Roth, J. (1987) Biochim. Biophys. Acta 906, 405-436.
- [653] Gleeson, P.A., Teasdale, R.D. and Burke, J. (1994) Glycoconjugate J. 11, 381-394.
- [654] Lepers, A., Shaw, L., Schneckenburger, P., Cacan, R., Verbert, A. and Schauer, R. (1990) Eur. J. Biochem. 193, 715–723.
- [655] Hayes, B.K., Freeze, H.H. and Varki, A. (1993) J. Biol. Chem. 268, 16139-16154.
- [656] Hayes, B.K. and Varki, A. (1993) J. Biol. Chem. 268, 16155-16169.
- [657] Dahdal, R.Y. and Colley, K.J. (1993) J. Biol. Chem. 268, 26310-26319.
- [658] Edwards, U., Müller, A., Hammerschmidt, S., Gerardy-Schahn, R. and Frosch, M. (1994) Mol. Microbiol. 14, 141–149.
- [659] Gottschalk, A. (Ed.) (1960) The Chemistry and Biology of Sialic Acids and Related Substances. University Press, Cambridge.
- [660] Schauer, R., Kelm, S., Reuter, G., Roggentin, P. and Shaw, L. (1995) In: A. Rosenberg (Ed.), Biology of the Sialic Acids. Plenum Publ. Corp. New York, pp. 7–67.
- [661] Kawai, T., Kato, A., Higashi, A., Kato, S. and Naiki, M. (1991) Cancer Res. 51, 1242-1246.
- [662] Schauer, R., Reuter, G., Mühlpfordt, H., Andrade, A.F.B. and Pereira, M.E.A. (1983) Hoppe-Seyler's Z. Physiol. Chem. 364, 1053–1057.
- [663] Schauer, R. (1985) Trends Biochem. Sci. 10, 357-360.
- [664] Kelm, S., Schauer, R., Manuguerra, J.-C., Gross, H.-J. and Crocker, P.R. (1994) Glycoconjugate J. 11, 576–585.
- [665] Kyogashima, M., Ginsburg, V. and Krivan, H.C. (1989) Arch. Biochem. Biophys. 270, 391-397.
- [666] Maget-Dana, R., Veh, R.W., Sander, M., Roche, A.-C., Schauer, R. and Monsigny, M. (1981) Eur. J. Biochem. 114, 11-16.
- [667] Ouadia, A., Karamanos, Y. and Julien, R. (1992) Glycoconjugate J. 9, 21-26.
- [668] Schauer, R. (1988) In: A.M. Wu (Ed.), The Molecular Immunology of Complex Carbohydrates, Adv. Exp. Med. Biol. Vol. 228. Plenum Press, New York, pp. 47–72.
- [669] Noguchi, A., Mukuria, C.J., Suzuki, E. and Naiki, M. (1995) J. Biochem. 117, 59-62.
- [670] Asaoka, H., Nishinaka, S., Wakamiya, N., Matsuda, H. and Murata, M. (1992) Immunol. Lett. 32, 91–96.
- [671] Griot-Wenk, M., Pahlsson, P., Chisholm-Chait, A., Spitalnik, P.F., Spitalnik, S.L. and Giger, U. (1993) Animal Genet. 24, 401–407.
- [672] Hashimoto, Y., Yamakawa, T. and Tanabe, Y. (1984) J. Biochem. 96, 1777-1782.
- [673] Andrews, G.A., Chavey, P.S., Smith, J.E. and Rich, L. (1992) Blood 79, 2485-2491.
- [674] Muchmore, E.A. (1992) Glycobiology 2, 337-343.
- [675] Bergwerff, A.A., Hulleman, S.H.D., Kamerling, J.P., Vliegenthart, J.F.G., Shaw, L., Reuter, G. and Schauer, R. (1993) In: J. Roth, U. Rutishauser and F.A. Troy II (Eds.), Polysialic Acid. Birkhäuser Verlag, Basel, pp. 201–212.
- [676] Sugita, M. (1979) J. Biochem. 86, 765-772.

- [677] Watarai, S., Kushi, Y., Shigeto, R., Misawa, N., Eishi, Y., Handa, S. and Yasuda, T. (1995) J. Biochem. 117, 1062–1069.
- [678] Schauer, R., Schoop, H.J. and Faillard, H. (1968) Hoppe-Seyler's Z. Physiol. Chem. 349, 645-652.
- [679] Schoop, H.J., Schauer, R. and Faillard, H. (1969) Hoppe-Seyler's Z. Physiol. Chem. 350, 155-162.
- [680] Schlenzka, W., Shaw, L. and Schauer, R. (1994) Glycobiology 4, 675-683.
- [681] Schneckenburger, P., Shaw, L. and Schauer, R. (1994) Glycoconjugate J. 11, 194-203.
- [682] Kawano, T., Kozutsumi, Y., Kawasaki, T. and Suzuki, A. (1994) J. Biol. Chem. 269, 9024–9029.
- [683] Schlenzka, W., Shaw, L. and Schauer, R. (1993) Biochim. Biophys. Acta 1161, 131-138.
- [684] Shaw, L. and Schauer, R. (1989) Biochem. J. 263, 355-363.
- [685] Takematsu, H., Kawano, T., Koyama, S., Kozutsumi, Y., Suzuki, A. and Kawasaki, T. (1994) J. Biochem. 115, 381–386.
- [686] Schlenzka, W., Shaw, L., Schauer, R., Schmidt, C.L., Bill, E. and Trautwein, A.X. (1994) J. Inorg. Biochem. 56, 29.
- [687] Kawano, T., Koyama, S., Takematsu, H., Kozutsumi, Y., Kawasaki, H., Kawashima, S., Kawasaki, T. and Suzuki, A. (1995) J. Biol. Chem. 270, 16458–16463.
- [688] Schlenzka, W., Shaw, L., Kelm, S., Schmidt, C.L., Bill, E., Trautwein, A.X., Lottspeich, F. and Schauer, R. (1996) FEBS Lett. 385, 197–200.
- [689] Mukuria, C.J., Mwangi, W.D., Noguchi, A., Waiyaki, G.P., Asano, T. and Naiki, M. (1995) Biochem. J. 305, 459-464.
- [690] Schauer, R. (1991) Glycobiology 1, 449-452.
- [691] Malykh, Ya.N., Dyatlovitskaya, É.V., Lemenovskaya, A.F., Sorokin, M.M., Indzhiya, L.V., Yakovleva, L.A. and Bergel'son, L.D. (1991) Biochemistry 56, 723-728.
- [692] Müthing, J., Steuer, H., Peter-Katalinić, J., Marx, U., Bethke, U., Neumann, U. and Lehmann, J. (1994) J. Biochem. 116, 64–73.
- [693] Hubbard, S.C., Walls, L., Ruley, H.E. and Muchmore, E.A. (1994) J. Biol. Chem. 269, 3717-3724.
- [694] Sjoberg, E.R., Manzi, A.E., Khoo, K.-H., Dell, A. and Varki, A. (1992) J. Biol. Chem. 267, 16200– 16211.
- [695] Campbell, F., Appleton, M.A.C., Fuller, C.E., Greeff, M.P., Hallgrimsson, J., Katoh, R., Ng, O.L.I., Satir, A., Williams, G.T. and Williams, E.D. (1994) J. Pathol. 174, 169–174.
- [696] Engstler, M., Hubl, U. and Schauer, R. (1995) Glycoconjugate J. 12, 520.
- [697] Kiehne, K. and Schauer, R. (1992) Biol. Chem. Hoppe-Seyler 373, 1117-1123.
- [698] Pacitti, A.F. and Gentsch, J.R. (1987) J. Virol. 61, 1407-1415.
- [699] Higa, H.H., Rogers, G.N. and Paulson, J.C. (1985) Virology 144, 279-282.
- [700] Klotz, F.W., Orlandi, P.A., Reuter, G., Cohen, S.J., Haynes, J.D., Schauer, R., Howard, R.J., Palese, P. and Miller, L.H. (1992) Mol. Biochem. Parasitol. 51, 49–54.
- [701] Varki, A. and Kornfeld, S. (1980) J. Exp. Med. 152, 532-544.
- [702] Koistinen, V. (1992) Biochem. J. 283, 317-319.
- [703] Shugaba, A., Umar, I., Omage, J., Ibrahim, N.D.G., Andrews, J., Ukoha, A.I., Saror, D.I. and Esievo, K.A.N. (1994) J. Comp. Pathol. 110, 91–95.
- [704] Larkin, M., Knapp, W., Stoll, M.S., Mehmet, H. and Feizi, T. (1991) Clin. Exp. Immunol. 85, 536-541.
- [705] Herrler, G., Rott, R., Klenk, H.-D., Müller, H.-P., Shukla, A.K. and Schauer, R. (1985) EMBO J. 4, 1503–1506.
- [706] Schultze, B., Gross, H.-J., Brossmer, R. and Herrler, G. (1991) J. Virol. 65, 6232-6237.
- [707] Schultze, B., Gross, H.-J., Brossmer, R., Klenk, H.-D. and Herrler, G. (1990) Virus Res. 16, 185-194.
- [708] Frasa, H., Procee, J., Torensma, R., Verbruggen, A., Algra, A., Rozenberg-Arska, M., Kraaijeveld, K. and Verhoef, J. (1993) J. Clin. Microbiol. 31, 3174–3178.
- [709] Zeng, F.Y. and Gabius, H.-J. (1992) Z. Naturforsch. 47c, 641-653.
- [710] Ravindranath, M.H. and Paulson, J.C. (1987) Methods Enzymol. 138, 520-527.
- [711] Sen, G. and Mandal, C. (1995) Carbohydrate Res. 268, 115-125.
- [712] Herrler, G., Reuter, G., Rott, R., Klenk, H.-D. and Schauer, R. (1987) Biol. Chem. Hoppe-Seyler 368, 451–454.
- [713] Chou, D.K.H., Flores, S., Jungalwala, F.B. (1990) J. Neurochem. 54, 1598-1607.
- [714] Varki, A., Hooshmand, F., Diaz, S., Varki, N.M. and Hedrick, M. (1991) Cell 65, 65-74.

- [715] Sen, G., Chowdhury, M. and Mandal, C. (1994) Mol. Cell. Biochem. 136, 65-70.
- [716] Reivinen, J., Holthöfer, H. and Miettinen, A. (1994) Int. Immunol. 6, 1409-1416.
- [717] Cheresh, D.A., Reisfeld, R.A. and Varki, A.P. (1984) Science 225, 844-846.
- [718] Paller, A.S., Arnsmeier, S.L., Robinson, J.K. and Bremer, E.G. (1992) J. Invest. Dermatol. 98, 226– 232.
- [719] Hutchins, J.T. and Reading, C.L. (1988) J. Cell. Biochem. 37, 37-48.
- [720] Huang, C.-B., Xu, J., Wu, K.-R. and Wang, M.-W. (1988) Anticancer Res. 8, 829-832.
- [721] Blum, A.S. and Barnstable, C.J. (1987) Proc. Natl. Acad. Sci. USA 84, 8716-8720.
- [722] Jass, J.R. and Roberton, A.M. (1994) Pathol. Intern. 44, 487-504.
- [723] Agawa, S. and Jass, J.R. (1990) J. Clin. Pathol. 43, 527-532.
- [724] Rahmann, H., Hilbig, R., Marx, J., Beitinger, H. and Mehlfeld, R. (1987) J. Therm. Biol. 12, 81-85.
- [725] Suzuki, Y., Nakao, T., Ito, T., Watanabe, N., Toda, Y., Guiyun, X., Suzuki, T., Kobayashi, T., Kimura, Y., Yamada, A., Sugawara, K., Nishimura, H., Kitame, F., Nakamura, K., Deya, E., Kiso, M. and Hasegawa, A. (1992) Virology 189, 121–131.
- [726] Reuter, G., Schauer, R. and Bumm, P. (1988) In: R. Schauer and T. Yamakawa (Eds.), Sialic Acids, Proc. Jap.-German Symp. Verlag Wissenschaft und Bildung, Kiel, pp. 258-259.
- [727] Harms, G., Corfield, A.P., Schauer, R. and Reuter, G. (1993) Biol. Chem. Hoppe-Seyler 374, 946-947.
- [728] Herrler, G. and Klenk, H.-D. (1991) Adv. Virus Res. 40, 213-234.
- [729] Vlasak, R., Krystal, M., Nacht, M. and Palese, P. (1987) Virology 160, 419-425.
- [730] Harms, G., Reuter, G., Corfield, A.P. and Schauer, R. (1996) Glycoconjugate J. 13, 621-630.
- [730a] Klein, A., Krishna, M., Varki, N.M. and Varki, A. (1994) Proc. Natl. Acad. Sci. USA 91, 7782-7786.
- [731] Schauer, R. (1970) Hoppe-Seyler's Z. Physiol. Chem. 351, 595-602.
- [732] Schauer, R. (1970) Hoppe-Seyler's Z. Physiol. Chem. 351, 749-758.
- [733] Schauer, R., Casals-Stenzel, J., Corfield, A.P. and Veh, R.W. (1988) Glycoconjugate J. 5, 257-270.
- [734] Higa, H.H., Butor, C., Diaz, S. and Varki, A. (1989) J. Biol. Chem. 264, 19427-19434.
- [735] Butor, C., Diaz, S. and Varki, A. (1993) J. Biol. Chem. 268, 10197-10206.
- [736] Sjoberg, E.R. and Varki, A. (1993) J. Biol. Chem. 268, 10185-10196.
- [737] Corfield, A.P., Wagner, S.A., Safe, A., Mountford, R.A., Clamp, J.R., Kamerling, J.P., Vliegenthart, J.F.G. and Schauer, R. (1993) Clin. Sci. 84, 573–579.
- [738] Schauer, R. and Wember, M. (1985) In: E.A. Davidson, J.C. Williams and N.M. Di Ferrante (Eds.), Glycoconjugates, Proc. 8th Int. Symp., Vol. 1. Praeger, New York, pp. 266–267.
- [739] De Freese, A., Shaw, L., Reuter, G. and Schauer, R. (1993) Glycoconjugate J. 10, 330.
- [740] Inoue, Y. (1993) In: J. Roth, U. Rutishauser and F.A. Troy II (Eds.), Polysialic Acid. Birkhäuser Verlag, Basel, pp. 171–181.
- [741] Strecker, G., Wieruszeski, J.-M., Plancke, Y. and Michalski, J.-C. (1993) Glycoconjugate J. 10, 345.
- [742] Sander, M., Veh, R.W. and Schauer, R. (1979) In: R. Schauer, P. Boer, E. Buddecke, M.F. Kramer, J.F.G. Vliegenthart and H. Wiegandt (Eds.), Glycoconjugates, Proc. 5th Int. Symp. on Glycoconjugates. Thieme, Stuttgart, pp. 358–359.
- [743] Schauer, R., Reuter, G., Stoll, S., Posadas del Rio, F., Herrler, G. and Klenk, H.-D. (1988) Biol. Chem. Hoppe-Seyler 369, 1121–1130.
- [744] Schauer, R., Reuter, G., Stoll, S. and Shukla, A.K. (1989) J. Biochem. 106, 143-150.
- [745] Butor, C., Higa, H.H. and Varki, A. (1993) J. Biol. Chem. 268, 10207-10213.
- [746] Higa, H.H., Manzi, A. and Varki, A. (1989) J. Biol. Chem. 264, 19435-19442.
- [747] Varki, A., Muchmore, E. and Diaz, S. (1986) Proc. Natl. Acad. Sci. USA 83, 882-886.
- [748] Corfield, A.P., Wagner, S.A., O'Donnell, L.J.D., Durdey, P., Mountford, R.A. and Clamp, J.R. (1993) Glycoconjugate J. 10, 72–81.
- [749] García-Sastre, A., Villar, E., Manuguerra, J.C., Hannoun, C. and Cabezas, J.A. (1991) Biochem. J. 273, 435–441.
- [750] Herrler, G., Gross, H.J., Imhof, A., Brossmer, R., Milks, G. and Paulson, J.C. (1992) J. Biol. Chem. 267, 12501–12505.
- [751] Manzi, A.E., Sjoberg, E.R., Diaz, S. and Varki, A. (1990) J. Biol. Chem. 265, 13091-13103.
- [752] Muñoz-Barroso, I., García-Sastre, A., Villar, E., Manuguerra, J.-C., Hannoun, C. and Cabezas, J.A. (1992) Virus Res. 25, 145–153.

- [753] Rosenberg, A. and Schengrund, C.-L. (1976) In: A. Rosenberg and C.-L. Schengrund (Eds.), Biological Roles of Sialic Acid. Plenum Press, New York, pp. 295–359.
- [754] Müller, H.E. (1974) Behring Inst. Mitt. 55, 34-56.
- [755] Engstler, M., Reuter, G. and Schauer, R. (1993) Mol. Biochem. Parasitol. 61, 1-14.
- [756] Colman, P.M. (1994) Prot. Sci. 3, 1687-1696.
- [757] Long, G.S., Taylor, P.W. and Luzio, J.P. (1993) In: J. Roth, U. Rutishauser and F.A. Troy II (Eds.), Polysialic Acid. Birkhäuser Verlag, Basel, pp. 137–144.
- [758] Deijl, C.M., Kamerling, J.P. and Vliegenthart, J.F.G. (1984) Carbohydr. Res. 126, 338-342.
- [759] Guo, X. and Sinnott, M.L. (1993) Biochem. J. 296, 291-292.
- [760] Guo, X., Laver, W.G., Vimr, E. and Sinnott, M.L. (1994) J. Am. Chem. Soc. 116, 5572-5578.
- [761] Bossart-Whitaker, P., Carson, M., Babu, Y.S., Smith, C.D., Laver, W.G. and Air, G.M. (1993) J. Mol. Biol. 232, 1069–1083.
- [762] Janakiraman, M.N., White, C.L., Laver, W.G., Air, G.M. and Luo, M. (1994) Biochemistry 33, 8172– 8179.
- [763] Schauer, R., Shukla, A.K., Schröder, C. and Müller, E. (1984) Pure Appl. Chem. 56, 907-921.
- [764] Kwiatkowski, B., Boscheck, B., Thiele, H. and Stirm, S. (1982) J. Virol. 43, 697-704.
- [765] Rougon, G. (1993) Eur. J. Cell Biol. 61, 197-207.
- [766] Li, Y.-T., Yuziuk, J.A., Li, S.-C., Nematalla, A., Hasegawa, A., Tsutsumi, M. and Nakagawa, H. (1993) Glycobiology 10, 525.
- [767] Kean, E.L. and Bighouse, K.J. (1974) J. Biol. Chem. 249, 7813-7823.
- [768] Vimr, E.R. (1994) Trends Microbiol. 2, 271-277.
- [769] Roggentin, P., Rothe, B., Lottspeich, F. and Schauer, R. (1988) FEBS Lett. 238, 31-34.
- [769a] Chien, C.H., Shann, Y.J. and Shen, S.Y. (1996) Enzyme Microbiol. Technol. 19, 267-276.
- [770] Galen, J.E., Ketley, J.M, Fasano, A., Richardson, S.H., Wasserman, S.S. and Kaper, J.B. (1992) Infect. Immun. 60, 406–415.
- [771] Rothe, B., Roggentin, P., Frank, R., Blöcker, H. and Schauer, R. (1989) J. Genet. Microbiol. 135, 3087–3096.
- [772] Hoyer, L.L., Hamilton, A.C., Steenbergen, S.M. and Vimr, E.R. (1992) Mol. Microbiol. 6, 873-884.
- [773] Roggentin, P., Rothe, B., Kaper, J.B., Galen, J., Lawrisuk, L., Vimr, E.R. and Schauer, R. (1989) Glycoconjugate J. 6, 349-353.
- [774] Miyagi, T., Konno, K., Emori, Y., Kawasaki, H., Suzuki, K., Yasui, A. and Tsuiki, S. (1993) J. Biol. Chem. 268, 26435–26440.
- [775] Ferrari, J., Harris, R. and Warner, T.G. (1994) Glycobiology 4, 367-373.
- [776] Pereira, M.E.A., Mejia, J.S., Ortega-Barria, E., Matzilevitch, D. and Prioli, R.P. (1991) J. Exp. Med. 174, 179–191.
- [777] Buschiazzo, A., Cremona, M.L., Campetella, O., Frasch, A.C.C. and Sanchez, D.D. (1993) Mol. Biochem. Parasitol. 62, 115–116.
- [778] Henningsen, M., Roggentin, P. and Schauer, R. (1991) Biol. Chem. Hoppe-Seyler 372, 1065-1072.
- [779] Yeung, M.K. (1993) Infect. Immun. 61, 109-116.
- [780] Akimoto, S., Ono, T., Tsutsui, H., Kinouchi, T., Kataoka, K. and Ohnishi, Y. (1994) Biochem. Biophys. Res. Commun. 203, 914–921.
- [781] Rothe, B., Rothe, B., Roggentin, P. and Schauer, R. (1991) Mol. Genet. 226, 190-197.
- [782] Traving, C., Schauer, R. and Roggentin, P. (1994) Glycoconjugate J. 11, 141-151.
- [783] Sakurada, K., Ohta, T. and Hasegawa, M. (1992) J. Bacteriol. 174, 6896-6903.
- [784] Cámara, M., Boulnois, G.J., Andrew, P.W. and Mitchell, T.J. (1994) Infect. Immun. 62, 3688–3695.
- [785] Galen, J.E., Vimr, E.R., Lawrisuk, L. and Kaper, J.B. (1990) In: R.B. Sack and Y. Zinnaka (Eds.), Advances in Research on Cholera and Related Diarrheas. Vol. 7. KTK Scientific Publishers, Tokyo, pp. 143–153.
- [786] Air, G.M., Webster, R.G., Colman, P.M. and Laver, W.G. (1987) Virology 160, 346-354.
- [787] Bryant, J.M., Luzio, J.P. and Taylor, P. (1995) Glycoconjugate J. 12, 509.
- [788] Prioli, R.P., Ortega-Barria, E., Mejia, J.S. and Pereira, M.E.A. (1992) Mol. Biochem. Parasitol. 52, 85–96.
- [789] Roggentin, T., Kleineidam, R.G., Schauer, R. and Roggentin, P. (1992) Glycoconjugate J. 9, 235-240.

- [790] Crennell, S.J., Garman, E.F., Laver, W.G., Vimr, E.R. and Taylor, G.L. (1993) Proc. Natl. Acad. Sci. USA 90, 9852–9856.
- [791] Crennell, S., Garman, E., Laver, G., Vimr, E. and Taylor, G. (1994) Structure 2, 535-544.
- [792] Varghese, J.N., Laver, W.G. and Colman, P.M. (1983) Nature 303, 35-40.
- [793] Colman, P.M., Varghese, J.N. and Laver, W.G. (1983) Nature 303, 41-44.
- [794] Varghese, J.N., McKimm-Breschkin, J.L., Caldwell, J.B., Kortt, A.A. and Colman, P.M. (1992) Proteins 14, 327–332.
- [795] Corfield, A.P., Lambré, C.R., Michalski, J.-C. and Schauer, R. (1992) Conférences Philippe Laudat 1991. INSERM, Paris, pp. 111–175.
- [796] Simpson, N.E. and Cann, H.M. (1991) Cytogenet. Cell Genet. 58, 428-458.
- [797] Schauer, R. and Vliegenthart, J.F.G. (1982) In: R. Schauer (Ed.), Sialic Acids Chemistry, Metabolism and Function. Springer Verlag, Wien, New York, pp. 1–3.
- [798] Schauer, R. and Wember, M. (1989) Biol. Chem. Hoppe-Seyler 370, 183-190.
- [799] Roggentin, P., Schauer, R., Hoyer, L.L. and Vimr, E.R. (1993) Mol. Microbiol. 9, 915-921.
- [800] Canard, B., and Cole, S.T. (1990) FEMS Lett. 66, 323-326.
- [801] Pereira, M.E.A. (1983) Science 219, 1444-1446.
- [802] Prioli, R.P., Mejia, J.S., Aji, T., Aikawa, M. and Pereira, M.E.A. (1991) Trop. Med. Parasitol. 42, 146-150.
- [803] Previato, J.O., Andrade, A.F.B., Pessolani, M.C.V. and Mendonça-Previato, L. (1985) Mol. Biochem. Parasitol. 16, 85-96.
- [804] Zingales, B., Carniol, C., de Lederkremer, R. and Colli, W. (1987) Mol. Biochem. Parasitol. 26, 135– 144.
- [805] Schenkman, S., Jiang, M.-S., Hart, G.W. and Nussenzweig, V. (1991) Cell 65, 1117-1125.
- [806] Engstler, M., Schauer, R. and Brun, R. (1995) Acta Tropica 59, 117-129.
- [807] Medina-Acosta, E., Paul, S., Tomlinson, S. and Pontes-de-Carvalho, L.C. (1994) Mol. Biochem. Parasitol. 64, 273-282.
- [808] Esievo, K.A.N. (1979) Proc. 16th Meeting of OAU/STRC International Council for Trypanosomiasis Research and Control, Yaunde, Cameroun, pp. 205–210.
- [809] Reuter, G., Schauer, R., Prioli, R. and Pereira, M.E.A. (1987) Glycoconjugate J. 4, 339-348.
- [810] Engstler, M. and Schauer, R. (1993) Parasitol. Today 9, 222-225.
- [811] Medina-Acosta, E., Franco, A.M.R., Jansen, A.M., Sampol, M., Nevés, N., Pontes-de-Carvalho, L., Grimaldi Jr., G. and Nussenzweig, V. (1994) Eur. J. Biochem. 225, 333–339.
- [812] Pontes de Carvalho, L.C., Tomlinson, S., Vandekerckhove, F., Bienen, E.J., Clarkson, A.B., Jiang, M.-S., Hart, G.W. and Nussenzweig, V. (1993) J. Exp. Med. 177, 465–474.
- [813] Engstler, M., Ferrero-García, M.A., Parodi, A.J., Schauer, R., Storz-Eckerlin, T., Vasella, A., Witzig, C. and Zhu, X. (1994) Helv. Chim. Acta 77, 1166–1174.
- [814] Parodi, A.J., Pollevick, G.D., Mautner, M., Buschiazzo, A., Sanchez, D.O. and Frasch, A.C.C. (1992) EMBO J. 11, 1705–1710.
- [815] Cross, G.A.M. and Takle, G.B. (1993) Annu. Rev. Microbiol. 47, 385-411.
- [816] Uemura, H., Schenkman, S., Nussenzweig, V. and Eichinger, D. (1992) EMBO J. 11, 3837-3844.
- [817] Ruef, B.J., Dawson, B.D., Tewari, D., Fouts, D.L. and Manning, J.E. (1994) Mol. Biochem. Parasitol. 63, 109–120.
- [818] Colli, W. (1993) FASEB J. 7, 1257-1264.
- [819] Cremona, M.L., Sánchez, D.O., Frasch, A.C.C. and Campetella, O. (1995) Gene 160, 123-128.
- [820] Schenkman, S., Chaves, L.B., Pontes de Carvalho, L.C. and Eichinger, D. (1994) J. Biol. Chem. 269, 7970–7975.
- [821] Campetella, O.E., Uttaro, A.D., Parodi, A.J. and Frasch, A.C.C. (1994) Mol. Biochem. Parasitol. 64, 337–340.
- [822] Pontes-de-Carvalho, L.C., Tomlinson, S. and Nussenzweig, V. (1993) Mol. Biochem. Parasitol. 62, 19-26.
- [823] Leguizamón, M.S., Campetella, O.E., Cappa, S.M.G. and Frasch, A.C.C. (1994) Infect. Immun. 62, 3441–3446.

- [824] Leguizamón, M.S., Campetella, O., Russomando, G., Almiron, M., Guillen, I., Cappa, S.M.G. and Frasch, A.C.C. (1994) J. Infect. Dis. 170, 1570–1574.
- [825] Schenkman, S., Eichinger, D., Pereira, M.E.A. and Nussenzweig, V. (1994) Annu. Rev. Microbiol. 48, 499-523.
- [826] Frasch, A.C.C. (1994) Parasitology 108, S37-S44.
- [827] Shimmoto, M., Fukuhara, Y., Itoh, K., Oshima, A., Sakuraba, H. and Suzuki, Y. (1993) J. Clin. Invest. 91, 2393–2398.
- [828] Hakomori, S.-I. (1994) In: K. Bock and H. Clausen (Eds.), Complex Carbohydrates in Drug Research. Munksgaard, Copenhagen, pp. 337–349.
- [829] Saito, M. and Yu, R.K. (1993) J. Neurosci. Res. 36, 127-132.
- [830] Previato, J.O., Jones, C., Gonçalves, L.P.B., Wait, R., Travassos, L.R. and Mendonça-Previato, L. (1994) Biochem. J. 301, 151–159.
- [831] Schenkman, S. and Eichinger, D. (1993) Parasitol. Today 9, 218-222.
- [832] Do Carmo Ciavaglia, M., de Carvalho, T.U. and de Souza, W. (1993) Biochem. Biophys. Res. Commun. 193, 718-721.
- [833] Tomlinson, S., Pontes de Carvalho, L.C., Vandekerckhove, F. and Nussenzweig, V. (1994) J. Immunol. 153, 3141–3147.
- [834] Prioli, R.P., Mejia, J.S. and Pereira, M.E.A. (1991) Eur. J. Epidemiol. 7, 344–348.
- [835] Chuenkova, M. and Pereira, M.E.A. (1995) J. Exp. Med. 181, 1693-1703.
- [836] Wessels, M.R., Rubens, C.E., Benedi, V.-J. and Kasper, D.L. (1989) Proc. Natl. Acad. Sci. USA 86, 8983–8987.
- [837] Frangipane, J.V. and Rest, R.F. (1993) Infect. Immun. 61, 1657-1666.
- [838] Schenkman, S., Kurosaki, T., Ravetch, J.V. and Nussenzweig, V. (1992) J. Exp. Med. 175, 1635–1641.
- [839] Araya, J.E., Cano, M.I., Yoshida, N. and da Silveira, J.F. (1994) Mol. Biochem. Parasitol. 65, 161–169.
- [840] Reyes, M.B., Pollevick, G.D. and Frasch, A.C.C. (1994) Gene 140, 139-140.
- [841] Kahn, S., Kahn, M., Van Voorhis, W.C., Goshorn, A., Strand, A., Hoagland, N., Eisen, H. and Pennathur, S. (1993) Mol. Biochem. Parasitol. 60, 149–152.
- [842] De Araujo-Jorge, T.C., Barbosa, H.S. and Meirelles, M.N.L. (1992) Mem. Inst. Oswaldo Cruz, Rio de Janeiro 87, Suppl. V, 43-56.
- [843] Ramirez, M.I., de Cassia Ruiz, R., Araya, J.E., da Silveira, J.F. and Yoshida, N. (1993) Infect. Immun. 61, 3636–3641.
- [844] Fenton Hall, B., Webster, P., Ma, A.K., Joiner, K.A. and Andrews, N.W. (1992) J. Exp. Med. 176, 313–325.
- [845] Fenton Hall, B. and Joiner, K.A. (1993) J. Euk. Microbiol. 40, 207-213.
- [846] Andrews, N.W. (1994) Braz. J. Med. Biol. Res. 27, 471-475.
- [847] Vermelho, A.B. and Meirelles, M.N.L. (1994) Mem. Inst. Oswaldo Cruz, Rio de Janeiro 89, 69-79.
- [848] Pellegrin, J.-L.J., Ortega-Barria, E., Barza, M., Baum, J. and Pereira, M.E.A. (1991) Invest. Ophthalmol. Visual Sci. 32, 3061–3066.
- [849] Godoy, V.G., Miller Dallas, M., Russo, T.A. and Malamy, M.H. (1993) Infect. Immun. 61, 4415-4426.
- [850] Nakato, H., Shinomiya, K. and Mikawa, H. (1986) Pathol. Res. Pract. 181, 311-319.
- [851] Roggentin, T., Kleineidam, R.G., Majewski, D.M., Tirpitz, D., Roggentin, P. and Schauer, R. (1993)
 J. Immunol. Methods 157, 125–133.
- [852] Roggentin, T., Roggentin, P., Kleineidam, R.G., Majewski, D.M. and Schauer, R. (1992) Trends Glycosci. Glycotechnol. 4, 263–268.
- [853] Schauer, R. (1992) Nachr. Chem. Tech. Lab. 40, 1227-1231.
- [854] Corfield, A.P., Wagner, S.A., Clamp, J.R., Kriaris, M.S. and Hoskins, L.C. (1992) Infect. Immun. 60, 3971–3978.
- [855] Boureau, H., Decré, D., Carlier, J.P., Guichet, C. and Bourlioux, P. (1993) Res. Microbiol. 144, 405– 410.
- [856] McGregor, J.A., French, J.I., Jones, W., Milligan, K., McKinney, P.J., Patterson, E. and Parker, R. (1994) Am. J. Obstet. Gynecol. 170, 1048–1060.
- [857] Homer, K.A., Whiley, R.A. and Beighton, D. (1994) J. Med. Microbiol. 41, 184-190.

- [858] Silva Filho, F.C., Breier-Saraiva, E.M., Tosta, M.X. and de Souza, W. (1989) Mol. Biochem. Parasitol. 35, 73-78.
- [859] Paton, J.C., Andrew, P.W., Boulnois, G.J. and Mitchell, T.J. (1993) Annu. Rev. Microbiol. 47, 89-115.
- [860] Boulnois, G.J. (1992) J. Genet. Microbiol. 138, 249-259.
- [861] Straus, D.C., Unbehagen, P.J. and Purdy, C.W. (1993) Infect. Immun. 61, 253-259.
- [862] Nuss, J.M. and Air, G.M. (1994) J. Mol. Biol. 235, 747-759.
- [863] Herrler, G., Hausmann, J. and Klenk, H.-D. (1995) In: A. Rosenberg (Ed.), Biology of the Sialic Acids. Plenum Publ. Corp., New York, pp. 315–336.
- [864] Hu, X., Ray, R. and Compans, R.W. (1992) J. Virol. 66, 1528-1534.
- [865] Schauer, R. and Corfield, A.P. (1981) In: F.G. de las Heras and S. Vega (Eds.), Medicinal Chemistry Advances. Pergamon Press, Oxford, New York, pp. 423–434.
- [866] Meindl, P., Bodo, G., Palese, P., Schulman, J. and Tuppy, H. (1974) Virology 58, 457-463.
- [867] Von Itzstein, M., Wu, W.-Y., Kok, G.B., Pegg, M.S., Dyason, J.C., Jin, B., Van Phan, T., Smythe, M.L., White, H.F., Oliver, S.W., Colman, P.M., Varghese, J.N., Ryan, D.M., Woods, J.M., Bethell, R.C., Hotham, V.J., Cameron, J.M. and Penn, C.R. (1993) Nature 363, 418–423.
- [868] Woods, J.M., Bethell, R.C., Coates, J.A.V., Healy, N., Hiscox, S.A., Pearson, B.A., Ryan, D.M., Ticehurst, J., Tilling, J., Walcott, S.M. and Penn, C.R. (1993) Antimicrob. Agents Chemother. 37, 1473–1479.
- [869] Zbiral, E., Schreiner, E., Christian, R., Kleineidam, R.G. and Schauer, R. (1989) Liebigs Ann. Chem. 159–165.
- [870] Zbiral, E., Schreiner, E., Salunkhe, M.M., Schulz, G., Kleineidam, R.G. and Schauer, R. (1989) Liebigs Ann. Chem. 519–526.
- [871] Suzuki, Y., Sato, K., Kiso, M. and Hasegawa, A. (1990) Glycoconjugate J. 7, 349-356.
- [872] Umezawa, H., Aoyagi, T., Komiyama, H., Morishima, H., Hamada, M. and Takeuchi, T. (1974) J. Antibiot. 27, 963-969.
- [873] Kudo, T., Nishimura, Y., Kondo, S. and Takeuchi, T. (1993) J. Antibiot. 46, 300-309.
- [874] Nishimura, Y., Umezawa, Y., Kondo, S., Takeuchi, T., Mori, K., Kijima-Suda, I., Tomita, K., Sugawara, K. and Nakamura, K. (1993) J. Antibiot. 46, 1883–1889.
- [875] Nagai, T., Miyaichi, Y., Tomimori, T., Suzuki, Y. and Yamada, H. (1992) Antiviral Res. 19, 207-217.
- [876] Messer, M. (1974) Biochem. J. 139, 415-420.
- [877] Taylor, N.R. and von Itzstein, M. (1994) J. Med. Chem. 37, 616-624.
- [878] Pegg, M.S. and von Itzstein, M. (1994) Biochem. Mol. Biol. Int. 32, 851-858.
- [879] Corfield, T. (1993) Glycobiology 3, 413-422.
- [880] Taylor, G. (1993) Nature 363, 401-402.
- [881] Taylor, G., Vimr, E., Garman, E. and Laver, W.G. (1992) J. Mol. Biol. 226, 1287-1290.
- [882] Ryan, D.M., Ticehurst, J., Dempsey, M.H. and Penn, C.R. (1994) Antimicrob. Agents Chemother. 38, 2270–2275.
- [883] Hayden, F.G., Rollins, B.S. and Madren, L.K. (1994) Antiviral Res. 25, 123-131.
- [884] Thomas, G.P., Forsyth, M., Penn, C.R. and McCauley, J.W. (1994) Antiviral Res. 24, 351-356.
- [885] Nagai, T., Miyaichi, Y., Tomimori, T., Suzuki, Y. and Yamada, H. (1990) Chem. Pharm. Bull. 38, 1329–1332.
- [886] Sabesan, S., Neira, S., Davidson, F., Duus, J.Ø. and Bock, K. (1994) J. Am. Chem. Soc. 116, 1616– 1634.
- [887] Driguez, P.-A., Barrere, B., Chantegrel, B., Deshayes, C., Doutheau, A. and Quash, G. (1992) Bioorg. Med. Chem. Lett. 2, 1361–1366.
- [888] Yamamoto, T., Kumazawa, H., Inami, K., Teshima, T. and Shiba, T. (1992) Tetrahedron Lett. 33, 5791– 5794.
- [889] Sabesan, S. (1994) Bioorg. Med. Chem. Lett. 4, 2457-2460.
- [890] Smythe, M.L. and von Itzstein, M. (1994) J. Am. Chem. Soc. 116, 2725-2733.
- [891] Prioli, R.P., Rosenberg, I. and Pereira, M.E.A. (1990) Mol. Biochem. Parasitol. 38, 191-198.
- [891a] Schauer, R. and Wember, M. (1996) Biol. Chem. Hoppe-Seyler 377, 293-299.
- [892] Zbiral, E., Kleineidam, R.G., Schreiner, E., Hartmann, M., Christian, R. and Schauer, R. (1992) Biochem. J. 282, 511–516.

- [893] Schauer, R., Stoll, S., Zbiral, E., Schreiner, E., Brandstetter, H.H., Vasella, A. and Baumberger, F. (1987) Glycoconjugate J. 4, 361–369.
- [894] Uchida, Y., Tsukada, Y. and Sugimori, T. (1984) J. Biochem. 96, 507-522.
- [895] Aisaka, K. and Uwajima, T. (1986) Appl. Environ. Microbiol. 51, 562-565.
- [896] Kawakami, B., Kudo, T., Narahashi, Y. and Horikoshi, K. (1986) Agric. Biol. Chem. 50, 2155-2158.
- [897] Ohta, Y., Shimosaka, M., Murata, K., Tsukada, Y. and Kimura, A. (1986) Appl. Microbiol. Biotechnol. 24, 386–391.
- [898] Aisaka, K., Igarashi, A., Yamaguchi, K. and Uwajima, T. (1991) Biochem. J. 276, 541-546.
- [899] Izard, T., Lawrence, M.C., Malby, R.L., Lilley, G.G. and Colman, P.M. (1994) Structure 2, 361–369.
- [900] Horiuchi, T. and Kurokawa, T. (1989) Clin. Chim. Acta 182, 117-122.
- [901] Hommes, F.A. (1991) In: F.A. Hommes (Ed.), Techniques in Diagnostic Human Biochemica Genetics: A Laboratory Manual. Wiley-Liss, New York, pp. 233–237.
- [902] Nöhle, U., Beau, J.-M. and Schauer, R. (1982) Eur. J. Biochem. 126, 543-548.
- [903] Nees, S. and Schauer, R. (1974) Behring Inst. Mitt. 55, 68-78.
- [904] Vimr, E.R. and Troy, F.A. (1985) J. Bacteriol. 164, 845-853.
- [905] Burnet, F.M., McCrea, J.F. and Stone, J.D. (1946) In: A. Gottschalk (Ed.), The Chemistry and Biology of Sialic Acids and Related Substances. University Press, Cambridge, pp. 98–105.
- [906] Faillard, H. and Schauer, R. (1972) In: A. Gottschalk (Ed.), Glycoproteins, Their Composition, Structure and Function. Elsevier, Amsterdam, pp. 1246–1267.
- [907] Varki, A. (1993) Glycobiology 3, 97-130.
- [908] Reutter, W., Köttgen, E., Bauer, C. and Gerok, W. (1982) In: R. Schauer (Ed.), Sialic Acids Chemistry, Metabolism and Function. Springer, Wien, pp. 263–305.
- [909] Rosenberg, A. and Schengrund, C.-L. (Eds.) (1976) Biological Roles of Sialic Acid, Plenum Press, New York.
- [910] Shimamura, M., Shibuya, N., Ito, M. and Yamagata, T. (1994) Biochem. Mol. Biol. Int. 33, 871-878.
- [911] Böhler, T. and Linderkamp, O. (1993) Clin. Hemorheol. 13, 775-778.
- [912] Izumida, Y., Seiyama, A. and Maeda, N. (1991) Biochim. Biophys. Acta 1067, 221-226.
- [913] Rogers, M.E., Williams, D.T., Niththyananthan, R., Rampling, M.W., Heslop, K.E. and Johnston, D.G. (1992) Clin. Sci. 82, 309–313.
- [914] Litvinov, S.V. and Hilkens, J. (1993) J. Biol. Chem. 268, 21364-21371.
- [915] Wibawan, I.W.T. and Lämmler, C. (1991) J. Genet. Microbiol. 137, 2721-2725.
- [916] Rahmann, H. (1992) In: N.N. Osborne (Ed.), Current Aspects of the Neurosciences. The Macmillan Press Ltd., pp. 87–125.
- [917] Chapman, S.A., Bonshek, R.E., Stoddart, R.W., Mackenzie, K.R. and McLeod, D. (1994) Br. J. Ophthalmol. 78, 632-637.
- [918] Aubin, Y. and Prestegard, J.H. (1993) Biochemistry 32, 3422-3428.
- [919] Jaenicke, R. (1993) Philos. Trans. R. Soc. London B 339, 287-295.
- [920] Fukuda, M. (1991) J. Biol. Chem. 266, 21327-21330.
- [921] Klenk, H.-D. and Garten, W. (1994) Trends Microbiol. 2, 39-43.
- [922] Yamamotogoshima, F. and Maeno, K. (1994) Microbiol. Immunol. 38, 819-822.
- [923] Corfield, A. (1993) In: B. Renner and G. Sawatzki (Eds.), New Perspectives in Infant Nutrition. Georg Thieme Verlag, Stuttgart, New York, pp. 57–65.
- [924] Verma, M. and Davidson, E.A. (1994) Glycoconjugate J. 11, 172-179.
- [925] Schauer, R. (1992) In: H. Ogura, A. Hasegawa and T. Suami (Eds.), Carbohydrates Synthetic Methods and Applications in Medicinal Chemistry. Kodansha, Tokyo, pp. 340–354.
- [926] Alviano, C.S., Pereira, M.E.A., Souza, W., Oda, L.M. and Travassos, L.R. (1982) FEMS Microbiol. Lett. 15, 223–227.
- [927] Soares, R.M.A., Alviano, C.S., Angluster, J. and Travassos, L.R. (1993) FEMS Microbiol. Lett. 108, 31-34.
- [928] Souza, E.T., Silva-Filho, F.C., de Souza, W., Alviano, C.S., Angluster, J. and Travassos, L.R. (1986) J. Med. Vet. Mycol. 24, 145–153.
- [929] Oda, L.M., Kubelka, C.F., Alviano, C.S. and Travassos, L.R. (1983) Infect. Immun. 39, 497-504.
- [930] Sharon, N. (1994) Struct. Biol. 1, 843-845.

- [931] Hughes, R.C. (1994) Glycobiology 4, 5-12.
- [932] Drickamer, K. (1994) In: M. Fukuda and O. Hindsgaul (Eds.), Molecular Glycobiology. IRL Press, Oxford, pp. 53-87.
- [933] Geffen, I. and Spiess, M. (1992) Int. Rev. Cytol. 137B, 181-219.
- [934] Liener, I.E., Sharon, N., Goldstein, I.J. (Eds.) (1986) The Lectins. Academic Press, Orlando.
- [935] Müller, W.E.G., Schröder, H.C., Schäcke, H., Müller, I.M. and Gamulin, V. (1993/94) Endocytobiosis and Cell Res. 10, 185–204.
- [936] Gamulin, V., Rinkevich, B., Schäcke, H., Kruse, M., Müller, I.M. and Müller, W.E.G. (1994) Biol. Chem. Hoppe-Seyler 375, 583-588.
- [937] Ashwell, G. and Morell, A.G. (1974) Adv. Enzymol. 41, 99-128.
- [938] Lodish, H.F., Kong, N. and Wikström, L. (1992) J. Biol. Chem. 267, 12753-12760.
- [939] Treichel, U., Meyer zum Büschenfelde, K.-H., Stockert, R.J., Poralla, T. and Gerken, G. (1994) J. Genet. Virol. 75, 3021–3029.
- [940] Thotakura, N.R., Szkudlinski, M.W. and Weintraub, B.D. (1994) Glycobiology 4, 525-533.
- [941] Martinuk, S.D., Manning, A.W., Black, W.D. and Murphy, B.D. (1991) Biol. Reproduct. 45, 598-604.
- [942] Jenkins, N. and Curling, E.M.A. (1994) Enzyme Microbiol. Technol. 16, 354-364.
- [943] Henkin, J., Dudlak, D., Beebe, D.P. and Sennello, L. (1991) Thrombosis Res. 63, 215-225.
- [944] Dong, Y.J., Kung, C. and Goldwasser, E. (1992) J. Cell. Biochem. 48, 269-276.
- [945] Matteri, R.L., Warikoo, P.K. and Bavister, B.D. (1992) Am. J. Primatol. 27, 205-213.
- [946] Lascols, O., Cherqui, G., Munier, A., Picard, J. and Capeau, J. (1994) Cell. Mol. Biol. 40, 359-371.
- [947] Amano, J. and Kobata, A. (1993) Arch. Biochem. Biophys. 305, 618-621.
- [948] Müller, E., Schröder, C., Sharon, N. and Schauer, R. (1983) Hoppe-Seyler's Z. Physiol. Chem. 364, 1410–1420.
- [949] Fischer, C., Kelm, S., Ruch, B. and Schauer, R. (1991) Carbohydr. Res. 213, 263-273.
- [950] Schauer, R. (1992) In: E. Windler and H. Greten (Eds.), Hepatic Endocytosis of Lipids and Proteins.
 W. Zuckschwerdt Verlag, München, pp. 127–136.
- [951] Sato, S. and Hughes, R.C. (1994) J. Biol. Chem. 269, 4424-4430.
- [952] Kelm, S. and Schauer, R. (1988) Biol. Chem. Hoppe-Seyler 369, 693-704.
- [953] Bratosin, D., Mazurier, J., Debray, H., Lecocq, M., Boilly, B., Alonso, C., Moisei, M., Motas, C. and Montreuil, J. (1995) Glycoconjugate J. 12, 258–267.
- [954] Rolfes-Curl, A., Ogden, L.L., Omann, G.M. and Aminoff, D. (1991) Exp. Gerontol. 26, 327-345.
- [955] Stenberg, P.E., Levin, J., Baker, G., Mok, Y. and Corash, L. (1991) J. Cell Physiol. 147, 7-16.
- [956] Savill, J., Fadok, V., Henson, P. and Haslett, C. (1993) Immunol. Today 14, 131-136.
- [957] Dini, L., Autuori, F., Lentini, A., Oliverio, S. and Piacentini, M. (1992) FEBS Lett. 296, 174-178.
- [958] Froman, D.P. and Thursam, K.A. (1994) Biol. Reproduct. 50, 1094–1099.
- [959] Lassalle, B. and Testart, J. (1994) J. Reproduct. Fertil. 101, 703-711.
- [960] Calzada, L., Salazar, E.L. and Pedron, N. (1994) Arch. Androl. 33, 87-92.
- [961] Kelm, S. and Schauer, R. (1986) Biol. Chem. Hoppe-Seyler 367, 989-998.
- [962] Ruch, B., Kelm, S. and Schauer, R. (1988) In: R. Schauer and T. Yamakawa (Eds.), Sialic Acids 1988, Proc. Jap.-German Symp. Sialic Acids. Verlag Wissenschaft und Bildung, Kiel, pp. 268–269.
- [963] Kuiper, J., Bakkeren, H.F., Biessen, E.A.L. and van Berkel, T.J.C. (1994) Biochem. J. 299, 285-290.
- [964] Kelm, S., Shukla, A.K., Paulson, J.C. and Schauer, R. (1986) Carbohydr. Res. 149, 59-64.
- [965] Reuter, G., Kelm, S. and Schauer, R. (1988) Acta Histochem. Suppl. XXXVI, 51-79.
- [966] Okuda, K., Ono, M. and Kato, T. (1989) Infect. Immun. 57, 1635-1637.
- [967] Sheth, H.B., Lee, K.K., Wong, W.Y., Srivastava, G., Hindsgaul, O., Hodges, R.S., Paranchych, W. and Irvin, R.T. (1994) Mol. Microbiol. 11, 715-723.
- [968] Devaraj, N., Sheykhnazari, M., Warren, W.S. and Bhavanandan, V.P. (1994) Glycobiology 4, 307-316.
- [969] Ourth, D.D. and Bachinski, L.M. (1987) Dev. Comp. Immunol. 11, 551-564.
- [970] Smith, H. (1991) Proc. R. Soc. London B 246, 97-105.
- [971] Jarvis, G.A. (1995) Trends Microbiol. 3, 198-201.
- [972] Elkins, C., Carbonetti, N.H., Varela, V.A., Stirewalt, D., Klapper, D.G. and Sparling, P.F. (1992) Mol. Microbiol. 6, 2617-2628.
- [973] Rest, R.F. and Frangipane, J.V. (1992) Infect. Immun. 60, 989-997.

- [974] Mandrell, R.E., Smith, H., Jarvis, G.A., Griffiss, J.M. and Cole, J.A. (1993) Microb. Pathogen. 14, 307–313.
- [975] Nahori, M.-A., Etievant, M., Terzidis-Trabelsi, H., David, B. and Lambre, C.R. (1994) Immunol. Lett. 42, 67-73.
- [976] Haddad, E.-B. and Gies, J.-P. (1992) Eur. J. Pharmacol. 211, 273-276.
- [977] Paulson, J.C. (1985) In: M. Conn (Ed.), The Receptors, Vol. 2. Academic Press, New York, pp. 131– 219.
- [978] Rogers, G.N., Herrler, G., Paulson, J.C., and Klenk, H.-D. (1986) J. Biol. Chem. 261, 5947-5951.
- [979] Vlasak, R., Luytjes, W., Spaan, W. and Palese, P. (1988) Proc. Natl. Acad. Sci. USA 85, 4526-4529.
- [980] Markwell, M.A.K. and Paulson, J.C. (1980) Proc. Natl. Acad. Sci. USA 77, 5693-5697.
- [981] Cahan, L.D. and Paulson, J.C. (1980) Virology 103, 505-509.
- [982] Svensson, L. (1992) J. Virol. 66, 5582-5585.
- [983] Loomes, L.M., Uemura, K.-I., Childs, R.A., Paulson, G.N., Rogers, G.N., Scudder, P.R., Michalski, J.-C., Hounsell, E.F., Taylor-Robinson, D. and Feizi, T. (1984) Nature 307, 560–563.
- [984] Murray, P.A., Levine, M.J., Tabak, L.A. and Reddy, M.S. (1982) Biochem. Biophys. Res. Commun. 106, 390–396.
- [985] Teneberg, S., Willemsen, P.T.J., de Graaf, F.K., Stenhagen, G., Pimlott, W., Jovall, P.-A., Angström, J. and Karlsson, K.-A. (1994) J. Biochem. 116, 560–574.
- [986] Parkkinen, A., Rogers, G.N., Korhonen, T., Dahr, W. and Finne, J. (1986) Infect. Immun. 54, 37-42.
- [987] Hanisch, F.-G., Hacker, J. and Schroten, H. (1993) Infect. Immun. 61, 2108-2115.
- [988] Hacker, J., Kestler, H., Hoschützky, H., Jann, K., Lottspeich, F. and Korhonen, T.K. (1993) Infect. Immun. 61, 544-550.
- [989] Hacker, J. and Morschhäuser, J. (1994) In: P. Klemm (Ed.), Fimbriae. Adhesion, Genetics, Biogenesis, and Vaccines. CRC Press, Boca Raton, Ann Arbor, London, Tokyo, pp. 27–36.
- [990] Morschhäuser, J., Vetter, V., Korhonen, T., Uhlin, B.E. and Hacker, J. (1993) Zbl. Bakt. 278, 165-176.
- [991] Ishikawa, H. and Isayama, Y. (1987) Infect. Immun. 55, 1607-1609.
- [992] Ko, H.L., Beuth, J., Sölter, J., Schroten, H., Uhlenbruck, G. and Pulverer, G. (1987) Infection 15, 237-240.
- [993] Hirmo, S., Kelm, S., Schauer, R., Nilsson, B. and Wadström, T. (1996) Glycoconjugate J. 13, 1005– 1011.
- [994] Liukkonen, J., Haataja, S., Tikkanen, K., Kelm, S. and Finne, J. (1992) J. Biol. Chem. 267, 21105– 21111.
- [995] Perkins, M.E. and Rocco, L.J. (1988) J. Immunol. 141, 3190-3196.
- [996] Pakandl, M. and Grubhoffer, L. (1994) Comp. Biochem. Physiol. 108B, 529-536.
- [997] Schengrund, C.-L. and Ringler, N.J. (1989) J. Biol. Chem. 264, 13233-13237.
- [998] Brennan, M.J., David, J.L., Kenimer, J.G. and Manclark, C.R. (1988) J. Biol. Chem. 263, 4895-4899.
- [999] Armstrong, G.D., Clark, C.G. and Heerze, L.D. (1994) Infect. Immun. 62, 2236-2243.
- [1000] Schiavo, G., Demel, R. and Montecucco, C. (1991) Eur. J. Biochem. 199, 705-711.
- [1001] Konami, Y., Ishida, C., Yamamoto, K., Osawa, T. and Irimura, T. (1994) J. Biochem. 115, 767-777.
- [1002] Goldhar, J. (1994) Methods Enzymol. 236, 211-231.
- [1003] Mandrell, R.E., Apicella, M.A., Lindstedt, R. and Leffler, H. (1994) Methods Enzymol. 236, 231-254.
- [1004] Sauter, N.K., Bednarski, M.D., Wurzburg, B.A., Hanson, J.E., Whitesides, G.M., Skehel, J.J. and Wiley, D.C. (1989) Biochemistry 28, 8388-8396.
- [1005] Weis, W., Brown, J.H., Cusack, S., Paulson, J.C., Skehel, J.J. and Wiley, D.C. (1988) Nature 333, 426-431.
- [1006] Watowich, S.J., Skehel, J.J. and Wiley, D.C. (1994) Structure 2, 719-731.
- [1007] Mochalova, L.V., Tuzikov, A.B., Marinina, V.P., Gambaryan, A.S., Byramova, N.E., Bovin, N.V. and Matrosovich, M.N. (1994) Antiviral Res. 23, 179–190.
- [1008] Malby, R.L., Tulip, W.R., Harley, V.R., McKimm-Breschkin, J.L., Laver, W.G., Webster, R.G. and Colman, P.M. (1994) Structure 2, 733-746.
- [1009] Lelwala-Guruge, J., Ascencio, F., Kreger, A.S., Ljungh, Å. and Wadström, T. (1993) Zbl. Bakt. 280, 93-106.

- [1010] Schroten, H., Plogmann, R., Hanisch, F.-G., Hacker, J., Nobis-Bosch, R. and Wahn, V. (1993) Acta Paediatr. 82, 6-11.
- [1011] Holder, A.A., Blackman, M.J., Borre, M., Burghaus, P.A., Chappel, J.A., Keen, J.K., Ling, I.T., Ogun, S.A., Owen, C.A. and Sinha, K.A. (1994) Biochem. Soc. Trans. 22, 291–295.
- [1012] Titani, K., Takio, K., Kuwada, M., Nitta, K., Sakakibara, F., Kawauchi, H., Takayanagi, G. and Hakomori, S.-I. (1987) Biochemistry 26, 2189-2194.
- [1013] Chakraborty, I., Mandal, C. and Chowdhury, M. (1993) Mol. Cell. Biochem. 126, 77-86.
- [1014] Popoli, M. and Mengano, A. (1988) Neurochem. Res. 13, 63-67.
- [1015] Tiemeyer, M., Yasuda, Y. and Schnaar, R.L. (1989) J. Biol. Chem. 264, 1671-1681.
- [1016] Banerjee, M. and Chowdhury, M. (1994) Human Reproduct. 9, 1497-1504.
- [1017] Ahmed, H. and Gabius, H.-J. (1989) J. Biol. Chem. 264, 18673-18678.
- [1018] Meri, S. and Pangburn, M.K. (1990) Proc. Natl. Acad. Sci. USA 87, 3982-3986.
- [1019] Crocker, P.R., Kelm, S., Dubois, C., Martin, B., McWilliam, B.S., Shotten, D.M., Paulson, J.C. and Gordon, S. (1991) EMBO J. 10, 1661–1669.
- [1020] Zeng, F.Y. and Gabius, H.J. (1991) Arch. Biochem. Biophys. 289, 137-144.
- [1021] Zeng, F.Y. and Gabius, H.J. (1992) Biochem. Int. 26, 17-24.
- [1022] Sgroi, D., Varki, A., Braesch-Andersen, S. and Stamenkovic, I. (1993) J. Biol. Chem. 268, 7011-7018.
- [1023] Powell, L.D., Sgroi, D., Sjoberg, E.R., Stamenkovic, I. and Varki, A. (1993) J. Biol. Chem. 268, 7019– 7027.
- [1024] Crocker, P.R., Mucklow, S., Bouckson, V., McWilliam, A., Willis, A.C., Gordon, S., Milon, G., Kelm, S. and Bradfield, P. (1994) EMBO J. 13, 4490–4503.
- [1025] Kelm, S., Pelz, A., Schauer, R., Filbin, M.T., Tang, S., de Bellard, M.-E., Schnaar, R.L., Mahoney, J.A., Hartnell, A., Bradfield, P. and Crocker, P.R. (1994) Curr. Biol. 4, 965–972.
- [1026] Fujita, N., Sato, S., Kurihara, T., Kuwano, R., Sakimura, K., Inusuka, T., Takahashi, Y. and Miyatake, T. (1989) Biochem. Biophys. Res. Commun. 165, 1162–1169.
- [1027] Stamenkovic, I. and Seed, B. (1990) Nature 345, 74-77.
- [1028] Hanasaki, K., Varki, A. and Powell, L.D. (1995) J. Biol. Chem. 270, 7533-7542.
- [1029] Hanasaki, K., Powell, L.D. and Varki, A. (1995) J. Biol. Chem. 270, 7543-7550.
- [1030] Powell, L.D., Jain, R.K., Matta, K.L., Sabesan, S. and Varki, A. (1995) J. Biol. Chem. 270, 7523-7532.
- [1031] Simmons, D. and Seed, B. (1988) J. Immunol. 141, 2797-2800.
- [1032] Freeman, S.D., Kelm, S., Barber, E.K. and Crocker, P.R. (1995) Blood 85, 2005-2012.
- [1033] Bevilacqua, M., Butcher, E., Furie, B., Furie, B., Gallatin, M., Gimbrone, M., Harlan, J., Kishimoto, K., Lasky, L., McEver, R., Paulson, J., Rosen, S., Seed, B., Siegelman, M., Springer, T., Stoolman, L., Tedder, T., Varki, A., Wagner, D., Weissman, I. and Zimmerman, G. (1991) Cell 67, 233.
- [1034] Varki, A. (1994) Proc. Natl. Acad. Sci. USA 91, 7390-7397.
- [1035] Hughes, R.C. (1992) Curr. Opin. Struct. Biol. 2, 687-692.
- [1036] Stad, R.K. and Buurman, W.A. (1994) Cell Adhes. Commun. 2, 261-268.
- [1037] Lowe, J.B. (1994) In: M. Fukuda and O. Hindsgaul (Eds.), Molecular Glycobiology. IRL Press, Oxford, pp. 163–205.
- [1038] Feizi, T. (1994) In: F. Wieland and W. Reutter (Eds.), 44th Colloquium Mosbach 1993, Glyco- and Cellbiology. Springer-Verlag, Berlin, Heidelberg, pp. 145–160.
- [1039] Bischoff, J. (1994) Trends Glycosci. Glycotechnol. 6, 351-365.
- [1040] Ito, K., Handa, K. and Hakomori, S.-I. (1994) Glycoconjugate J. 11, 232-237.
- [1041] Hanski, C., Drechsler, K., Hanisch, F.-G., Sheehan, J., Manske, M., Ogorek, D., Klussmann, E., Hanski, M.-L., Blank, M., Xing, P.-X., McKenzie, I.F.C., Devine, P.L. and Riecken, E.-O. (1993) Cancer Res. 53, 4082–4088.
- [1042] Hakomori, S.-i. (1991) Curr. Opin. Immunol. 3, 646-653.
- [1043] Basu, M., Basu, S.S., Li, Z., Tang, H. and Basu, S. (1993) Indian J. Biochem. Biophys. 30, 324-332.
- [1044] Fukuda, M. (1994) In: K. Bock and H. Clausen (Eds.), Complex Carbohydrates in Drug Research. Munksgaard, Copenhagen, pp. 353–365.
- [1045] Hanisch, F.-G., Hanski, C. and Hasegawa, A. (1992) Cancer Res. 52, 3138-3144.
- [1046] Hemmerich, S. and Rosen, S.D. (1994) Biochemistry 33, 4830-4835.

- [1047] Scudder, P.R., Shailubhai, K., Duffin, K.L., Streeter, P.R. and Jacob, G.S. (1994) Glycobiology 4, 929-933.
- [1048] Huang, K., Kikuta, A. and Rosen, S.D. (1994) J. Neuroimmunol. 53, 133-141.
- [1049] Graves, B.J., Crowther, R.L., Chandran, C., Rumberger, J.M., Li, S., Huang, K.-S., Presky, D.H., Familletti, P.C., Wolitzky, B.A. and Burns, D.K. (1994) Nature 367, 532–538.
- [1050] Drickamer, K. (1994) Glycobiology 4, 245-248.
- [1051] Hajela, K., Kayestha, R. and Kumar, G.P. (1994) FEBS Lett. 354, 217-219.
- [1052] Kotovuori, P., Tontti, E., Pigott, R., Shepherd, M., Kiso, M., Hasegawa, A., Renkonen, R., Noramo, P., Altieri, C.D. and Gahmberg, C.G. (1993) Glycobiology 3, 131–136.
- [1053] Diamond, M.S. and Springer, T.A. (1994) Curr. Biol. 4, 506-517.
- [1054] Böggemeyer, E., Stehle, T., Schaible, U.E., Hahne, M., Vestweber, D. and Simon, M.M. (1994) Cell Adhes. Commun. 2, 145–157.
- [1055] Simmons, R.D. and Cattle, B.A. (1992) J. Neuroimmunol. 41, 123-130.
- [1056] Mulligan, M.S., Paulson, J.C., De Frees, S., Zheng, Z.-L., Lowe, J.B. and Ward, P.A. (1993) Nature 364, 149–151.
- [1057] Buerke, M., Weyrich, A.S., Zheng, Z.-L., Gaeta, F.C.A., Forrest, M.J. and Lefer, A.M. (1994) J. Clin. Invest. 93, 1140–1148.
- [1058] Ichikawa, Y. and Wong, C.-H. (1994) In: K. Bock and H. Clausen (Eds.), Complex Carbohydrates in Drug Research. Munksgaard, Copenhagen, pp. 118–130.
- [1059] Chen, L.Y., Nichols, W.W., Hendricks, J.B., Yang, B.C. and Mehta, J.L. (1994) Cardiovasc. Res. 28, 1414–1422.
- [1060] Winn, R.K., Liggitt, D., Vedder, N.B., Paulson, J.C. and Harlan, J.M. (1993) J. Clin. Invest. 92, 2042– 2047.
- [1061] Stamenkovic, I., Sgroi, D. and Aruffo, A. (1992) Cell 68, 1003-1004.
- [1062] Van den Berg, T.K., Brevé, J.J.P., Damoiseaux, J.G.M.C., Döpp, E.A., Kelm, S., Crocker, P.R., Dijkstra, C.D. and Kraal, G. (1992) J. Exp. Med. 176, 647–655.
- [1063] Meyer-Franke, A. and Barres, B. (1994) Curr. Biol. 4, 847-850.
- [1064] Crocker, P.R., Freeman, S., Gordon, S. and Kelm, S. (1995) J. Clin. Invest. 95, 635-643.
- [1065] Engel, P., Nojima, Y., Rothstein, D., Zhou, L.-J., Wilson, G.L., Kehrl, J.H. and Tedder, T.F. (1993) J. Immunol. 150, 4719–4732.
- [1066] Braesch-Andersen, S. and Stamenkovic, I. (1994) J. Biol. Chem. 269, 11783-11786.
- [1067] Umemori, H., Sato, S., Yagi, T., Aizawa, S. and Yamamoto, T. (1994) Nature 367, 572-576.
- [1068] Beppu, M., Takahashi, T., Hayashi, T. and Kikugawa, K. (1994) Biochim. Biophys. Acta 1223, 47-56.
- [1069] Devi, S.J.N., Robbins, J.B. and Schneerson, R. (1991) Proc. Natl. Acad. Sci. USA 88, 7175-7179.
- [1070] Duk, M., Sticher, U., Brossmer, R. and Lisowska, E. (1994) Glycobiology 4, 175-181.
- [1071] Younes-Chennoufi, A.B., Léger, J.-M., Hauw, J.-J., Preud'homme, J.-L., Bouche, P., Aucouturier, P., Ratinahirana, H., Lubetzki, C., Lyon-Caen, O. and Baumann, N. (1992) Ann. Neurol. 32, 18–23.
- [1072] Kanamori, A., Inoue, S., Xulei, Z., Zuber, C., Roth, J., Kitajima, K., Ye, J., Troy II, F.A. and Inoue, Y. (1994) Histochemistry 101, 333–340.
- [1073] Bast, B.J.E.G., Zhou, L.-J., Freeman, G.J., Colley, K.J., Ernst, T.J., Munro, J.M. and Tedder, T.F. (1992) J. Cell Biol. 116, 423–435.
- [1074] Pörtner, A., Peter-Katalinić, J., Brade, H., Unland, F., Büntemeyer, H. and Müthing, J. (1993) Biochemistry 32, 12685–12693.
- [1075] Mukuria, C.J., Noguchi, A., Suzuki, E. and Naiki, M. (1994) Jpn. J. Med. Sci. Biol. 47, 253-264.
- [1076] Kimber, S.J., Bentley, J., Ciemerych, M., Moller, C.J. and Bock, E. (1994) Eur. J. Cell Biol. 63, 102– 113.
- [1077] Roth, J., Kempf, A., Reuter, G., Schauer, R. and Gehring, W.J. (1992) Science 256, 673-675.
- [1078] Roth, J. (1993) Histochem. J. 25, 687-710.
- [1079] Jowett, A.K., Kimber, S.J. and Ferguson, M.W.J. (1994) J. Anat. 185, 89-94.
- [1080] Skacel, P.O., Edwards, A.J., Harrison, C.T. and Watkins, W.M. (1991) Blood 78, 1452-1460.
- [1081] Vertino-Bell, A., Ren, J., Black, J.D. and Lau, J.T.Y. (1994) Dev. Biol. 165, 126-136.
- [1082] Nagai, Y. and Tsuji, S. (1994) Prog. Brain Res. 101, 119-126.

- [1083] Rahmann, H., Rösner, H., Körtje, K.-H., Beitinger, H. and Seybold, V. (1994) Prog. Brain Res. 101, 127–145.
- [1084] Yamashita, T., Tsuji, S. and Nagai, Y. (1991) Glycobiology 1, 149-154.
- [1085] Watanabe, M., Timm, M. and Fallah-Najmabadi, H. (1992) Dev. Dynam. 194, 128-141.
- [1086] Zuber, C., Lackie, P.M., Catterall, W.A. and Roth, J. (1992) J. Biol. Chem. 267, 9965-9971.
- [1087] Cantz, M. (1982) In: R. Schauer (Ed.), Sialic Acids Chemistry, Metabolism and Function. Springer Verlag Wien, New York, pp. 307–320.
- [1088] D'Agrosa, R.M., Hubbes, M., Zhang, S., Shankaran, R. and Callahan, J.W. (1992) Biochem. J. 285, 833-838.
- [1089] Okamura-Oho, Y., Zhang, S. and Callahan, J.W. (1994) Biochim. Biophys. Acta 1225, 244-254.
- [1090] Sasagasako, N., Miyahara, S., Saito, N., Shinnoh, N., Kobayashi, T. and Goto, I. (1993) Clin. Genet. 44, 8–11.
- [1091] Schmid, B., Paton, B.C., Sandhoff, K. and Harzer, K. (1992) Hum. Genet. 89, 513-518.
- [1092] Mueller, O.T., Henry, W.M., Haley, L.L., Byers, M.G., Eddy, R.L. and Shows, T.B. (1986) Proc. Natl. Acad. Sci. USA 83, 1817–1821.
- [1093] Grzeschik, K.H. and Skolnick, M.H. (1991) Cytogenet. Cell Genet. 58, 785-799.
- [1094] Elsliger, M.-A. and Potier, M. (1994) Prot. Struct. Funct. Genet. 18, 81-93.
- [1095] Pshezhetsky, A.V. and Potier, M. (1993) Biochem. Biophys. Res. Commun. 195, 354-362.
- [1096] Mancini, G.M.S., Hu, P., Verheijen, F.W., van Diggelen, O.P., Janse, H.C., Kleijer, W.J., Beemer, F.A. and Jennekens, F.G.I. (1992) Eur. J. Pediatr. 151, 590–595.
- [1097] Pisoni, R.L. and Thoene, J.G. (1991) Biochim. Biophys. Acta 1071, 351-373.
- [1098] Haataja, L., Schleutker, J., Laine, A.-P., Renlund, M., Savontaus, M.-L., Dib, C., Weissenbach, J., Peltonen, L. and Aula, P. (1994) Am. J. Hum. Genet. 54, 1042–1049.
- [1099] Maguire, T.M., Gillian, A.M., O'Mahony, D., Coughlan, C.M., Dennihan, A. and Breen, K.C. (1994) Neurobiol. Aging 15, 99-102.
- [1100] Maguire, T.M., O'Mahony, D., Gillian, A.M., Dennihan, A. and Breen, K.C. (1994) Neurodegeneration 3, 129–133.
- [1101] Narayanan, S. (1994) Ann. Clin. Lab. Sci. 24, 376-384.
- [1102] Vavasseur, F., Dole, K., Yang, J., Matta, K., Myerscough, N., Corfield, A., Paraskeva, C. and Brockhausen, I. (1994) Eur. J. Biochem. 222, 415–424.
- [1103] Vierbuchen, M., Schröder, S., Larena, A., Uhlenbruck, G. and Fischer, R. (1994) Virchows Arch. Int. J. Pathol. 424, 205–211.
- [1104] Kemmner, W., Krück, D. and Schlag, P. (1994) Clin. Exp. Metastasis 12, 245-254.
- [1105] Sata, T., Roth, J., Zuber, C., Stamm, B. and Heitz, P.U. (1991) Am. J. Pathol. 139, 1435-1448.
- [1106] Dall'Olio, F., Malagolini, N., Guerrini, S., Serafini-Cessi, F. (1993) Biochem. Biophys. Res. Commun. 196, 714-720.
- [1107] Patel, P.S., Adhvaryu, S.G., Balar, D.B., Parikh, B.J. and Shah, P.M. (1994) Anticancer Res. 14, 747– 751.
- [1108] Harvey, B.E., Toth, C.A., Wagner, H.E., Steele Jr., G.D. and Thomas, P. (1992) Cancer Res. 52, 1775– 1779.
- [1109] Mostafapour, M.K. and Goldstein, I.J. (1993) Arch. Biochem. Biophys. 303, 255-259.
- [1110] David, L., Nesland, J.M., Funderud, S. and Sobrinho-Simões (1993) Cancer 72, 1522-1527.
- [1111] Easton, E.W., Bolscher, J.G.M. and van den Eijnden, D.H. (1991) J. Biol. Chem. 266, 21674–21680.
- [1112] Miyagi, T., Sato, K., Hata, K. and Taniguchi, S. (1994) FEBS Lett. 349, 255-259.
- [1113] Takano, R., Muchmore, E. and Dennis, J.W. (1994) Glycobiology 4, 665-674.